PROSPECTUS SUPPLEMENT NO. 11 DATED July 27, 2007 (To Prospectus Dated August 7, 2006)

LIQUIDMETAL TECHNOLOGIES, INC.

11,614,322 Shares of Common Stock

This prospectus supplement supplements information contained in, and should be read in conjunction with, that certain Prospectus, dated August 7, 2006, of Liquidmetal Technologies, Inc., as supplemented by Supplement #1, dated August 9, 2006, Supplement #2, dated August 16, 2006, Supplement #3, dated October 12, 2006, Supplement #4, dated October 24, 2006, Supplement #5, dated November 14, 2006, Supplement #6 dated January 4, 2007, Supplement #7 dated March 16, 2007, Supplement #8 dated April 25, 2007, Supplement #9 dated May 15, 2007, and Supplement #10 dated June 6, 2007.

This prospectus supplement is not complete without, and may not be delivered or used except in connection with, the original Prospectus and Supplements #1, #2, #3, #4, #5, #6, #7, #8, #9, and #10 thereto. The Prospectus relates to the public sale, from time to time, of up to 11,614,322 shares of our common stock by the selling shareholders identified in the Prospectus.

The information attached to this prospectus supplement modifies and supersedes, in part, the information in the Prospectus, as supplemented. Any information that is modified or superseded in the Prospectus shall not be deemed to constitute a part of the Prospectus, except as modified or superseded by this prospectus supplement or Prospectus Supplements #1, #2, #3, #4, #5, #6, #7, #8, #9, and #10.

This prospectus supplement includes the attached Current Report on Form 8-K, as filed by us with the Securities and Exchange Commission on July 27, 2007.

We may amend or supplement the Prospectus, as supplemented, from time to time by filing amendments or supplements as required. You should read the entire Prospectus and any amendments or supplements carefully before you make an investment decision.

The Securities and Exchange Commission and state securities regulators have not approved or disapproved these securities or determined if this Prospectus Supplement (or the original Prospectus, as previously supplemented) is truthful or complete. Any representation to the contrary is a criminal offense.

The date of this prospectus supplement is July 27, 2007.

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT
Pursuant to Section 13 or 15(d) of the
Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): July 24, 2007

LIQUIDMETAL TECHNOLOGIES, INC.

(Exact name of Registrant as Specified in its Charter)

Delaware (State or other jurisdiction of incorporation or organization)

000-31332 (Commission File Number) 33-0264467 (I.R.S. Employer Identification No.)

30452 Esperanza
Rancho Santa Margarita, California 92688
(Address of Principal Executive Offices; Zip Code)

Registrant's telephone number, including area code: (949) 635-2100

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2-(b))
- Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

LIQUIDMETAL TECHNOLOGIES, INC.

FORM 8-K

Item 1.01. Material Contracts.

On July 24, 2007, Liquidmetal Technologies, Inc. (the "Company") completed an \$11.1 million financing transaction (the "Transaction") that provided the Company with funding to repay convertible notes previously issued by the Company that were scheduled to become due in July and August 2007. In the Transaction, the Company transferred substantially all of the assets (the "Transferred Assets") of the Company's Calciudmetal Coatings, LLC, a Delaware limited liability company ("LMC"), and LMC assumed substantially all of the liabilities of the Division (the "Assumed Liabilities"). LMC was capitalized through a \$6.5 million subordinated debt and equity investment by C₃ Capital Partners and a \$5.0 million senior credit facility with Bank Midwest, N.A., both out of Kansas City, Missouri. The Company will retain a 69.5% in LMC, and the remaining 30.75% of the equity of LMC will be held by C₃ Capital Partners, Larry Buffington (who will also serve as the President and CEO of LMC), and CRESO Capital Partners (the financial stations in the Transaction).

Asset Purchase and Contribution Agreement

In connection with the Transaction, the Company and LMC entered into an Asset Purchase and Contribution Agreement, dated July 24, 2007 (the "Asset Purchase Agreement"), under which the Company sold and contributed the Transferred Assets to LMC in exchange for the 69.25% membership interest in LMC, a cash purchase price of \$11,095,000, and the assumption by LMC of the Assumed Liabilities. Under the Asset Purchase Agreement, the Company agreed that it would not compete or assist other competitors in the non-coatings amorphous or semi-amorphous metallic materials industry. Pursuant to the Asset Purchase Agreement, the Company and LMC also entered into a Services Agreement, under which the Company will continue to provide certain transition services for LMC following the closing of the Transaction, and a License and Technical Support Agreement, under which the Company and LMC entered into certain cross-licenses and covenants relating to the post-closing operation of their respective businesses. Also under the Asset Purchase Agreement, LMC entered into an Employment Agreement with Larry Buffington, who will also serve as President and Chief Executive Officer of LMC. The Employment Agreement has a 5-year term, provides for an annual base salary of \$250,000, and provides for potential performance bonuses of up to 50% of Mr. Buffington's base salary. The Employment Agreement provides that, in the event that Mr. Buffington's employment is terminated by LMC in a "Termination Without Cause" (as defined in the agreement), then Mr. Buffington will be entitled to severance in the amount of 12-months salary continuation (or salary until the end of the 5-year term if less than 12 months is left in the term).

In the Transaction, LMC also entered into a Securities Purchase Agreement, dated July 24, 2006 (the "Securities Purchase Agreement"), with C₃ Capital Partners, L.P. ("C₃"), C₃ Capital Partners II, L.P. ("C₃ I"), and with C₃, the "C₃ entities"), and Liquidmetal Coatings Solutions, LLC, a wholly owned subsidiary of LMC that will operate the thermal spray coatings business ("LMCS"). Pursuant to the Securities Purchase Agreement, LMC sold and issued to the C₃ entities subordinated promissory notes in the aggregate principal amount of \$5.53 million (the "Notes") and a 19% membership interest in LMC. Under the Securities Purchase Agreement, the C₃ entities have the right, beginning on the 5th anniversary of the closing of the Transactions (or, if earlier, upon a default by LMC under the Notes or Securities Purchase Agreement) to require LMC to purchase the C₃ entities' membership interests in LMC for a purchase price equal to their pro rata portion of the greater of (i) the appraised fair market value of LMC or (ii) six times LMC's trailing 12-month earnings before interest taxes, depreciation, and amortization, less funded debt.

The Notes have a maturity date of July 20, 2012 with no required principal payments before maturity other than upon specified triggering events, such as a change in control of LMC. Interest accrues under the Notes at a rate of 14% per anum, with 12% interest being payable monthly beginning September 2007 and the remaining 2% interest being payable at maturity of the Notes. In connection with the Securities Purchase Agreement and the Notes, the Company and LMC entered into pledge agreements with the C_3 entities in which the Company pledged its membership interest in LMC to secure the obligations under the notes and LMC pledged its membership interests in LMCs to secure their obligations under the Notes.

Loan Agreement with Bank Midwest, N.A.

As part of the Transaction, the Company entered into a Loan Agreement, dated July 24, 2007 (the "Loan Agreement"), with Bank Midwest, N.A. ("Midwest"). The Loan Agreement provides for total loan availability of \$5.5 million, consisting of a \$4 million term loan and a revolving loan of up to \$1.5 million. The term loan portion of the Loan Agreement is evidenced by a Term Note (the "Term Note") issued by LMC to Midwest in the principal amount of \$4 million and having a maturity date of July 20, 2011. The revolving loan portion of the Loan Agreement is evidenced by a Revolving Note (the "Revolving Note") issued by LMC to Midwest of up to \$1.5 million and having a maturity date of July 20, 2008. Borrowing availability under the Revolving Note is based in a percentage of LMC's eligible receivables and eligible inventory, and the initial advance under the Revolving Note was \$1.0 million. LMC's obligations under the Loan Agreement are secured by a blanket security interest in all of LMC's and LMCS's assets, and pursuant to an intercreditor agreement between the C₃ entities and Midwest, Midwest's security interest in the assets is senior to C₃'s security interest in the same assets.

Interest on the Term Note accrues on the outstanding principal amount at the rate of 8.48% per annum. LMC is required to make monthly payments of principal and interest under the Term Note, with monthly payments of (i) \$62,500 during months 1 through 12, (ii) \$83,333.33 during months 13 through 36, and (iii) \$104,166.67 during months 37 through 48. All remaining principal and interest shall be due and payable upon the maturity date. If the Term Note is prepaid within the first twenty-four (24) months after July 24, 2007, LMC will pay a prepayment penalty of 1% of the outstanding principal balance, together with accrued interest. After the twenty-four month period has lapsed, LMC will have the right to prepay the Term Note without penalty. Interest accrues on the Revolving Note at the prime rate of interest. LMC will make monthly interest payments on the Revolving Note until July 20, 2008, at which point LMC will pay all remaining principal and interest.

3

LMC Operating Agreement

As a part of the Transaction and immediately prior to the transferred Assets and Assumed Liabilities pursuant to the Asset Purchase Agreement, the Company, the C₃ entities, Larry Buffington, CRESO Capital Partners, and LMC entered into an Operating Agreement, dated July 24, 2007, relating to LMC (the "Operating Agreement"). Under the Operating Agreement, LMT was issued its 69.25% membership interest in LMC, the C₃ funds were issued their collective 19% membership interest, Mr. Buffington was issued his 10% membership interest (subject to repurchase a various prices depending upon his length of employment with LMC), and CRESO Capital Partners was issued a 1.75% membership interest. The Operating Agreement provides that LMC will be managed by a Board of Managers comprised of one manager designated by the Company, one manager designated by C₃, and Larry Buffington (but Mr. Buffington will only serve as a manager while he serves as President and CEO of the Company). The Operating Agreement designates Larry Buffington as LMC's initial President and CEO.

Item 2.01. Completion of Disposition of Assets.

The information included in Item 1.01 of this Form 8-K is hereby incorporated by reference into this Item 2.01.

Item 2.03. Creation of a Direct Financial Obligation.

The information included in Item 1.01 of this Form 8-K is hereby incorporated by reference into this Item 2.03 to the extent that such information would be required to be disclosed pursuant to Item 2.03 of Form 8-K.

Item 7.01. Regulation FD Disclosure.

The following information is being furnished under Item 7.01 of Form 8-K: Press release, dated July 26, 2007, announcing the Transaction. A copy of this press release is attached as Exhibit 99.1 to this Form 8-K.

Item 9.01. Financial Statements and Exhibits.

See the Exhibit Index set forth below for a list of exhibits included with this Form 8-K.

Signature

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunder duly authorized.

LIQUIDMETAL TECHNOLOGIES, INC.

By: /s/ Larry Buffington

Larry Buffington, President and Chief Executive Officer

Date: July 27, 2007

Evhibi

5

EXHIBIT INDEX

Number	Document Description
2.1	Asset Purchase and Contribution Agreement, dated July 24, 2007 between Company and Liquidmetal Coatings, LLC. (includes Liquidmetal Coatings, LLC Operating Agreement).
2.2	Loan Agreement, dated July 24, 2007 by and among Liquidmetal Coatings, LLC, Liquidmetal Coatings Solutions, LLC and Bank Midwest, N.A.
2.3	Securities Purchase Agreement, dated July 24, 2007 by and among Liquidmetal Coatings, LLC, C3 Capital Partners, L.P., C3 Capital Partners II, L.P. and Liquidmetal Coatings Solutions, LLC.
99.1	Press Release, dated July 26, 2007
	6

Exhibit 2.1

ASSET PURCHASE AND CONTRIBUTION AGREEMENT

by and among

LIQUIDMETAL TECHNOLOGIES, INC.

and

LIQUIDMETAL COATINGS, LLC.

July 24, 2007

ASSET PURCHASE AND CONTRIBUTION AGREEMENT

TABLE OF CONTENTS

		THE OF CONTENTS	
1.	PURCHASE, SAI 1.1. 1.2.	E AND CONTRIBUTION OF ASSETS Assets to be Transferred Excluded Assets	1 1 3
2.	ASSUMPTION O 2.1. 2.2.	F LIABILITIES Liabilities to be Assumed Liabilities Not to be Assumed	3 3 4
3.	ISSUANCE OF L 3.1. 3.2. 3.3. 3.4. 3.5.	MC LLC MEMBERSHIP INTERESTS AND PAYMENT OF CASH PURCHASE PRICE Consideration for Contributed Assets Purchase Price for Purchased Assets Payment of Purchase Price Purchase Price Adjustment Tax Allocation	4 4 4 4 5 7
4.	REPRESENTATION 4.1. 4.2. 4.3. 4.4.	ONS AND WARRANTIES OF LMT Corporate Authority Title to and Condition of Properties Incorporation of Securities Purchase Agreement Representations and Warranties	7 7 7 8 8
5.	REPRESENTATIO 5.1. 5.2.	ONS AND WARRANTIES OF LMC Corporate Authority	8 8 8
6.	COVENANTS OF 6.1. 6.2. 6.3. 6.4. 6.5.	CLMT Conduct of Business Pending the Closing Consents Other Action Cooperation Restrictions on LMT	9 9 9 10 10
7.	COVENANTS OF 7.1. 7.2.	CLMC Restrictions on LMC Post-Closing Access to LMC Documents and Records	11 11 12
8.	CONDITIONS PF 8.1. 8.2. 8.3.	RECEDENT TO LMC'S OBLIGATIONS Representations and Warranties True on the Closing Date No Business Material Adverse Effect Compliance With Agreement	13 13 13 13
	8.4. 8.5. 8.6. 8.7. 8.8. 8.9. 8.10. 8.11. 8.12. 8.13.	Absence of Litigation Senior Credit Facility Transactions Contemplated by Securities Purchase Agreement LMC Operating Agreement Escrow Agreement Buffington Employment Agreement Services Agreement License and Technical Support Agreement Opinions of PCE Consents and Approvals No Late Payables	13 13 13 13 14 14 14 14 14 14
9.	CONDITIONS PR 9.1. 9.2. 9.3. 9.4. 9.5. 9.6. 9.7. 9.8. 9.9. 9.10. 9.11.	Representations and Warranties True on the Closing Date Compliance With Agreement Absence of Litigation Senior Credit Facility Transactions Contemplated by Securities Purchase Agreement LMC Operating Agreement Escrow Agreement Buffington Employment Agreement Services Agreement License and Technical Support Agreement Opinions of PCE Consents and Approvals	14 15 15 15 15 15 15 15 15 16 16
10.	SURVIVAL; INDI 10.1. 10.2. 10.3. 10.4. 10.5.	EMNIFICATION LMT's Obligation to Indemnify LMC Obligation to Indemnify Procedures Relating to Indemnification Limitations on Indemnification Characterization of Indemnification Payments	16 16 16 17 17
11.	CLOSING 11.1. 11.2.	Documents to be Delivered by LMT Documents to be Delivered by LMC	18 18 19
12.	TERMINATION 12.1. 12.2.	Right of Termination Without Breach Termination for Breach	20 20 20 20
13.	MISCELLANEOU 13.1. 13.2. 13.3. 13.4. 13.5.	Further Assurance Disclosures and Announcements Assignment; Parties in Interest Equitable Relief Governing Law	21 21 21 21 21 21 21
		ii	
	13.6. 13.7. 13.8. 13.9. 13.10.	Amendment and Modification Notice Expenses Entire Agreement Counterparts Headings	22 22 23 23 23 23

Disclosure Schedule

Real Property Leases Schedule 1.1(a) Schedule 1.1(b) Tangible Personal Property Schedule 1.1(c) Personal Property Leases Schedule 1.1(d) Insurance Policies Schedule 1.1(e) Contracts Schedule 1.1(f) Division Intellectual Property Schedule 2.0() Assumed Contracts

Tax Allocation Schedule 5.0 Foreign Corporation Qualification Schedule 1.1(g)

Schedule 3.0(a) Pre-Closing Liens

Schedule 3.0(b) Post-Closing Liens

Exhibit B Form of Escrow Agreement

Exhibit A

follows:

Exhibit C Form of Buffington Employment Agreement

Exhibit D Form of Services Agreement

Form of License and Technical Support Agreement Exhibit E

Exhibit F Form of Promissory Note for Upward Purchase Price Adjustment

Form of LMC Limited Liability Company Agreement

iii

Exhibits

ASSET PURCHASE AND CONTRIBUTION AGREEMENT

THIS ASSET PURCHASE AND CONTRIBUTION AGREEMENT (this "Agreement") is made and effective as of July 24, 2007 by and among LIQUIDMETAL TECHNOLOGIES, INC., a Delaware corporation ("LMT"), and LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company ("LMC").

WHEREAS, LMT operates a business division that markets and sells metallic industrial coatings and powders under the Liquidmetal® Coatings and/or ArmacorTM Coatings brand names (the "Business"); and

WHEREAS, LMT desires to sell and contribute substantially all of its operating assets that are used, held for use or acquired or developed for use primarily in the Business to LMC, and LMC desires to purchase such assets and accept such contribution. As used in this Agreement, "Division" means LMT with respect to the Business; and

WHEREAS, LMC has entered into a Securities Purchase Agreement of even date herewith (the "Securities Purchase Agreement") with C3 Capital Partners, L.P., a Delaware limited partnership ("C3"), and C3 Capital Partners II, L.P., a Delaware limited partnership ("C3 II"), pursuant to which C3 and C3 II have agreed to purchase from LMC 14% Subordinated Notes (the "Notes") and membership interests (the "Membership Interests") in LMC.

NOW, THEREFORE, in consideration of the foregoing and the representations, warranties, covenants, agreements and conditions set forth in this Agreement, and intending to be legally bound, the Parties agree as

PURCHASE, SALE AND CONTRIBUTION OF ASSETS

Assets to be Transferred

Upon the terms and subject to the conditions set forth in this Agreement, LMT shall, on the Closing Date, contribute or sell, transfer, convey, assign, and deliver to LMC, and LMC shall, on the Closing Date, accept such contribution and purchase from LMT, all of the following assets used in the Business, together with all rights and privileges associated with such assets, but not including the Excluded Assets (collectively, the "Transferred Assets"):

- (a) Leased Real Property. The leases of real property identified in Schedule 1.1(a) (the "Leased Real Property").
- (b) Tangible Personal Property. The machinery, equipment, furniture, computer hardware, supplies, spare parts, materials, vehicles and other items of tangible personal property identified in Schedule 1.1(b).
- Personal Property Leases. The leases of machinery, equipment, vehicles, furniture and other personal property identified in Schedule 1.1(c).
- (d) Insurance Policies. All LMT's rights to benefits under the insurance policies identified in Schedule 1.1(d) with respect to claims directly relating to the Business.
- Contracts. All LMT's rights in, to and under all contracts, purchase orders and sales orders identified in Schedule 1.1(e) (collectively, the "Contracts"). To the extent that any Contract for which assignment to LMC is provided herein is not assignable without the consent of another party, this Agreement shall not constitute an assignment or an attempted assignment thereof if such assignment or attempted assignment would constitute a breach thereof. LMT and LMC agree to use their reasonable best efforts (without any requirement on the part of LMC to pay any money or agree to any change in the terms of any such Contract) to obtain the consent of such other party to the assignment of any such Contract to LMC in all cases in which such consent is or may be required for such assignment. If any such consent shall not be obtained, LMT agrees to cooperate with LMC in any reasonable arrangement designed to provide for LMC the benefits intended to be assigned to LMC under the relevant Contract, including enforcement at the cost and for the account of LMC of any and all rights of LMT against the other party thereto arising out of the breach or cancellation thereof by such other party or otherwise.
- Coatings Intellectual Property Rights. All LMT's interest in any Intellectual Property Rights used or held for use primarily in the Division, including without limitation the Intellectual Property Rights identified in Schedule 1.1(f) (collectively, referred to as the "Coatings Intellectual Property"). For purposes of this Agreement, "Intellectual Property Rights" means all: (i) patents, patent applications, patent disclosures and inventions; (ii) trademarks, service marks, trade dress, trade names, internet domain names, logos and corporate names and registrations and applications for registration thereof, together with all of the goodwill associated therewith; (iii) copyrights (registered or unregistered) and copyrightable works and registrations and applications for registration thereof; (iv) mask works and registrations and applications for registration thereof; (v) computer software, data, data bases and documentation thereof; and (vi) trade secrets and other confidential information (including, without limitation, ideas, formulas, compositions, inventions (whether patentable or unpatentable and whether or not reduced to practice), knowhow, manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, proposals, technical data, copyrightable works, financial and marketing plans and customer and supplier lists and information).
 - Permits. All licenses, permits, approvals, certifications and listings of the Business. (g)
 - (h) Literature. All sales literature, promotional literature, catalogs and similar materials used or held for use in the Business
- Records and Files. All books, records, files or other embodiments of information of the Business, including all diagrams, prints, surveys, drawings, customer data, training materials, operations manuals and other records, data and materials, whether relating to past or current operations (provided that LMT shall have the right to retain copies thereof).

2

- Notes and Accounts Receivable. All notes, drafts and accounts receivable aged sixty (60) days or less (including unbilled receivables) derived from the Business and the full benefit of all security for such rights (j) to payment (collectively, the "Transferred Receivables").
- (k) Inventory. Inventories of raw materials, work-in-process and finished goods (including all such in transit), and service and repair parts, supplies and components held for resale in the Business by LMT on the Closing Date, together with related packaging materials (collectively the "Inventory").
 - General Intangibles. All prepaid items, causes of action arising out of occurrences before or after the Closing, and other intangible rights and assets, relating to the Business.

12 Excluded Assets

Notwithstanding anything to the contrary in Section 1.1, LMT shall not contribute or sell, convey, assign, transfer or deliver to LMC, and LMC shall not purchase or acquire from LMT, the following (collectively, "Excluded Assets"):

- Trademarks. All of LMT's rights in, to and under the Liquidmetal® trademark and any and all derivations thereof. (a)
- (b) Cash. Subject to the provisions of Section 3.4 hereof, any cash and cash equivalents of the Division.
- (c) Aged Over 60 Days. All notes, drafts and accounts receivable aged more than sixty (60) days derived from the Business and the full benefit of all security for such rights to payment ("Excluded" Receivables")
 - Certain Records, Any books, records, files or other embodiments of information relating exclusively to any Excluded Asset.

ASSUMPTION OF LIABILITIES

2.1. <u>Liabilities to be Assumed.</u>

As used in this Agreement, the term "<u>Liability</u>" shall mean and include any direct or indirect indebtedness, guaranty, endorsement, claim, loss, damage, deficiency, cost, expense, obligation or responsibility, fixed or unfixed, known or unknown, asserted or unasserted, liquidated or unliquidated, secured or unsecured. Subject to the terms and conditions of this Agreement, on the Closing Date, LMC shall assume and agree to perform and discharge the following, and only the following Liabilities of the Division (collectively the "<u>Assumed Liabilities</u>"):

(a) Final Closing Balance Sheet Liabilities. The accounts payable and accrued Liabilities of the Division reflected or reserved against on the Final Closing Balance Sheet (as hereinafter defined), but only in the amounts so reflected or reserved.

3

(b) Contractual Liabilities. The Division's Liabilities arising from and after the Closing Date under and pursuant to the Contracts specifically described in any of Schedules 1.1(a), 1.1(c) or 1.1(e).

The Contracts described in subsection 2.1(b) above are hereinafter collectively described as the "Assumed Contracts."

(c) Liabilities Under Permits and Licenses. The Division's Liabilities arising from and after the Closing Date under any permits or licenses described in 1.1(g) and assigned to LMC at the Closing.

Liabilities Not to be Assumed.

Except as and to the extent specifically set forth in Section 2.1, LMC is not assuming any Liabilities of LMT and all such Liabilities shall be and remain the responsibility of LMT.

3. ISSUANCE OF LMC LLC MEMBERSHIP INTERESTS AND PAYMENT OF CASH PURCHASE PRICE

3.1. Consideration for Contributed Assets.

In consideration of the Transferred Assets contributed to LMC, subject to the terms and conditions of this Agreement, at Closing LMC shall issue to LMT limited liability company membership interests of LMC ("Membership Interests") equal to 69.25% of the total outstanding Membership Interests of LMC outstanding immediately after giving effect to such issuance. The Membership Interests shall be issued pursuant to the form of Limited Liability Company Agreement attached as Exhibit "A".

3.2. Purchase Price for Purchased Assets

In consideration of the Transferred Assets purchased by LMC, subject to the terms and conditions of this Agreement, LMC shall assume the Assumed Liabilities and authorize its lenders to pay to the Escrow Agent pursuant to the Escrow Agreement cash equal to the sum of Eleven Million Ninety Five Thousand and No/100 Dollars (\$11,095,000) as adjusted upward or downward by the Upward Purchase Price Adjustment or Downward Purchase Price Adjustment, if any (the "Purchase Price"). The term "Escrow Agreement" shall mean an Escrow Agreement in substantially the form attached hereto as Exhibit B, and "Escrow Agent" shall have the meaning set forth in the Escrow Agreement in Substantially the form attached hereto as Exhibit B, and "Escrow Agent" shall have the meaning set forth in the Escrow Agreement in Substantially the form attached hereto as Exhibit B, and "Escrow Agent" shall have the meaning set forth in the Escrow Agreement in Substantially the form attached hereto as Exhibit B, and "Escrow Agent" shall have the meaning set forth in the Escrow Agreement in Substantially the form attached hereto as Exhibit B, and "Escrow Agent" shall have the meaning set forth in the Escrow Agreement in Substantially the form attached hereto as Exhibit B, and "Escrow Agent" shall have the meaning set forth in the Escrow Agreement in Substantially the form attached hereto as Exhibit B, and "Escrow Agent" shall have the meaning set forth in the Escrow Agreement in Substantial shall be added to the Subs

3.3. Payment of Purchase Price.

The Purchase Price shall be paid by LMC as follows:

(a) Assumption of Liabilities. At the Closing, LMC shall deliver to LMT such documents and instruments as are reasonably required to evidence the assumption of the Assumed Liabilities.

4

- (b) Cash to LMT. At the Closing, LMC shall direct its lenders to deliver to the Escrow Agent pursuant to the Escrow Agreement cash in an amount equal to Eleven Million Ninety Five Thousand and No/100 Dollars (\$11,095,000), plus or minus, as the case may be, the amount of the Estimated Purchase Price Adjustment.
- (c) Adjustment of Final Cash Purchase Price. On or before the fifth business day following the final determination of the Final Closing Balance Sheet (as hereinafter defined) (such date being hereinafter referred to as the "Settlement Date"), either (i) LMT shall pay to LMC the amount of any Downward Differential in cash or (ii) LMC shall pay to LMT the amount of any Upward Differential by a promissory note in the principal amount of the Upward Differential and in substantially the form attached hereto as Exhibit F. (an "Adjustment Note").
- (d) <u>Issuance of LMC Membership Interest</u>. At Closing, LMC shall issue to LMT Membership Interests of LMC equal to 69.25% of the total outstanding Membership Interests of LMC outstanding immediately after giving effect to such issuance.

3.4. Purchase Price Adjustment.

- (a) Pursuant to the provisions of this Section 3.4, the Purchase Price shall be decreased, dollar for dollar, if the following calculation yields a negative number or increased, dollar for dollar, if the following calculation yields a positive number:
 - (i) Current assets (excluding the Thermal Spray Assets and the Excluded Receivables) minus current liabilities (subject to Section 8.4 hereof and excluding the Thermal Spray Liabilities), as of the Closing Date, as set forth on the Final Closing Balance Sheet, minus:
 - (ii) \$800,000
 - (the negative difference, if any, the "<u>Downward Purchase Price Adjustment</u>" and the positive difference, if any, the "<u>Upward Purchase Price Adjustment</u>"). For purposes of this Agreement, the term "<u>Thermal Spray Assets</u>" refers to the assets acquired by LMT on June 1, 2007 from Foster Wheeler Energy Services, Inc., and "<u>Thermal Spray Liabilities</u>" means the liabilities assumed by LMT on June 1, 2007 from Foster Wheeler Energy Services, Inc.
 - (b) On the Closing Date, LMT shall prepare and deliver to LMC a good faith estimate of the balance sheet of the Division as of the close of business on the Closing Date (the "Closing Balance Sheet Estimate").
- (c) If based on the Closing Balance Sheet Estimate the calculation in Section 3.4(a) (the "Estimated Calculation") would result in a Downward Purchase Price Adjustment, then the cash consideration to be delivered by LMC pursuant to Section 3.3(b) shall be reduced by the amount of such Downward Purchase Price Adjustment. If the Estimated Calculation would result in an Upward Purchase Price Adjustment, then in addition to the consideration set forth in Section 3.3, LMC shall deliver an Adjustment Note in favor of LMT in the amount of

5

such Upward Purchase Price Adjustment which shall be due and payable in full forty-five (45) days after the Closing Date.

- (d) As promptly as practical, and in any event not more than 90 days after the Closing Date, LMC shall prepare and deliver to LMT (i) its statement setting forth its good faith determination of the balance sheet of the Division as of the close of business on the Closing Date (the "Closing Balance Sheet Statement"), and (ii) a calculation of the amount of the Upward Purchase Price Adjustment or Downward Purchase Price Adjustment or Downward Purchase Price Adjustment ("LMC's Calculation"). In connection with the preparation of the Closing Balance Sheet Statement, LMT shall cooperate, and cause its agents and representatives to cooperate, with LMC and its agents and representatives. The parties agree that Tarsus CFO Services, LLC shall assist in the preparation of the Closing Balance Sheet Statement on behalf of LMC.
- (e) The Closing Balance Sheet Statement and LMC's Calculation shall be conclusive and binding on the parties and the Closing Balance Sheet Statement shall be deemed the Final Closing Balance Sheet, unless LMT delivers to LMC a notice specifying in reasonable detail the respects in which it objects to the Closing Balance Sheet Statement (the "Objection Notice") within 30 days following LMT's receipt thereof. In connection with the LMT's review of the Closing Balance Sheet Statement, LMC shall cooperate, and cause its agents and representatives, to cooperate with LMT and its agents and representatives and shall provide such agents and representatives access to such books, records and information relating to the Closing Balance Sheet Statement, as may be reasonably requested from time to time. If LMT delivers a timely Objection Notice, LMC, LMT and their respective accountants shall negotiate among themselves for a period of 45 days thereafter in an attempt to resolve the objections stated therein. If no resolution is reached within such 45-day period, then LMC and LMT shall select an independent accounting firm mutually agreeable to both parties (the "Accounting Firm") to review LMT's books and related information to determine (the "Independent Determination") the balance sheet of the Division as of the close of business on the Closing Date (the "Final Closing Balance Sheet") and to calculate the amount of the Upward Purchase Price Adjustment (the "Definitive Calculation"), if any. The Accounting Firm shall be supplied such information, books and records and access to such individuals as it may reasonably require from any party. The Accounting Firm to deliver the Final Closing Balance Sheet and the Definitive Calculation within 30 days of its selection. The Final Closing Balance Sheet and the Definitive Calculation within the Independent Determination.
- (f) If the amount of LMC's Calculation or the Definitive Calculation, as the case may be, is greater (the "<u>Upward Differential</u>") than the amount of the Estimated Purchase Price Adjustment, then LMC shall pay to LMC an amount equal to the Upward Differential. If the amount of LMC's Calculation or the Definitive Calculation, as the case may be, is less (the "<u>Downward Differential</u>") than the amount of the Estimated Purchase Price Adjustment, then LMT shall pay to LMC an amount equal to the Downward Differential.

3.5. Tax Allocation.

The allocation of the Transferred Assets between the Transferred Assets to be sold by LMT to LMC and the Transferred Assets to be contributed by LMT to LMC is set forth on Schedule 3.5, and the allocation of the Purchase Price among the Transferred Assets to be sold is also set forth on Schedule 3.5. LMT and LMC will follow and use such allocation in all tax returns, filings or other related reports made by them to any governmental agencies.

4. REPRESENTATIONS AND WARRANTIES OF LMT

LMT makes the following representations and warranties to LMC, each of which is true and correct on the date hereof, shall remain true and correct to and including the Closing Date, shall be unaffected by any investigation heretofore or hereafter made by LMC, or any knowledge of LMC other than as specifically disclosed in the Disclosure Schedules delivered to LMC at the time of the execution of this Agreement, and shall survive the Closing of the transactions provided for herein.

4.1. <u>Corporate</u>

- (a) Organization. LMT is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- (b) <u>Corporate Power</u>. LMT has all requisite corporate power and authority to carry on the Business as and where such is now being and as heretofore conducted, to enter into this Agreement and the other documents and instruments to be executed and delivered by LMT pursuant hereto and to carry out the transactions contemplated hereby and thereby.
- (c) Qualification. LMT is duly licensed or qualified to do business as a foreign corporation, and is in good standing, in each jurisdiction wherein the character of the properties owned or leased by the Division, or the nature of the Business, makes such licensing or qualification necessary, except where the failure to qualify would not result or constitute any event, circumstance, change or effect that is or would reasonably be likely to be materially adverse to the Business or the assets, liabilities, condition (financial or otherwise) or results of operations of the Division (a "Business Material Adverse Effect"). The states in which the Division is licensed or qualified to do business are listed in Schedule 4.1(c).

4.2. Authority

The execution and delivery of this Agreement and the other documents and instruments to be executed and delivered by LMT pursuant hereto and the consummation of the transactions contemplated hereby and thereby have been duly authorized by LMT. No other or further corporate act or proceeding on the part of LMT is necessary to authorize this Agreement or the other documents and instruments to be executed and delivered by LMT pursuant hereto or the consummation of the transactions contemplated hereby and thereby. This Agreement constitutes, and when executed and delivered, the other documents and instruments to be executed and delivered by LMT pursuant hereto will constitute, valid binding agreements of LMT, enforceable in accordance with their respective terms, except as such enforceability may

7

be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other similar laws now or hereafter in effect relating to creditors' rights and by general equitable principles (regardless of whether enforceability is considered in a proceeding in equity or at law).

4.3. <u>Title to and Condition of Properties.</u>

LMT has good and marketable title to all the Transferred Assets, free and clear of all mortgages, liens (statutory or otherwise), security interests, claims, pledges, licenses, equities, options, conditional sales contracts, assessments, levies, easements, covenants, reservations, restrictions, rights-of-way, exceptions, limitations, charges or encumbrances of any nature whatsoever (collectively, "Liens") except those described in Schedule 4.3(a), and municipal and zoning ordinances and easements for public utilities, none of which interfere with the use of the Transferred Assets as currently utilized or adversely affect the marketablity of the Transferred Assets. At Closing, LMC will receive good and marketable title to all the Transferred Assets, free and clear of all Liens of any nature whatsoever except those described in Schedule 4.3(b).

4.4. Incorporation of Securities Purchase Agreement Representations and Warranties.

The representations and warranties made by LMC in the Sections 6.5, 6.6, 6.7, 6.8, 6.10, 6.11, 6.12, 6.13, 6.14, 6.15, 6.17, 6.18, 6.19, 6.21, 6.23, 6.24 and 6.25 of the Securities Purchase Agreement (the "Incorporated SPA Representations and Warranties") are incorporated herein by reference as representations and warranties of LMT as if fully set forth herein and LMT hereby adopts and ratifies all of the Incorporated SPA Representations and Warranties. For purposes hereof, any reference to "LMC" or "Company" in the SPA Representations and Warranties shall be deemed to refer to the Division.

. REPRESENTATIONS AND WARRANTIES OF LMC

LMC makes the following representations and warranties to LMT, each of which is true and correct on the date hereof, shall remain true and correct to and including the Closing Date, shall be unaffected by any investigation heretofore or hereafter made by LMT or any notice to LMT, and shall survive the Closing of the transactions provided for herein.

5.1. Corporate

- (a) Organization. LMC is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Delaware.
- (b) <u>Corporate Power.</u> LMC has all requisite company power to enter into this Agreement and the other documents and instruments to be executed and delivered by LMC and to carry out the transactions contemplated hereby and thereby.

5.2. Authority

The execution and delivery of this Agreement and the other documents and instruments to be executed and delivered by LMC pursuant hereto and the consummation of the transactions contemplated hereby and thereby have been duly authorized by LMC. No other

8

company act or proceeding on the part of LMC or its members is necessary to authorize this Agreement or the other documents and instruments to be executed and delivered by LMC pursuant hereto or the consummation of the transactions contemplated hereby and thereby. This Agreement constitutes, and when executed and delivered, the other documents and instruments to be executed and delivered by LMC pursuant hereto will constitute, valid and binding agreements of LMC, enforceable in accordance with their respective terms, except as such may be limited by bankruptcy, insolvency, reorganization or other laws affecting creditors' rights generally, and by general equitable principles.

6. COVENANTS OF LMT

operation

LMC

LMT covenants and agrees as follows:

6.1. <u>Conduct of Business Pending the Closing</u>.

From the date hereof until the Closing, except as otherwise approved in writing by LMC:

- (a) No Changes. LMT will carry on the Business in materially the same manner as heretofore and will not make or institute any material changes in its methods of purchase, sale, management, accounting or
- (b) <u>Maintain Organization</u>. LMT will take such action as may be necessary to maintain, preserve, renew and keep in favor and effect the existence, rights and franchises of the Division and will use its best efforts to preserve the business organization of the Division intact, to keep available to LMC the present officers and employees, and to preserve for LMC its present relationships with suppliers and customers and others having business relationships with the Division.
- (c) Maintenance of Insurance. LMT shall maintain all of the insurance with respect to the Business in effect as of the date hereof and shall procure such additional insurance as shall be reasonably requested by
 - (d) <u>Maintenance of Property</u>. LMT shall use, operate, maintain and repair all property of the Division in a normal business manner.
- (e) No Negotiations. LMT will not directly or indirectly (through a representative or otherwise) solicit or furnish any information to any prospective buyer, commence, or conduct presently ongoing, negotiations with any other party or enter into any agreement with any other party concerning the sale of the Division, the Division's assets or Business (an "acquisition proposal"), and LMT shall immediately advise LMC of the receipt of any acquisition proposal with respect to the Business.

6.2. Consents

LMT will use its best efforts prior to Closing to obtain all consents necessary for the consummation of the transactions contemplated hereby.

9

6.3. Other Action.

6.4. <u>Cooperation</u>.

LMT and LMC agree to cooperate with each other in connection with, including furnishing or causing to be furnished, upon request and as promptly as practicable, such information and assistance relating to the transactions contemplated by the Agreement as is reasonably necessary for, the preparation or filing of all Tax Returns, the making of any elections related to Taxes, and the conduct of any audits, examinations, litigation, suits or other proceedings relating the Taxes, including refunds.

6.5. Restrictions on LMT

- (a) In order to protect adequately the interest of LMC in the Transferred Assets and in consideration of the consideration paid or issued by LMC to LMT for the Transferred Assets, LMT agrees that it shall not, and shall cause its Affiliates (other than LMC) not to, during the four-year period beginning on the Closing Date anywhere in the world:
 - (i) directly or indirectly, whether as an owner, shareholder, member, investor, partner, joint venturer, licensor, financier, operator, consultant, agent, distributor, independent contractor, participant, creditor or otherwise, invest in (other than ownership as a passive investor of less than five percent of the voting stock of a publicly-traded company), own, manage, operate, finance, control or participate in the ownership, management, operation, financing, control of, or act as a consultant to, be associated with, lend its or their name or any trade name to, any of its or their credit to, or otherwise render services or advice to or on behalf of, any business that engages in the manufacture, marketing, sale, distribution, or application of metallic coating material in the form of a wire or powder that is sprayed in liquid form on industrial equipment for use as a coating to protect against wear, corrosion, or abrasion;
 - (ii) directly or indirectly induce or attempt to induce any customer of LMC in relation to the Business to reduce such customer's purchases of products of the Business from LMC or its Affiliates after the Closing Date or induce or attempt to induce any supplier to LMC with respect to the Business to reduce such supplier's deliveries of materials of the Business after the Closing Date;
 - (iii) directly or indirectly, except as expressly permitted by LMC or its successors or assigns in advance in writing, solicit any employee to leave the employ of LMC or its successors and assigns; or
- (b) LMT acknowledges and agrees that LMC's remedies at law for any violation or attempted violation of LMT's obligations under this Section 6.5 would be inadequate and incomplete, and agrees that in the event of any such violation or attempted

10

violation, LMC shall be entitled to a temporary restraining order, temporary and permanent injunctions, and other equitable relief, without the necessity of posting any bond or proving any actual damage, in addition to all other rights and remedies that may be available to LMC from time to time.

- (c) If a judicial or arbitral determination is made that any of the provisions of this Section 6.5 constitutes an unreasonable or otherwise unenforceable restriction against LMT or any of its Affiliates, the provisions of this Section 6.5 shall be rendered void only to the extent that such judicial or arbitral determination finds such provisions to be unreasonable or otherwise unenforceable with respect to LMT or any Affiliate thereof. In this regard, the parties hereby agree that any judicial authority construing this Agreement shall be empowered to sever any territory or portion thereof, any prohibited business activity or any time period from the coverage of this Section 5.0 and to apply the provisions of this Section 6.5 to the remaining portion of the coverage activities and the remaining time period not so severed by such judicial or arbitral authority. The time period during which the prohibitions set forth in this Section 6.5 shall apply shall be tolled and suspended for a period equal to the aggregate time during which LMT or any Affiliate thereof (other than LMC) violates such prohibitions in any respect.
- (d) For purposes of this Agreement, the term "Affiliate" means, as to any specified person or entity, any other person or entity that directly or indirectly controls, or is under common control with, or is controlled by, such specified person or entity and, if such other person is an individual, any member of the immediate family of such individual. As used in this definition, "control" (including, with its correlative meanings, "controlled by," and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of the management or policies (whether through ownership of securities or partnership or other ownership interests, by contract, or otherwise) and "immediate family" shall mean any parent, child, grandchild, spouse, or sibling.

7. COVENANTS OF LMC

LMC covenants and agrees as follows:

7.1. Restrictions on LMC.

- (a) In order to protect adequately the interest of LMT in entering into the transactions contemplated by this Agreement and in consideration of the obligations of LMT herein, LMC agrees that it shall not, and shall cause its Affiliates (other than LMT) not to, during the four-year period beginning on the Closing Date anywhere in the world:
 - (i) directly or indirectly, whether as an owner, shareholder, member, investor, partner, joint venturer, licensor, financier, operator, consultant, agent, distributor, independent contractor, participant, creditor or otherwise, invest in (other than ownership as a passive investor of less than five percent of the voting stock of a publicly-traded company), own, manage, operate, finance, control or participate in the ownership, management, operation, financing, control of, or act as a consultant to, be associated with, lend its or their name or any trade name to,

11

any of its or their credit to, or otherwise render services or advice to or on behalf of, any business that engages manufacture, marketing, sale, distribution, or application of amorphous or semi-amorphous metallic materials (or composite materials including amorphous or semi-amorphous metallic materials) or products made therefrom, but excluding the Business (the "LMT Business");

- (ii) directly or indirectly induce or attempt to induce any customer of LMT in relation to the LMT Business to reduce such customer's purchases of products of the LMT Business from LMT or its Affiliates after the Closing Date or induce or attempt to induce any supplier to LMT with respect to the LMT Business to reduce such supplier's deliveries of materials of the LMT Business after the Closing Date;
- (iii) directly or indirectly, except as expressly permitted by LMT or its successors or assigns in advance in writing, solicit any employee to leave the employ of LMT, its Affiliates (other than LMC), or their respective successors and assigns; or
- (b) LMC acknowledges and agrees that LMT's remedies at law for any violation or attempted violation of the LMC's obligations under this Section 7.1 would be inadequate and incomplete, and agrees that in the event of any such violation or attempted violation, LMT shall be entitled to a temporary restraining order, temporary and permanent injunctions, and other equitable relief, without the necessity of posting any bond or proving any actual damage, in addition to all other rights and remedies that may be available to LMT from time to time.
- (c) If a judicial or arbitral determination is made that any of the provisions of this Section 7.1 constitutes an unreasonable or otherwise unenforceable restriction against LMC or any of its Affiliates, the provisions of this Section 7.1 shall be rendered void only to the extent that such judicial or arbitral determination finds such provisions to be unreasonable or otherwise unenforceable with respect to LMC or any Affiliate thereof. In this regard, the parties hereby agree that any judicial authority construing this Agreement shall be empowered to sever any territory or portion thereof, any prohibited business activity or any time period from the coverage of this Section 7.1 and to apply the provisions of this Section 7.1 to the remaining portion of the covered territory, the remaining business activities and the remaining time period not so severed by such judicial or arbitral authority. The time period during which the prohibitions set forth in this Section 7.1 shall apply shall be tolled and suspended for a period equal to the aggregate time during which an LMC violates such prohibitions in any respect.

7.2. <u>Post-Closing Access to LMC Documents and Records.</u>

After the Closing, LMC shall cooperate, and cause its agents and representatives to cooperate, with LMT and its agents and representatives to provide such agents and representatives access to such books, records and information relating to LMC, as may be reasonably requested from time to time.

12

8. CONDITIONS PRECEDENT TO LMC'S OBLIGATIONS

Each and every obligation of LMC to be performed on the Closing Date shall be subject to the satisfaction prior to or at the Closing of each of the following conditions:

8.1. <u>Representations and Warranties True on the Closing Date.</u>

Each of the representations and warranties made by LMT in this Agreement, and the statements contained in the Disclosure Schedules or in any instrument, list, certificate or writing delivered by LMT pursuant to this Agreement, shall be true and correct when made and shall be true and correct at and as of the Closing Date as though such representations and warranties were made or given on and as of the Closing Date, except for any changes permitted by the terms of this Agreement or consented to in writing by LMC.

8.2. No Business Material Adverse Effect.

No events, circumstances or developments shall have occurred since the date of the Recent Balance Sheet which, individually or in the aggregate, results or would reasonably be expected to result in a Business Material Adverse Effect.

8.3. <u>Compliance With Agreement.</u>

LMT shall have performed and complied with all of its agreements and obligations under this Agreement which are to be performed or complied with by LMT prior to or on the Closing Date, including the delivery of the closing documents specified in Section 11.1.

8.4. Absence of Litigation

No Litigation shall have been commenced or threatened, and no investigation by any Government Entity shall have been commenced, against LMC, the Division or any of the Affiliates, officers or directors of any of them, with respect to the transactions contemplated hereby.

8.5. Senior Credit Facility.

Bank Midwest, N.A. shall have advanced or otherwise made available to LMC an aggregate of no less than \$5,000,000 and no more than \$5,500,000 pursuant to a secured credit facility on customary terms and conditions reasonably satisfactory to LMC.

8.6. Transactions Contemplated by Securities Purchase Agreement.

LMC, C3, and C3 II shall have consummated the purchase and sale of the Notes and Membership Interests pursuant to the Securities Purchase Agreement.

8.7. LMC Operating Agreement.

LMT, C3, C3 II, Larry Buffington, and CRESO Capital Partners shall have each executed and delivered to LMC a duly executed counterpart of a Limited Liability Company

13

Operating Agreement of LMC in substantially the form attached hereto as Exhibit A (the "Operating Agreement")

8.8. Escrow Agreement.

LMT, C3, C3 II, and Bank Midwest shall have each executed and delivered to LMC a duly executed counterpart of an Escrow Agreement.

8.9. <u>Buffington Employment Agreement</u>

Larry Buffington shall have delivered to LMC a duly executed counterpart of an Employment Agreement in substantially the form attached hereto as Exhibit C (the "Buffington Employment Agreement").

8.10. Services Agreement

LMT shall have delivered to LMC a duly executed counterpart of a Services Agreement in substantially the form attached hereto as Exhibit D ("Services Agreement").

8.11. <u>License and Technical Support Agreement.</u>

LMT shall have delivered to LMC a duly executed counterpart of a License and Technical Support Agreement in substantially the form attached hereto as Exhibit E (the "License Agreement").

8.12. Opinions of PCE

The Board of Directors of LMT shall have received from PCE Valuations, LLC (i) an opinion as to the fair value of the Transferred Assets and the consideration received therefore by LMT and (ii) an opinion as to the fairness, from a financial point of view to the creditors of LMT, of the transactions contemplated by this Agreement (the "PCE Opinions").

8.13. Consents and Approvals.

LMT shall have received all consents and/or approvals from third parties necessary for the consummation of the transactions contemplated by this Agreement.

8.14. <u>No Late Payables</u>.

The Business will have no notes payable or accounts payable outstanding that are past due according to the payment terms set forth on the applicable invoice.

CONDITIONS PRECEDENT TO LMT'S OBLIGATIONS

Each and every obligation of LMT to be performed on the Closing Date shall be subject to the satisfaction prior to or at the Closing of the following conditions:

14

9.1. Representations and Warranties True on the Closing Date.

Each of the representations and warranties made by LMC in this Agreement shall be true and correct when made and shall be true and correct at and as of the Closing Date as though such representations and warranties were made or given on and as of the Closing Date.

9.2. Compliance With Agreement.

LMC shall have performed and complied with all of LMC's agreements and obligations under this Agreement which are to be performed or complied with by LMC prior to or on the Closing Date, including the delivery of the closing documents specified in Section 11.2.

9.3. Absence of Litigation.

No Litigation shall have been commenced or threatened, and no investigation by any Government Entity shall have been commenced, against LMC, LMT or any of the Affiliates, officers or directors of any of them, with respect to the transactions contemplated hereby; provided that the obligations of LMT shall not be affected unless there is a reasonable likelihood that as a result of such action, suit, proceeding or investigation LMT will be unable to retain substantially all the consideration to which it is entitled under this Agreement.

Senior Credit Facility.

Bank Midwest, N.A. shall have advanced or otherwise made available to LMC an aggregate of no less than \$5,000,000 and no more than \$5,500,000 pursuant to a secured credit facility on customary terms and conditions reasonably satisfactory to LMC.

9.5. Transactions Contemplated by Securities Purchase Agreement.

LMC, C3, and C3 II shall have consummated the purchase and sale of the Notes and Membership Interests pursuant to the Securities Purchase Agreement.

9.6. <u>LMC Operating Agreement.</u>

LMC, C3, C3 II, and Larry Buffington shall have each executed and delivered to LMT a duly executed counterpart of the Operating Agreement.

9.7. <u>Escrow Agreement.</u>

LMC, C3, C3 II, and Bank Midwest shall have each executed and delivered to LMT a duly executed counterpart of an Escrow Agreement.

9.8. <u>Buffington Employment Agreement.</u>

Larry Buffington shall have delivered to LMC a duly executed counterpart of the Buffington Employment Agreement.

9.9. <u>Services Agreement</u>.

LMC shall have delivered to LMT a duly executed counterpart of a Services Agreement.

9.10. License and Technical Support Agreement.

LMC shall have delivered to LMT a duly executed counterpart of the License Agreement.

9.11. Opinions of PCE.

The Board of Directors of LMT shall have received the PCE Opinions from PCE Valuations.

9.12. Consents and Approvals.

LMT shall have received all consents and/or approvals from third parties necessary for the consummation of the transactions contemplated by this Agreement.

10. SURVIVAL; INDEMNIFICATION

10.1. LMT's Obligation to Indemnify.

Subsequent to the Closing, subject to the terms and conditions of this Article 10, LMT shall indemnify, defend and hold harmless LMC and its directors, officers, employees, agents, Affiliates and assigns (collectively, the "LMC Indemnified Persons") from and against all losses, liabilities, damages, deficiencies, costs or expenses, including interest and penalties imposed or assessed by any judicial or administrative body and reasonable attorneys' fees, whether or not arising out of third-party claims and including all amounts paid in investigation, defense or settlement of the foregoing, net of all tax savings for the foregoing (collectively, "Losses") suffered or incurred by any LMC Indemnified Person based upon, arising out of or otherwise in respect of (i) any inaccuracy in or breach of (without regard to any knowledge, dollar threshold or materiality qualification contained therein) any representation or warranty of LMT in this Agreement or in any document delivered pursuant hereto, (ii) any breach of (without regard to any knowledge, dollar threshold or materiality qualification contained herein) any covenant or agreement of LMT in this Agreement or in any document delivered pursuant hereto, and (iii) any Liability of the Division other than Assumed Liabilities.

10.2. <u>LMC Obligation to Indemnify.</u>

Subsequent to the Closing, subject to the terms and conditions of this Article 10, LMC shall indemnify, defend and hold harmless LMT, its directors, officers, employees, agents, Affiliates and assigns from and against all Losses suffered or incurred by any such Person based upon, arising out of or otherwise in respect of (i) any inaccuracy in or breach of any representation or warranty of LMC in this Agreement or in any document delivered pursuant hereto, (ii) any breach of any covenant or agreement of LMC in this Agreement or in any document delivered pursuant hereto or (iii) any Assumed Liabilities.

16

10.3. Procedures Relating to Indemnification.

- (a) An indemnified Person under Sections 10.1 or 10.2 (the "<u>Indemnified Party</u>") shall give prompt written notice to the indemnifying party (the "<u>Indemnifying Party</u>") of any Loss in respect of which such Indemnified Party is seeking indemnification under Sections 10.1 or 10.2, specifying in reasonable detail the nature of such Loss, the section or sections of this Agreement to which the Loss relates, and the amount of such Loss (or if not then determinable, its best estimate of the amount of such Loss), except that any delay or failure to so notify the Indemnifying Party shall only relieve the Indemnifying Party of its obligations hereunder to the extent, if at all, that it is prejudiced by reason of such delay or failure.
- (b) If a Loss is suffered or incurred for or on account of or arises from or in connection with any demand, claim, suit, action, cause of action, investigation or inquiry by a Person not party to this Agreement (a "Third Party Claim"), the Indemnifying Party shall assume the defense thereof, including the employment of counsel reasonably satisfactory to the Indemnified Party and the payment of all expenses. The Indemnified Party shall have the right to employ separate counsel in such Third Party Claim and participate in such defense thereof at its own expense. The Indemnifying Party shall not, without the Indemnified Party's prior written consent, which shall not be unreasonably withheld, settle or compromise any Third Party Claim or consent to the entry of any judgment with respect to any Third Party Claim. If the Indemnifying Party fails to assume the defense of any Third Party Claim within 20 business days after notice thereof, the Indemnifying Party.
- (c) With respect to any Loss (other than any Loss suffered or incurred for or on account of or arising from or in connection with any Third Party Claim), the Indemnifying Party shall have 30 business days from receipt of notice from the Indemnified Party of such Loss within which to respond thereto. If the Indemnifying Party does not respond within such 30 business day period, the Indemnifying Party shall be deemed to have accepted responsibility to make payment and shall have no further right to contest the validity of such Loss.
 - (d) The Indemnifying Party shall promptly pay the Indemnified Party any amount due under this Section 10.

10.4. <u>Limitations on Indemnification</u>

Except for fraud, as to which claims may be brought without limitation as to time, amount or the other provisions of this Section 10:

- (a) Time Limitation. No claim or action shall be brought under this Article 10 for breach of a representation or warranty after the lapse of eighteen (18) months following the Closing Date. Regardless of the foregoing, however, or any other provision of this Agreement:
 - (i) There shall be no time limitation on claims or actions brought for breach of any representation or warranty made in or pursuant to Sections 4.1 (Corporate), 4.2 (Authority), 4.3(a) (Marketable Title to Properties), 5.1 (LMC's Corporate) and 5.2 (LMC's Authority).

17

- (ii) Any claim made by a party hereunder by delivery of written notice giving reasonable notice of a claim and specifying the factual basis of the claim in reasonable detail to the extent then known, for breach of a representation or warranty prior to the termination of the survival period for such claim shall be preserved despite the subsequent termination of such survival period.
- (iii) If any act, omission, disclosure or failure to disclose shall form the basis for a claim for breach of more than one representation or warranty, and such claims have different periods of survival hereunder, the termination of the survival period of one claim shall not affect a party's right to make a claim based on the breach of representation or warranty still surviving.
- (b) Amount Limitation. Except with respect to claims for breaches of representations or warranties contained in Sections 4.1 (Corporate), 4.2 (Authority), 4.3(a) (Marketable Title to Properties), 5.1 (LMC's Corporate) and 5.2 (LMC's Authority), the maximum, aggregate liability for all indemnification claims under this Section 10 (other than (i) LMT's failure to pay and perform LMT's Liabilities (including, without limitation the Liabilities not being assumed as described in Section 2.2) other than Assumed Liabilities, and (ii) claims for fraud) against LMT, on the one hand, or against LMC, on the other, shall not exceed the Purchase Price, as adjusted pursuant to Section 3.4.
- (c) Exclusive Remedy. The right to indemnification provided by this Section 10 shall be the exclusive remedy available to the parties with respect to claims described in this Section 10, and no other rights or remedies at law or in equity will accrue with respect to such claims.

10.5. <u>Characterization of Indemnification Payments.</u>

Any payments made pursuant to Section 10 of this Agreement shall be treated for all Tax purposes as adjustments to the consideration to be paid hereunder and no party or any of its Affiliates shall take any position on a Tax Return or in any proceeding with any taxing authority contrary to such treatment, unless otherwise required by law.

11. CLOSING

The closing of this transaction ("the <u>Closing"</u>) shall take place at the offices of Foley & Lardner LLP, 100 North Tampa Street, Suite 2700, Tampa, Florida at 10:00 a.m. (local time) on the date hereof or at such other time and place as the parties hereto shall agree upon. Such date is referred to in this Agreement as the "<u>Closing Date</u>".

11.1. <u>Documents to be Delivered by LMT.</u>

At the Closing, LMT shall deliver to LMC the following documents, in each case duly executed or otherwise in proper form:

(a) <u>Bills of Sale</u>. Bills of sale and such other instruments of assignment, transfer, conveyance and endorsement as will be sufficient in the opinion of LMC and its

- (b) Compliance Certificate. A certificate signed by the chief executive officer of LMT that each of the representations and warranties made by LMT in this Agreement is true and correct in all material respects on and as of the Closing Date with the same effect as though such representations and warranties had been made or given on and as of the Closing Date (except for any changes permitted by the terms of this Agreement or consented to in writing by LMC), and that LMT have performed and complied with all of LMT's obligations under this Agreement which are to be performed or complied with on or prior to the Closing Date.
 - (c) Certified Resolutions. A certified copy of the resolutions of the Board of Directors of LMT authorizing and approving this Agreement and the consummation of the transactions contemplated by this Agreement
 - (d) Incumbency Certificate. Incumbency certificates relating to each person executing any document executed and delivered to LMC pursuant to the terms hereof.
- (e) Other Documents. All other documents, instruments or writings required to be delivered to LMC at or prior to the Closing pursuant to this Agreement and such other certificates of authority and documents as LMC may reasonably request.

11.2. Documents to be Delivered by LMC

At the Closing, LMC shall deliver to LMT the following documents, in each case duly executed or otherwise in proper form:

- (a) Cash Purchase Price. To LMT a certified or bank cashier's check (or wire transfer) as required by Section 3.3(b) hereof.
- (b) <u>Assumption of Liabilities</u>. Such undertakings and instruments of assumption as will be reasonably sufficient in the opinion of LMT and its counsel to evidence the assumption of Assumed Liabilities as provided for in Section 2.
- (c) Compliance Certificate. A certificate signed by the chief executive officer of LMC that the representations and warranties made by LMC in this Agreement are true and correct on and as of the Closing Date with the same effect as though such representations and warranties had been made or given on and as of the Closing Date (except for any changes permitted by the terms of this Agreement or consented to in writing by LMT), and that LMC has performed and complied with all of LMC's obligations under this Agreement which are to be performed or complied with on or prior to the Closing Date.
 - (d) Certified Resolutions. A certified copy of the resolutions of the Board of Managers of LMC authorizing and approving this Agreement and the consummation of the transactions contemplated by this Agreement.
 - (e) Incumbency Certificate. Incumbency certificates relating to each person executing any document executed and delivered to LMT by LMC pursuant to the terms hereof.

19

- (f) Certificated Membership Interests of LMC. Certificated Membership Interests of LMC equal to 69.25% of the total outstanding Membership Interests of LMC outstanding immediately after giving effect to such issuance.
- (g) Other Documents. All other documents, instruments or writings required to be delivered to LMT at or prior to the Closing pursuant to this Agreement and such other certificates of authority and documents as LMT may reasonably request.

12. TERMINATION

12.1. <u>Right of Termination Without Breach.</u>

This Agreement may be terminated without further liability of any party at any time prior to the Closing by mutual written agreement of LMC and LMT.

12.2. <u>Termination for Breach</u>.

- (a) Termination by LMC. If (i) there has been a material violation or breach by LMT of any of the agreements, representations or warranties contained in this Agreement which has not been waived in writing by LMC, or (ii) there has been a failure of satisfaction of a condition to the obligations of LMC which has not been so waived, or (iii) LMT shall have attempted to terminate this Agreement under this Section 12 or otherwise without grounds to do so, then LMC may, by written notice to LMT at any time prior to the Closing that such violation, breach, failure or wrongful termination attempt is continuing, terminate this Agreement with the effect set forth in Section 12.2(c) hereof.
- (b) <u>Termination by LMT</u>. If (i) there has been a material violation or breach by LMC of any of the agreements, representations or warranties contained in this Agreement which has not been waived in writing by LMT, or (ii) there has been a failure of satisfaction of a condition to the obligations of LMT which has not been so waived, or (iii) LMC shall have attempted to terminate this Agreement under this Section 12 or otherwise without grounds to do so, then LMT may, by written notice to LMC at any time prior to the Closing that such violation, breach, failure or wrongful termination attempt is continuing, terminate this Agreement with the effect set forth in Section 12.2(c) hereof.
- (c) Effect of Termination. Termination of this Agreement pursuant to this Section 12.2 shall not result in liability to any of the parties if the violation, breach or failure by LMT (in the case of termination pursuant to Section 12.2(a)(i) or (ii)) or by LMC (in the case of termination pursuant to Section 12.2(b)(i) or (ii)) is due to circumstances beyond such party's control. Termination of this Agreement pursuant to this Section 12.2 for reasons other than those set forth in the preceding sentence shall entitle the terminating party to recover from the other party the terminating party's costs and expenses, including reasonable attorney's fees and the reasonable value of in-house personnel's attention to the transaction, incurred in connection with the negotiation and execution of this Agreement. Subject to the foregoing, the parties' obligations under Section 13.8 of this Agreement shall survive termination.

20

13. MISCELLANEOUS

13.1. Further Assurance.

From time to time, at LMC's request and without further consideration, LMT will execute and deliver to LMC such documents, instruments and consents and take such other action as LMC may reasonably request in order to consummate more effectively the transactions contemplated hereby, to discharge the covenants of LMT and to vest in LMC good, valid and marketable title to the business and assets being transferred hereunder.

13.2. <u>Disclosures and Announcements</u>

Both the timing and the content of all disclosure to third parties and public announcements concerning the transactions provided for in this Agreement by either LMT or LMC shall be subject to the approval of the other in all essential respects, except that LMC's approval shall not be required as to any statements and other information which LMT may submit to the Securities and Exchange Commission or LMT's stockholders or be required to make pursuant to any rule or regulation of the Securities and Exchange Commission or national securities exchange or national over-the-counter market or otherwise required by law.

Assignment; Parties in Interest.

- (a) Assignment. Except as expressly provided herein, the rights and obligations of a party hereunder may not be assigned, transferred or encumbered without the prior written consent of the other parties.
- (b) <u>Parties in Interest</u>. This Agreement shall be binding upon, inure to the benefit of, and be enforceable by the respective successors and permitted assigns of the parties hereto. Nothing contained herein shall be deemed to confer upon any other person any right or remedy under or by reason of this Agreement.

13.4. <u>Equitable Relief</u>

LMC and LMT agree that any breach of any party's obligation to consummate the contribution, purchase and sale of the Transferred Assets on the Closing Date, will result in irreparable injury to the other parties for which a remedy at law would be inadequate; and that, in addition to any relief at law which may be available to the injured party or parties for such breach and regardless of any other provision contained in this Agreement, the injured party or parties shall be entitled to injunctive and other equitable relief as a court may grant. This Section 13.4 shall not be construed to limit any injured party's right to obtain equitable relief for other breaches of this Agreement under general equitable standards.

13.5. Governing Law

This Agreement shall be governed by and be construed in accordance with the laws of the State of Delaware without giving effect to the principles of conflicts of laws thereof. Any legal suit, action or proceeding brought by LMT or LMC, or any of their respective Affiliates, arising out of or based upon this Agreement shall be instituted in the courts of the

21

State of Texas or the Courts of the United States of America located in the State of Texas (collectively, the "Courts"), and each of LMT and LMC (on its behalf and on behalf of its Affiliates) waives any objection which it may now or hereafter have to the laying of venue of any such proceeding, and irrevocable submits to the exclusive jurisdiction of the Courts in any such suit, action or proceeding. EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY SUIT, ACTION OR PROCEEDING (INCLUDING ANY COUNTERCLAIM) ARISING OUT OF OR BASED UPON THIS AGREEMENT.

13.6. <u>Amendment and Modification</u>

13.7. <u>Notice</u>.

All notices, requests, demands and other communications hereunder shall be given in writing and shall be: (a) personally delivered; (b) sent by telecopier, facsimile transmission or other electronic means of transmitting written documents; or (c) sent to the parties at their respective addresses indicated herein by registered or certified U.S. mail, return receipt requested and postage prepaid, or by private overnight mail courier service. The respective addresses to be used for all such notices, demands or requests are as follows:

(a) If to LMC, to:

30452 Esperanza, Rancho Santa Margarita, CA 92688 Attention: Larry Buffington, Chief Executive Officer Facsimile: (949) 635-2108

(with a copy to)

C₃ Capital, LLC 4520 Main Street Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: 816-756-5552

or to such other person or address as LMC shall furnish to LMT in writing.

(b) If to LMT, to:

30452 Esperanza, Rancho Santa Margarita, CA 92688 Attention: John Kang, Chairman

22

Facsimile: (949) 635-2108

(with a copy to)

Curt Creely, Esq. Foley & Lardner LLP 100 North Tampa Street Suite 2700 Tampa, FL 33602-5810 Facsimile: 813-221-4210 Telephone: 813-225-4122

or to such other person or address as LMT shall furnish to LMC in writing.

If personally delivered, such communication shall be deemed delivered upon actual receipt; if electronically transmitted pursuant to this paragraph, such communication shall be deemed delivered the next business day after transmission (and sender shall bear the burden of proof of delivery); if sent by overnight courier pursuant to this paragraph, such communication shall be deemed delivered upon receipt; and if sent by U.S. mail pursuant to this paragraph, such communication shall be deemed delivered as of the date of delivery indicated on the receipt issued by the relevant postal service, or, if the addressee fails or refuses to accept delivery, as of the date of such failure or refusal. Any party to this Agreement may change its address for the purposes of this Agreement by giving notice thereof in accordance with this Section.

13.8. Expenses.

Regardless of whether or not the transactions contemplated hereby are consummated, each party shall pay all fees and expenses of its legal, accounting, investment banking and other professional counsel and agents in connection with the transactions contemplated hereby, subject to Sections 10 and 12.2(c) hereof.

13.9. Entire Agreement.

This instrument embodies the entire agreement between the parties hereto with respect to the transactions contemplated herein, and there have been and are no agreements, representations or warranties between the parties other than those set forth or provided for herein.

13.10. Counterparts.

This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

13.11. Headings.

The headings in this Agreement are inserted for convenience only and shall not constitute a part hereof.

23

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date and year first above written.

LIQUIDMETAL TECHNOLOGIES, INC.

sy: /s/ John Kang

John Kang, Chairman of the Board

LIQUIDMETAL COATINGS, LLC

By: /s/ Larry Buffington

Larry Buffington, President and Chief Executive Officer

24

Exhibit A

Form of LMC Limited Liability Company Agreement

Adopted as of July 24, 2007

TABLE OF CONTENTS

Article I Section 1.1 Section 1.2 Section 1.3 Section 1.4 Section 1.5	FORMATION AND ORGANIZATION NAME AND FORMATION PRINCIPAL PLACE OF BUSINESS REGISTERED OFFICE AND REGISTERED AGENT TERM NO STATE LAW PARTNERSHIP	
Article II Section 2.1 Section 2.2	PURPOSE AND POWERS OF THE COMPANY PURPOSE POWERS OF THE COMPANY	
Article III Section 3.1 Section 3.2 Section 3.3 Section 3.4 Section 3.5 Section 3.6 Section 3.7	capital structure, contributions to capital and capital accounts INITIAL CONTRIBUTIONS ADDITIONAL CAPITAL CONTRIBUTIONS RETURN OF CONTRIBUTIONS CAPITAL ACCOUNTS CAPITAL STRUCTURES AND PERCENTAGE INTERESTS LOANS BY MEMBERS PROFITS INTEREST OF CLASS C HOLDERS	
Article IV Section 4.1 Section 4.2 Section 4.3 Section 4.4 Section 4.5 Section 4.6 Section 4.7 Section 4.8 Section 4.9	allocations of taxable profits and losses DETERMINATION OF PROFIT OR LOSS COSTS AND EXPENSES ALLOCATION SPECIAL ALLOCATIONS BUILT-IN GAIN OR LOSS INCOME CHARACTERIZATION CREDITS CHANGE IN INTERESTS CREDITING ACCOUNTS	
Article V Section 5.1 Section 5.2 Section 5.3 Section 5.4	DISTRIBUTIONS DISTRIBUTABLE AMOUNTS ALLOCATION LIQUIDATING DISTRIBUTIONS MANDATORY TAX DISTRIBUTIONS	
Article VI Section 6.1 Section 6.2 Section 6.3	duration of business; termination DURATION OF COMPANY DEATH, ETC., OF MEMBER LIQUIDATION	
Article VII Section 7.1	RIGHTS AND DUTIES OF MEMBERS LIABILITIES OF MEMBERS	10 10

i

Section 7.2	LIMITATIONS ON POWERS OF MEMBERS	10
Section 7.3	PROHIBITION AGAINST PARTITION	10
Article VIII	meetings of members	10
Section 8.1	ANNUAL MEETING	10
Section 8.2	SPECIAL MEETING	10
Section 8.3	PLACE OF MEETING	11
Section 8.4	NOTICE OF MEETING	11
Section 8.5	WAIVER OF NOTICE	11
Section 8.6	FIXING OF RECORD DATE	11
Section 8.7	QUORUM	11
Section 8.8	PROXIES	11
Section 8.9	VOTING OF MEMBERSHIP UNITS	12
Section 8.10	VOTING OF MEMBERSHIP UNITS BY CERTAIN HOLDERS	12
Section 8.11	ACTION BY WRITTEN CONSENT OR TELEPHONE CONFERENCE	12
Section 8.12	ANNUAL REPORTS, ETC.	13
Article IX	BOARD OF MANAGERS	13
Section 9.1	MANAGEMENT AND CONTROL	13
Section 9.2	NUMBER AND ELECTION	13
Section 9.3	RESIGNATION	13
Section 9.4	REMOVAL OF MANAGER	13
Section 9.5	EXPRESSLY AUTHORIZED RIGHTS AND POWERS	14
Section 9.6	CERTAIN LIMITATIONS	14
Section 9.7	MEETINGS OF MANAGERS	15
Section 9.8	COMPENSATION	16
Article X	OFFICERS	16
Section 10.1	NUMBER	16
Section 10.2	ELECTION AND TERM OF OFFICE	17
Section 10.3	REMOVAL AND RESIGNATION	17
Section 10.4	VACANCIES	17
Section 10.5	CHAIR	17
Section 10.6	PRESIDENT	17
Section 10.7	VICE PRESIDENTS	17
Section 10.8	SECRETARY	17
Section 10.9	TREASURER	18
Section 10.10	SALARIES	18
Article XI	INDEMNIFICATION OF MEMBERS, managers and directors	18
Section 11.1	RIGHT TO INDEMNIFICATION	18
Section 11.2	ADVANCES	18
Section 11.3	SAVINGS CLAUSE	18

ii

Section 12.2 Section 12.3 Section 12.4 Section 12.5 Section 12.6 Section 12.7 Section 12.8 Section 12.9	RIGHT OF FIRST REFUSAL INVOLUNTARY TRANSFER TERMINATION OF EMPLOYMENT WITHDRAWALS RIGHTS OF ASSIGNEE OF MEMBERSHIP UNITS SUBSTITUTED MEMBER SECURITIES LAWS INVALID TRANSFER	19 21 22 23 23 23 24 24
Section 12.10 Section 12.11	DISTRIBUTIONS AND ALLOCATIONS IN RESPECT OF A TRANSFERRED INTEREST DRAG-ALONG RIGHTS	24 24
Section 12.12	APPLICABILITY OF PROVISIONS TO CLASS A HOLDERS	25
Article XIII	CONVERSION TO CORPORATE SOLUTION	26
Article XIV Section 14.1 Section 14.2 Section 14.3 Section 14.4 Section 14.5 Section 14.6 Section 14.7 Section 14.8 Section 14.9	BOOKS AND RECORDS BOOKS AND RECORDS CUSTODY OF MEMBER FUNDS; BANK ACCOUNTS ACCOUNTANTS SECTION 754 ELECTION FISCAL YEAR TAX MATTERS PARTNER ANNUAL FINANCIAL STATEMENTS AND TAX RETURN INFORMATION MONTHLY FINANCIAL STATEMENTS INSPECTION RIGHTS	26 26 27 27 27 27 27 27 27 28 28
Article XV	employment agreements	28
Article XVI Section 16.1	definitions DEFINITIONS	28 28
Article XVII	agreement prepared by attorney for company	33
Article XVIII	amendments	33
Article XIX	manager and member relations to the company	33
Article XX Section 20.1 Section 20.2	CHOICE OF LAW; SUBMISSION TO JURISDICTION; AND WAIVER OF JURY TRIAL LAW WAIVER OF JURY TRIAL	34 34 34

iii

Article XXI	MISCELLANEOUS	34
Section 21.1	DELAY OR OMISSIONS	34
Section 21.2	WAIVER	34
Section 21.3	WAIVER OF PARTITION	34
Section 21.4	SUCCESSORS AND ASSIGNS	34
Section 21.5	NOTICES	35
Section 21.6	ENTIRE AGREEMENT	35
Section 21.7	PRONOUNS	35
Section 21.8	TITLES AND SUBTITLES	35
Section 21.9	SEVERABILITY	35
Section 21.10	BINDING EFFECT	36
Section 21.11	CREDITORS	36
Section 21.12	EXECUTION OF ADDITIONAL INSTRUMENTS	36
Section 21.13	RIGHTS AND REMEDIES CUMULATIVE	36
Section 21.14	COUNTERPARTS	36

iv

OPERATING AGREEMENT

OF

LIQUIDMETAL COATINGS, LLC

THIS OPERATING AGREEMENT ("Agreement") is hereby entered into effective as of the 24th day of July, 2007, by and between the persons identified as Members on Exhibit "A" attached hereto and executing this Agreement on the signature pages hereof (each of whom is sometimes hereinafter referred to individually as a "Member" and collectively as "Members") and LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company (the "Company").

RECITALS

The Company was organized in accordance with the Delaware Limited Liability Company Act upon the filing of a Certificate of Formation with the Secretary of State of the State of Delaware effective July 9, 2007. The parties hereby provide for the Company's capitalization and management and its affairs and the conduct of its business.

NOW, THEREFORE, in consideration of the mutual promises set forth herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

Article I. FORMATION AND ORGANIZATION

- Section 1.1. NAME AND FORMATION. The name of the Company is: LIQUIDMETAL COATINGS, LLC. All Company business shall be conducted in the name of "LIQUIDMETAL COATINGS, LLC" or such other names that comply with applicable law as the Board of Managers may select from time to time.
- Section 1.2. PRINCIPAL PLACE OF BUSINESS. The initial principal office of the Company will be located at 30452 Esperanza, Rancho Santa Margarita, California 92688, and may be changed to such other place within or without the State of Delaware as may be determined from time to time by the Board of Managers. It is contemplated that the initial principal office will move to the Houston, Texas metropolitan area when a location for such office is leased in that area.
- Section 1.3. REGISTERED OFFICE AND REGISTERED AGENT. The Company's initial registered agent and office shall be CorpDirect Agents, Inc., 615 South DuPont Highway, Kent County, Dover, Delaware 19901. The Company may change its registered agent or registered office to any other place or places in the State of Delaware as may be determined from time to time by the Board of Managers.
 - Section 1.4. TERM. The term of the Company shall be perpetual, unless the Company is dissolved in accordance with the provisions of this Agreement.
- Section 1.5. NO STATE LAW PARTNERSHIP. The Members intend that the Company (i) shall be taxed as a partnership for all applicable federal, state and local income tax purposes and (ii) shall not be a partnership or joint venture for any other purpose, and that no Member or any Manager shall, by virtue of this Agreement, be a partner or joint venturer of any other Member or Manager.

Article II. PURPOSE AND POWERS OF THE COMPANY

- Section 2.1. PURPOSE. The Company is formed for the purpose of manufacturing and applying metallic coating to various products and engaging in any other business activity permitted under the Act as the Board of Managers may determine from time to time.
- Section 2.2. POWERS OF THE COMPANY. Subject to the provisions of this Agreement, the Company shall have the power and authority to take any and all actions necessary, appropriate, proper, advisable, incidental or convenient to or for the furtherance of the purposes set forth in Section 2.1, and all powers reasonably connected with such activities and businesses that may be legally exercised by limited liability companies under the Act, and may engage in all activities necessary, customary, convenient, or incident to any of the foregoing.

Article III. <u>CAPITAL STRUCTURE, CONTRIBUTIONS TO CAPITAL</u> <u>AND CAPITAL ACCOUNTS</u>

- Section 3.1. INITIAL CONTRIBUTIONS. The Members have previously contributed, or shall contribute simultaneously with the execution hereof, capital to the Company in accordance with the schedule attached hereto as Exhibit "A" (the "Initial Capital Contributions") in exchange for their respective Membership Units.
- Section 3.2. <u>ADDITIONAL CAPITAL CONTRIBUTIONS</u>. The Member may make additional contributions of capital to the Company as the Member determines are necessary, appropriate or desirable; provided, however, that the Member shall have no obligation to contribute any additional capital to the Company, and except as set forth in the Act, the Member shall have no personal liability for any obligations of the Company.
- Section 3.3. RETURN OF CONTRIBUTIONS. Until the dissolution and liquidation of the Company, no Member shall have the right to demand or receive any part of the Member's capital contribution and there is no right given to any Member to demand and receive property other than cash in return for the Member's capital contribution.
- Section 3.4. CAPITAL ACCOUNTS. An individual Capital Account shall be maintained for each Member in accordance with Section 704(b) of the Code, paragraph 1.704-1(b)(2)(iv) of the accompanying Treasury Regulations, and the following rules:
 - (a) Computation of Capital Account Balance. The Capital Account of a Member shall consist of the amount of money and the fair market value of any property (other than money) comprising the Member's proportionate share of the Initial Capital Contribution pursuant to Section 3.1 hereof, as increased by: (i) the amount of money and the fair market value of any property (other than money) comprising any additional capital

2

contributions made by the Member, (ii) any amount credited to the Capital Account of a Member pursuant to Section 4.9 hereof as a result of any Company income, profits or gains allocated to the Member (and as adjusted pursuant to Section 1.704-1(b)(2)(iv) of the Treasury Regulations), and (iii) the amount of any Company liabilities assumed by the Member or that are secured by any Company property distributed to that Member, and decreased by: (i) the amount of money and the fair market value of any property (other than money) comprising any distributions to the Member pursuant to Article V hereof, (ii) any amount debited to the Capital Account of a Member pursuant to Section 4.9 hereof as a result of any Company expenses, deductions, losses and credits allocated to the Member (and as adjusted pursuant to Section 1.704-1(b)(2)(iv) of the Treasury Regulations), and (iii) the amount of any liabilities of such Member that are assumed by the Company or that are secured by any property contributed by that Member to the Company.

- (b) <u>Built-In Gain or Loss</u>. The Capital Account of a Member shall not be increased or decreased, as the case may be, with regard to any built-in gain or loss allocated to the Member pursuant to Section 4.5 hereof.
- (c) Transferee's Capital Account. In the event of a transfer of any Membership Units, the transferee shall assume the Capital Account balance of the transferor.
- (d) <u>Interest</u>. No interest shall be paid on any present or future Capital Account balance.
- (e) Conformance with Regulations. The provisions of this Section 3.4 are intended to comply with Treasury Regulation Section 1.704-1(b) regarding the maintenance of the Capital Accounts of the Members and this Section 3.4 shall be interpreted and applied in a manner consistent with such Treasury Regulations. In the event that the Board of Managers shall determine that it is prudent to modify the manner in which the Capital Accounts, or any debits or credits thereto, are computed in order to comply with such Treasury Regulations, the Board of Managers may make such modifications, provided that it is not likely to have a material effect on any amounts distributable to any Member upon the dissolution of the Company. The Board of Managers shall also make any appropriate modifications in the event unanticipated events might otherwise cause this Agreement not to comply with Treasury Regulation Section 1.704-1(b).

Section 3.5. CAPITAL STRUCTURES AND PERCENTAGE INTERESTS

- (a) <u>Capital Structure</u>. Ownership rights in the Company are divided into and represented by membership units ("<u>Membership Units</u>"). The Membership Units are further divided into three classes: (i) Class A Units; (ii) Class B Units; and (iii) Class C Units.
- (b) <u>Initial Percentage Interest</u>. Upon their admission to the Company, the Members shall own the number and class of Membership Units and the corresponding Percentage Interests in the Company as set forth on <u>Exhibit "A"</u> attached hereto.
- (c) Adjustments to Percentage Interests. In the event of the issuance of any additional Membership Units to an existing Member(s), including the admission of an additional member(s), each Member's Percentage Interest in the Company shall be deemed to be equal to a fraction, the numerator of which is the number of Membership Units issued and

3

outstanding to the Member, and the denominator of which is the number of Membership Units then issued and outstanding to all Members. All adjustments to a Member's Percentage Interest in the Company shall be permanent and shall not otherwise be altered except as provided in this Agreement.

- (d) <u>Effect of Percentage Interests</u>. The Percentage Interests of the Members, as determined and adjusted pursuant to this Section 3.5, are maintained solely for the purposes of determining the amount of certain allocations of taxable profits and losses and cash distributions allocable to the Members pursuant to Articles IV and V hereof, and for other purposes set forth in this Agreement. The Percentage Interest of a Member shall not reflect that Member's proportionate interest in the capital of the Company at any time.
- Section 3.6. LOANS BY MEMBERS. A Member may at any time lend funds to the Company as may be agreed upon by the Board of Managers. Such funds shall represent a debt, payable on demand, unless otherwise specifically provided, from the Company to the Member making the loan, and interest, at a rate agreed upon by the Board of Managers, shall be paid thereon and charged as an expense to the Company.
- **Section 3.7. PROFITS INTEREST OF CLASS C HOLDERS.** The Class C Holders shall own a percentage of the profits of the Company equal to the right to receive their respective percentage interest as shown on Exhibit "A", attached hereto, of the available cash distributed among the Members pursuant to Section 5.2 and Section 5.3. In addition, taxable income shall be allocated to the Class C Holders equal to their respective percentage interest as shown on Exhibit "A", attached hereto, of the Net Profits allocated among the Members pursuant to Section 4.3(a)(ii).

It is the agreement of the Members that the rights of the Class C Holders under this Section 3.7 shall be limited to receiving allocations of taxable income and distributions of cash as described herein, but that the Class C Holders shall not receive any other rights in and shall not participate in any other allocations of taxable income, loss, credit, deduction, gain, or other tax items or distributions of cash from the Company except as specifically provided in this Section 3.7.

Article IV. ALLOCATIONS OF TAXABLE PROFITS AND LOSSES

- Section 4.1. <u>DETERMINATION OF PROFIT OR LOSS.</u> The items of income, gains, expenses, deductions, losses and credits generated by the Company for federal income tax purposes shall be determined in accordance with a generally accepted method of accounting as soon as practicable after the close of the fiscal year of the Company.
- Section 4.2. <u>COSTS AND EXPENSES</u>. The Company shall pay all expenses (which expenses shall be billed directly to the Company) of the Company which may include but are not limited to: (i) legal, audit, accounting, and other fees; (ii) expenses and taxes incurred in connection with the issuance, distribution and transfer of documents evidencing ownership of Membership Units in the Company or in connection with the business of the Company; (iii) expenses of organizing, revising, amending, converting, modifying or terminating the Company; (iv) expenses in connection with distributions made by the Company to, and communications and bookkeeping work necessary in maintaining relations with, the Members; and (v) costs of any accounting, statistical or bookkeeping equipment necessary for the maintenance of the books and records of the Company.

4

- Section 4.3. ALLOCATION. The net profits, net gains and net losses generated by the Company for federal income tax purposes for a year shall be allocated among the Members as follows:
 - (a) <u>Profits</u>. Any net taxable income and gain (collectively, "Net <u>Profits</u>") of the Company for the year shall be allocated among the Members as follows:
 - (i) First, 100% of the Net Profits shall be allocated pro rata (relative to the entire amount to be allocated to all Members in that year pursuant to this Section 4.3(a)(i)) among the Members in an amount equal to the excess, if any, of (a) the cumulative Losses allocated pursuant to Section 4.3(b) hereinbelow for all prior years, over (b) the cumulative Profits allocated pursuant to this Section 4.3(a)(i) for all prior years, until such time as each Member has been allocated an aggregate amount of the Net Profits in all years pursuant to this Section 4.3(a)(i) equal to the aggregate amount of Losses allocated to the Member pursuant to Section 4.3(b) hereof in all prior years; and
 - (ii) Any remaining Net Profits shall be allocated among the Members according to their respective Percentage Interests in the Company at that time, as set forth in Section 3.5 hereof.
 - (b) Losses. Any net taxable losses and deductions (collectively, "Losses") of the Company for the year shall be allocated among the Members in the following order of priority:
 - (i) First, all Losses shall be allocated among the Members in proportion to the positive balances in their respective Capital Accounts at that time, after a reduction for any cash distributed from the Company to the Members pursuant to Article V in that year, until the balance of all Members' Capital Accounts have been reduced to zero (0); and
 - (ii) Any remaining Losses shall be allocated among the Members according to their respective Percentage Interests in the Company at that time, as set forth in Section 3.5 hereof.
- Section 4.4. SPECIAL ALLOCATIONS. Notwithstanding any other provision of this Agreement to the contrary, the following special allocations shall be made in the following order:

(a) Minimum Gain Chargeback. Notwithstanding any other provision of this Section 4.4, if there is a net decrease in the Partnership Minimum Gain (as described in Article XVI hereof) during any Company fiscal year, each Member will be specially allocated, before any other allocation is made, items of income and gain for such year (and, if necessary, subsequent years) in an amount equal to that Member's share of the net decrease in Partnership Minimum Gain, determined as provided in Section 1.704-2(g)(2) of the Regulations.

Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be allocated shall be determined in accordance with Regulations Section 1.704-2(f), and this Section 4.4(a) is

5

intended to comply with the minimum gain chargeback requirement in that Section of the Regulations and shall be interpreted consistently therewith.

(b) Partner Minimum Gain Chargeback. Notwithstanding any other provision of this Article IV except Section 4.4(a), above, if there is a net decrease in Partner Minimum Gain (as described in Article XVI hereof) attributable to a Partner Nonrecourse Debt (as described in Article XVI hereof) during any Company fiscal year, each Member who has a share of the Partner Minimum Gain attributable to such Partner Nonrecourse Debt (determined as provided in Section 1.704-2(i)(5) of the Regulations) as of the beginning of the year shall be specially allocated items of Company income and gain for such year (and, if necessary, subsequent years) in an amount equal to that Member's share of the net decrease in Partner Minimum Gain attributable to such Partner Nonrecourse Debt, as provided in Section 1.704-2(i)(4) of the Regulations.

Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with Section 1.704-2(i)(4) of the Regulations. This Section 4.4(b) is intended to comply with the minimum gain chargeback requirement in that Section of the Regulations and shall be interpreted consistently therewith.

- (c) Qualified Income Offset. In the event any Member unexpectedly receives an adjustment, allocation or distribution of a nature described in Treasury Regulations Sections 1.704-1(b)(2)(ii)(d)(4), (5) or (6), then each such Member will be specially allocated items of Company taxable income and gain in an amount and manner sufficient to eliminate, to the extent required by the Regulations, that Member's Deficit Capital Account Balance as quickly as possible; provided that an allocation pursuant to this Section 4.4(c) shall be made if and only to the extent that such Member would have a Deficit Capital Account Balance after all other allocations provided for in this Article IV have been tentatively made as if this Section 4.4(c) were not in this Agreement.
- (d) Partner Nonrecourse Deductions. Any Partner Nonrecourse Deductions (as described in Article XVI hereof) for any fiscal year or other period shall be specially allocated to the Member who bears the economic risk of loss with respect to the Partner Nonrecourse Debt (as described in Article XVI hereof) to which such Partner Nonrecourse Deductions are attributable in accordance with Section 1.704-2(i)(2) of the Regulations.
- (e) <u>Curative Allocations</u>. The allocations set forth in this Section 4.4 (the "<u>Regulatory Allocations</u>") are intended to comply with certain requirements of Regulations Sections 1.704-1(b) and 1.704-2. Notwithstanding any other provision of this Article IV (other than the Regulatory Allocations), the Regulatory Allocations shall be taken into account in allocating other items of profits, losses, income, gains and deductions among the Members so that, to the extent possible, the net amount of such allocations of other taxable items and the Regulatory Allocations to each Member shall be equal to the net amount which would have been allocated to each such Member if the Regulatory Allocations had not occurred. For purposes of applying the previous sentence:
 - (i) No allocations of Nonrecourse Deductions shall be made pursuant to this Section 4.4(e) with respect to Regulatory Allocations required pursuant to Section 4.4(a) hereof prior to the Company fiscal year during which there is a net decrease in Partnership

6

Minimum Gain, and then only to the extent necessary to avoid any potential economic distortions caused by such net decrease in Partnership Minimum Gain;

- (ii) Allocations pursuant to this Section 4.4(e) shall be deferred with respect to allocations of Nonrecourse Deductions to the extent the Board of Managers determine that such allocations are likely to be offset by subsequent allocations pursuant to subsection (a) of Section 4.4 hereof;
- (iii) No allocations of Partner Nonrecourse Deductions shall be made pursuant to this Section 4.4(e) with respect to Regulatory Allocations required under subsection (b) or (d) of this Section 4.4 prior to the Company fiscal year during which there is a net decrease in Partner Minimum Gain attributable to such Partner Nonrecourse Debt, and then only to the extent necessary to avoid any potential economic distortions caused by such net decrease in Partner Minimum Gain; and
- (iv) Allocations pursuant to this Section 4.4(e) shall be deferred with respect to allocations pursuant to subsection (d) of this Section 4.4 relating to a particular Partner Nonrecourse Debt to the extent the Board of Managers reasonably determine that such allocations are likely to be offset by subsequent allocations pursuant to subsection (b) of this Section 4.4.
- Section 4.5. BUILT-IN GAIN OR LOSS. Notwithstanding any other provision of this Article IV, in accordance with Code Section 704(c) and the accompanying Treasury Regulations thereunder, income, gain, loss and deduction with respect to any property other than cash contributed to the capital of the Company shall, solely for tax purposes, be allocated among the Members so as to take into account any variation between (i) the adjusted basis of such property to the Company for federal income tax purposes on the date of contribution and (ii) the fair market value of such property on the date contributed to the Company, as determined by the Board of Managers.
- Section 4.6. INCOME CHARACTERIZATION. For purposes of determining the character (as ordinary income or capital gain) of any taxable income of the Company allocated to the Members pursuant to this Article IV, such portion of the taxable income of the Company allocated pursuant to this Article IV which is treated as ordinary income attributable to the recapture of depreciation shall, to the extent possible, be allocated among the Members in the proportion which (i) the amount of depreciation previously allocated to each Member bears to (ii) the total of such depreciation allocated to all Members. This Section 4.6 shall not alter the amount of allocations among the Members pursuant to this Article IV, but merely the character of income so allocated.
 - **Section 4.7. CREDITS.** Tax credits shall be allocated among the Members in accordance with Section 4.3 hereof.
- Section 4.8. CHANGE IN INTERESTS. Notwithstanding the foregoing, in the event of a change in the Members' Percentage Interests in the Company during a year, whether occasioned by admission of a new Member, additional contributions, assignments of Membership Units or otherwise, the allocation of items of income and expense shall be made so as to reflect the Members' varying Percentage Interests in the Company during the year. Profits and losses for the year shall be prorated on a daily basis and allocated among the Members based upon the period of time during which they held their respective Percentage Interests.

7

Section 4.9. <u>CREDITING ACCOUNTS</u>. Items of income, gains, expenses, deductions, losses and credits shall be credited or debited, as the case may be, to each Member's Capital Account created pursuant to Section 3.4 as provided in this Article.

Article V. DISTRIBUTIONS

- Section 5.1. <u>DISTRIBUTABLE AMOUNTS</u>. The Company may make distributions of any amounts in excess of its reasonable operating requirements as determined by the Board of Managers. The amounts available for distribution may be generated by operations of the Company through sale, condemnation, financing or refinancing of assets of the Company, by collection of amounts owed to the Company or by any other transaction. In the case of amounts attributable to a Capital Transaction, the net proceeds to be distributed hereunder shall be the net cash proceeds received by the Company after payment of, or provision for, all Company debts, obligations and reserves required or permitted to be paid upon, or incurred or established in connection with, the receipt by the Company of such proceeds (said reserves to be established in the reasonable discretion of the Board of Managers), and all expenses incurred by the Company in connection with the Capital Transaction giving rise to such proceeds. Notwithstanding the foregoing, no distribution shall be made unless after the distribution the Company retains assets sufficient to pay all its debts as they become due and such distribution, if made, would not cause the Company to otherwise become insolvent.
 - Section 5.2. ALLOCATION. Distributions of available cash pursuant to Section 5.1 hereof shall be made to the Members according to their respective Percentage Interests in the Company at that time.
- Section 5.3. <u>LIQUIDATING DISTRIBUTIONS</u>. After giving effect to all Member contributions, distributions, allocations, and capital account adjustments for all taxable years of the Company (including the taxable year during which the Company is liquidated), the Company shall distribute the liquidation proceeds derived from the dissolution, winding-up, liquidation, and termination of the Company (plus any remaining assets of the Company that are to be distributed in kind to the Members) in the following priority and proportions:
 - (a) To pay in the order of priority required by the Act all the liabilities and obligations of the Company that are then due and payable to creditors (including Members), including all costs and expenses directly relating to the liquidation, winding-up, and termination of the Company, but excluding Capital Contributions and distributions payable to the Members; and then
 - (b) To establish any reasonable cash reserve that the Members may determine unanimously by resolution to be appropriate for any contingent, conditional, or unmatured liabilities or obligations of the Company and as an allowance for doubtful accounts receivable or the unrealized portion of any promissory note, installment obligation, or other deferred revenue payable to the Company; and then
 - (c) To the Members, *pro rata*, in accordance with the sum of their Initial Capital Contributions plus any additional Capital Contributions which have not yet been returned to the respective Member (the "<u>Unreturned Capital Contributions</u>"), in an amount equal to each Member's Unreturned Capital Contributions; and then

8

(d) To the Members, *pro rata*, in accordance with the Members' respective Percentage Interests at the time of such liquidation..

Section 5.4. MANDATORY TAX DISTRIBUTIONS. In order to permit the Members to pay taxes on Company taxable income, the Company shall, to the extent of Available Cash, distribute cash at least annually and no later than 75 days after the end of a Fiscal Year, an amount which, at the time of such distribution, when added to all prior distributions pursuant to this Section 5.4, equals the cumulative maximum tax payable by an individual taxpayer at the highest combined marginal tax rates of federal and state income taxes imposed on resident individuals in the United States on the taxable income of the Company. Such distribution shall be made ratably to the Members in proportion to their respective Member Percentages.

Article VI. DURATION OF BUSINESS; TERMINATION

- **Section 6.1. DURATION OF COMPANY**. The Company shall continue until the earlier of:
 - (a) The decision of the Board of Managers to dissolve the Company;
 - (b) The expiration of the term of the Company set forth in Section 1.4 hereof; or
 - (c) The sale or other disposition of all Company Property and the distribution of all sales proceeds resulting therefrom to the Members in accordance with this Agreement.
- Section 6.2. DEATH, ETC., OF MEMBER. The Company shall not be dissolved upon the death, insanity, total disability, bankruptcy, dissolution or withdrawal of any Member, or by the assignment by any Member of all of the Membership Units or Economic Interest in the Company.
 - Section 6.3. <u>LIQUIDATION</u>. In the event of termination of the Company:

Company

- (a) The Board of Managers shall wind up the affairs of the Company, shall sell all the Company assets as promptly as is consistent with obtaining the fair value thereof, and shall apply and distribute the proceeds of liquidation in the following order of priority:
 - (i) To the payment of debts and liabilities of the Company (including to Members to the extent otherwise permitted by law) and the expenses of liquidation; then
 - (ii) To the setting up of such reserves as the Board of Managers winding up the Company's affairs may reasonably deem necessary or appropriate for any dispute, contingent or unforeseen liabilities or obligations of the Company; then
 - (iii) The remainder to the Members in accordance with Section 5.3 above.

9

- (b) Each Member shall look solely to the assets of the Company for the return of the Member's capital contribution, and if the Company Property remaining after payment or discharge of the debts and liabilities of the Company is insufficient to return the contributions of each Member, a Member shall have no recourse against any other Member(s).
 - (c) The proceeds of liquidation shall be distributed to the Members pursuant to Section 5.3 hereof.
- (d) Upon the liquidation of the Company, if any Member has a Deficit Balance in the Member's Capital Account (after giving effect to all contributions, distributions and allocations for all taxable years, including the year in which such liquidation occurs), that Member shall have no obligation to make any contributions to the capital of the Company with respect to such deficit and such deficit shall not be considered a debt owed to the Company or any other person or entity for any purpose whatsoever.

Article VII. RIGHTS AND DUTIES OF MEMBERS

Section 7.1. <u>LIABILITIES OF MEMBERS</u>. No Member shall be obligated to make capital contributions to the Company except as provided in Article III. No Member shall have any personal liability with respect to the liabilities or obligations of the Company except as provided in Article III. Failure of the Company to observe any formalities or requirements relating to the exercise of its powers or the management of its business or affairs under this Agreement or the Act shall not be grounds for imposing personal liability on the Members for liabilities or obligations of the Company.

Section 7.2. <u>LIMITATIONS ON POWERS OF MEMBERS</u>. Except as expressly authorized by this Agreement, no Member may, directly or indirectly, (a) resign, retire or withdraw from the Company, (b) dissolve, terminate or liquidate the Company, (c) petition a court for the dissolution, termination or liquidation of the Company, or (d) cause any property of the Company to be subject to the authority of any court, trustee or receiver (including suits for partition and bankruptcy, insolvency, and similar proceedings).

Section 7.3. PROHIBITION AGAINST PARTITION. Each Member irrevocably waives any and all rights the Member may have to maintain an action for partition with respect to any property of the

Article VIII. MEETINGS OF MEMBERS

Section 8.1. ANNUAL MEETING. The annual meeting of the Members shall be held between January 1st and December 31st of each year, on such date and at such hour as may be specified in the Notice of Meeting or in a duly executed Waiver of Notice thereof, for the purpose of electing Managers to serve for the ensuing year and the transaction of such other business as may come before the meeting. If the day fixed for the annual meeting shall be a legal holiday in the State of Texas, such meeting shall be held on the next succeeding business day.

Section 8.2. SPECIAL MEETING. Special meetings of the Members, for any purpose or purposes, may be called by the President or by a majority of the Board of Managers or by any Member owning Membership Units in the Company.

10

- Section 8.3. PLACE OF MEETING. The Board of Managers may designate any place, either within or without the State of Delaware, unless otherwise prescribed by statute, as the place of meeting for any annual or special meeting of Members. If no designation is made by the Board of Managers, the place of meeting shall be the principal office of the Company in the State of Texas.
- Section 8.4. NOTICE OF MEETING. Written or printed notice stating the place, day and hour of the meeting and, in the case of a special meeting, the purpose or purposes for which the meeting is called, shall be delivered to each Member of record entitled to vote at such meeting not less than ten (10) nor more than sixty (60) days before the date of the meeting, either personally or by first-class mail, by or at the direction of the President or the Secretary or by the Members calling the meeting. If mailed, such notice shall be deemed to be delivered at the earliest date of the following: (a) when received; (b) five days after its deposit in the United States mail addressed to the Member at his address as it appears on the records of the Company, with the postage thereon prepaid; or (c) on the date shown on the return receipt if sent by registered or certified mail, return receipt requested.
- Section 8.5. WAIVER OF NOTICE. Members may waive notice of a meeting before or after the date and time specified in the written notice of meeting. All waivers of notice must be in writing, be signed by the Member entitled to the notice and be delivered to the Company for inclusion in the appropriate records. None of the business to be transacted at, nor the purpose of, a Members' meeting must be specified in a written waiver of notice. Attendance of a Member at a meeting shall constitute a waiver of notice of the meeting, unless the Member at the beginning of the meeting objects to holding the meeting or transacting business at the meeting.
- **Section 8.6.** FIXING OF RECORD DATE. The Board of Managers may fix a date, not less than ten (10) nor more than sixty (60) days before the date set for any meeting of the Members, as the record date as of which the Members of record entitled to notice of and to vote at such meeting and any adjournment thereof shall be determined.
- Section 8.7. QUORUM. A majority of the Membership Units of the Company, represented in person or by proxy, shall constitute a quorum at a meeting of the Members. When a meeting is adjourned to another time or place, it shall not be necessary to give any notice of the adjourned meeting if the time and place to which the meeting is adjourned are announced at the meeting at which the adjournment is taken, and any business may be transacted at the adjourned meeting that might have been transacted at the original date of the meeting. If, however, after the adjournment, the Board of Managers fix a new record date for the adjourned meeting, a notice of the adjourned meeting, a notice of the adjourned meeting shall be given in compliance with 4.0 to each Member of record on the new record date entitled to vote at such meeting. After a quorum has been established at a Members' meeting, the subsequent withdrawal of Members, so as to reduce the Membership Units entitled to vote at the meeting below the percentage required for a quorum, shall not affect the validity of any action taken at the meeting or any adjournment thereof.
- Section 8.8. PROXIES. Every Member entitled to vote at a meeting of Members or to express consent or dissent without a meeting, or his duly authorized attorney-in-fact, may authorize another person or persons to act for him by proxy. The proxy must be executed in writing by the Member or his duly authorized attorney-in-fact. Such proxy shall be filed with the Company before or at the time of such meeting or at the time of expressing such consent or dissent without a meeting. No proxy shall be valid after the expiration of eleven (11) months after the date thereof unless provided otherwise in the proxy.

11

- Section 8.9. VOTING OF MEMBERSHIP UNITS. Except as provided in 6.0, each Member shall be entitled to one (1) vote per one (1) Membership Unit (and a corresponding fractional vote for each fraction of a Membership Unit) upon each matter submitted to a vote at a meeting of the Members. If a quorum is present, the affirmative vote of the Members owning a majority of the Membership Units represented at the meeting and entitled to vote on the subject matter shall be the act of the Members unless a greater percentage is required by the Delaware Statutes, this Agreement or the Company's Certificate of Formation.
- Section 8.10. <u>VOTING OF MEMBERSHIP UNITS BY CERTAIN HOLDERS.</u> Membership Units standing in the name of a corporation may be voted by the officer, agent or proxy designated by the bylaws of the corporate Member or, in the absence of any applicable bylaws, by such person as the managers of the corporate Member may designate. Proof of such designation may be made by presentation of a certified copy of the bylaws or other instrument of the corporate Member. In the absence of any such designation or, in case of conflicting designation by the corporate Member, the chairman of the board, the president, any vice president, the secretary, and the treasurer of the corporate Member shall be presumed to possess, in that order, authority to vote such Membership Units.

Membership Units held by an administrator, executor, guardian or conservator may be voted by him, either in person or by proxy, without a transfer of such Membership Units into his name.

Membership Units standing in the name of a trustee may be voted by him, either in person or by proxy, but no trustee shall be entitled to vote Membership Units held by him without a transfer of such Membership Units into his

Membership Units standing in the name of a receiver may be voted by such receiver, and Membership Units held by or under the control of a receiver may be voted by such receiver without the transfer thereof into his name, if authority so to do be contained in an appropriate order of the court by which such receiver was appointed.

A Member whose Membership Units are pledged shall be entitled to vote such Membership Units until the Membership Units have been transferred into the name of the pledgee, and thereafter the pledgee or his nominee shall be entitled to vote the Membership Units so transferred only in the event the pledgee becomes a Substituted Member (as defined in Section 12.3 hereinbelow).

Section 8.11. ACTION BY WRITTEN CONSENT OR TELEPHONE CONFERENCE

(a) Any action required or permitted to be taken at any meeting of Members may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by Members having not less than the minimum number of Membership Units necessary to authorize or take such action at a meeting at which all Members entitled to vote were present and voted. A telegram, telex, cablegram or similar transmission by a Member, or a photographic, photostatic, facsimile or similar reproduction of a writing signed by a Member, shall be regarded as signed by the Member for purposes of this Section 8.11. Prompt notice of the taking of any action by Members without a meeting by less than unanimous written consent shall be given to those Members who did not consent in writing to the action.

12

- (b) The record date for determining Members entitled to consent to action in writing without a meeting shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Company by delivery to its registered office, its principal place of business or the Board of Managers.
- (c) Members may participate in and hold a meeting by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can speak to and hear each other. Participation in such meeting shall constitute attendance and presence in person at such meeting, except where a person participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened.
- Section 8.12. ANNUAL REPORTS, ETC.. The Company shall file any necessary reports with the State of Delaware and pay the annual tax for Limited Liability Companies to the State of Delaware.

Article IX. BOARD OF MANAGERS

- Section 9.1. MANAGEMENT AND CONTROL. The management and control of the Company shall be vested in the Board of Managers. The Board of Managers shall have, except as specifically limited in this Agreement, full and exclusive authority in the management and control of the Company, and shall have all the rights and powers which are otherwise conferred by law or are necessary or advisable for the discharge of their duties and the management of the business and affairs of the Company.
- Section 9.2. NUMBER AND ELECTION. The Board of Managers shall consist of three Managers (unless it has only two Managers as a result of the last two sentences of this paragraph). Managers shall be elected as follows: Members holding Class A Units shall have the right to elect one Manager, Members holding Class B Units shall have the right to elect one Manager, and Larry Buffington shall serve as a Manager. In the event of the death, resignation or removal of a Manager, the Member who elected such Manager may elect a successor Manager. Notwithstanding the foregoing, in the event that Larry Buffington ceases to be actively employed by the Company on a full-time basis as the Company's principal executive officer, then as of the date of such cessation, he shall cease to serve as a Manager, and the size of the Board of Managers will thereupon be reduced to two Managers. In addition, Larry Buffington shall not have the right to serve as a Manager during any period of time during which he is serving as an officer or employee of Liquidmetal Technologies, Inc. ("LMT")). The Member holding Class A Units hereby elects John Kang to serve as an initial Manager, and the Members holding Class B Units hereby elect Robert Smith to serve as an initial Manager.
- Section 9.3. RESIGNATION. A Manager may resign at any time by delivering written notice to the Board of Managers or to the Company. A resignation is effective when notice is delivered unless the notice specifies a later effective date.
 - Section 9.4. REMOVAL OF MANAGER. Each Member may remove its elected Manager, with or without cause, at any time.

13

- Section 9.5. <u>EXPRESSLY AUTHORIZED RIGHTS AND POWERS</u>. Without limiting the generality of Section 9.1, but subject to the provisions of Section 9.6, the Board of Managers are expressly authorized on behalf of the Company to:
 - (a) elect officers to manage the day-to-day operations of the Company;
 - (b) procure and maintain with responsible companies such insurance as may be advisable in such amounts and covering such risks as are deemed appropriate by the Board of Managers;
 - (c) take and hold any assets of the Company in the Company name, or in the name of a nominee of the Company;
 - (d) execute and deliver on behalf of and in the name of the Company, or in the name of a nominee of the Company, all instruments necessary or incidental to the conduct of the Company's business;
 - (e) protect and preserve the assets of the Company and incur indebtedness;
 - (f) sell, dispose of, trade, exchange, convey, quitclaim, surrender, release or abandon, upon terms and conditions which the Board of Managers may negotiate and deem appropriate, personal property of the Company;
 - (g) execute and deliver documents and instruments on behalf of the Company in connection with the acquisition and disposition of its assets, and to execute, terminate, modify, enforce, continue or otherwise deal with any Company indebtedness and security interests, to sell Company assets, and to take any other action with respect to agreements made between the Company and a lender or any affiliate thereof, all subject to the limitations of Section 9.6;
 - (h) open Company bank accounts in which all Company funds shall be deposited and from which payments shall be made; and
 - (i) invest Company funds and working capital reserves.
- Section 9.6. CERTAIN LIMITATIONS. Notwithstanding the generality of the foregoing, and in addition to other acts expressly prohibited by this Agreement or by law, the Board of Managers, without the prior written consent or approval of the Members holding a Supermajority (as defined in Article XVI) of the issued and outstanding Membership Units, may not cause the Company to do any of the following:
 - (a) amend or restate the Certificate of Formation;
 - (b) sell, lease, exchange or otherwise dispose of all or substantially all the Company's property and assets;
 - (c) be a party to (i) a merger, (ii) interest exchange or (iii) other transaction affecting the ownership or structure of the Company:
 - (d) do any act in contravention of this Agreement;

. .

- (e) do any act which would make it impossible to carry on the ordinary business of the Company, except as expressly provided in this Agreement;
- (f) confess a judgment against the Company;
- (g) file a bankruptcy petition on behalf of the Company;
- (h) execute or deliver any general assignment for the benefit of the creditors of the Company;
- (i) Convert the Company to a Corporation pursuant to Article XIII;
- (j) assign rights in specific Company property for other than a Company purpose; or
- (k) knowingly or willingly do any act (except an act expressly required by this Agreement) which would cause the Company to become an association taxable as a corporation.
- Section 9.7. <u>MEETINGS OF MANAGERS</u>.

- (a) Following Annual Meeting of Members. In connection with any annual meeting of Members at which Managers were appointed, the Board of Managers may, if a quorum is present, hold its first meeting for the transaction of business immediately after and at the same place as such annual meeting of the Members. Notice of such meeting at such time and place shall not be required.
- (b) Regular Meetings. Regular meetings of the Board of Managers shall be held at such times and places as shall be designated from time to time by resolution of the Board of Managers. Notice of such regular meetings shall not be required after the initial notice of the schedule of meetings.
- (c) Special Meetings. Special meetings of the Board of Managers may be called by the President or by any Manager on at least forty-eight (48) hours notice to each Manager. Such notice shall state the purpose or purposes of, or the business to be transacted at, such meeting.
- (d) Attendance and Place of Meeting. Meetings of the Board of Managers may be held at such place or places as shall be determined from time to time by resolution of the Board of Managers. The chair of the Board of Managers, if one has been designated by the Board of Managers, shall preside when present at meetings of the Board of Managers. Attendance of a Manager at a meeting shall be presumed to constitute a waiver of notice of such meeting, except where a Manager attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened or unless he or she shall file a written objection to such action with the Person acting as secretary of the meeting before the adjournment thereof on the ground that the meeting is not lawfully called or convened.
 - (e) Quorum and Action. A quorum for the transaction of business of the Board of Managers shall require the presence of a majority of the number of Managers fixed

15

pursuant to Section 9.2 of this Article. At all meetings of the Board of Managers, business shall be transacted in such order as shall from time to time be determined by resolution of the Board of Managers. The act of the majority of the Managers present at a meeting at which a quorum is present shall be the act of the Board of Managers at which action on any Company matter is taken shall be presumed to have assented to the action unless he or she shall either voice his or her dissent at the meeting or unless he or she shall file a written dissent to such action with the person acting as secretary of the meeting before the adjournment thereof. Such right to dissent shall not apply to a Manager who voted in favor of such action.

(f) Action by Written Consent or Telephone Conference. Any action permitted or required to be taken at a meeting of the Board of Managers or of any committee designated by the Board of Managers may be taken without a meeting, without prior notice and without a vote if a consent or consents in writing, setting forth the action so taken, is signed by a quorum of the Managers as set forth in Section 9.7(e). Such consent shall have the same force and effect as a vote taken at a meeting and may be stated as such in any document or instrument filled with the Secretary of State of Delaware, and the execution of such consent or consents shall constitute attendance or presence in person at a meeting of the Board of Managers or any such committee, as the case may be. A telegram, telex, cablegram or similar transmission by a person, or a photographic, photostatic, facsimile or similar reproduction of a writing signed by a person, shall be regarded as signed by that person for the purposes of this Section 9.7(f).

Subject to the requirements of the Act, the Certificate of Formation or this Agreement, the Board of Managers, or members of any committee designated by the Board of Managers, may participate in and hold a meeting of the Managers or any committee of Managers, as the case may be, by means of a conference telephone or similar communications equipment by means of which all persons participating in the meeting can speak to and hear each other, and participation in such meeting shall constitute attendance and presence in person at such meeting, except where a person participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened.

Section 9.8. COMPENSATION. If approved by the Board of Managers and the Members, one or more Managers, or members of any committee designated by the Board of Managers, may be reasonably compensated for services rendered. Each Manager and members of any committee designated by the Board of Managers shall be reimbursed for actual expenses incurred in attending meetings.

Article X. OFFICERS

Section 10.1. NUMBER. The Officers of the Company shall be the President, a Secretary and a Treasurer, each of whom shall be elected by the Board of Managers. Such other Officers and assistant Officers and agents as may be deemed necessary may be elected or appointed by the Board of Managers. Any two (2) or more offices may be held by the same person. The Board of Managers hereby elects Larry Buffington as the initial President and Chief Executive Officer of the Company to serve until his successor his duly elected or until his earlier resignation or removal.

16

- Section 10.2. ELECTION AND TERM OF OFFICE. The Officers of the Company shall be elected annually by the Board of Managers at the regular meeting of the Managers held after each annual meeting of the Members. If the election of Officers shall not be held at such meeting, such election shall be held as soon thereafter as conveniently may be. Each Officer shall hold office until his successor shall have been duly elected and shall have qualified or until his earlier resignation, removal from office or death.
- **Section 10.3. REMOVAL AND RESIGNATION.** Any Officer or agent elected or appointed by the Board of Managers may be removed by the Board of Managers with or without cause, but such removal shall be without prejudice to the contract rights, if any, of the person so removed. Election or appointment of an Officer or agent shall not of itself create contract rights. Any Officer of the Company may resign at any time by giving written notice to the Board of Managers. Any such resignation shall take effect after time specified therein, or, if the time is not specified therein, upon its acceptance by the Board of Managers.
 - Section 10.4. <u>VACANCIES</u>. A vacancy, however occurring, in any office may be filled by the Board of Managers for the unexpired portion of the term.
- Section 10.5. CHAIR. The Chair of the Board of Managers, if one is elected or appointed, shall preside at all meetings of the Board of Managers and shall have such other powers and duties as may from time to time be prescribed by the Board of Managers, upon written directions given to him or her pursuant to resolutions duly adopted by the Board of Managers.
- Section 10.6. PRESIDENT. The President shall be the principal executive Officer of the Company and, subject to the control of the Board of Managers, shall in general supervise and control all of the business affairs of the Company. The President shall, when present, preside at all meetings of the Members and of the Board of Managers, unless the Board of Managers has elected a Chair of the Board of Managers and the Chair is present at such meeting. The President may sign deeds, mortgages, bonds, contracts, or other instruments which the Board of Managers has authorized to be executed, except in cases where the signing and execution thereof shall be expressly delegated by the Board of Managers or by this Agreement to some other Officer or agent of the Company, or shall be required by law to be otherwise signed or executed; and in general shall perform all duties as from time to time may be assigned to him by the Board of Managers.
- Section 10.7. <u>VICE PRESIDENTS</u>. If a Vice-President is elected or appointed, in the absence of the President or in the event of his death, inability or refusal to act, the Vice-President shall have the duties of the President, and when so acting, shall have all the powers of, and be subject to all the restrictions upon, the President. The Vice-President shall perform such other duties as from time to time may be assigned to him by the President or the Board of Managers.
- Section 10.8. SECRETARY. The Secretary shall: (a) keep the minutes of all the meetings of the Members and the Board of Managers in one or more books provided for that purpose; (b) see that all notices are duly given in accordance with the provisions of this Agreement or as required by law; (c) be custodian of the Company records; (d) keep a register of the post office address of each Member; (e) have general charge of the Membership Unit transfer books of the Company, if the Company issues certificates representing such Membership Units; and (f) in general perform all duties incident to the office of Secretary and such other duties as from time to time may be assigned to him by the President or by the Board of Managers.

17

- Section 10.9. TREASURER. The Treasurer shall: (a) have charge and custody of and be responsible for all funds and securities of the Company; receive and give receipts for moneys due and payable to the Company from any source whatsoever, and deposit all such moneys in the name of the Company in such banks, trust companies or other depositories as shall be selected by the Board of Managers; and (b) in general perform all of the duties incident to the office of Treasurer and such other duties as from time to time may be assigned to him by the President or by the Board of Managers. If required by the Board of Managers, the Treasurer shall give a bond for the faithful discharge of his duties in such sum and with such surety or sureties as the Board of Managers shall determine.
- Section 10.10. SALARIES. The salaries of the Officers shall be fixed from time to time by the Board of Managers and no Officer shall be prevented from receiving such salary by reason of the fact that he is also a Manager of the Company.

Article XI. INDEMNIFICATION OF MEMBERS, MANAGERS AND DIRECTORS

- Section 11.1. RIGHT TO INDEMNIFICATION. The Company shall indemnify the Members, Managers and Officers to the fullest extent allowed by law from all claims brought by third parties relating to or arising out of the Company's business. The Company shall indemnify and hold harmless the Members, Managers and Officers to the fullest extent permitted or authorized by the Act or future legislation or by current or future judicial or administrative decision (but, in the case of future legislation or decision, only to the extent that it permits the Company to provide broader indemnification rights than permitted prior to the legislation or decision, against all fines, liabilities, settlements, losses, damages, costs and expenses, including attorneys' fees, asserted against the Member, Manager and/or Officer or incurred by any of them in their capacity as Member, Manager and/or Officer or arising out of their istatus as Member, Manager and/or Officer, as the case may be. The foregoing right of indemnification shall not be exclusive of other rights to which those seeking indemnification may be entitled. The Company may maintain insurance, at its expense, to protect itself and the indemnified persons against all fines, liabilities, costs and expenses, including attorneys' fees, whether or not the Company would have the legal power to indemnify him directly against such liability.
- Section 11.2. <u>ADVANCES.</u> Costs, charges and expenses (including attorneys' fees) incurred by a person referred to in Section 11.1 in defending a civil or criminal suit, action or proceeding shall be paid by the Company in advance of the final disposition thereof upon receipt of an undertaking to repay all amounts advanced if it is ultimately determined that the person is not entitled to be indemnified by the Company as authorized by this Article.
- Section 11.3. SAVINGS CLAUSE. If this Article XI or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Company shall nevertheless indemnify and hold harmless the Members, Managers, Officers or any other person indemnified pursuant to this Article XI as to costs, charges and expenses (including attorneys' fee), judgments, fines and amounts paid in settlement with respect to any action, suit or proceeding, whether civil, criminal, administrative or investigative to the full extent permitted by any applicable portion of this Article XI that shall not have been invalidated and to the fullest extent permitted by applicable law.

Article XII. ASSIGNMENT OF MEMBERSHIP UNITS

Section 12.1. GENERAL RESTRICTION ON ASSIGNMENT. Except as expressly permitted hereunder, no Member or Economic Interest Owner may Transfer all or any portion of, or any interest or rights in, their Membership Units or Economic Interest without the prior written consent of the Board of Managers (excluding Managers who are Affiliates of the transferring Member) (the "Non-Affiliated Managers") or, if there are no Non-Affiliated Managers, then the consent of all of the Members other than the transferring Member (the "Non-Affiliated Members"). Each Member hereby acknowledges the reasonableness of this prohibition in view of the purposes of the Company and the relationship of the Members.

The Transfer of any Membership Units or Economic Interests in violation of the prohibition contained in this Section 12.1 shall be deemed invalid, null and void, and of no force and effect. Any Person to whom Membership Units are attempted to be Transferred in violation of this Section 12.1 shall not be entitled to vote on matters coming before the Members, participate in the management of the Company, act as an agent of the Company, receive distributions from the Company or have any other rights in or with respect to the Membership Units. Any person to whom an Economic Interest is attempted to be transferred in violation of this Section 12.1 shall not be entitled to receive distributions from the Company or have any other rights in or with respect to the Economic Interest.

For purposes of this Agreement, the term "Transfer", or "Transferred", when used in this Agreement with respect to the Membership Units or Economic Interests, includes a sale, assignment, gift, pledge, encumbrance or any other disposition, whether voluntary, by operation of law or otherwise, and "transferee" and "transferer" have corresponding meanings. Further, any such proposed Transfer must be accomplished by written instrument satisfactory in form and content to the Non-Affiliated Managers or Non-Affiliated Members (as the case may be), accompanied by such assurances of genuineness and effectiveness of signatures and the obtaining of any governmental approvals or legal opinions as the Non-Affiliated Managers or Non-Affiliated Members (as the case may be) may reasonably require and payment of any reasonable costs of transfer as the Non-Affiliated Managers or Non-Affiliated Members (as the case may be) may require and all transfer taxes as may be imposed.

In the event of a "<u>change of control</u>" (as defined hereinbelow) of the Class A Holder, the Class C Holder shall be deemed to have made an offer to sell all of his Membership Units and Economic Interests to the Company immediately prior to the effectuation of the change of control in accordance with the terms of Section 12.2 hereinbelow; *provided, however*, that for purposes of this provision, the per Membership Unit purchase price for the Offered Interest shall be the Fair Market Value (as defined in Article XVI hereinbelow) of the Offered Interest and the terms of payment shall be the payment terms set forth in Section 12.2(d) below. For purposes of this provision, the term "<u>change of control</u>" shall mean a transaction or series of transactions (related or unrelated) which result in the Class A Holder no longer owning, immediately after such transaction or series of transactions, in excess of fifty percent (50%) (determined by vote and value) of the issued and outstanding equity securities or beneficial interests of the Company.

Section 12.2. RIGHT OF FIRST REFUSAL. In addition to the general restrictions contained in Section 12.1 above, no Member or Economic Interest Owner may voluntarily Transfer any right, title or interest in its Membership Units or Economic Interest, or any part thereof, to any Person, unless the Member or Economic Interest Owner desiring to make the transfer, hereinafter referred to

19

as the "<u>Transferor</u>", shall (a) have received a *bona fide* written offer to purchase such interest in the Company by an unrelated third party (the "<u>Proposed Transferee</u>") and (b) have made an offer to tender to the Company for redemption all of the Transferor's interest in the Company which the Transferor desires to sell, give or otherwise transfer to the Proposed Transferee (the "<u>Offered Interest</u>") in the manner hereinafter described in this Section 12.2, and the offer shall not have been accepted.

- (a) Offer by Transferor. The offer by a Transferor shall be given in writing to the Company and shall be accompanied by a copy of the written offer, proposal or contract, between the Transferor and the Proposed Transferee (hereinafter referred to as the "Written Offer"), and shall set forth the nature of the transaction (whether sale, gift, or other transfer), the name and address of the Proposed Transferee and the terms of the transaction, including an identification of the interest involved, the number of Membership Units involved, the purchase price, and the payment terms (hereinafter referred to as the "Proposed Transaction"). The offer by the Transferor shall consist of an offer to tender to the Company for redemption all of the Offered Interest at the price and upon the terms set forth in this Section 12.2 below.
- (b) Exercise of Option. Within fifteen (15) days after the receipt of the offer, the Company shall have the option, but not the duty, to redeem all or any portion of the Offered Interest; provided, however, that the Company's decision to accept or reject the offer shall be made by a vote of a majority of the Non-Affiliated Managers or, if there are no Non-Affiliated Managers, then by a Majority Vote of the Non-Affiliated Members. The election to redeem shall be exercised by the giving of notice thereof to the Transferor.

The notice of exercise of option shall specify a date for the closing of the redemption of the Offered Interest (hereinafter referred to as the "Closing" or the "Closing Date"), which shall not be less than thirty (30) days nor more than sixty (60) days after the expiration of the time within which the Company may exercise its option.

- (c) Redemption/Purchase Price. If the Company elects to redeem all of the Offered Interest, the price for each Membership Unit of the Offered Interest shall be the price per Membership Unit set forth in the Written Offer
 - (d) Payment of Redemption/Purchase Price. Upon any redemption under this Section 12.2, the redemption price shall be paid in accordance with the terms provided in the Written Offer.
- (e) Closing. The Closing of a purchase under this Section 12.2 shall take place at the principal office of the Company on a date specified in writing in the written acceptance by the Company to the Transferor, unless the Transferor and the Company otherwise mutually agree on another place or date. At the Closing, the Transferor shall deliver, in exchange for the total purchase price, whether in cash or partially in cash and partially by promissory notes, as the case may be, the certificate, if any, representing the Offered Interest being transferred, duly endorsed, and such other documents as shall be necessary and reasonably required to conclude the transfer.
- (f) Release from Restriction. If the Transferor's offer to tender or sell is not accepted by the Company as to the entire interest of the Transferor, the Transferor may make a transfer to the prospective bona fide transferee of the entire Offered Interest, the transfer to

20

be made only in strict accordance with the terms of the Written Offer. If the Transferor shall fail to make the transfer within thirty (30) days following the expiration of the time hereinabove provided for the exercise of the election to purchase by the Company, the Membership Units of the Transferor shall again be subject to all the restrictions of this Agreement.

- (g) Continuance of Restrictions Upon Subsequent Owners. In the event the Transferor makes a bona fide transfer of the Offered Interest under the provisions of this Section 12.2, then the Offered Interest transferred to the Proposed Transferee shall be subject to all the provisions of this Agreement. No Membership Units or Economic Interest shall be transferred on the books of the Company and no certificate evidencing such Membership Units or Economic Interest shall be issued to the Proposed Transferee unless and until the Proposed Transferee has executed a counterpart to this Agreement, the original of which shall be retained as part of the Company's records. Failure of the Proposed Transferee to execute a counterpart of this Agreement, however, shall not affect the applicability of this Agreement to the Offered Interest, it being the intention of each Member and the Company that any and all subsequent owners of Membership Units and Economic Interests voluntarily transferred shall only receive and own the Membership Unit or Economic Interest subject to the same restrictions upon transfer and encumbrance as set forth in this Agreement, to which the Transferor was subject, including, without limitation, all of the provisions of this Article XII.
- Section 12.3. INVOLUNTARY TRANSFER. In the event any Membership Units or Economic Interest are the subject of an involuntary Transfer, whether due to bankruptcy, assignment for benefit of creditors, judicial order, legal process, divorce, execution, attachment, enforcement of a pledge or other encumbrance, or otherwise (hereinafter referred to as the "Affected Interest"), the Member or Economic Interest Owner owning the Affected Interest shall be deemed to have made, immediately prior to such involuntary transfer, an offer first to tender to the Company for redemption all of the Affected Interest in the manner hereinafter described in this Section 12.3. There shall be no obligation or requirement that the Company redeem any of the Affected Interest under this Section 12.3, any redemption of the Affected Interest being solely upon election to do so. The Company may redeem all or any portion of the Affected Interest.
 - (a) Exercise of Option. If the Company elects to redeem all or any portion of the Affected Interest in accordance with this Section 12.3, the Company shall serve notice in writing of its election upon the Member owning the Affected Interest and the creditor(s) of the Member, spouse (in the event of a divorce) or other person or entity who is to be the recipient of the Affected Interest (hereinafter referred to collectively as the "Transferor") within ninety (90) days after the Board of Managers shall have received actual notice of the involuntary transfer. The Company's decision to accept or reject the offer shall be made by a majority vote of the Non-Affiliated Managers, then by a Majority Vote of the Non-Affiliated Members.

The notice of exercise of option shall specify a date for the closing of the redemption of the Affected Interest (hereinafter referred to as the "Closing" or the "Closing Date"), which shall not be less than thirty (30) days nor more than ninety (90) days after the expiration of the time within which the Company may exercise its option.

(b) Redemption/Purchase Price. If the Company elects to redeem all or any portion of the Affected Interest, the price for each Membership Unit of the Affected Interest

21

shall be the lesser of: (i) the Fair Market Value (as defined in Article XVI hereinbelow) of the Affected Interest, or (ii) the total amount, including acquisition costs, if any, which had been due to the creditor of the Member who was to be the recipient of the Affected Interest.

- (c) Payment of Redemption/Purchase Price. Upon any redemption under this Section 12.3, the redemption price shall be paid in the same manner provided in Section 12.2(d) above.
- (d) Closing. The closing of a purchase under this Section 12.3 shall take place at the principal office of the Company on a date specified in writing in the written acceptance by the Company to the Transferor, unless the Transferor and the Company otherwise mutually agree on another place or date. At the Closing, the Transferor shall deliver, in exchange for the total purchase price, whether in cash or partially in cash and partially by promissory notes, as the case may be, the certificate, if any, representing the Affected Interest being transferred, duly endorsed, and such other documents as shall be necessary and reasonably required to conclude the transfer.
- (e) Continuance of Restrictions Upon Transferor. In the event the "deemed offer" of the Transferor is not accepted by the Company as to all of the Affected Interest, then the portion of the Affected Interest not redeemed under this Section 12.3 may be transferred to the transferee/creditor subject to all of the provisions of this Agreement. The transferee/creditor shall execute a counterpart of this Agreement, the original of which shall be retained as part of the Company's records. The failure of the transferee/creditor to execute a counterpart to this Agreement shall not affect the applicability of this Agreement to the Affected Interest, it being the intention of each Member and the Company that any and all subsequent owners of Membership Units or Economic Interests acquired pursuant to an involuntary transfer shall only receive and own the Membership Units or Economic Interests subject to the restrictions upon transfer and encumbrance as set forth in this Agreement to which the Original Member was subject.

- Section 12.4. TERMINATION OF EMPLOYMENT. In the event that Larry Buffington's ("Buffington") employment with the Company is terminated for any reason (including but not limited to death, voluntary severance or permanent disability) (the "Terminating Event"), then the Company may redeem, within six (6) months after the Terminating Event, all of the Class C Units in the Company owned by Buffington on the following terms:
 - (a) For twenty-percent (20%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within one year of the effective date of this Agreement;
 - (b) For forty-percent (40%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within the second year after the effective date of this Agreement;
 - (c) For sixty-percent (60%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within the third year after the effective date of this Agreement;

22

- (d) For eighty-percent (80%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within the fourth year after the effective date of this Agreement.
- (e) For the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within or after the fifth year after the effective date of this Agreement.

The redemption price shall be paid within ninety (90) days after the Company notifies Buffington or his successor of the election to redeem his Class C Units. The redemption price shall be paid in cash to the extent it will not impair the operations of the Company, and the balance shall be paid by a promissory note payable on the first anniversary of the Termination Event with interest at Prime Rate published in the Wall Street Journal, adjusted quarterly.

- Section 12.5. WITHDRAWALS. No Member may withdraw from the Company, except upon the prior written consent of the Board of Managers.
- Section 12.6. RIGHTS OF ASSIGNEE OF MEMBERSHIP UNITS. Unless the assignee of a Membership Unit is admitted as a Substituted Member as provided in Section 12.7, the assignee will be merely an Economic Interest Owner, and the assignee's rights shall be limited to sharing in the profits to which the assignor would otherwise have been entitled and to receiving the assignor's share of any proceeds and an accounting upon dissolution. The dissepanded for purposes of determining whether Members owning the required Membership Units have voted on any matter requiring a vote of the Members or in determining the total of the Membership Units outstanding for voting purposes. For example, in the event a judgment creditor obtains a charge against a Member's Membership Unit, pursuant to Section 608.433(4) of the Act, or any successor provision, then unless the judgment creditor is admitted as a Substituted Member, the judgment creditor will be merely an Economic Interest Owner and will not acquire any other rights of a Member. All remaining rights and interest in the Membership Units which were owned by the Transferring Member immediately prior to the Transfer and that were associated with the assigned Economic Interest Owner or to a successor or transferree of such Economic Interest Owner.
- Section 12.7. SUBSTITUTED MEMBER. An assignee of the whole or any portion of a Member's Membership Interests in the Company, validly assigned under this Article XII, may become a Substituted Member in the place of his assignor(s), to the extent of the Membership Interests validly assigned, if all of the following conditions are satisfied:
 - (a) A fully executed and acknowledged written instrument of assignment has been filed with the Company which sets forth the intention of the assignor(s) that the assignee become a Substituted Member in his/their place, to the extent of the Membership Interests assigned.

23

- (b) The assignee executes, acknowledges and delivers to the Managers a written acceptance and adoption of the provisions of this Agreement, in form and substance acceptable to the Managers in their sole discretion
 - (c) The assignee pays a transfer fee to the Company in an amount sufficient to cover all reasonable expenses connected with the admission of such person as a Substituted Member.
- Section 12.8. SECURITIES LAWS. The Membership Units have not been registered under the Federal or state securities laws of any state and, therefore, may not be resold unless appropriate Federal and state securities laws. as well as the other provisions of this 25 have been complied with.
- Section 12.9. INVALID TRANSFER. No Transfer of a Membership Unit or Economic Interest that is in violation of this Article XII shall be valid or effective, and the Company shall not recognize any improper transfer for the purposes of making allocations, payments of profits, return of capital contributions or other distributions with respect to such Membership Unit or Economic Interest. The Company may enforce the provisions of this Article XII either directly or indirectly or through its agents by entering an appropriate stop transfer order on its books or otherwise refusing to register or transfer or permit the registration or transfer on its books of any proposed transfers not in accordance with this Article XII.
- Section 12.10. <u>DISTRIBUTIONS AND ALLOCATIONS IN RESPECT OF A TRANSFERRED INTEREST</u>. If any Member Transfers any part of a Membership Unit or Economic Interest in the Company during any accounting period in compliance with the provisions of this Article XII, Company income, gain, deductions and losses attributable to such interest for the respective period shall be divided and allocated between the transferor and the transferee by taking into account their varying interests during the applicable accounting period in accordance with Code section 706(d), using the daily proration method. All Company distributions on or before the effective date of such transfer shall be made to the transferor and all such Company distributions, the Company shall recognize a transfer on the day following the day of transfer. Neither the Company nor any Member shall incur any liability for making Company allocations and distributions in accordance with the provisions of this Section 12.10.
 - Section 12.11. <u>DRAG-ALONG RIGHTS</u>. Notwithstanding anything in this Agreement to the contrary, in the event that either the Class A Holders or the Class B Holders elect to sell all of their Membership Units in a single transaction to a <u>bona fide</u> third party purchaser, either the Class A Holders or the Class B Holders, as the case may be, may require all of the other Members (the "<u>Affected Members</u>") to sell all of their Membership Units for the same price and on the same terms as those to be received by either the Class A Holders or the Class B Holders, as the case may be. In such case, the provisions of Sections 12.1 and 12.2 hereof shall not apply to such Transfer and the Affected Members shall be deemed to have made an offer to sell all of their Membership Units on such terms and conditions as those under which either the Class A Holder or the Class B Holder, as the case may be, except representations relating to the Affected Members' ownership of the Membership

24

Units, their ability to deliver marketable title to such Membership Units and similar representations.

- (b) Notwithstanding anything in this Agreement to the contrary, in the event that either the Class A Holders or the Class B Holders elect to sell all of their Membership Units in a single transaction to a bona fide third party purchase, either the Class A Holders or the Class B Holders, as the case may be, shall require the bona fide third party purchaser to offer to the Affected Members to purchase all their Membership Units for the same price and on the same terms as those to be received by either the Class B Holders, as the case may be. If the third party purchaser will not make such offer to the Affected Members, then the Class A Holders or the Class B Holders, as the case may be, shall not be allowed to sell their Membership Units to such third party purchaser.
- Section 12.12. <u>APPLICABILITY OF PROVISIONS TO CLASS A HOLDERS</u> The provisions of Sections 12.1 and 12.2 shall not apply to the Membership Units owned by any Class A Holder, and any Class A Holder may transfer any or all of its Membership Units without the consent of the Board of Managers or the Members, subject to the provisions of Section 12.12(b) below.
- (b) At any time a Class A Holder reasonably anticipates that it has a good faith desire to sell some or all of its Membership Units, such Class A Holder shall notify the Class B Holders of that desire and determine whether the Class B Holders have an interest in purchasing all the Membership Units owned by such Holders (the "Offered Units"). If the Class B Holders, or any of them, have such an interest, eagreement within respect thereto. If the parties have not entered into a definitive purchase agreement with respect thereto. If the parties have not entered into a definitive purchase greement with respect thereto. If the parties have not entered into a definitive purchase greement within innety (90) days after the original notice from the Class A Holder, the Class A Holder may offer and sell the Offered Units to a third party purchaser (the "Third Party Purchaser"). If the Class A Holder shall fail to locate a Third Party Purchaser to purchase the Offered Units within one hundred eighty (180) days following the expiration of the time hereinabove provided for, the Offered Units shall again be subject to all the restrictions of this Agreement.
- (c) Continuance of Restrictions Upon Subsequent Owners. In the event a Class A Holder sells the Offered Units to a Third Party Purchaser under the provisions of this Section 12.12, then the Offered Units transferred to the Third Party Purchaser shall be subject to all the provisions of this Agreement. No Membership Units or Economic Interest shall be transferred on the books of the Company and no certificate evidencing such Membership Units or Economic Interest shall be issued to the Third Party Purchaser unless and until the Third Party Purchaser has executed a counterpart to this Agreement, the original of which shall be retained as part of the Company's records. Failure of the Third Party Purchaser to execute a counterpart of this Agreement to the Offered Units, it being the intention of each Member and the Company that any and all subsequent owners of Membership Units and Economic Interests voluntarily transferred shall only receive and own the Membership Units or Economic Interests subject to the same restrictions upon transfer and encumbrance as set forth in this Agreement, to which the Class A Holder was subject, including the applicable provisions of this Article XII.

25

Subject to Section 9.6, the Board of Managers shall have the power and authority to effect the conversion of the Company's business form from a limited liability company to a corporation (for any reason whatsoever, including without limitation, a public offering of the Company's Capital Securities) or the merger of the Company with or into a new or previously-established but dormant corporation having no assets or liabilities, debts or other obligations of any kind whatsoever other than those associated with its formation and initial capitalization (such a conversion or merger is referred to as a "Conversion" and such corporation is referred to as "Newco").

Upon the consummation of a Conversion, the Units held by each holder thereof shall thereupon be converted into a number of shares of Newco's Capital Securities containing the economic and other terms and rights relative to each other holder of Units as the Board of Managers' determine to be as nearly as practicable in all material respects the same as such holder's Units as provided herein. The Board of Managers' determination of the class (and the terms thereof and rights associated therewith) and number of shares of Newco Capital Securities that each Member receives upon a Conversion shall be final and binding on the holders of Units absent manifest arithmetic error.

In connection with a Conversion, each Member hereby covenants and agrees to take any and all such action and execute and deliver any and all such instruments and other documents as the Board of Managers may reasonably request in order to effect or evidence such Conversion. Without limiting the generality of the foregoing, no Member shall have or be entitled to exercise any dissenter's rights, appraisal rights or other similar rights in connection with such Conversion.

Article XIV. BOOKS AND RECORDS

Section 14.1. BOOKS AND RECORDS. The Board of Managers shall keep or cause to be kept complete books and records of the Company and supporting documentation of the transactions with respect to the conduct of the Company's business. These and all other records of the Company, including information relating to the Company's activities, information with respect to the sale by a Member or any Affiliate of goods or services to the Company, and a list of the names and business addresses of all Members shall be kept at the offices of the Company, or at such other location as may be determined by the Board of Managers, and shall be available for examination there by any Member, or his duly authorized representative, at reasonable times upon reasonable notice. Any Member, or his duly authorized representative, upon paying the costs of collection, duplication and mailing, shall be entitled to a copy of the list of names and addresses of the Members. The books and records shall be maintained in accordance with sound accounting practices.

The Company will also cause Liquidmetal Coatings Solutions, LLC, a Delaware limited liability company ("LMCS"), its wholly owned subsidiary, to keep complete books and records in the same manner as required above.

26

Section 14.2. CUSTODY OF MEMBER FUNDS; BANK ACCOUNTS

- (a) The Board of Managers shall have fiduciary responsibility for the safekeeping and use of all funds and assets of the Company, whether or not in the immediate possession or control of the Board of Managers. The funds of the Company shall not be commingled with the funds of any other person and the Board of Managers shall not employ, or permit any other person to employ, such funds in any manner except for the benefit of the Company.
- (b) All funds of the Company not otherwise invested shall be deposited in one or more accounts maintained in such banking institutions as the Board of Managers shall determine, and withdrawals shall be made only in the regular course of Company business on such signature or signatures as the Board of Managers may, from time to time, determine.
- Section 14.3. ACCOUNTANTS. The accountants for the Company shall be such certified public accountants as shall be selected by the Board of Managers.
- Section 14.4. SECTION 754 ELECTION. In the event of a distribution of Company Property (other than money) to a Member or upon a transfer of all or any part of the Membership Units of a Member, the Board of Managers may in their sole and absolute discretion, upon the written request of the Member receiving the distribution or the transferee of the Membership Units, as the case may be (the "Electing Member"), elect pursuant to Section 754 of the Code, to adjust the basis of the Company's property in the manner provided in Sections 754 and 743 thereof, respectively. Each Member agrees to furnish the Company with all information necessary to give effect to such election. The election will be filed with the Company information tax return for the first taxable year to which the election applies. If the Board of Managers decide to make the Section 754 election, the Electing Member will be responsible for all additional accounting costs incurred by the Company as a result of the Electing Member's request to make the election under Code Section 754.
 - Section 14.5. FISCAL YEAR. The fiscal year of the Company shall be the calendar year.
- Section 14.6. TAX MATTERS PARTNER. LMT shall be the "Tax Matters Partner" of the Company for purposes of Section 6231(a)(7) of the Code. The Tax Matters Partner shall have the power and authority, subject to the review and control of the Board of Managers, to manage and control, on behalf of the Company, any administrative proceeding at the Company level with the Internal Revenue Service relating to the determination of any item of Company income, gain, loss, deduction or credit for federal income tax purposes. The Tax Matters Partner may be removed, and a new Tax Matters Partner appointed, by the Board of Managers in accordance with the Code and the Treasury Regulations.
- Section 14.7. ANNUAL FINANCIAL STATEMENTS AND TAX RETURN INFORMATION. The Company shall provide to C³ Capital Partners, L.P. ("C³"), and C³ Capital Partners II, L.P. ("C³ II"), as soon as available, and, in any event, within one hundred and twenty (120) days after the end of each fiscal year of the Company, beginning with the fiscal year ending December 31, 2007, a copy of the annual consolidated financial statements of the Company and LMCS for such fiscal year containing balance sheets, statements of income, retained earnings and cash flows at the end of such fiscal year, in each, except for the fiscal year ending December 31, 2007, setting forth in comparative form the figures for the preceding fiscal year (if any), all in reasonable detail and audited and certified by independent certified public accountants of recognized

27

standing reasonably acceptable to C³ and C³ II, to the effect that such report has been prepared in accordance with GAAP. Within ninety (90) days after the end of each Tax year, Company shall provide sufficient information with respect to the Company for such Tax year necessary for C³ and C³ II to prepare their federal, state and local income Tax Returns. The Company shall provide copies of the federal, state and local income Tax Returns of the Company for each Tax year promotive after the filling thereof.

Section 14.8. MONTHLY FINANCIAL STATEMENTS. The Company shall provide to C³ and C³ II, as soon as available and, in any event, within thirty (30) days after the end of each month, a copy of an unaudited consolidated financial report of the Company and LMCS as of the end of such month and for the portion of the fiscal year then ended, including balance sheets, statements of income, retained earnings and cash flows, setting forth in each case comparisons to the Company's and LMCS's Annual Budget and to the corresponding period in the preceding fiscal year and a brief narrative explaining the results of the Company's and LMCS's operations for such period; all such statements shall have been prepared in accordance with GAAP (absent footnotes and customary year-end adjustments) and fairly present the financial condition and results of operations of the Company and LMCS at the date and for the periods indicated therein.

Section 14.9. INSPECTION RIGHTS. At any reasonable time and from time to time, the Company shall permit representatives of C³ and C³ II to examine, copy and make extracts from its books and records, to visit and inspect the Company's and LMCS's properties, and to discuss Company's and LMCS's business, operations, and financial condition with Company's or LMCS's officers, employees and independent certified public accountants, provided that C³ and C³ II agree to keep such information confidential and to use such information solely in connection with its investment in the Membership Units.

Article XV. EMPLOYMENT AGREEMENTS

The Company shall neither (i) modify, amend or terminate the confidentiality, nonsolicitation and noncompetition agreements between the Company and Larry Buffington (the "Employment Agreements") without the prior written consent of C³ and C³ II, which consent shall not be unreasonably withheld or delayed nor (ii) provide any employee, who is a party to the Employment Agreements, with any compensation, benefit or reimbursement for any expenses related to the performance of such employee's duties as contemplated by such employee's Employment Agreements, except as provided by the Employment Agreements.

Article XVI. DEFINITIONS

- Section 16.1. DEFINITIONS. Unless the context otherwise requires, the terms defined in this 29 shall, for the purposes of this Agreement, have the meanings herein specified.
- "Act" means the Delaware Limited Liability Company Act, as amended.
- "Affiliate" or "Affiliate" or "Affiliate" or "Affiliate" means, with respect to any Member, a partner of a Member; any member of the immediate family of any Member; any shareholder, officer or director of a Member or any member of their respective immediate families; any person, firm or entity which, directly or indirectly, controls, is controlled by, or is under common control with a Member, any

28

partner of any Member or any shareholder, officer or director of a partner of any Member or their respective families; or any person, firm or entity which is associated with a Member, any partner of a Member, any officer, director or shareholder of a Member or any member of their respective immediate families in a joint venture, partnership or other form of business association. In this definition, the term "control" shall mean the ownership of ten percent (10%) or more of the beneficial interest in the firm or entity referred to, and the term "immediate family" shall mean the spouse, ancestors, lineal descendants, brothers and sisters of the person in question, including those adopted. To the extent the term "Affiliated Party" is used in the context of an affiliation with a Person (the "Subject") other than a Member then such term shall have the same meaning above; however, the term "Member" within such definition shall be replaced with the Subject.

- "Agreement" means this Limited Liability Company Operating Agreement, as amended, modified, supplemented or restated from time to time in accordance with the terms hereof.
- "Business Day," means any day other than a Saturday, a Sunday, or a holiday on which national banking associations in the State of Delaware are closed.
- "Capital Account" means, with respect to any Member, the account maintained for such Member in accordance with the provisions of Section 3.4.
- "Capital Contribution" means, with respect to any Member, the aggregate amount of money and the initial Gross Asset Value of any property (other than money) contributed to the Company pursuant to Section 3.2.

"Capital Securities" means as to any Person that is a corporation, the authorized shares of such Person's capital stock, including all classes of common, preferred, voting and nonvoting capital stock, and, as to any Person that is not a corporation or an individual, the ownership or membership interests in such Person, including, without limitation, the right to share in profits and losses, the right to receive distributions of cash and property, and the right to receive allocations of items of income, gain, loss, deduction and credit and similar items from such Person, whether or not such interests include voting or similar rights entitling the holder thereof to exercise control over such Person.

"Capital Transaction" means any of the following events: (A) any sale or other disposition of all or any part of any capital assets of the Company, (B) any loans secured by all or any part of the capital assets of the Company, (C) the refinancing of any Company indebtedness, (D) the condemnation of all or any part of the capital assets of the Company, or (E) any insurance recovery relating to any capital assets owned by the Company.

"Class A Holder" means a Member holding Class A Units.

"Class B Holder" means a Member holding Class B Units.

"Class C Holder" means a Member holding Class C Units.

"Class A Percentage Interest" means, with respect to a Class A Holder at any time and from time to time, a percentage equal to a fraction, the numerator of which is the number of Class A

29

Units owned by such Class A Holder and the denominator of which is the aggregate number of Common Units owned by all Members.

"Class B Percentage Interest" means, with respect to a Class B Holder at any time and from time to time, a percentage equal to a fraction, the numerator of which is the number of Class B Units owned by such Class B Holder and the denominator of which is the aggregate number of Common Units owned by all Members.

"Class C Percentage Interest" means, with respect to a Class C Holder at any time and from time to time, a percentage equal to a fraction, the numerator of which is the number of Class C Units owned by such Class C Holder and the denominator of which is the aggregate number of Common Units owned by all Members;

"Class A Unit" means a Unit representing a fractional part of the Member Interests of the Members and having the rights and obligations specified with respect to the Class A Units in this Agreement.

"Class B Unit" means a Unit representing a fractional part of the Member Interests of the Members and having the rights and obligations specified with respect to the Class B Units in this Agreement.

"Class C. Unit" means a Unit representing a fractional part of the Member Interests of the Members and having the rights and obligations specified with respect to the Class C. Units in this Agreement.

" $\underline{\textbf{Code}}$ " means the U.S. Internal Revenue Code of 1986, as amended, and the Treasury Regulations promulgated thereunder.

"Company" means the Delaware limited liability company that is the subject of this Agreement.

"Company Property" means all real and personal property acquired by the Company and any improvements thereto, including, without limitation, any tangible or intangible property of the Company.

"Conversion" has the meaning specified in Article XIII of this Agreement.

"Deficit Capital Account Balance" means, with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant fiscal year, after giving effect to the following adjustments:

- (a) A credit to such Capital Account of any amount which such Member is obligated to restore or is deemed to be obligated to restore pursuant to Treasury Regulations Sections 1.704-2(g) and 1.704-2(j)(5); and
- (b) A debit to such Capital Account of the items described in Sections 1.704-1(b)(2)(ii)(d)(4), (5) and (6) of the Treasury Regulations.

"Economic Interest" shall mean a Member's or Economic Interest Owner's pro rata share of the Company's profits and distributions of the Company's assets pursuant to this Operating

30

Agreement and the Act, but shall not include any right to participate in the management of affairs of the Company, including, the right to vote on, consent to or otherwise participate in any decision of the Members or Managers.

"Economic Interest Owner" shall mean the owner of an Economic Interest who is not a Member.

"Fair Market Value" of a Membership Interest shall mean the amount the Selling Member or Transferor would receive with respect to its Membership Interest upon the dissolution and termination of the Company assuming (A) such dissolution or termination occurred on the date of the Notice, and (B) the assets of the Company were sold for their fair market value without compulsion for the Company to sell such assets. The fair market value of the Company is assets shall be agreed upon by the Selling Member or Transferor, as the case may be, and the Company. If they are unable to agree upon a value within ten (10) days, then they shall agree upon an appraiser who shall determine the value. However, if they are unable to agree upon an appraiser within five (5) working days after either party serves written demand on the other, then each shall select one appraiser and the two appraisers so selected shall select a third appraiser whose determination shall be conclusive and binding for this purpose. In the event that the Company or the Selling Member or Transferor, as the case may be, fails to designate an appraiser pursuant to the preceding sentence within five (5) working days after either party serves written demand on the other, then the appraisal shall be divided equally between the Company and the Selling Member or Transferor, as the case may be.

"Fiscal Year" means the fiscal year of the Company which shall be the calendar year unless otherwise required by the Code

"Lien" means (a) any encumbrance, mortgage, pledge, lien, charge or other security interest of any kind upon any property or assets of any character, or upon the income or profits therefrom; (b) any acquisition of or agreement to have an option to acquire any property or assets upon conditional sale or other title retention agreement, device or arrangement (including a capitalized lease); or (c) any sale, assignment, pledge or other transfer for security of any accounts, general intangibles or chattel paper, with or without recourse; excluding in each instance the lien of this Agreement.

"Manager" has the meaning specified in Article IX of this Agreement.

"Member" means the persons listed on Exhibit "A" hereto, and includes any Person admitted as an additional Member or a substitute Member pursuant to the provisions of this Agreement, in such Person's capacity as a member of the Company, and "Members" means two (2) or more of such Persons when acting in their capacities as members of the Company.

"Membership Rights" means all legal and beneficial ownership interests in, and rights and duties as a Member of, the Company, including, without limitation, the right to share in Profits and Losses, the right to receive distributions of cash and other property from the Company, and the right to receive allocations of items of income, gain, loss, deduction and credit and similar items from the Company.

"Membership Units" shall mean the units into which the ownership interests of the Members in the Company are divided, including such Member's Economic Interest and the right of such Member to any and all benefits to which such Member may be entitled as provided in this

31

Agreement or under the Act, together with the obligation of such Member to comply with all of the provisions of this Agreement and of the Act.

"Newco" has the meaning specified in Article XIII of this Agreement.

"Nonrecourse Deductions" has the meaning set forth in Sections 1.704-2(b) and 1.704-2(c) of the Regulations. The amount of Nonrecourse Deductions for a Company fiscal year equals the net increase in Partnership Minimum Gain during that fiscal year determined pursuant to Section 1.704-2(d) of the Regulations reduced (but not below zero) by the aggregate distributions during that fiscal year of proceeds of a Nonrecourse Liability that are allocable to an increase in Partnership Minimum Gain, determined according to the provisions of Sections 1.704-2(h) of the Regulations.

"Nonrecourse Liability" has the meaning set forth in Section 1.752-1(a)(2) of the Regulations

"Partner Minimum Gain" means an amount, with respect to each Partner Nonrecourse Debt, equal to the Partnership Minimum Gain that would result if such Partner Nonrecourse Debt were treated as a Nonrecourse Liability, determined in accordance with Section 1.704-2(i) of the Regulations.

"Partner Nonrecourse Debt" has the meaning set forth in Section 1.704-2(b)(4) of the Regulations.

"Partner Nonrecourse Deductions" has the meaning set forth in Section 1.704-2(i)(2) of the Regulations. The amount of Partner Nonrecourse Deductions with respect to a Partner Nonrecourse Debt for a Company fiscal year equals the net increase during the year in Partner Minimum Gain attributable to such Partner Nonrecourse Debt reduced (but not below zero) by proceeds of the liability distributed during that fiscal year to the Member bearing the economic risk of loss for the liability that are both attributable to the liability and allocable to an increase in Partner Minimum Gain attributable, determined according to the provisions of Section 1.704-2(h) of the Regulations.

"Partnership Minimum Gain" has the meaning set forth in Regulations Sections 1.704-2(b)(2) and 1.704-2(d).

"Percentage Interest" means, with respect to a holder of Common Units at any time and from time to time, the Class A Percentage Interest, Class B Percentage and/or Class C Percentage held by such holder

"Person" includes any individual, corporation, association, partnership (general or limited), joint venture, trust, estate, limited liability company, or other legal entity or organization.

"Requisite Majority" means the approval or vote of the holders owning at least a majority of the Company's issued and outstanding Membership Units at any given time.

"Supermajority" means the approval or vote of the holders owning at least eighty-five percent (85%) of the Company's issued and outstanding Membership Units at any given time.

"Tax Matters Partner" has the meaning specified in Section 14.6.

32

"Transfer" means any sale, assignment, encumbrance, hypothecation, pledge, conveyance in trust, gift, transfer by bequest, devise or descent, or other transfer or disposition of any kind, including, but not limited to, transfers to receivers, levying creditors, trustees or receivers in bankruptcy proceedings or general assignees for the benefit of creditors, whether voluntary or by operation of law, directly or indirectly.

"Treasury Regulations" means such regulations implemented by the Internal Revenue Service pursuant to the Code.

Article XVII. AGREEMENT PREPARED BY ATTORNEY FOR COMPANY

The parties acknowledge that the counsel for the Company, Foley & Lardner, prepared this Agreement on behalf of and in the course of its representation of the Company, and that:

- (a) Each Member and Manager has been advised that a conflict of interest may exist among the Members, the Board of Managers and the Company; and
- (b) Each Member and Manager has been urged and has had the opportunity to seek the advice of independent legal counsel.

Article XVIII. AMENDMENTS

Except as otherwise expressly provided herein, the written approval of the Members holding a Supermajority, as that term is defined herein, of the Membership Units in the Company shall be required to alter, modify or amend this Agreement; provided, however, that no alteration, modification or amendment of Articles III, IV, VI or VII hereof or this Article XVIII which would materially and adversely affect the economic interest of one or more Members, or their successors or assigns, may be made without the unanimous written consent of all such Members so adversely affected. Notwithstanding the above, this Agreement may be amended from time to time by the Board of Managers, without the consent of any of the Members, (i) to add to the representations, duties or obligations of the members of the Board of Managers, (ii) to cure any ambiguity, to correct or supplement any provision herein which may be inconsistent with any other provisions with respect to matters or questions arising under this Agreement and (iii) to delete or add any provision of this Agreement required to be so deleted or added by the staff of the Securities and Exchange Commission or by any state securities commission or similar official, which addition or deletion is deemed by such commission or official to be for the benefit or protection of the Members.

Article XIX. MANAGER AND MEMBER RELATIONS TO THE COMPANY

No Manager or Member shall be restricted in any way from engaging in any other business venture or activity and no Manager or Member shall be accountable to the Company or to any other Manager or Member because of any activity or venture which does not directly involve the Company. Neither the Company nor the Members shall have any right under this Agreement in and to the other activities of any Manager or Member or to their income or profits from such business venture or activity.

33

Article XX. CHOICE OF LAW; SUBMISSION TO JURISDICTION; AND WAIVER OF JURY TRIAL

Section 20.1. LAW. This Agreement shall be governed by and construed under the laws of the State of Delaware. The parties agree that any action brought by any party under or in relation to this Agreement, including without limitation to interpret or enforce any provision of this Agreement, shall be brought in, and each party agrees to and does hereby submit to the jurisdiction and venue of, any state or federal court located in Delaware.

Section 20.2. WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY VOLUNTARILY AND IRREVOCABLY WAIVES TRIAL BY JURY IN ANY ACTION OR OTHER PROCEEDING BROUGHT IN CONNECTION WITH THIS AGREEMENT, ANY OF THE RELATED AGREEMENTS, DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY.

Article XXI. MISCELLANEOUS

Section 21.1. DELAY OR OMISSIONS. It is agreed that no delay or omission to exercise any right, power or remedy accruing to any party, upon any breach, default or noncompliance by another party under this Agreement shall impair any such right, power or remedy, nor shall it be construed to be a waiver of any such breach, default or noncompliance, or any acquiescence therein, or of or in any similar breach, default or noncompliance thereafter occurring. It is further agreed that any waiver, permit, consent or approval of any kind or character on any party's part of any breach, default or noncompliance under this Agreement or any waiver on such party's part of any provisions or conditions of the Agreement must be in writing and shall be effective only to the extent specifically set forth in such writing. All remedies, either under this Agreement, by law, or otherwise afforded to any party, shall be cumulative and not alternative.

Section 21.2. WAIVER. No waivers of any breach of this Agreement extended by any party hereto to any other party shall be construed as a waiver of any rights or remedies of any other party hereto or with respect to any subsequent breach.

Section 21.3. WAIVER OF PARTITION. The Members hereby agree that no Member, nor any successor in interest to any Member, shall have the right, while this Agreement remains in effect, to have any Company property partitioned, or to file a complaint or institute any proceeding at law or in equity to have such property partitioned, and all Members, on behalf of themselves and their heirs, successors and assigns, hereby waive any such right.

Section 21.4. SUCCESSORS AND ASSIGNS. Except as otherwise expressly provided herein, the provisions hereof shall inure to the benefit of, and be binding upon the parties hereto and their respective successors, assigns, heirs, executors and administrators and shall inure to the benefit of and be enforceable by each Person who shall be a holder of the Company from time to time; provided, however, that prior to the receipt by the Company of adequate written notice of the transfer of any Capital Securities of the Company specifying the full name and address of the transferee's satisfaction of all requirements to be a Permitted Transferee hereunder, the Company may deem and treat the person listed as the holder of such Capital Securities

34

of the Company in its records as the absolute owner and holder of such Capital Securities of the Company for all purposes.

Section 21.5. NOTICES. All notices required in connection with this Agreement shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient; if not, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written notification of receipt. All communications shall be sent to the Company at the address below and to each Member at the address as set forth on Exhibit "A" hereto or at such other address as such party may designate by ten (10) days advance written notice to the other parties hereto:

If to the Company, addressed to:

LIQUIDMETAL COATINGS, LLC 30452 Esperanza Rancho Santa Margarita, California 92688 Attention: Larry Buffington, Chief Executive Officer Fax No.: 813.314.0270

with a copy to:

Foley & Lardner LLP 100 North Tampa Street, Suite 2700 Tampa, FL 33602-5804 Tel: (813) 225-4122 Facsimile: (813) 221-4210| Attention: Curt P. Creely, Esq.

Section 21.6. ENTIRE AGREEMENT. This Agreement and the Exhibits hereto, along with the other documents delivered pursuant thereto, including but not limited to that certain Securities Purchase Agreement of even date herewith, constitute the full and entire understanding and agreement between the parties with regard to the subjects hereof and hor party shall be liable or bound to any other in any manner by any oral or written representations, warranties, covenants and agreements except as specifically set forth herein and therein. Each party expressly represents and warrants that it is not relying on any oral or written representations, warranties, covenants or agreements outside of this Agreement.

Section 21.7. PRONOUNS. All pronouns contained herein, and any variations thereof, shall be deemed to refer to the masculine, feminine or neutral, singular or plural, as to the identity of the parties hereto may require.

Section 21.8. TITLES AND SUBTITLES. The titles of the sections and subsections of the Agreement Section 21.9. SEVERABILITY. In the event one or more of the provisions of this Agreement should be a section 21.9.	ment are for convenience of reference only and are not to be considered in construing this Agreement. uld, for any reason, be held by any court of competent jurisdiction to be invalid,
	35
egal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other prover been contained herein.	visions of this Agreement, and this Agreement shall be construed as if such invalid, illegal or unenforceable provision had
Section 21.10. BINDING EFFECT . Except as provided to the contrary, the terms and provisions of irs, successors and assigns.	this Agreement shall be binding upon and shall inure solely to the benefit of all the Members, their personal representative
Section 21.11. CREDITORS. The provisions of this Agreement are not for the benefit of and may n	tot be specifically enforced by any creditors of the Company.
Section 21.12. <u>EXECUTION OF ADDITIONAL INSTRUMENTS</u> . Each Member hereby agrees cessary to comply with any applicable laws, rules, or regulations.	to execute such other and further statements of interest and holdings, designations, powers of attorney, and other instrume
Section 21.13. <u>RIGHTS AND REMEDIES CUMULATIVE</u> . The rights and remedies provided by e any or all other remedies. Said rights and remedies are given in addition to any other rights the parties may he	this Agreement are cumulative and the use of any one right or remedy by any party shall not preclude or waive the right to ave by law, statute, ordinance, or otherwise.
	each of which will be deemed an original, but all of which together shall constitute one instrument. Counterparts of this on shall be deemed to constitute signed original counterparts hereof and shall bind the parties signing and delivering in suc
IN WITNESS WHEREOF, the parties have executed this Limited Liability Company Operating Agreem	nent as of the date first set forth hereinabove.
	COMPANY:
	LIQUIDMETAL COATINGS, LLC
	By:
	Name: Title:
	MEMBERS:
	CLASS A HOLDER:
	LIQUIDMETAL TECHNOLOGIES, INC.
	By:
	Title:
	36
	CLASS B HOLDERS:
	C CAPITAL PARTITION A P

CLASS B HOLDERS:
C₃ CAPITAL PARTNERS, L.P.
By: Its General Partner
C ₃ Capital Partners, LLC, a Delaware limited liability company
By:
Print Name:
Title:
C₃ CAPITAL PARTNERS II, L.P.
By: Its General Partner
C ₃ Capital Partners II, LLC, a Delaware

limited liability company

CLASS C HOLDERS:

LARRY BUFFINGTON

Larry Buffington

Global Strategy & Capital Group, Inc. d.b.a CRESO Capital Partners

By:
Print Name:
Title:

37

Exhibit "A"

 Initial Capital
 Number and Percentage

 Members
 Contribution
 Class of Units
 Interest

Liquidmetal Technologies, Inc., a Delaware corporation 30452 Esperanza Rancho Santa Margarita, California 92688 Attention: Legal Department Fax No.: 813.314.0270	\$ 3,577,210	6,986 Class A Units*	69.25%
C ₃ Capital Partners, LP, a Delaware limited partnership C ³ Capital, LLC 4520 Main Street Suite 1600 Suansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: 816-756-5552	\$ 564,218	1,102 Class B Units**	11.02%
C ₃ Capital Partners II, LP, a Delaware limited partnership C ³ Capital, LLC 4520 Main Street Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: 816-756-5552	\$ 408,572	798 Class B Units**	7.98%
Larry Buffington 2205 Riva Row, Unite 2410 The Woodlands, TX 77380 Fax No. (832) 585-0341	\$ 0.00	1,000 Class C Units***	10.00%

Members		Initi Cor	al Capital tribution	Number and Class of Units	Percentage Interest
Global Strategy & Capital Group, Inc. d.b.a CRESO Capital Partners, a 660 Newport Center Drive, Suite 800 Newport Beach, CA 92660 Fax No. (949) 209-5441 Attn: Thomas Papa	corporation	\$	0.00	114 Class C Units****	1.75%
TOTALS		\$	4,550,000	10,000 Units	100%

- * Issued pursuant to that certain Asset Purchase and Contribution Agreement of even date herewith between the Company and Liquidmetal Technologies, Inc.
- ** Issued pursuant to that certain Securities Purchase Agreement of even date herewith between the Company, C3 Capital Partners, L.P., and C3 Capital Partners II, L.P.
- *** Issued pursuant to that certain Employment Agreement of even date herewith between the Company and Larry Buffington.
- **** Issued as consideration for placement agent services provided by Cresco Capital Partners.

2

Exhibit B

Form of Escrow Agreement

ESCROW AGREEMENT

This Escrow Agreement dated this 24th day of July, 2007 (this "Escrow Agreement"), is entered into by and among C₃ Capital Partners, LP, C₃ Capital Partners II, LP (together "C₃"), Bank Midwest, N.A. (when acting in its capacity as a lender, "BMW"), Liquidmetal Technologies, Inc., a Delaware corporation ("LMT"), Liquidmetal Coatings, LLC, a Delaware limited liability company, Liquidmetal Coatings Solutions, LLC, a Delaware limited liability company (together "LMC") and Bank Midwest, N.A. (when acting in its capacity as escrow agent, the "Escrow Agent").

WHEREAS, LMT is entering into a transaction (the "Transaction") wherein substantially all of the assets of its Liquidmetal Coatings business unit ("Transferred Assets") will be transferred to a new majority owned subsidiary, LMC, through a combination of sale and contribution, and LMC will incur indebtedness secured by the Transferred Assets and any other assets thereafter acquired by LMC; and,

WHEREAS, to help finance the Transaction, C_3 will loan to and invest monies in the aggregate amount of six million five hundred thousand dollars (\$6,500,000.00) to LMC, and contemporaneously therewith LMC will execute a subordinated note in favor of C_3 , and grant C_3 a second in priority security interest (subordinate only to the security interest of BMW) in the assets of LMC. In addition, C_3 will receive units of LMC equal to nineteen percent (19%) of the common equity of LMC (" C_3 Contribution"); and,

WHEREAS, to further help finance the Transaction, BMW will loan monies in the aggregate amount of five million five hundred thousand dollars (\$5,500,000.00) to LMC and contemporaneously therewith LMC will execute notes in favor of BMW and grant BMW a first in priority security interest in the assets of LMC ("BMW Contribution"); and,

WHEREAS, LMC will use the C₃ Contribution and BMW Contribution net of the loan fees, legal fees and other costs incurred by C₃ and BMW (collectively the "Transaction Monies") to pay the amounts required by LMT to consummate the Transaction and convey the Transferred Assets to LMC; and,

WHEREAS, LMT in connection with the Transaction is required to pay in full, from the received Transaction Monies, all of the indebtedness and liabilities owed to the holders of the 7% Senior Secured Convertible Notes Due August 2, 2007, the 6% Senior Secured Notes Due July 29, 2007 (collectively "Notes"), and the amounts due to the other parties listed on the Funds Flow Memorandum attached hereto as Exhibit A (collectively, "Other Creditors");

WHEREAS, LMT in connection with the Transaction is required to pay in full, from the received Transaction Monies, all of the indebtedness and liabilities owed to the

holders of the 8% Unsecured Subordinated Convertible Notes Due August 17, 2007, (the "Unsecured Notes"); and,

WHEREAS, to assure that the Transaction Monies are paid to the holders of the Notes and the Other Creditors, all parties hereto are agreeing to deposit the Transaction Monies with the Escrow Agent for disbursement by the Escrow Agent pursuant to the terms of this Escrow Agreement ("Escrow Funds").

WHEREAS, the parties agree and acknowledge that Bank Midwest, N.A. has duplicate roles under the terms of this Escrow Agreement, and will be acting as both lender and escrow agent and will fulfill the duties of each in its capacity herein.

NOW THEREFORE, in consideration of the foregoing, of the mutual promises and covenants set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

1. Operation of the Escrow. C₃ and BMW will cause to be wired directly to Escrow Agent, the Escrow Funds in order to complete the Transaction. Escrow Agent shall deposit the Escrow Funds in an interest bearing account at an institution which is federally insured ("Account") and any interest amounts accrued in said Escrow Account will be paid to the party or parties receiving the principal Escrow Funds at such time as final disbursement of the Escrow Funds is made. Escrow Agent shall disburse such funds upon the following terms and conditions.

- (a) Receipt by Escrow Agent of: (i) an executed copy of the Consent Agreement from the holders of the Convertible Subordinated Notes issued on January 3, 2007 and amendments thereto dated April 23, 2007, (ii) an executed copy of Consent from Hana Financial, Inc; (iii) executed copies of the payoff letters from the holders of the Notes and each of the Other Creditors; and (iv) release of security interests by the holders of the Notes, and the Other Creditors, if applicable ("Required Documents").
 - (b) Upon the receipt of the Required Documents the Escrow Agent shall disburse to the holders of the Notes and the Other Creditors the amounts set forth on the Funds Flow Memorandum attached hereto as Exhibit A.
 - (c) Upon the receipt of joint written directions from C3, BMW, LMC and LMT, the Escrow Agent shall disburse to the holders of the Unsecured Notes the amount set forth in such written instructions.
- 2. Payment of Escrow Agent's Fees and Expenses. All fees and expenses, including reasonable attorneys' fees, charged by Escrow Agent for its activities pursuant to this Agreement shall be borne equally and jointly by LMT and LMC, provided, however, that LMT, and LMC, shall be solely responsible for their own expenses incurred in connection with their respective performances hereunder, except as provided in other prior agreements between the parties; and provided, further, that any expenses, including reasonable attorneys' fees, charged by Escrow Agent in connection with any dispute between LMT, LMC, C₃, and BMW with respect to this Agreement shall be borne by the nonprevailing party as among LMT, LMC, C₃ and BMW.

2

- 3. Termination. This Escrow Agreement shall be terminated upon disbursement of all the Escrow Funds in accordance with the terms of this Escrow Agreement and may be terminated prior to that date by written consent signed by all the parties hereto, including the Escrow Agreement shall be terminated prior to that date by written termination shall be effective on the date specified in the termination document, or, if not specified, then on the date when signed by all parties hereto, including the Escrow Agreet by September 1, 2007, or an earlier date which is agreed upon in writing by all parties. In the event the Escrow Agent does not receive the Required Documents with respect to one or more holders of Notes or Other Creditors by September 1, 2007, the Escrow Agent shall return the Escrow Funds to G₃ and BMW.
- 4. <u>Limitation of Liability</u>, IN NO EVENT SHALL ESCROW AGENT BE LIABLE TO LMT, LMC, C₃, OR BMW, DIRECTLY OR INDIRECTLY, FOR ANY (I) DAMAGES OR EXPENSES ARISING OUT OF THE SERVICES PROVIDED HEREUNDER, OTHER THAN DAMAGES WHICH RESULT FROM THE ESCROW AGENT'S GROSS NEGLIGENCE OR WILLFUL MISCONDUCT, OR (II) SPECIAL OR CONSEQUENTIAL DAMAGES, EVEN IF THE ESCROW AGENT HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.
 - 5. Escrow Agent's Rights and Responsibilities. To induce Escrow Agent to act hereunder, LMT, LMC, C3 and BMW hereby agree that:
- (a) Escrow Agent shall be entitled to rely upon any opinion, order, decision, notice, affidavit, direction, or other writing delivered to it hereunder without being required to determine the authenticity or the correctness of any fact stated therein that is believed by Escrow Agent in good faith to be genuine and to have been signed or presented by the proper person. Escrow Agent shall not be responsible for the sufficiency of the execution or for the genuineness of documents or instruments delivered to it under this Agreement or for any lack of endorsement thereon or for a description therein, nor shall Escrow Agent be responsible or liable on account of the identity, authority, or rights of the person executing or delivering or purporting to execute or deliver any such document or instrument;
- (b) Escrow Agent may act, and shall be fully protected and shall incur no liability in acting, in reliance upon advice of counsel in reference to any matter connected with this Agreement and shall not be liable for any mistake of fact or error in judgment, or for any acts or omissions of any kind, unless caused by its bad faith or willful misconduct;
- (c) LMT, LMC, C₃ and BMW, jointly and severally, shall indemnify Escrow Agent from, and hold Escrow Agent harmless against, any loss, liability, cost, or expense relating to breach of this Agreement, including, without limitation, reasonable attorneys' fees and expenses, incurred by Escrow Agent, unless caused by its bad faith or willful misconduct arising out of, or in connection with, Escrow Agent carrying out its duties and responsibilities under this Agreement.

3

- 6. <u>Disputes.</u> In the event of any disagreement between any of the parties hereto resulting in adverse claims and demands being made in connection with the Escrow Funds, the Escrow Agent shall be entitled to refuse to comply with any demand or claim, as long as such disagreement shall continue, and in so refusing to make any delivery or other disposition of the Escrow Funds, the Escrow Agent shall not be or become liable to any of the parties hereto for its refusal to comply with such conflicting or adverse demands, and the Escrow Agent shall be entitled to refuse and refrain to act until:
 - (a) The rights of the adverse claimants shall have been fully and finally adjudicated in a court having jurisdiction over the parties and the Escrow Funds, or
 - (b) All differences shall have been reconciled by a written agreement signed by LMT, LMC, C3, and BMW, and the Escrow Agent shall have received an original execution copy of same.
- 7. <u>Resignation of Escrow Agent</u>. The Escrow Agent may resign or be removed, at any time, for any reason, by written notice of its resignation or removal to the proper parties at their respective addresses as set forth herein, at least 10 business days before the date specified for such resignation or removal to take effect. Upon the effective date of such resignation or removal:
- (a) The Escrow Funds shall be delivered by the Escrow Agent to such successor escrow agent as may be designated in writing by LMT, LMC, C₃ and BMW, whereupon the Escrow Agent's obligations hereunder shall cease and terminate; or
- (b) If no such successor Escrow Agent has been designated by such date, the Escrow Agent's sole obligation thereafter shall be to keep the Escrow Funds and to deliver the same to a person designated in writing by LMT, LMC, C₃ and BMW or in accordance with the directions of a final order or judgment of a court of competent jurisdiction.
 - 8. Severability. Any invalidity, in whole or in part, of any provision of this Escrow Agreement shall not affect the validity or enforceability of any other provisions of this Escrow Agreement.
 - 9. Amendment of Agreement. The Escrow Agent shall not be bound by any amendment of this Escrow Agreement and no amendment shall be effective unless:
 - (a) The amendment is in writing; and
 - (b) The amendment is executed by all the parties hereto.
- 10. Notices. All notices shall be deemed conclusively to have been given and delivered if the same is in writing and (a) mailed, by registered or certified mail, postage prepaid, or (b) hand delivered, or (c) by overnight delivery via a reputable national overnight delivery service, to the addresses first set forth herein or to such other address as any of the parties hereto may designate by notice given as above provided. Any item sent by registered or certified mail, as above

4

provided, will be deemed given two days following deposit in the United States mail. The addresses for the purposes herein are:

Liquidmetal Technologies, Inc.

30452 Esperanza

Rancho Santa Margarita, California 92688

Attn: Legal Department

Liquidmetal Coatings, LLC 30452 Esperanza

Rancho Santa Margarita, California 92688

Attn: Legal Department

Liquidmetal Coatings Solutions, LLC

30452 Esperanza

Rancho Santa Margarita, California 92688

Attn: Legal Department

C₃ Capital Partners, LP

4520 Main Street, Suite 1600 Kansas City, Missouri 64111

Attn: Robert L. Smith

C₃ Capital Partners II, LP

4520 Main Street, Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith

Bank Midwest, N.A

1100 Main Street, Suite 350 Kansas City, Missouri 64105 Attn: Judson Stanion

Bank Midwest, N.A. (as Escrow Agent)

1100 Main Street, Suite 350 Kansas City, Missouri 64105 Attn: Judson Stanion

11. <u>Miscellaneous</u>. This Escrow Agreement and the related documents covering the Transaction contains the entire understanding of the parties hereto and supersedes all prior oral and written understandings, agreements or undertakings of the parties with respect to the escrowing of funds by LMT, LMC, C₃, BMW in connection with the Transaction. This Escrow Agreement and all actions taken hereunder in accordance with its terms shall be binding upon and inure to the benefit of LMT, LMC, C₃, BMW, the Escrow Agent and their respective heirs,

5

successors and assigns. This Escrow Agreement and the Escrow Agreement may be executed in one or more counterparts with the same effect as if all signatures and all counterparts were written upon the same instrument.

IN WITNESS WHEREOF, the parties hereto have each caused this Escrow Agreement to be executed as of the date first above written.

Liquidmetal Technologies, Inc.	C ₃ Capital Partners II, LP By: C ₃ Capital Partners II, LLC, its general partner
By: Printed Name: Title:	By: Printed Name: Title:
Liquidmetal Coatings, LLC	Bank Midwest, N.A.
By: Printed Name: Title:	By: Printed Name: Title:
Liquidmetal Coatings Solutions, LLC	Bank Midwest, N.A. in its capacity as Escrow Agent
By: Printed Name: Title:	By: Printed Name: Title:
C_3 Capital Partners, LP By: C_3 Capital Partners, LLC, its general partner	
By: Printed Name: Title:	
	6

Exhibit C

Form of Buffington Employment Agreement

EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT (this "Agreement") is made and entered into effective as of July 24, 2007 (the "Effective Date"), by and between LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company (the "Company"), and LARRY BUFFINGTON, an individual residing in the State of Texas (the "Employee").

RECITALS

WHEREAS, the Employee desires to be employed by the Company upon the terms and conditions set forth in this Agreement; and

WHEREAS, the Company desires to assure itself of the Employee's continued employment in the capacities set forth herein.

NOW, THEREFORE, in consideration of the foregoing recitals and for other good and valuable consideration, the parties hereto covenant and agree as follows:

- 1. <u>EMPLOYMENT.</u> THE COMPANY HEREBY EMPLOYS EMPLOYEE, AND THE EMPLOYEE HEREBY ACCEPTS SUCH EMPLOYMENT, UPON THE TERMS AND CONDITIONS SET FORTH IN THIS AGREEMENT
- 2. TERM. SUBJECT TO THE TERMS AND CONDITIONS OF THIS AGREEMENT, INCLUDING, BUT NOT LIMITED TO, THE PROVISIONS FOR TERMINATION SET FORTH IN SECTION 5 HEREOF, THE EMPLOYMENT OF THE EMPLOYEE UNDER THIS AGREEMENT SHALL COMMENCE ON THE EFFECTIVE DATE AND SHALL CONTINUE THROUGH THE CLOSE OF BUSINESS ON FIFTH (5TH) ANNIVERSARY OF THE EFFECTIVE DATE (THE "INTIAL TERM"). UPON THE EXPIRATION OF THE INITIAL TERM, THE EMPLOYEE'S EMPLOYMENT WITH THE COMPANY WILL CONTINUE ON AN "AT-WILL" BASIS AND MAY BE TERMINATED BY EMPLOYEE OR THE COMPANY FOR ANY REASON AND AT ANY TIME, PROVIDED THAT THE TERMINATING PARTY SHALL PROVIDE AT LEAST NINETY (90) DAYS PRIOR WRITTEN NOTICE OF THE TERMINATION TO THE OTHER PARTY (UNLESS THE TERMINATION IS FOR CAUSE AS DEFINED IN THIS AGREEMENT, IN WHICH CASE THE EMPLOYEE'S EMPLOYMENT MAY BE TERMINATED IMMEDIATELY). NOTWITHSTANDING THE EXPIRATION OR TERMINATION OF EMPLOYEE'S EMPLOYMENT, THE PROVISIONS OF THIS AGREEMENT OTHER THAN THOSE OF SECTIONS 2, 4, AND 5 SHALL REMAIN IN FULL FORCE AND EFFECT. ALL OTHER PROVISIONS OF THIS AGREEMENT, INCLUDING BUT WITHOUT LIMITATION, SECTIONS 1, 6, AND 7, SHALL SURVIVE THE EXPIRATION OF THE THIS AGREEMENT OR THE TERMINATION OF EMPLOYMENT BY ANY MEANS BY ANY PARTY, SECTIONS 2, 6, AND 7 OF THIS AGREEMENT SHALL SURVIVE AND REMAIN FULLY ENFORCEABLE.
- 3. DUTIES. EMPLOYEE WILL INITIALLY SERVE AS PRESIDENT AND CHIEF EXECUTIVE OFFICER OF THE COMPANY AND REPORT TO THE BOARD OF MANAGERS OF THE COMPANY. EMPLOYEE WILL BE BASED AT THE COMPANY'S PRINCIPAL EXECUTIVE OFFICES LOCATED IN OR AROUND HOUSTON, TEXAS. EMPLOYEE WILL DEVOTE THE NECESSARY BUSINESS TIME, ATTENTION, SKILL, AND ENERGY TO THE BUSINESS OF THE COMPANY, WILL USE THE EMPLOYEE'S BEST EFFORTS TO PROMOTE THE SUCCESS OF THE COMPANY'S BUSINESS, AND WILL COOPERATE FULLY WITH THE BOARD OF MANAGERS IN THE ADVANCEMENT OF THE BEST INTERESTS OF THE COMPANY. FURTHERMORE, THE EMPLOYEE SHALL ASSUME AND COMPETENTLY PERFORM SUCH REASONABLE RESPONSIBILITIES AND DUTIES AS MAY BE ASSIGNED TO THE EMPLOYEE FROM TIME TO TIME BY THE BOARD OF MANAGERS OF THE COMPANY OR THEIR DESIGNEE. TO THE EXTENT THAT THE COMPANY SHALL HAVE ANY PARENT COMPANY, SUBSIDIARIES, AFFILIATED CORPORATIONS, PARTNERSHIPS, OR JOINT VENTURES OTHER THAN LIQUIDMETAL TECHNOLOGIES, INC. (COLLECTIVELY "RELATED ENTITIES"), THE EMPLOYEE SHALL PERFORM SUCH DUTIES TO PROMOTE THESE ENTITIES AND TO PROMOTE AND PROTECT THEIR RESPECTIVE INTERESTS TO THE SAME EXTENT AS THE INTERESTS OF THE COMPANY WITHOUT ADDITIONAL COMPENSATION. IN ADDITION, EMPLOYEE MAY, UPON THE AGREEMENT OF LIQUIDMETAL TECHNOLOGIES, INC., CONTINUE TO SERVE AS PRESIDENT AND CHIEF EXECUTIVE OFFICER OF LIQUIDMETAL TECHNOLOGIES, INC.

COMPENSATION.

- (a) Annual Base Salary. As compensation for Employee's services and in consideration for the Employee's covenants contained in this Agreement, the Company shall pay the Employee an annual base salary of \$250,000.00. The annual base salary may be adjusted upward (but not downward) in the sole discretion of the Board of Managers of the Company. For purposes of this Agreement, the term "Salary Year" means the one year, 365-day period (or 366 day period for a leap year) that begins on the Effective Date and each successive one, year period thereafter.
- (b) Bonuses. In addition to the Employee's annual base compensation, during each fiscal year during the term of the Employee's employment hereunder, the Employee shall be entitled to an annual performance bonus upon terms and conditions more particularly established by the Board of Managers of the Company. The annual performance bonus will be in an amount up to 50% of Employee's base salary. (the "Maximum Bonus Amount"), payable as follows: (i) 25% of the Maximum Bonus Amount will be paid if the Company meets its EBITDA target (as established by the Board of Managers) for the applicable fiscal year, (ii) an additional 25% of the Maximum Bonus Amount will be paid if the Company meets its debt repayment target (as established by the Board of Managers) for the applicable fiscal year, and (iii) the remaining Maximum Bonus Amount (or any portion thereof) will be paid only at the discretion of the Board of Managers based on any factor or consideration that the Board of Managers, in its sole discretion, deems relevant.

- (c) Reimbursement of Expenses. The Employee shall be reimbursed for all reasonable and customary travel and other business expenses incurred by Employee in the performance of Employee's duties hereunder, provided that such reimbursement shall be subject to, and in accordance with, any expense reimbursement policies and/or expense documentation requirements of the Company that may be in effect from time to time.
- (d) Other Benefits. During the term of the Employee's employment hereunder, the Employee shall be eligible to participate in a health insurance plan selected by the Company. The Company shall pay 100 % of the premium of such health insurance plan. Also, Employee will, during his employment hereunder, be entitled to an automobile allowance in the amount of \$800.00 per month.
- (e) Vacation. The Employee shall be entitled to four (4) weeks paid vacation during each Salary Year during the term of the Employee's employment hereunder. Vacation shall be taken at such times and with such notice so as to not disrupt or interfere with the business of the Company. Unused vacation from a particular Salary Year will carry over to succeeding Salary Years, but only up to a maximum of four (4) weeks may be carried into any succeeding Salary Year.

<u>TERMINATION</u>.

- (a) Death. The Employee's employment under this Agreement shall terminate immediately upon Employee's death. In the event of a termination pursuant to this Section 5(a), the Employee's estate shall be entitled to receive any unpaid base salary, bonuses or benefits owing to Employee up through and including the date of the Employee's death.
- (b) <u>Disability. If, during the term of the Employee's employment hereunder, the Employee becomes physically or mentally disabled in the determination of a physician appointed or selected by the Company, or, if due to any physical or mental condition, the Employee becomes unable for a period of more than sixty. (60) days during any six-month period to perform Employee's duties hereunder on substantially a full-time basis as determined by a physician selected by the Company, the Company may, at its option, terminate the Employee's employment upon not less than thirty. (30) days written notice. In the event of a termination pursuant to this Section 5(b), the Employee shall be entitled to receive any unpaid base salary, bonuses or benefits owing to Employee up through and including the effective date of termination.</u>
- (c) Termination By Company With Cause. The Company may terminate the Employee's employment at any time with Cause. As used in this Agreement, "Cause" shall include the following: (1) the Employee's failure or inability to perform Employee's duties under this Agreement; (2) dishonesty or other serious misconduct, (3) the commission of an unlawful act material to Employee's employment, (4) a material violation of the Company's policies or practices which reasonably justifies immediate termination: (5) committing, pleading, guilty, noto contendre or no contest (or their equivalent) to, entering into a pretrial intervention or diversion program regarding, or conviction of, a felony or any crime or act involving moral turpitude, fraud. dishonesty, or misrepresentation: (6) the commission by the Employee of any act which could reasonably affect or impact to a material degree the interests of the Company or Related

3

Entities or in some manner injure the reputation, business, or business relationships of the Company or Related Entities; (7) the Employee's inability to perform an essential function of Employee's position; or (8) any material breach by Employee of this Agreement. The Company may terminate this Agreement for Cause at any time without notice. In the event of a termination for Cause, the Company shall be relieved of all its obligations to the Employee provided for by this Agreement as of the effective date of termination, and all payments to the Employee hereunder shall immediately cease and terminate as of such date, except that Employee shall be entitled to receive any unpaid base salary, bonuses or benefits owing to Employee up to and including the effective date of termination, provided, however, that the Employee's obligations under Sections 6 and 7 shall survive such a termination for Cause and any liabilities or obligations which have accrued and are owed by the Employee to the Company shall not be extinguished or released thereby.

(d) Termination By Company Without Cause. In addition to the other termination provisions of this Agreement, the Company may terminate the Employee's employment at any time and for any reason and without cause (a "Termination Without Cause"). In the event of a Termination Without Cause, the Employee shall continue to receive the Employee's base salary. (as then in effect) through the later of one year after the date of the Termination Without Cause on the end of the Initial Term (the "Severance Period"). In addition to the severance pay described in the preceding sentence, the Employee shall continue to receive, during the Severance Period, all employee health and welfare benefits that Employee would have received during the Severance Period in the absence of such termination, but only to the extent that the Company is permitted to continue providing such benefits after termination in accordance with the applicable employee benefit plans. Employee agrees and acknowledges, however, that Employee will forfeit the right to receive base salary and benefits during the Severance Period immediately upon the Employee's breach of any covenant set forth in Section 6 of this Agreement. The Employee will also forfeit the right to salary and benefits during the Severance Period upon accepting employment with another employer with comparable salary and benefits hereunder shall be forfeited and shall case upon the Employee benefits from the Employee's new employer. Notwithstanding the foregoing, the termination of the Employee's employment pursuant to the second sentence of Section 2 of this Agreement shall not constitute a Termination Without Cause and shall not give rise to any severance payment or other benefits pursuant to this Section 5 (d).

4

NONCOMPETITION, NONSOLICITATION, AND NONDISCLOSURE COVENANTS.

- (a) Rationale for Restrictions. Employee acknowledges that Employee's services hereunder are of a special, unique, and extraordinary character, and Employee's position with the Company places Employee in a position of confidence and trust with customers, suppliers, and other persons and entities with whom the Company and its Related Entities have a business relationship. The Employee further acknowledges that the rendering of services under this Agreement will require the disclosure to Employee of Confidential Information (as defined below) including Trade Secrets of the Company and relating to the Company and the Company hereby, agrees to provide such Confidential Information. As a consequence, the Employee agrees that it is reasonable and necessary for the protection of the goodwill and legitimate business interests of the Company and Related Entities that the Employee make the covenants contained in this Section 6, that such covenants are a material inducement for the Company to employ the Employee and to enter into this Agreement, and that the covenants are given as an integral part of and incident to this Agreement.
- (b) Noncompetition and Nonsolicitation Covenants. As used herein, the term "Restrictive Period" means the time period commencing on the Effective Date of this Agreement and ending on the second (2nd) anniversary of the date on which the Employee's employment by the Company (or any Related Entity) expires or is terminated for any reason, including both a termination by the Company for Cause and a Termination Without Cause. In addition, the term "Covered Business" means any business which is the same as, or similar to, any business conducted by the Company or any of the Related Entities at any time during the Restrictive Period. The Employee agrees that the Employee will not engage in any of the following acts anywhere in the world during the Restrictive Period:
 - (i) directly or indirectly engage or invest in; own, manage, operate, finance, control, or participate in the ownership, management, operation, financing, or control of; be employed by, associated with, or in any manner connected with; lend the Employee's name or any similar name to; lend Employee's credit to; or render services or advice to, any business which competes with, is engaged in, or carries on any aspect of a Covered Business;
 - (ii) directly or indirectly assist, promote or encourage any existing or potential employees, customers, clients, or vendors of the Company or any Related Entity, as well as any other parties which have a business relationship with the Company or a Related Entity, to terminate, discontinue, or reduce the extent of their relationship with the Company or a Related Entity;
 - (iii) directly or indirectly solicit business of the same or similar type as a Covered Business, from any person or entity known by the Employee to be a customer or client of the Company, whether or not the Employee had contact with such person or entity during the Employee's employment with the Company;

5

- (iv) disparage the Company, any Related Entities, and/or any shareholder, director, officer, employee, or agent of the Company or any Related Entity; and/or
- (v) engage in any practice the purpose of which is to evade the provisions of this Section 6 or commit any act which adversely affects the Company, any Related Entity, or their respective businesses.

Employee acknowledges that Employee's services hereunder are of a special, unique, and extraordinary character, and Employee's position with the Company places Employee in a position of confidence and trust with customers, suppliers, and other persons and entities with whom the Company and its Related Entities have a business relationship. The Employee further acknowledges that the rendering of services under this Agreement will likely require the disclosure to Employee of Confidential Information (as defined below) and Trade Secrets (as defined below) of the Company relating to the Company and/or Related Entities. As a consequence, the Employee agrees that it is reasonable and necessary for the protection of the goodwill and legitimate business interests of the Company and Related Entities that the Employee make the covenants contained in this Section 6, that such covenants are a material inducement for the Company to employ the Employee and to enter into this Agreement, and that the covenants are given as an integral part of and incident to this Agreement. Accordingly, the Employee agrees that the geographic scope of the above covenants is a reasonable means of protecting the Company's (and the Related Entities') legitimate business interests. Notwithstanding the foregoing covenants, nothing set forth in this Agreement shall prohibit the Employee from owning the securities of (i) corporations which are listed on a national securities exchange or traded in the national over-the-counter market in an amount which shall not exceed 5% of the outstanding shares of any such corporation or (ii) any corporation, partnership, firm or other form of business organization which does not compete with, is not engaged in, and does not carry on any aspect of, either directly or indirectly through a subsidiary or otherwise, any Covered Business.

(c) Disclosure of Confidential Information. The Employee acknowledges that the inventions, innovations, software, Trade Secrets, business plans, financial strategies, finances, and all other confidential or proprietary, information with respect to the business and operations of the Company and Related Entities are valuable, special, and unique assets of the Company, Accordingly, the Employee agrees not to, at any time whatsoever either during or after the Employee's term of employment with the Company, disclose, directly or indirectly, to any person or entity, or use or authorize any person or entity to use, any confidential or proprietary information with respect to the Company or Related Entities without the prior written consent of the Company, including, without limitation, information as to the financial condition, results of operations, identities of clients or prospective clients, products under development, acquisition strategies or acquisitions under consideration, pricing or cost information, marketing strategies, passwords or codes or any other information relating to the Company or any of the Related Entities which could be reasonably regarded as confidential (collectively referred to as "Confidential Information"). However, the term "Confidential Information" does not include any information which is or shall become generally available to the public other than as a result of disclosure by the Employee or by any person or entity which the Employee reasonably should know) has a duty of confidentiality to the Company or a

6

- (d) Removal and Return of Proprietary Items. The Employee will not remove from the Company's premises (except to the extent such removal is for purposes of the performance of the Employee's duties at home or while traveling, and under such conditions and restrictions as are specifically authorized and/or required by the Company) or transmit by any means, electronic or otherwise, any document, record, notebook, plan, model, component, device, computer software or code, or Confidential Information or Trade Secret whether embodied in a disk or in any other form, including electronic form (collectively, the "Proprietary Items"). The Employee recognizes that, as between the Company and the Employee, all of the Proprietary Items, whether or not developed by the Employee, are the exclusive property of the Company. Upon termination of Employee's employment with the Company by either party (regardless of the reason for termination), or upon the request of the Company during the term of employment, the Employee will return to the Company all of the Proprietary Items in the Employee's possession or subject to the Employee's control, and the Employee shall not retain any copies, abstracts, sketches, or other physical embodiment of any of the Proprietary Items, Confidential Information, Trade Secret or any part thereof.
- (e) Enforcement and Remedies. In the event of any breach of any of the covenants set forth in this Section 6, the Employee recognizes that the remedies at law will be inadequate and that in addition to any relief at law which may be available to the Company for such violation or breach and regardless of any other provision contained in this Agreement, the Company shall be entitled to equitable remedies (including an injunction) and such other relief as a court may grant after considering the intent of this Section 6. Additionally, the period of time applicable to any covenant set forth in this Section 6 will be extended by the duration of any violation by Employee of such covenant. In the event a court of competent jurisdiction determines that any of the covenants set forth in this Section 6 are excessively broad as to duration, geographic scope, prohibited activities or otherwise, the parties agree that this covenant shall be reduced or curtailed to the extent, but only to the extent, necessary to render it enforceable.

7

7. EMPLOYEE INVENTIONS

- (a) Definition. For purposes of this Agreement, "Employee Invention" means any idea, invention, technique, modification, process, or improvement (whether patentable or not), any industrial design (whether registerable or not), any mask work, however fixed or encoded, that is suitable to be fixed, embedded or programmed in a semiconductor product (whether recordable or not), and any work of authorship (whether or not copyright protection may be obtained for it) created, conceived, or developed by the Employee, either solely or in conjunction with others, during the Employee's employment with the Company or during the ninety (90) day period following such employment, that relates in any way to, or is useful in any manner in, the businesses then being conducted or proposed to be conducted by the Company or any Related Entity.
- (b) Ownership of Employee Inventions. Employee agrees and acknowledges that all Employee Inventions will belong exclusively to the Company and that all Employee Inventions are works made for hire and the property of the Company, including any copyrights, patents, semiconductor mask protection, or other intellectual property rights pertaining thereto. If it is determined that any such works are not works made for hire, the Employee hereby assigns to the Company's right, title, and interest, including all rights of copyright, patent, semiconductor mask protection, and other intellectual property rights, to or in such Employee Inventions. The Employee covenants that the Employee will promptly:
 - (i) disclose to the Company in writing any Employee Invention;
 - (ii) assign to the Company or to a party designated by the Company, at the Company's request and without additional compensation, all of the Employee's right to the Employee Invention for the United States and all foreign jurisdictions;
 - (iii) execute and deliver to the Company such applications, assignments, and other documents as the Company may request in order to apply for and obtain patents or other registrations with respect to any Employee Invention in the United States and any foreign jurisdictions;
 - (iv) sign all other papers necessary to carry out the above obligations; and
 - (v) give testimony and render any other assistance in support of the Company's rights to any Employee Invention.
- 8. <u>ESSENTIAL AND INDEPENDENT COVENANTS</u>. THE EMPLOYEE'S COVENANTS IN SECTIONS 6 AND 7 OF THIS AGREEMENT ARE INDEPENDENT COVENANTS, AND THE EXISTENCE OF ANY CLAIM BY THE EMPLOYEE AGAINST THE COMPANY UNDER THIS AGREEMENT OR OTHERWISE WILL NOT EXCUSE THE EMPLOYEE'S BREACH OF ANY COVENANT IN SECTION 6 OR 7. THE COVENANTS OF SECTIONS 6 AND 7 SHALL SURVIVE THE TERMINATION, EXTINGUISHMENT, OR LAPSE OF THIS AGREEMENT UNDER ANY CIRCUMSTANCES, EVEN IF THIS AGREEMENT IS TERMINATED BY EITHER PARTY, WHETHER FOR CAUSE OR NOT FOR CAUSE.

8

- 9. REPRESENTATIONS AND WARRANTIES BY THE EMPLOYEE. THE EMPLOYEE REPRESENTS AND WARRANTS TO THE COMPANY THAT THE EXECUTION AND DELIVERY BY THE EMPLOYEE OF THIS AGREEMENT DO NOT, AND THE PERFORMANCE BY THE EMPLOYEE OF THE EMPLOYEE'S OBLIGATIONS HEREUNDER WILL NOT, WITH OR WITHOUT THE GIVING OF NOTICE OR THE PASSAGE OF TIME, OR BOTH: (A) VIOLATE ANY JUDGMENT, WRIT, INJUNCTION, OR ORDER OF ANY COURT, ARBITRATOR, OR GOVERNMENTAL AGENCY APPLICABLE TO THE EMPLOYEE, OR (B) CONFLICT WITH, RESULT IN THE BREACH OF ANY PROVISIONS OF OR THE TERMINATION OF, OR CONSTITUTE A DEFAULT UNDER, ANY AGREEMENT TO WHICH THE EMPLOYEE IS A PARTY OR BY WHICH THE EMPLOYEE IS OR MAY BE BOUND, INCLUDING, WITHOUT LIMITATION, ANY NONCOMPETITION AGREEMENT OR SIMILAR AGREEMENT. EMPLOYEE FURTHER REPRESENTS AND WARRANTS THAT HE FULLY AND COMPLETELY UNDERSTANDS THIS AGREEMENT AND THAT HE HAS ENGAGE IN NEGOTIATIONS WITH THE COMPANY AND HAS EITHER CONSULTED WITH AN ATTORNEY OF HIS CHOICE OR HAS HAD AMPLE OPPORTUNITY TO DO SO AND IS FULLY SATISFIED WITH THE OPPORTUNITY HE HAS HAD.
- 10. NOTICES. FOR PURPOSES OF THIS AGREEMENT, NOTICES AND ALL OTHER COMMUNICATIONS PROVIDED FOR HEREIN SHALL BE IN WRITING AND SHALL BE DEEMED TO HAVE BEEN DULY GIVEN WHEN HAND-DELIVERED, SENT BY FACSIMILE TRANSMISSION (AS LONG AS RECEIPT IS ACKNOWLEDGED), OR MAILED BY UNITED STATES CERTIFIED OR REGISTERED MAIL, RETURN RECEIPT REQUESTED, POSTAGE PREPAID, ADDRESSED TO THE ADDRESS OR FACSIMILE NUMBER FOR EACH PARTY SET FORTH ON THE SIGNATURE PAGE HERETO, OR TO SUCH OTHER ADDRESS OR FACSIMILE NUMBER AS EITHER PARTY MAY HAVE FURNISHED TO THE OTHER IN WRITING IN ACCORDANCE HEREWITH, EXCEPT THAT A NOTICE OF CHANGE OF ADDRESS SHALL BE EFFECTIVE ONLY UPON RECEIPT.
- 11. MISCELLANEOUS. NO PROVISION OF THIS AGREEMENT MAY BE MODIFIED OR WAIVED UNLESS SUCH WAIVER OR MODIFICATION IS AGREED TO IN WRITING SIGNED BY BOTH OF THE PARTIES HERETO. NO WAIVER BY ANY PARTY HERETO OF ANY BREACH BY ANY OTHER PARTY HERETO SHALL BE DEEMED A WAIVER OF ANY SIMILAR OR DISSIMILAR TERM OR CONDITION AT THE SAME OR AT ANY PRIOR OR SUBSEQUENT TIME. THIS AGREEMENT IS THE ENTIRE AGREEMENT BETWEEN THE PARTIES HERETO WITH RESPECT TO THE EMPLOYEE'S EMPLOYMENT BY THE COMPANY, AND THERE ARE NO AGREEMENTS OR REPRESENTATIONS, ORAL OR OTHERWISE, EXPRESSED OR IMPLIED, WITH RESPECT TO OR RELATED TO THE EMPLOYMENT OF THE EMPLOYEE WHICH ARE NOT SET FORTH IN THIS AGREEMENT. THIS AGREEMENT SHALL BE BINDING UPON, AND INURE TO THE BENEFIT OF, THE COMPANY, ITS RESPECTIVE SUCCESSORS AND ASSIGNS, AND THE EMPLOYEE AND EMPLOYEE'S HEIRS, EXECUTORS, ADMINISTRATORS AND LEGAL REPRESENTATIVES. THE DUTIES AND COVENANTS OF THE EMPLOYEE UNDER THIS AGREEMENT. BEING

9

PERSONAL, MAY NOT BE DELEGATED OR ASSIGNED BY THE EMPLOYEE WITHOUT THE PRIOR WRITTEN CONSENT OF THE COMPANY, AND ANY ATTEMPTED DELEGATION OR ASSIGNMENT WITHOUT SUCH PRIOR WRITTEN CONSENT SHALL BE NULL AND VOID AND WITHOUT LEGAL EFFECT. THE PARTIES AGREE THAT IF ANY PROVISION OF THIS AGREEMENT SHALL UNDER ANY CIRCUMSTANCES BE DEEMED INVALID OR INOPERATIVE, THE AGREEMENT SHALL BE CONSTRUED WITH THE INVALID OR INOPERATIVE PROVISION DELETED AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES SHALL BE CONSTRUED AND ENFORCED ACCORDINGLY. THIS AGREEMENT MAY BE ASSIGNED BY THE COMPANY WITHOUT THE CONSENT OF THE EMPLOYEE, PROVIDED, HOWEVER, THAT THE EMPLOYEE IS GIVEN NOTICE OF THE ASSIGNMENT.

- 12. GOVERNING LAW; RESOLUTION OF DISPUTES. THE VALIDITY, INTERPRETATION, CONSTRUCTION, AND PERFORMANCE OF THIS AGREEMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF TEXAS WITHOUT REGARD TO PRINCIPLES OF CHOICE OF LAW OR CONFLICTS OF LAW THEREUNDER. ANY ACTION OR PROCEEDING SEEKING TO ENFORCE ANY PROVISION OF, OR BASED ON ANY RIGHT ARISING OUT OF, THIS AGREEMENT MAY BE BROUGHT AGAINST EITHER OF THE PARTIES IN THE COURTS OF THE STATE OF TEXAS, COUNTY OF HARRIS OR MONTGOMERY, OR, IF IT HAS OR CAN ACQUIRE JURISDICTION, IN THE FEDERAL COURTS LOCATED IN, HARRIS COUNTY OR MONTGOMERY COUNTY, TEXAS, AND EACH OF THE PARTIES CONSENTS TO THE JURISDICTION OF SUCH COURTS (AND OF THE APPROPRIATE APPELLATE COURTS) IN ANY SUCH ACTION OR PROCEEDING AND WAIVES ANY OBJECTION TO VENUE LAID THEREIN. PROCESS IN ANY ACTION OR PROCEEDING REFERRED TO IN THE PRECEDING SENTENCE MAY BE SERVED ON EITHER PARTY ANYWHERE IN THE WORLD. THE PARTIES HERETO AGREE THAT HAVING VENUE AND JURISDICTION SOLELY IN TEXAS IS REASONABLE IN THAT THE HEADQUARTERS FOR THE COMPANY WILL BE IN HARRIS COUNTY OR MONTGOMERY COUNTY, TEXAS AND THAT SITE FOR LITIGATION IS THE MOST CENTRAL FOR SUCH MATTERS. THE PARTIES HEREBY WAIVE A JURY TRIAL IN ANY LITIGATION ARISING UNDER OR RELATING TO THIS AGREEMENT OR THE EMPLOYMENT OF THE EMPLOYEE WITH THE COMPANY. THIS AGREEMENT OR ANY PART THEREOF.
- 13. <u>COUNTERPARTS; EACSIMILE SIGNATURES.</u> THIS AGREEMENT MAY BE EXECUTED IN ONE OR MORE COUNTERPARTS, EACH OF WHICH SHALL BE DEEMED AN ORIGINAL, BUT ALL OF WHICH TOGETHER SHALL CONSTITUTE ONE AND THE SAME INSTRUMENT. THIS AGREEMENT MAY BE EFFECTIVE UPON THE EXECUTION AND DELIVERY BY ANY PARTY HERETO OF FACSIMILE COPIES OF SIGNATURE PAGES HERETO DULY EXECUTED BY SUCH PARTY; <u>PROVIDED</u>, <u>HOWEVER</u>, THAT ANY PARTY DELIVERING A FACSIMILE SIGNATURE PAGE COVENANTS AND AGREES TO DELIVER

10

PROMPTLY AFTER THE DATE HEREOF TWO (2) ORIGINAL COPIES TO THE OTHER PARTY HERETO.

14. MODIFICATION BY THE COURT. IN THE EVENT THAT ANY PROVISION OR SECTION OF THIS AGREEMENT VIOLATES ANY LAW OF THE STATE OF TEXAS OR IS FOR SOME OTHER REASON UNENFORCEABLE AS WRITTEN IN THE STATE OF TEXAS, THE EMPLOYEE AND THE COMPANY AGREE THAT THE UNENFORCEABLE PROVISION OR SECTION SHOULD NOT CAUSE THE ENTIRE AGREEMENT TO BECOME UNENFORCEABLE UNLESS IT IS CAUSED TO FAIL IN ITS ESSENTIAL PURPOSE. IN THE EVENT THAT ANY PROVISION OR SECTION OF THIS AGREEMENT VIOLATES ANY LAW OF THE STATE OF TEXAS OR IS FOR SOME OTHER REASON UNENFORCEABLE AS WRITTEN IN THE STATE OF TEXAS, THE EMPLOYEE AGREES THAT THE PROVISION SHOULD BE REDUCED IN SCOPE OR LENGTH OR OTHERWISE MODIFIED BY THE COURT, IF POSSIBLE UNDER THE LAW, TO CAUSE THE PROVISION OR SECTION OF THE AGREEMENT TO BE LEGAL AND ENFORCEABLE BUT TO STILL PROVIDE TO THE COMPANY THE MAXIMUM PROTECTION AVAILABLE TO IT UNDER THE LAW.

LIQUIDMETAL COATINGS, LLC

IN WITNESS WHEREOF, the parties have executed this Agreement as of the day and year first above written.

Ву:
Liquidmetal Coatings, LLC 30452 Esperanza
Rancho Santa Margarita, CA 92688
Attention: Chief Financial Officer
Fax No.: (949) 635-2108 (initial address until Houston office is opened)
(initial address diffil flousion office is opened)
EMPLOYEE
Ву:
Printed Name: Larry Buffington
Address and Facsimile Number:
Larry Buffington
2205 Riva Row, Unit 2410
The Woodlands, TX 77380

Exhibit D

Form of Services Agreement

Fax: 832-585-0341

SERVICES AGREEMENT

THIS SERVICES AGREEMENT is made as of the 24th day of July, 2007 (the "Agreement"), by and between **Liquidmetal Coatings**, **LLC**, a Delaware limited liability company, ("LMC"), and **Liquidmetal Technologies**, **Inc.** (or "LMT"). LMT and LMC may collectively be referred to as the "Parties. Capitalized terms not otherwise defined herein shall have the meanings given to such terms in the Asset Purchase Agreement (as defined below).

WITNESSETE

WHEREAS, the Parties have entered into an Asset Purchase Agreement of even date herewith (the "Asset Purchase Agreement"), pursuant to which, among other things, LMC purchased from the LMT the Purchased Assets (as defined in the Asset Purchase Agreement); and

WHEREAS, LMC manufacturers and sells industrial coatings materials and provides application services of industrial coatings from its facilities and offices (the "Business"); and

WHEREAS, subject to the terms and provisions hereof, LMC desires to engage LMT to provide certain services to LMC and LMT desires to provide such services to LMC.

NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto covenant and agree as follows:

- 1. <u>ADMINISTRATIVE AND INFORMATION TECHNOLOGY ("IT") SERVICES.</u> SUBJECT TO THE TERMS AND PROVISIONS OF THIS AGREEMENT, LMC HEREBY RETAINS LMT TO PROVIDE THE FOLLOWING SERVICES DURING THE TERM OF THIS AGREEMENT (THE "<u>ADMINISTRATIVE SERVICES</u>") AND LMT HEREBY ACCEPTS THE ENGAGEMENT ON THE TERMS SET FORTH IN THIS AGREEMENT:
- 1.1. LMC hereby engages LMT, and LMT hereby accept such engagement, to provide day-to-day administrative and technology services to and for LMC, including but not limited to financial management, audit and Sarbanes-Oxley support, IT support, human resources support and other general or administrative support.
- 1.2. LMC shall maintain one or more bank accounts. LMC may provide LMT with authority to deposit customer receipts into such accounts, make payments from such accounts or to invest such funds in connection with the cash management system employed by LMC. The handling of receipts and disbursements with respect to such bank accounts shall be in accordance with customary business practices.
 - 1.3. LMT shall maintain, supervise and from time to time revise and amend management, financial and informational systems in support of the operations of LMC, including

1

but not limited to those required for accounting, inventory management, purchasing, sales order processing and e-mail.

- 1.4. LMT may negotiate agreements and contracts for and on behalf of LMC in the usual course of business in such areas as general liability insurance, property insurance and directors and officers insurance.
- 1.5. LMT shall have the right to act as the agents of LMC in the procuring of licenses, permits and other approvals.
- 1.6. Notwithstanding any provision in this Agreement, the full authority and responsibility for the day-to-day operation of LMC shall, during the term of this Agreement, be and remain in LMC, and LMT shall have no rights or duties except as set forth in this Agreement. LMT shall exercise commercially reasonable efforts at all times during the term of this Agreement in providing administrative and IT services as efficiently and economically as practicable. LMT shall exercise good faith efforts not to expend more than a reasonable sum of money for any goods purchased or services engaged on behalf of LMC.
- 1.7. LMC's engagement of LMT with respect to the Administrative and IT Services is not exclusive. As between LMC and LMT, LMT will have the primary day-to-day responsibilities with respect to the Administrative and IT services, but LMC will retain the authority to direct LMT. Therefore, notwithstanding any provision in this Agreement, LMC will have the right, but not the obligation, to oversee, review and monitor LMT in its performance of services and its exercise of rights and powers pursuant to this Agreement, and LMT will follow all policies and directives of LMC provided to LMT in writing with respect to the Administrative and IT Srvices.
 - 1.8. LMT shall, subject to the direction of LMC, perform the duties set forth on Exhibit A hereto
- 2. REIMBURSEMENT OF EXPENSES. LMC WILL REIMBURSE LMT FOR ITS DIRECT, OUT-OF—POCKET EXPENSES INCURRED IN CONNECTION WITH THE PROVISION OF THE SERVICES REQUIRED BY THIS AGREEMENT, INCLUDING (I) ITS REASONABLE OUT OF POCKET EXPENSES IN CONNECTION WITH TRAVEL, LODGING AND METHALS; (II) ITS REASONABLE LEGAL FEES, ACCOUNTING, AND OTHER PROFESSIONAL FEES INCURRED ON BEHALF OF LMC (PROVIDED, HOWEVER, THAT LMT WILL NOT ENGAGE ATTORNEYS, ACCOUNTANTS OR OTHER PROFESSIONALS WITHOUT THE PRIOR APPROVAL OF LMC), AND (III) OTHER DIRECT EXPENSES INCURRED ON BEHALF OF LMC. EXCEPT AS SET FORTH IN SECTION 6 BELOW, LMT WILL NOT BE REIMBURSED FOR ANY OF ITS INDIRECT OR OVERHEAD EXPENSES. LMC MAY ESTABLISH REASONABLE POLICIES IN CONNECTION WITH THE SUBMISSION OF VOUCHERS AND EVIDENCE REGARDING EXPENSES TO BE REIMBURSED, BUT ANY CHANGE IN POLICY WILL NOT APPLY TO ANY EXPENSES INCURRED BY LMT PRIOR TO THE DATE IT RECEIVES WRITTEN NOTICE OF SUCH CHANGE TO THE EXTENT THAT THE CHANGE WOULD LIMIT OR RESTRICT THEIR REIMBURSEMENT RIGHT. LMT WILL NOT BE

3. RELATIONSHIP OF THE PARTIES. LMT and LMC affirmatively state that they do not have the intention and are not forming a joint venture, partnership, association or other relationship for tax or any other purposes. The relationship between LMT and LMC created under this Agreement is that of an agent (LMT) contracting with a principal (LMC) as independent contractors.

4. INDEMNIFICATION.

- (a) LMC agrees to indemnify and hold harmless LMT and its members, officers, employees and agents (collectively, the "LMT Indemnified Parties") from and against any and all losses, claims, damages, liabilities, costs and expenses (including reasonable attorneys' fees and expenses related to the defense of any claims) (a "Loss"), which may be asserted against any of the LMT Indemnified Parties as a result of the provision of services pursuant to this Agreement (including acts and omissions relating thereto), but only if such Loss has not been caused by the willful misconduct (including a willful breach of this Agreement) of LMT.
- (b) LMT agrees to indemnify and hold harmless LMC and its members, officers, employees and agents (collectively, the "LMC Indemnified Parties") from and against all Losses which may be asserted against any of LMC Indemnified Parties as a result of the willful misconduct (including a willful breach of this Agreement) of LMT in connection with the performance by LMT of its duties hereunder or the exercise by LMT of its rights and powers hereunder if such Loss has not been caused by the gross negligence or willful misconduct (including a willful breach of this Agreement) of any of LMC Indemnified Parties.

5. TERM AND TERMINATION.

- 5.1. The term of this Agreement shall commence as of the date hereof (the "Effective Date") and shall continue in effect until the 180th calendar day after the Effective Date, unless terminated in accordance with this Section 5.
 - 5.2. LMC shall have the right to terminate immediately this Agreement for cause in the event of the occurrence of any of the following:
- (a) If LMT breaches or defaults in a material manner in the performance of any term, condition or undertaking set forth herein or breaches or fails to follow in a material manner any policy established or directive issued by LMC and fails to cure such breach or default within 30 days after LMT's receipt of written notice from LMC describing the occurrence and nature of the breach or default, or fails to submit a plan for curing the breach or default that is reasonably acceptable to LMC within such 30 day period and thereafter diligently and continuously proceeds to cure the breach or default pursuant to the plan if the breach or default cannot reasonably be cured within the 30 day period;

3

- (b) Upon the filing of a voluntary petition in bankruptcy or for reorganization under any bankruptcy law, or a petition for the appointment of a receiver for all or any substantial portion of the property of LMT;
- (c) Upon the consent by LMT to an order for relief under the federal bankruptcy laws or the failure to vacate such an order, judgment or decree, by any court of competent jurisdiction, on the application of a creditor, adjudicating LMT as bankrupt, or to be insolvent, or approving a petition seeking reorganization or the appointment of a receiver, trustee or liquidator of all or a substantial part of either of LMT's assets, if such order, judgment or decree shall continue unstaved and in effect for any period of 60 days.
 - 5.3. LMT shall have the right to terminate this Agreement for cause in the event of the occurrence of any of the following:
- (a) If LMC breaches or defaults in a material manner in the performance of any term, condition or undertaking set forth herein (except for a breach or default of any fees due pursuant to Section 6 hereof) and fails to cure such breach or default within 30 days after LMC's receipt of written notice from LMT describing the occurrence and nature of the breach or default, or fails to submit a plan for curing the breach or default that is reasonably acceptable to LMT within such 30 day period and thereafter diligently and continuously proceeds to cure the breach or default pursuant to the plan if the breach or default cannot reasonably be cured within the 30 day period; or
 - If LMC fails to timely pay any amounts that become due pursuant to Section 6 hereof; or
 - (c) If LMC or its affiliate for any reason ceases to own its business.
- 6. FEE FOR SERVICES. LMT SHALL BE ENTITLED TO RECEIVE FEES FOR THE SERVICES PROVIDED PURSUANT TO THIS AGREEMENT ON AN HOURLY BASIS BASED ON THE HOURS EXPENDED BY LMT PERSONNEL AT THE HOURLY RATES AS NOTED ON EXHIBIT B. THESE FEES SHALL BE DUE UPON PRESENTMENT OF INVOICES.
- 7. NOTICES. ALL NOTICES AND OTHER COMMUNICATIONS HEREUNDER SHALL BE IN WRITING AND SHALL BE DEEMED TO HAVE BEEN DULY GIVEN (A) WHEN DELIVERED IN PERSON OR WHEN DELIVERED BY OVERNIGHT COURIER, (B) ON THE FIRST BUSINESS DAY FOLLOWING RECEIPT WHEN DELIVERED BY TELEGRAPHIC OR OTHER ELECTRONIC MEANS (INCLUDING FACSIMILE, TELECOPY AND TELEX), OR (C) IF MAILED, UPON THE DELIVERY OR REFUSAL DATE (AS SHOWN ON THE RECEIPT), IF DEPOSITED IN THE UNITED STATES MAIL, CERTIFIED OR REGISTERED MAIL, FIRST-CLASS POSTAGE PREPAID, RETURN RECEIPT REQUESTED, TO THE PARTIES AT THE FOLLOWING ADDRESSES OR FACSIMILE NUMBERS:

4

If to LMC, to: Liquidmetal Technologies, Inc.

30452 Espera

30452 Esperanza Rancho Santa Margarita, CA 92688 Attn: Larry Buffington, CEO Fax: (949) 635-2108

With a copy to (which shall not constitute notice): Foley & Lardner LLP

100 N. Tampa Street, Suite 2700 Tampa, FL 33602 Attention: Curt P. Creely Fax: (813) 221-4210

and with a copy to (which shall not constitute notice): Bryan

Bryan Cave LLP 3500 One Kansas City Place 1200 Main Street Kansas City, Missouri 64105 Attn: Thomas W. Van Dyke Facsimile: (816) 374-3300

If to LMT, to:

Liquidmetal Technologies, Inc. 30452 Esperanza

Rancho Santa Margarita, CA 92688 Attn: John Kang, Chairman Fax: (949) 635-2108

With a copy to (which shall not constitute notice):

Foley & Lardner LLP 100 N. Tampa Street, Suite 2700 Tampa, FL 33602 Attention: Curt P. Creely Fax: (813) 221-4210

Any party from time to time may change its address or facsimile number for the purpose of receipt of notices to that party by giving a similar notice specifying a new address or facsimile number to the other notice parties listed above in accordance with the provisions of this Section 7.

8. <u>ENTIRE AGREEMENT.</u> THIS AGREEMENT SUPERSEDES ALL PRIOR ORAL DISCUSSIONS AND WRITTEN AGREEMENTS BETWEEN THE PARTIES WITH RESPECT TO THE SUBJECT MATTER OF THIS AGREEMENT (INCLUDING ANY TERM SHEET OR SIMILAR AGREEMENT OR DOCUMENT RELATING TO THE TRANSACTIONS CONTEMPLATED HEREBY). THIS AGREEMENT, TOGETHER WITH THE ASSET PURCHASE AGREEMENT,

CONTAINS THE SOLE AND ENTIRE AGREEMENT BETWEEN THE PARTIES HERETO WITH RESPECT TO THE SUBJECT MATTER HEREOF.

- 9. WAIVER. ANY TERM OR CONDITION OF THIS AGREEMENT MAY BE WAIVED AT ANY TIME BY THE PARTY WHICH IS ENTITLED TO THE BENEFIT THEREOF. ANY SUCH WAIVER MUST BE IN WRITING AND MUST BE DULY EXECUTED BY SUCH PARTY. A WAIVER ON ONE OCCASION SHALL NOT BE DEEMED TO BE A WAIVER OF THE SAME OR ANY OTHER BREACH, PROVISION OR REQUIREMENT ON ANY OTHER OCCASION.
 - 10. AMENDMENT. THIS AGREEMENT MAY BE MODIFIED OR AMENDED ONLY BY A WRITTEN INSTRUMENT DULY EXECUTED BY EACH OF THE PARTIES HERETO.
- 11. COUNTERPARTS AND FACSIMILE SIGNATURES. THIS AGREEMENT MAY BE EXECUTED SIMULTANEOUSLY IN ANY NUMBER OF COUNTERPARTS, EACH OF WHICH SHALL BE DEEMED AN ORIGINAL, BUT ALL OF WHICH SHALL CONSTITUTE ONE AND THE SAME INSTRUMENT. FACSIMILE SIGNATURES ON THIS AGREEMENT SHALL BE DEEMED TO BE ORIGINAL SIGNATURES FOR ALL PURPOSES.
- 13. GOVERNING LAW. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF CALIFORNIA APPLICABLE TO A CONTRACT EXECUTED AND PERFORMED IN SUCH STATE. THE PARTIES HERETO AGREE THAT NO PROVISIONS OF THIS AGREEMENT OR ANY RELATED DOCUMENT SHALL BE CONSTRUED FOR OR AGAINST OR INTERPRETED TO THE ADVANTAGE OR DISADVANTAGE OF ANY PARTY HERETO BY ANY AUTHORITY OR ARBITRATOR BY REASON OF ANY PARTY'S HAVING OR BEING DEEMED TO HAVE STRUCTURED OR DRAFTED SUCH PROVISION, EACH PARTY HAVING PARTICIPATED EQUALLY IN THE STRUCTURING AND DRAFTING HEREOF.
- 14. BINDING EFFECT. THIS AGREEMENT SHALL BE BINDING UPON AND WILL INURE TO THE BENEFIT OF THE PARTIES AND THEIR RESPECTIVE SUCCESSORS AND PERMITTED ASSIGNS.
- 15. ASSIGNMENT. NEITHER THIS AGREEMENT NOR ANY RIGHT HEREUNDER OR PART HEREOF MAY BE ASSIGNED BY ANY PARTY HERETO WITHOUT THE PRIOR WRITTEN CONSENT OF THE OTHER PARTIES HERETO.
 - 16. HEADINGS. THE HEADINGS USED IN THIS AGREEMENT HAVE BEEN INSERTED FOR CONVENIENCE AND DO NOT CONSTITUTE PROVISIONS

6

TO BE CONSTRUED OR INTERPRETED IN CONNECTION WITH THIS AGREEMENT.

- 17. SEVERABILITY; INVALID PROVISIONS. IF ANY PROVISION OF THIS AGREEMENT IS HELD TO BE ILLEGAL, INVALID OR UNENFORCEABLE UNDER ANY PRESENT OR FUTURE LAW, (A) SUCH PROVISIONS WILL BE FULLY SEVERABLE, (B) THIS AGREEMENT WILL BE CONSTRUED AND ENFORCED AS IF SUCH ILLEGAL, INVALID OR UNENFORCEABLE PROVISION HAD NEVER COMPRISED A PART HEREOF, (C) THE REMAINING PROVISIONS OF THIS AGREEMENT WILL REMAIN IN FULL FORCE AND EFFECT AND WILL NOT BE AFFECTED BY THE ILLEGAL, INVALID OR UNENFORCEABLE PROVISION OR BY ITS SEVERANCE HEREFROM; AND (D) IN LIEU OF SUCH ILLEGAL, INVALID OR UNENFORCEABLE PROVISION, THERE WILL BE ADDED AUTOMATICALLY AS A PART OF THIS AGREEMENT A LEGAL, VALID AND ENFORCEABLE PROVISION AS SIMILAR IN TERMS TO SUCH ILLEGAL, INVALID OR UNENFORCEABLE PROVISION AS MAY BE POSSIBLE.
- 18. <u>FURTHER ASSURANCES.</u> ON AND AFTER THE DATE HEREOF, LMC AND LMT WILL TAKE ALL APPROPRIATE ACTION AND EXECUTE ALL DOCUMENTS, INSTRUMENTS OR CONVEYANCES OF ANY KIND WHICH MAY BE REASONABLY NECESSARY OR ADVISABLE TO CARRY OUT ANY OF THE PROVISIONS HEREOF.
- 19. CONFLICTS BETWEEN AGREEMENTS. THIS AGREEMENT IS MADE PURSUANT TO THE ASSET PURCHASE AGREEMENT AND IS SUBJECT TO THE TERMS AND CONDITIONS THEREOF. IF ANY CONFLICTS EXIST BETWEEN THIS AGREEMENT AND THE ASSET PURCHASE AGREEMENT, THE ASSET PURCHASE AGREEMENT SHALL CONTROL.
- 20. EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHTS IT MAY HAVE TO DEMAND THAT ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR IN ANY WAY RELATED TO THIS AGREEMENT OR THE RELATIONSHIPS OF THE PARTIES HERETO BE TRIED BY JURY. THIS WAIVER EXTENDS TO ANY AND ALL RIGHTS TO DEMAND A TRIAL BY JURY ARISING FROM ANY SOURCE INCLUDING, BUT NOT LIMITED TO, THE CONSTITUTION OF THE UNITED STATES OR ANY STATE THEREIN, COMMON LAW OR ANY APPLICABLE STATUTE OR REGULATIONS. EACH PARTY HERETO ACKNOWLEDGES THAT IT IS KNOWINGLY AND VOLUNTARILY WAIVING ITS RIGHT TO DEMAND TRIAL BY JURY.

7

IN WITNESS WHEREOF, the parties have execut	this Agreement on the date first above written.
LMC:	LIQUIDMETAL COATINGS, LLC
	Ву:
	Title:
LMT:	LIQUIDMETAL TECHNOLOGIES, INC.
	Ву:
	Title:
	8

Exhibit A

. Financial Management

- A. Prepare financial reports
- B. Support preparation and work with audit firm
- C. Support preparation and work for Sarbanes-Oxley Section 404 compliance
- D. Process Vendor Purchases, Invoices and Accounts Payable
- E. Process Customer Orders, Invoices and Accounts Receivable
- F. Perform Human Resource functions including
 - · Process new employees
 - · Process terminated employees
 - · Compute payroll calculation
 - · Distribute payroll checks and stubs
- 2. Information Technology Services

- A. Maintain SAP financial system
- B. Maintain e-mail system
- C. Provide remote desktop support
- General and Administrative Services
 - A. Assist in selection of insurance policies and coverage
 - B. Assist in selection and transition of financial system
 - C. Assist in selection and implementation of payroll system and service
 - D. Assist in selection of local IT support company

9

		Exhibit B Rates	
VP — Finance Controller Staff Accountant VP — R&D Director — R&D Senior Technician Technician Administrative VP — IT	<u>Position</u>	Hourly Rate \$75.00 \$45.00 \$22.50 \$90.00 \$50.00 \$42.50 \$22.50 \$18.00 \$75.00	

Exhibit E

Form of License and Technical Support Agreement

10

LICENSE AND TECHNICAL SUPPORT AGREEMENT

THIS LICENSE AND TECHNICAL SUPPORT AGREEMENT (this "Agreement") is made and entered into as of the 24th day of July, 2007 (the "Effective Date"), by and between LIQUIDMETAL TECHNOLOGIES, INC., a Delaware corporation having its principal place of business at 30452 Esperanza, Rancho Santa Margarita, CA 92688 ("LMT"), and LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company having its principal place of business at 30452 Esperanza, Rancho Santa Margarita, CA 92688 ("LMC").

RECITALS:

- A. Pursuant to an Asset Purchase and Contribution Agreement of even date herewith between LMT and LMC, LMT transferred substantially all of the assets (the "Transferred Assets") of LMT's Liquidmetal Coatings business unit (the "Coatings Business") to LMC through a combination of sale and capital contribution (the "Coatings Transaction").
 - B. The Transferred Assets include certain patent rights and other intellectual property related to the Coatings Business (the "Transferred Intellectual Property").
- C. After the date hereof, LMT will continue to employ certain individuals who may from time to time develop technology or inventions that may be useful in the Coatings Business, and LMC may develop technology or inventions that may be useful in LMT's Amorphous Alloy Business (as defined below), and the parties therefore desire to enter into certain technology licenses with one another as more particularly set forth herein.
 - D. After the date hereof and upon the terms and conditions set forth herein, LMC may also wish to procure certain technical support from LMT employees relating to Metallic Coatings (as defined below).

AGREEMENTS:

NOW, THEREFORE, in consideration of the foregoing recitals and the mutual covenants and agreements set forth herein, LMT and LMC agree as follows:

Article 1. CERTAIN DEFINITIONS

For purposes of this Agreement and except as otherwise specifically set forth herein, the following capitalized terms shall have the following meanings:

1.1. "Affiliate" means, with respect to any specified person or entity, any corporation, limited liability company or other legal entity which directly or indirectly controls, is controlled by, or is under common control with specified person or entity or its successors or assigns. For the purposes of this Agreement, "control" shall mean the direct or indirect ownership of more than fifty percent (50%) of the outstanding shares on a fully diluted basis or other voting rights of the specified entity to elect directors or managers, or the right to direct or cause the direction of the management and policies of the specified entity.

whether by contract or otherwise; and the terms "controlling" and "controlled" have meanings correlative to the foregoing.

- 1.2. "Amorphous Alloy" means any one or more amorphous alloys or bulk metallic glasses (or composite materials containing amorphous alloys or bulk metallic glasses). The term "Amorphous Alloy" includes, but is not limited to, any Liquidmetal. alloys and any amorphous alloys, bulk metallic glasses, or composite materials that may be developed or conceived under this Agreement. However, the term "Amorphous Alloy" specifically excludes any Metallic Coating.
- 1.3. "Amorphous Alloy Business" means any business that is conducted by LMT on the date hereof and any business that may be conducted by LMT from time to time hereafter, including but not limited to the manufacture, sale, or distribution of Amorphous Alloys.
- 1.4. "Confidential Information" shall mean any and all commercial, technical, financial, proprietary, and other information relating to a Discloser, its affiliates, and their respective business operations, including, but not limited to, samples, data, technical information, know-how, formulas, ideas, inventions, discoveries, patents, patent applications. Intellectual Property, product development plans, demonstrations, business and financial information, applications and designs, and all manifestations or embodiments relating to the foregoing and all improvements made thereto, in whatever form provided, whether oral, written, visual, machine-readable, electronic, or otherwise. "Confidential Information" also includes any information described above which a Discloser obtains from a third party and which the Discloser treats as proprietary or designates as confidential, whether or not owned or developed by the Discloser.
- 1.5. "Discloser" shall mean the party that is disclosing Confidential Information under this Agreement, regardless of whether such Confidential Information is being provided directly by such party, by a Representative of the party, or by any other person that has an obligation of confidentiality with respect to the Confidential Information being disclosed.
- 1.6. "Improvement" shall mean any modification of a process, machine, or composition of matter that relates to the LMT Field or LMC Field, as the case may be, developed at any time during the period beginning on the Effective Date and ending on the fifth (5th) anniversary of the Effective Date.
- 1.7. "Intellectual Property" means any and all inventions (whether or not protected or protectable under patent laws), works of authorship, information fixed in any tangible medium of expression (whether or not protected or protectable under copyright laws), moral rights, trade secrets, developments, designs, applications, processes, know-how, discoveries, ideas (whether or not protectable under trade secret laws), and all other subject matter protected or protectable under patent, copyright, moral right, trademark, trade secret, or other laws, including, without limitation, all new or useful art, combinations, formulae, manufacturing techniques, technical developments, applications, data, and research results
- 1.8. "Licensee" means a party to this Agreement acting in its capacity as the grantee of a license pursuant to Article 2 hereof, and "Licensor" means a party to this Agreement acting in its capacity as a grantor of a license pursuant to Article 2 hereof.
 - 1.9. "Licensed Product" means any LMC Licensed Product or LMT Licensed Product.

- 1.11. "LMC Field" shall mean the manufacture, sale, distribution, or application of Metallic Coatings or of equipment used in the manufacture or application of Metallic Coatings.
- 1.12. "LMC Licensed Patents" means any Patents of LMT or any Affiliate of LMT that have utility in the LMC Field and that are first created, conceived, or reduced to practice by LMT during the period beginning on the Effective Date and ending on the fifth (5th) anniversary of the Effective Date.
- 1.13. "LMC Licensed Products" means any Metallic Coatings or any Metallic Coating application services or any equipment used in the application or production of Metallic Coatings utilizing technology under LMC Licensed Patents.
- 1.14. "LMC Licensed Technical Information" means unpublished research and development information, unpatented inventions, know-how, trade secrets, and technical data now or hereafter in the possession of LMT that are reasonably necessary or useful for using the LMC Licensed Patents to produce LMC Licensed Products within the LMC Field, provided LMT has the right to disclose such items to LMC and provided that such items come into the possession of LMT on or before the fifth (5th) anniversary of the date of the Effective Date.
- 1.15. "LMT Field" shall mean the manufacture, sale, distribution, or application of Amorphous Alloys, products containing Amorphous Alloys, or equipment used in the production of Amorphous Alloys or products containing the same.
- 1.16. "LMT Licensed Patents" means any Patents of LMC or any Affiliate of LMC that have utility in the LMT Field and that are first created, conceived, or reduced to practice by LMC during the period beginning on the Effective Date and ending on the fifth (5th) anniversary of the Effective Date.
- 1.17. "LICENSED Products" means any Amorphous Alloys, products containing Amorphous Alloys, or equipment used in the production of Amorphous Alloys or products containing Amorphous Alloys utilizing technology under LMT Licensed Patents.
- 1.18. "LMT Licensed Technical Information" means unpublished research and development information, unpatented inventions, know-how, trade secrets, and technical data now or hereafter in the possession of LMC that are reasonably necessary or useful for using the LMT Licensed Patents to produce LMT Licensed Products within the LMT Field, provided LMC has the right to disclose such items to LMT and provided that such items come into the possession of LMC on or before the fifth (5th) anniversary of the Effective Date.
- 1.19. "Metallic Coating" means any metallic coating material in the form of a wire or powder that is sprayed in liquid form on industrial equipment for use as a coating to protect against wear, corrosion, or abrasion, including the coatings materials historically branded and sold by LMT prior hereto under the Liquidmetal and Armacor brands.
- 1.20. "Net Sales Price" means, for purposes of computing royalties under this Agreement, a Licensee's invoice price, after deduction of regular trade and quantity discounts, but before deduction of any other items, including, but not limited to, freight allowances, cash discounts, and agents' commissions.
- 1.21. "Patents" means any and all letters patent (including, but not limited to, patents of implementation, improvement, or addition, utility model and appearance design patents, and inventors certificates, as well as all divisionals, reissues, reexaminations, continuations, continuations-in-part, renewals, extensions, substitutions, foreign equivalents and counterparts, and any other forms of patent protection directed to the inventions covered by any of the foregoing), applications for letters patent

3

(including, but not limited to, all foreign counterpart patent applications), and letters patent that may issue on such applications.

- 1.22. "Recipient" shall mean the party receiving Confidential Information that is protected under this Agreement.
- 1.23. "Representatives" shall mean the respective directors, officers, employees, financial advisors, accountants, attorneys, agents, and consultants of a party.
- 1.24. "Sublicensing Revenue" shall mean the fair market cash value of any and all consideration received from sublicenses in consideration for sublicenses granted by Licensor or its Affiliates hereunder, including without limitation running royalties, license issue fees and other licensing fees, milestone payments, research grants, minimum annual royalties, equity or other payments of any kind whatsoever, irrespective of whether such revenues are received in the form of cash, barter, credit, stock, warrants, release from debt, goods or services, licenses back, or any other form whatever.

Article 2. LICENSE GRANT

- 2.1. <u>License Grant to LMC. Upon the terms and conditions set forth herein, LMT hereby grants to LMC a worldwide, exclusive, perpetual, royalty-bearing license (including the right to grant sublicenses) under the LMC Licensed Patents, LMC Licensed Technical Information and Improvements to make, have made, use, offer to sell, sell, export and import LMC Licensed Products within the LMC Field.</u>
- 2.2. License Grant to LMT. Upon the terms and conditions set forth herein, LMC hereby, grants to LMT a worldwide, exclusive, perpetual, royalty-bearing license (including the right to grant sublicenses) under the LMT Licensed Patents, LMT Licensed Technical Information and Improvements to make, have made, use, offer to sell, sell, export and import LMT Licensed Products within the LMT Field.
- 2.3. **Reservation of Rights.** All rights not specifically, granted to Licensee by this Agreement are expressly reserved by Licensor. Licensor reserves the right to continue its use of the Licensed Patents for purposes of research and development.
- 2.4. Notice of Improvements. During the Term, Licensor shall on at least a semiannual basis provide Licensee with written notice of all Improvements in the LMC Field or the LMT Field (as the case may be) conceived, discovered or developed by Licensor or any of its Affiliates since the Effective Date or last providing such notice to Licensee (an "Improvement Notice"). The Improvement Notices shall include a written description of each such Improvement in sufficient detail so as to allow its technical and proprietary merit to be promptly and fully evaluated. Such Improvements shall automatically be licensed to the Licensee pursuant to this Article 2.
 - 2.5 Sublicenses. Any sublicense granted by a Licensee shall only contain terms consistent with this Agreement, including, without limitation, Article 4 hereof.

Article 3. ROYALTY

3.1. **Royalty Rate.** LMC agrees to pay LMT a royalty equal to three percent (3.0%) of the Net Sales Price of LMC Licensed Products by LMC or any Affiliate of LMC, and LMT agrees to pay LMC a royalty equal to three percent (3.0%) of the Net Sales Price of LMT Licensed Products by LMT or any Affiliate of LMT.

4

- 3.2. **Royalties for Transactions Not at Arm's Length.** In order to assure to Licensor the full royalty payments contemplated by this Agreement, Licensee agrees that in the event any Licensed Product shall be sold or sublicensee (1) to a Licensee Affiliate, or (2) to a corporation, firm, or association with which, or individual with whom, Licensee or its stockholders or affiliates shall have any agreement, understanding or arrangement (such as, among other things, an option to purchase stock, or an arrangement, involving a division of profits or special rebates or allowances) without which agreement, understanding or arrangement, involving a division of profits or special rebates or allowances) without which agreement, understanding or arrangement, involving a division of profits or special rebates or allowances) without which agreement, understanding or arrangement results in extending to such corporation, firm, association or individual lower prices or royalties for Licensed Product than those charged to outside concerns buying similar products in similar amounts and under similar conditions, then, and in any such events, the royalties to be paid hereunder in respect of such Licensed Product shall be computed on the Net Sales Price at which (or the royalty applicable to) products of similar kind and quality, sold in similar quantities, are then currently being offered for sale by other manufacturers.
 - 3.3. Quarterly Payment. Royalties shall be paid to Licensor on a quarterly basis within twenty (20) days following the end of each calendar quarter.
- 3.4. Late Fees. In the event Licensee is late in making payments to Licensor under any provision of this Agreement, for each month (or portion thereof) that such payments are late, Licensee shall pay Licensor a late charge equal to the lesser of one and one-half percent (1.5%) of the late payment or the maximum rate permitted by law multiplied by the amount of the payment that is late (including any prior accumulated late charges).
- 3.5. Taxes. All payments due hereunder shall be paid without deduction for taxes, assessments, or other charges of any kind or description that may be imposed on Licensor by any government except the federal government of the United States with respect to any amounts payable to Licensor pursuant to this Agreement, and such taxes, assessments, and other charges shall be paid for and assumed by Licensee.
- 3.6. Currency. All payments specified in this Agreement shall be made in United States Dollars. Licensee shall pay all amounts owed to Licensor by wire transfer of immediately available funds unless otherwise agreed upon in writing between the parties.
 - 3.7. Expiration of Patent. Licensee's obligation to pay royalties in respect of a Licensed Patent shall end upon the expiration of such Licensed Patent.
- 3.8. Royalty on Materials Not Sold Separately. The parties acknowledge that some of the Licensed Products will not be sold separately, but rather will be sold in combination with or as parts of other products or services. For example, some products sold by Licensee may be include in application projects which involve services, materials and other costs. In all such cases, where Licensed Products are not sold separately, but are sold in combination with or as parts of other products or services, the Net Sales Price of the Licensed Products so sold shall be calculated for the purpose of computing royalties due by determining the portion of the total Net Sales Price attributable to the Licensed Products as if the Licensed Products had been sold independently in similar quantities and terms,

3.9. Sublicense Revenue. Licensee shall pay to Licenser an amount equal to fifty percent (50%) of Sublicensing Revenues actually received by Licensee or any Affiliate of Licensee. Consideration received by Licensee in the form of equity or other securities shall be paid to Licensor in kind (provided that such a transfer of securities is permitted), and consideration received in the form of goods shall be paid to Licensor in cash based upon the fair market value of such goods actually received.

Article 4.
REPORTS AND AUDITS

5

4.1. **Reports.** Licensee shall keep accurate and sufficient records to determine amounts owed to Licensor under this Agreement. Licensee shall make a written report detailing the basis for any computations to Licensor within thirty. (30) days following each calendar quarter. Along with such reports, Licensee shall transmit payment for the royalty shown to be due. In the event no royalty is due, the report shall so state. Records necessary for the computation of amounts payable by Licensee under this Agreement shall be maintained by Licensee for a period of five (5) years following each accounting report due hereunder.

4.2. Audits. Such records of Licensee shall be open to inspection by an auditor selected by Licensor during regular business hours of Licensee. Such records shall only be used by Licensor or its agents to determine the accuracy of the royalties paid and reports submitted. Licensor shall bear the expenses of the auditor it selects, except Licensee shall reimburse Licensor for such expenses in the event the total underpayment of royalties identified during the audit exceeds the cost of the audit. If royalties due as determined by the audit are greater than royalties actually paid by Licensee, Licensee will be invoiced for the difference, along with interest at the rate specified in Section 3.5 (Late Fees) hereof.

6

Article 5. NO WARRANTIES

EXCEPT FOR THE SPECIFIC PROVISIONS OF THIS AGREEMENT, EACH PARTY EXPRESSLY DISCLAIMS ALL REPRESENTATIONS AND WARRANTIES REGARDING THE TECHNOLOGY LICENSED HEREUNDER, WHETHER EXPRESS, IMPLIED, STATUTORY, OR ARISING OUT OF CUSTOM OR TRADE USAGE, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, AND NON-INFRINGEMENT. THE PARTIES MAKE NO WARRANTIES WITH RESPECT TO FREEDOM FROM ALLEGED INFRINGEMENT OF THIRD PARTY PATENTS OR FREEDOM FROM THIRD PARTY INFRINGERS, AND LICENSOR IS NOT UNDER ANY OBLIGATION TO HOLD LICENSEE HARMLESS AGAINST SUCH ALLEGED INFRINGEMENT OF THIRD PARTY PATENTS NOR TO ENFORCE ITS PATENT PROPERTIES AGAINST ALLEGED INFRINGERS. NOTHING IN THIS AGREEMENT SHALL BE DEEMED TO BE A REPRESENTATION OR WARRANTY BY LICENSOR OF THE ACCURACY, SAFETY, OR USEFULNESS FOR ANY PURPOSE OF ANY TECHNICAL INFORMATION, TECHNIQUES, OR PRACTICES AT ANY TIME MADE AVAILABLE BY LICENSOR. NO ORAL OR WRITTEN INFORMATION OR ADVICE GIVEN BY LICENSOR OR LICENSOR'S AUTHORIZED REPRESENTATIVES SHALL CREATE A WARRANTY OR REPRESENTATION. THIS SECTION SHALL BE ENFORCEABLE TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW.

IN NO EVENT SHALL A PARTY BE LIABLE TO THE OTHER PARTY UNDER THIS AGREEMENT OR ANY THIRD PARTY FOR ANY INCIDENTAL OR CONSEQUENTIAL DAMAGES (INCLUDING, WITHOUT LIMITATION, ANY INDIRECT, SPECIAL, PUNITIVE, OR EXEMPLARY DAMAGES FOR LOSS OF BUSINESS, LOSS OF PROFITS, LOSS OF REVENUES, BUSINESS INTERRUPTION, LOSS OF SOFTWARE, LOSS OF DATA, LOSS OF BUSINESS INFORMATION, THE INADEQUACY OF THE LICENSED MATERIALS, FOR ANY OTHER ITEMS RELIATING TO THE SUBJECT MATTER OF THIS AGREEMENT.

Article 6. TRADEMARK LICENSE

- 6.1. Trademark License. LMT hereby grants to LMC a perpetual, non-exclusive worldwide, royalty-free, fully paid up, non-transferable license to use the "Liquidmetal" trademark (the "Trademark") solely in connection with the marketing and sale of the products and services in its Coatings Business, subject to the following terms and conditions:
- (a) Except as otherwise agreed to by LMT in writing, all use of the Trademark by LMC is subject to LMT's standard trademark usage policy in effect from time to time (provided that LMT delivers a copy of such policy to LMC).
 - (b) All stylized use of the Trademark shall be solely in the original logotype identified by LMT, except as otherwise agreed in writing by LMT.
- (c) LMC agrees not to affix the Trademark to products other than the products in its Coatings Business. Furthermore, Licensee agrees not to attach any additional trademarks, logos, or designations to such without the prior written consent of LMT (which consent will not be unreasonably withheld). The "®" icon shall always follow the Trademark.
 - (d) LMC shall not challenge the validity of LMT's rights in and to the Trademark or the validity of the Trademark or any registration(s) thereof. LMC agrees that it shall not register or

7

attempt to register the Trademark or any other trademark or trade name of LMT, or use or register any other trademark or trade name which may be confusingly similar to the Trademark or any other trademark or trade name of LMT.

- (e) LMC shall promptly, upon receipt of notice thereof, fully inform LMT as to any actual or proposed action, by any governmental agency, consumer or environmental group, media or other organization, directed toward removing from the market any Liquidmetal-branded product based on alleged injury or death, alleged potential for harm, product defect, alleged contamination, tampering or similar occurrence, actual or alleged violation of law in connection with production, labeling, packaging, storage, shipment, advertising or sale, or for any other reason whatsoever. LMC shall likewise promptly inform LMT as to any proposal to remove from the market any such as described above on account of suspected nonconformity with applicable product quality or safety standards, improper labeling, possibility of consumer harm, and/or violation of any law or regulation.
- (f) LMC shall be permitted to sublicense the Trademark to any third party that complies with the usage conditions and restrictions set forth in this Section 6.1, including without limitation that the sublicensee may only use the Trademark in connection with the marketing and sale of the products and services in LMC's Coatings Business. LMC shall cause any such sublicense to comply with the terms, conditions, and restrictions of this Section 6.1.

Article 7. TECHNICAL ASSISTANCE AND SUPPORT

Upon the request of LMC and at reasonable times and upon reasonable notice, LMT agrees to make its employees available to provide technical support and assistance with respect to business, products, and technology of LMC. In consideration of such technical support and assistance, LMC will pay LMT an hourly charge in accordance with the fee schedule attached as Exhibit A hereto. The obligation to provide such technical support and assistance shall expire on the fifth (5²⁰) anniversary of the Effective Date unless the parties otherwise agree.

Article 8. TERM AND TERMINATION

- 8.1. Term. Except as set forth in Section 8.2 below, the rights and licenses set forth in this Agreement shall be perpetual in nature.
- 8.2. <u>Termination</u>. Notwithstanding any other provision contained herein, this Agreement may be terminated as follows:
- (a) Material Breach. This Agreement shall terminate on the thirtieth (30th) day after either party gives the other party written notice of a material breach by the other party of any term or condition of this Agreement, unless the breach is cured before that day. The right of a party to terminate this Agreement shall be in addition to and not in lieu of any other right or remedy that the terminating party may have at law or in equity.
- (b) Bankruptcy. This Agreement may be terminated immediately by a party in the event the other party becomes insolvent, files or has filed against it a petition under any chapter of the United States Bankruptcy Code (or any similar petition under the insolvency law of an applicable jurisdiction) and such petition is not dismissed within thirty (30) days, proposes any dissolution, liquidation, financial reorganization, or re-capitalization with creditors, or makes an assignment or trust

8

mortgage for the benefit of creditors, or if a receiver, trustee, custodian, or similar agent is appointed or takes possession of any property or business of such other party.

(c) Patent Expiration. The grant of rights in the LMT Licensed Patents shall terminate upon the expiration of the last to expire of the LMT Licensed Patents. The grant of rights in the LMC Licensed Patents shall terminate upon the expiration of the last to expire of the LMC Licensed Patents.

8.3. <u>Effect of Termination.</u>

- (a) Rights Upon Termination. Upon termination of this Agreement, the licenses and all other rights granted to a party under this Agreement shall immediately terminate.
- (b) Continuation of Obligations. In addition to any other provisions of this Agreement which by their terms are intended to survive any termination of this Agreement, after termination of this Agreement, the provisions concerning the parties' obligations and responsibilities under Article 9 (Confidentiality) shall continue in full force and effect for an additional period of ten (10) years, and indefinitely for trade secrets; and a Licensee's payment and other obligations under Section 3 (License Fees and Royalties) shall continue in effect until paid.

Article 9. CONFIDENTIALITY

9.1. <u>Terms of Agreement. Each party agrees not to disclose any terms of this Agreement to any third party without the consent of the other party: provided, however, that disclosures may be made as required by securities or other applicable laws; or by either party to its accountants, attorneys, and other professional advisors. Neither party shall release any publicity or information concerning this Agreement without the other party's prior written approval, which shall not be unreasonably withheld or delayed, provided that the parties agree to issue a jointly-written press release following the execution of this Agreement.</u>

9.2 Restrictions on Disclosure and Use

(a) Restrictions and Covenants. Except as otherwise provided herein, each party agrees that, in its capacity as the Recipient of Confidential Information, it will (i) hold the Discloser's Confidential Information in strict confidence, use a high degree of care in safeguarding the Discloser's Confidential Information, and take all precautions necessary to protect the Discloser's Confidential Information including, at a minimum, all precautions the Recipient normally employs with respect to its own Confidential Information, (ii) not divulge any of the Discloser's Confidential Information derived therefrom (including results of tests on material samples) to any other person (except as set forth in Section 9.2(b) (Disclosure to Representatives) hereof), (iii) not make any use whatsoever at any time of the Discloser's Confidential Information except as is necessary in the performance of Recipient's specific duties under this Agreement, (iv) not copy, reverse engineer, alter, modify, break down, melt down, disassemble or transmit any of the Discloser's Confidential Information, (v) not, within the meaning of United States or other export control laws or regulations, export or re-export, directly or indirectly, including but not limited to export on the Internet or other network service, any of the Discloser's Confidential Information, (vi) notify the Discloser in writing immediately upon discovery by the Recipient or its Representatives of any unauthorized use or disclosure of the Discloser's Confidential Information, and (vii) upon the termination or expiration of this Agreement, immediately return to the Discloser's Confidential Information, including all originals, copies and extracts.

9

- (b) <u>Disclosure to Representatives</u>. The Recipient may only disseminate the Discloser's Confidential Information to its Representatives who have been informed of the Recipient's obligations under this Agreement and are bound by an obligation of confidentiality and non-use with respect to the Discloser's Confidential Information at least as broad in scope as the Recipient's obligations under this Agreement. The Recipient agrees to reasonably restrict disclosure of the Discloser's Confidential Information to the smallest number of the Recipient's Representatives which have a need to know the Confidential Information. The Recipient shall be responsible for enforcing this Agreement as to the Recipient's Representatives and shall take such action (legal or otherwise) to the extent necessary to cause them to comply with this Agreement.
- (c) Trade Secrets. Any trade secrets of the Discloser will also be entitled to all of the protections and benefits of applicable trade secret law, and the Recipient agrees to be bound by all applicable trade secret laws, unfair competition laws, and any other similar laws with respect to the Discloser's Confidential Information. If any Confidential Information that the Discloser deems to be a trade secret is found by a court of competent jurisdiction not be a trade secret under applicable law, such Confidential Information will nevertheless still be protected by this Agreement.
- (d) Protection of Licensed Technical Information by Licensee. Licensee acknowledges and agrees that the Licensed Technical Information derives economic value from not being generally known to other persons who can obtain economic value from its disclosure or use. Therefore, without the express written consent of Licensor, Licensee covenants and agrees that it, its employees, contractors, representatives, successors, assigns, affiliates, parents, subsidiaries, officers, directors, and the like will (1) hold the Licensed Technical Information in strict confidence, use a high degree of care in safeguarding the Licensed Technical Information, and take all precautions reasonably necessary to protect the Licensed Technical Information, and precautions Licensee or many information or any information derived therefrom to any person other than Licensor, (3) not make any use whatsoever at any time of the Licensed Technical Information except in furtherance of Licensee's obligations to Licensor and as necessary to produce Licensed Products in accordance with the license granted under this Agreement, (4) not, within the meaning of United States or other export control laws or regulations, export or re-export, directly or indirectly, including but not limited to export on the Internet or other network service, any of the Licensed Technical Information, and (5) notify Licensor in writing immediately upon discovery of any unauthorized use or disclosure of the Licensed Technical Information by Licensee or its employees or any third party.
- (e) Enforcement. Licensee acknowledges and agrees that due to the unique nature of the Licensed Technical Information and other Confidential Information of Licensor, there can be no adequate remedy at law for any breach of its obligations hereunder, which breach may result in irreparable harm to Licensor, and therefore, that upon any such breach or any threat thereof, Licensor shall be entitled to appropriate equitable relief, including injunction, without the requirement of posting a bond, in addition to whatever remedies it might have at law.
 - Exceptions. The restrictions on the Recipient's disclosure and use of the Discloser's Confidential Information under this Section 9.2 will not apply to the extent of any Confidential Information:
 - (i) that becomes publicly known without breach of the Recipient's or its Representatives' obligations under this Agreement;

10

- (ii) that is rightfully acquired by Recipient from a third party which is not subject to any restriction or obligation (whether contractual, fiduciary, or otherwise) on disclosure or use of such Confidential Information:
- (iii) that is independently developed by employees of the Recipient without knowledge of or reference to such Confidential Information, as evidenced by written documentation or other tangible evidence of Recipient:
- (iv) that is required to be disclosed by law or by court order or government order, provided that the Recipient (a) promptly notifies the Discloser of any such disclosure requirement so that the Discloser may seek an appropriate protective order (or other appropriate protections), and (b) provides reasonable assistance (at no cost to the Recipient) in obtaining such protective order or other form of protection; or
 - (v) as to which and to the extent to which the Recipient has received express written consent from an authorized officer of the Discloser to disclose or use.
- 9.3. Third Party Information. Each party represents and warrants to the other that it is free to divulge, without any obligation to or violation of the rights of any third party, any and all information which it will demonstrate divulge, or in any other manner make known to the other pursuant to this Agreement. Each party shall indemnify and hold harmless the other from and against any and all liability, loss, cost, expense, damage, claim or demand for actual violation of the rights of any third party in any trade secret, proprietary know-how, or other confidential information by reason of the other party's receipt of information disclosed hereunder. The foregoing provision shall not be construed to affect or diminish the obligations of confidentiality and non-disclosure of the parties as provided in this Article 9 (Confidentiality).

Article 10. MISCELLANEOUS

- 10.1. Force Majeure. Excluding payment obligations, neither party shall be liable for, nor shall it be considered in breach of this Agreement due to, any failure to perform its obligations under this Agreement as a result of a cause beyond its control, including any act of God or public enemy, act of any military, civil or regulatory authority, terrorism or threat thereof, change in any law or regulation, fire, flood, earthquake, storm or other like event, disruption or outage of communications, power or other utility, labor problem, unavailability of supplies, or any other cause, whether similar or dissimilar to any of the foregoing, which could not have been prevented by the party with reasonable care.
- 10.2. Notices. All notice, requests, demands and other communications hereunder shall be in English and shall be given in writing and shall be: (i) personally delivered; (ii) sent by telecopier, facsimile transmission or other electronic means of transmitting written documents with confirmation of receipt; or (iii) sent to the parties at their respective addresses indicated herein by registered or certified mail, return receipt requested and postage prepaid, or by private overnight mail courier services with confirmation of receipt. The respective addresses to be used for all such notices, demands or requests are as follows:

()	If to LMC

11

Or to such other person or address as LMC shall furnish to LMT in writing.

() If to LMT:

Liquidmetal Technologies 30452 Esperanza Rancho Santa Margarita, CA 92688 Attention: John Kang Phone No.: (949) 635-2100 Fax No.: (949) 635-2108

With copies to:

Or to such other person or address as LMT shall furnish to LMC in writing.

If personally delivered, such communication shall be deemed delivered upon actual receipt by the "attention" addressees or persons authorized to accept for such addressees; if transmitted by facsimile pursuant to this paragraph, such communication shall be deemed delivered the next business day after transmission (and sender shall bear the burden of proof of delivery); if sent by overnight courier pursuant to this paragraph, such communication shall

be deemed delivered upon receipt by the "attention" addressees or persons authorized to accept for such addressees; and if sent by mail pursuant to this paragraph, such communication shall be deemed delivered as of the date of delivery indicated on the receipt issued by the relevant postal service, or, if the addressee fails or refuses to accept delivery, as of the date of such failure or refusal. Any party to this Agreement may change its address for the purposes of this Agreement by giving notice thereof in accordance with this paragraph.

- 10.3. **Independent Contractors.** In the performance of this Agreement, Licensor and Licensee are independent contractors. Neither party nor any of its employees or agents shall be considered an employee or agent of the other party. Nor shall any partnership, co-venture or joint-employer relationship be created or implied by virtue of this Agreement or of its performance. The parties intend that this Agreement shall not create a partnership for tax purposes.
- 10.4. Severability. Each provision contained in this Agreement is declared to constitute a separate and distinct covenant and provision and to be severable from all other separate, distinct covenants and provisions. It is agreed that should any clause, condition or term, or any part thereof, contained in this Agreement be unenforceable or prohibited by law or by any present or future legislation then such clause, condition, term or part thereof, shall be amended, and is hereby amended, so as to be in compliance with the said legislation or law, then such clause, condition, term or part thereof is severable from this Agreement, and all the rest of the clauses, terms and conditions or parts thereof contained in this Agreement shall remain unimpaired and continue in full force and effect.
 - 10.5. Amendment. This Agreement may not be amended or modified other than by a written agreement executed by the parties hereto or their respective successors and legal representatives
 - 10.6. Waiver. No waiver of a breach of any provision of this Agreement, whether or not similar, nor

12

nor shall such waiver constitute a continuing waiver of such breach unless otherwise expressly provided in such waiver.

- 10.7. **Governing Law.** This Agreement, the legal relations between the parties, and any action, whether contractual or non-contractual, instituted by any party with respect to matters arising under or growing out of or in connection with or in respect of this Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware (U.S.A.), excluding any choice of law rules that may direct the application of the laws of another jurisdiction, and except that questions affecting the construction and effect of any Patent shall be determined by the law of the country in which the Patent has been granted.
- 10.8. **Resolution of Disputes.** The parties irrevocably agree that any legal actions or proceedings brought by or against them with respect to this Agreement shall be brought exclusively in the state or federal courts in and for Orange County, California, and by execution and delivery hereof, the parties irrevocably submit to such jurisdiction and hereby irrevocably waive any and all objections which they may have with respect to venue in any of the above courts. THE PARTIES HEREBY EXPRESSLY WAIVE ANY AND ALL RIGHT TO A TRIAL BY JURY WITH RESPECT TO ANY ACTION, PROCEEDING OR OTHER LITIGATION RESULTING FROM OR INVOLVING THE ENFORCEMENT OF THIS AGREEMENT.
- 10.9. Attorneys' Fees. In any action between the parties for relief based in whole or in part on this Agreement (or the breach thereof), including actions to collect overdue royalty payments, the prevailing party shall be entitled to recover (in addition to any other relief awarded or granted) its reasonable costs and expenses (including attorneys' fees and expert witness fees) incurred in the proceeding,
- 10.10. Entire Agreement. This Agreement sets forth the complete agreement of the parties concerning the subject matter hereof. No claimed oral agreement in respect thereto shall be considered as any part hereof. No waiver of or change in any of the terms hereof subsequent to the execution hereof claimed to have been made by any representative of either party shall have any force or effect unless in writing, signed by duly authorized representatives of the parties.
 - 10.11. Recitals. The recitals set forth in the preamble to this Agreement are true and correct and are made a part of this Agreement.
- 10.12. Assignment. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Neither party shall assign its rights or duties under this Agreement in whole or in part, without the prior written consent of the other party, except that either party may assign this Agreement without the consent of the other party as a part of the sale or transfer of all or substantially all of the assets or business of the assigning party.
 - 10.13. Headings. The section and paragraph headings in this Agreement are for convenience only and are not intended to affect the meaning or interpretation of this Agreement.
- 10.14. Contract Interpretation. Ambiguities, inconsistencies, or conflicts in this Agreement shall not be strictly construed against the drafter of the language but will be resolved by applying the most reasonable interpretation under the circumstances, giving full consideration to the parties' intentions at the time this Agreement is entered into. Where the context of this Agreement requires, singular terms shall be considered plural, and plural terms shall be considered singular.
 - 10.15. Counterparts. This Agreement may be executed simultaneously in counterparts, each of which will be deemed an original, but all of which together will constitute the same Agreement.

13

LIQUIDMETAL TECHNOLOGIES, INC.

10.16. Language. In the event of controversy between the parties respecting the interpretation or application of this Agreement, the English language version of the Agreement shall be controlling,

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their duly authorized representatives as of the Effective Date set forth above:

By: Name: Title: LIQUIDMETAL COATINGS, LLC

14

By: Name: Title:

EXHIBIT A

 Level
 Hourly Cost

 Senior Scientist
 Technician

15

Exhibit F

Form of Promissory Note for Upward Purchase Price Adjustment

PROMISSORY NOTE

\$155,298.00 Effective as of , 2007

FOR VALUE RECEIVED, the undersigned LIQUIDMETAL COATINGS, LLC ("Maker"), promises to pay to the order of LIQUIDMETAL TECHNOLOGIES, INC. ("Holder") or any subsequent holder of this Promissory Note (this "Note"), at 30452 Esperanza, Rancho Santa Margarita, CA 92688 or at such other place as Holder may designate, the principal sum of the Upward Purchase Price Adjustment, as that term is defined in the Asset Purchase and Contribution Agreement between Maker and Holder, or One Hundred Fifty Five Thousand Two Hundred Ninety Eight Dollars (\$155,298.00), together with any accrued interest thereon, which shall be due and payable upon the following terms and conditions contained in this Note.

- 1. The outstanding principal balance of this Note shall be due and payable on the date that is forty-five (45) calendar days after the date hereof. This Note may be prepaid, either in whole or in part, at any time without penalty.
- 2. The outstanding principal amount of this Note shall bear interest at a rate equal to the Prime Rate (as defined below) plus .5%, calculated on a 360-day year basis. The "Prime Rate," as used in this Note, shall mean that rate of interest announced from time to time by Bank of America, N.A. as its "prime rate," it being understood and agreed that such rate shall not necessarily be the lowest rate then offered by Bank of America, N.A. to its most

creditworthy borrowers. Any change in the Prime Rate shall be effective as of the beginning of the day on which such change is announced to become effective.

- 3. The remedies under this Note shall be cumulative. Failure of the Holder to exercise any of its rights and remedies under this Note shall not constitute a waiver of the right to exercise the same at that or any other time. All rights and remedies of the Holder for default under this Note shall be cumulative to the greatest extent permitted by law. If there is any default under this Note, and this Note is placed in the hands of an attorney for collection or is collected through any court, including any bankruptcy court, Maker promises to pay to the Holder's reasonable attorneys' fees and court costs incurred in collecting or attempting to collect this Note or enforcing the Holder's rights hereunder.
- 4. This Note shall be governed by and construed in accordance with the laws of the State of Delaware, without reference to principles of choice of law thereunder. The venue for any judicial or arbitration proceedings arising out of this Note or the obligations hereunder shall be in the state courts of the State of Texas. As it is the intent of all parties to this transaction to abide by the interest limitations of any applicable usury law, it is expressly agreed, anything herein to the contrary notwithstanding, that the Holder shall not be allowed or entitled to collect any interest (or any sum which is considered interest by law) which is in excess of any legal rate

applicable hereto. Should any amount be collected hereunder which would cause the interest to exceed said lawful rate, such part of said amount in excess of the lawful rate shall automatically be credited to principal, or, if all principal amounts have been paid, shall be refunded to the Maker. The provisions of this Note are hereby modified to the extent necessary to conform with the limitations and provisions of this paragraph. This paragraph shall govern over all other provisions in any document or agreement now or hereafter existing. Maker shall pay all documentary stamp taxes on this Note. EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY SUIT, ACTION OR PROCEEDING (INCLUDING ANY COUNTERCLAIM) ARISING OUT OF OR BASED UPON THIS NOTE.

MAKER

LIQUIDMETAL COATINGS, LLC

By:

Name: Larry E. Buffington

Title: President and Chief Executive Officer

2

Exhibit 2.2

LOAN AGREEMENT

THIS LOAN AGREEMENT (as amended, modified or supplemented from time to time, this "Agreement") made this 24th day of July, 2007 (the "Effective Date"), by and between BANK MIDWEST, N.A. ("Lender"), with an address of 1100 Main Street, Kansas City, Missouri 64105, LIQUIDMETAL COATINGS, LLC ("LMC"), with an address of c/o Liquidmetal Technologies, Inc., 30452 Esperanza, Rancho Santa Margarita, California 92688 and LIQUIDMETAL COATINGS SOLUTIONS, LLC ("LMCS"), with an address of c/o Liquidmetal Technologies, Inc., 30452 Esperanza, Rancho Santa Margarita, California 92688. LMC and LMCS, jointly and severally, hereinafter referred to as "Borrowers", and each individually, "Borrower".

WITNESSETH

WHEREAS, Borrowers may from time to time request Loans from Lender, and the parties wish to provide for the terms and conditions upon which such Loans or other financial accommodations, if made by Lender, shall be made:

NOW, THEREFORE, in consideration of any Loan (including any Loan by renewal or extension) hereafter made to Borrowers by Lender, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by each Borrower, the parties agree as follows:

ARTICLE I DEFINITIONS

As used herein, the following definitions shall apply:

"Debt" means, as to any Person, without duplication, (a) all indebtedness of such Person; (b) all borrowed money of such Person (including principal, interest, fees and charges), whether or not evidenced by bonds, debentures, notes or similar instruments; (c) all obligations to pay the deferred purchase price of property or services; (d) all obligations, contingent or otherwise, with respect to the maximum face amount of all letters of credit (whether or not drawn), bankers' acceptances and similar obligations issued for the account of such Person (including the letters of credit), and all unpaid drawings in respect of such letters of credit, bankers' acceptances and similar obligations; (e) all indebtedness secured by any Lien on any property owned by such Person, whether or not such indebtedness has been assumed by such Person (provided, however, if such Person has not assumed or otherwise become liable in respect of such indebtedness, such indebtedness shall be deemed to be in an amount equal to the fair market value of the property subject to such Lien at the time of determination); (f) the aggregate amount of all capitalized lease obligations of such Person; (g) all contingent liabilities of such Person, whether or not reflected on its balance sheet; (h) all hedging obligations of such Person; (i) all Debt of any partnership of which such Person is a general partner; and (j) all monetary obligations of such Person under (i) a so-called synthetic, off-balance sheet or tax retention lease, or (ii) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the indebtedness of such Person (without regard to accounting

treatment). Notwithstanding the foregoing, Debt shall not include trade payables and accrued expenses incurred by such Person in accordance with customary practices and in the ordinary course of business of such Person

"Debtor Relief Proceeding" means any bankruptcy, reorganization, liquidation, conservatorship, receivership or other similar proceedings under any state or federal law now or hereafter in effect which are commenced by or against either or both of the Borrowers or if either or both of the Borrowers or if either or both of the Borrowers otherwise take any action to restrain, enjoin or otherwise impede the Lender in the exercise of the remedies afforded to Lender under this Agreement or under any of the obligations owed by each Borrower to the Lender, or at law or in equity.

"Depreciation" shall mean the total amounts added to depreciation, amortization, obsolescence, valuation and other proper reserves, as reflected on Borrower's financial statements and determined in accordance with

"EBITDA" means, for any period, the sum for such period of (i) Net Income, <u>plus</u> (ii) Interest Charges, <u>plus</u> (iii) federal and state income taxes, <u>plus</u> (iv) Depreciation, <u>plus</u> (v) all other non-cash charges, in each case to the extent included in determining Net Income for such period.

"Eligible Cash" means cash on deposit with Lender

GAAP.

"Eligible Inventory" means all inventory of Borrowers which meets each of the following requirements:

- (a) it is subject to a perfected, first priority Lien in favor of Lender and is not subject to any other assignment, claim or Lien other than Permitted Liens;
- (b) it is in the possession and control of any Borrower and it is stored and held in facilities owned by a Borrower or, if such facilities are not so owned, Lender is in possession of a collateral access agreement with respect thereto;
 - (c) it is not inventory produced in violation of the Fair Labor Standards Act and subject to the "hot goods" provisions contained in Title 29 U.S.C. §215;
 - (d) it is not subject to any agreement or license which would restrict Lender's ability to sell or otherwise dispose of such inventory;
 - (e) it is located in the United States or in any territory or possession of the United States that has adopted Article 9 of the Uniform Commercial Code;
 - (f) it is not supply items, packaging or any other similar materials;
 - (g) it is not identified to any purchase order or contract to the extent progress or advance payments are received with respect to such inventory;
 - (h) it does not breach any of the representations, warranties or covenants pertaining to inventory set forth in the Loan Documents; and

2

(i) Lender shall not have determined in good faith in accordance with Lender's reasonable and customary business practices that it is unacceptable due to age, type, category, quality, quantity and/or any other reason whatsoever.

Inventory which is at any time Eligible Inventory but which subsequently fails to meet any of the foregoing requirements shall forthwith cease to be Eligible Inventory.

"Eligible Receivables" means any right of payment for goods sold, leased or licensed or service rendered by a Borrower, whether or not earned by performance, and may, without limitation, in whole or in part, be in the form of an account, chattel paper or instrument as determined by Lender in good faith in accordance with Lender's reasonable and customary business practices, which are aged less than ninety (90) days.

"Event of Default" means any of the following in addition to any event of default defined in the Loan Documents:

- (a) A failure to pay when due any principal, interest, fee, escrow payment or expense reimbursement.
- (b) A failure to perform or pay when due any other material obligation, covenant, representation, warranty, or agreement under the terms of any Loan Document in strict accordance with the terms and provisions thereof.
- (c) Without Lender's prior written consent, a sale, conveyance, or encumbrance of any interest in the Collateral. Notwithstanding the foregoing, Lender's prior written consent shall not be required for any sale or conveyance which results in the Term Note and Revolving Note being paid in full.
 - (d) The occurrence of any default under any other promissory note, guaranty, security document, or loan document executed by Borrower and owned by Lender at the time of default.
- (e) The occurrence of any default under any document executed by Borrower which (i) is owned by any person other than Lender, (ii) constitutes a lien on the Collateral (no permission for creation of such liens being implied), (iii) remains uncured after the expiration of any applicable cure period and (iv) could reasonably result in a material impairment of Borrower's ability to pay or perform its obligations under the Loan Documents.
- (f) A violation, whether discovered or asserted before or after closing of any federal, state, or local law, rule, regulation, or order issued by a governmental agency or court which would or could have a material adverse impact on Borrower's business or properties (including without limitation the Collateral) including but not limited to (i) any provisions of the Comprehensive Environmental Response, Compensation and Liability Act of 1980, or any similar law which prohibits or restricts the storage, maintenance, or discharge of hazardous materials or waste, or (ii) any provisions of the Americans with Disabilities Act of 1990 or any similar federal, state, or local law imposing requirements relating to the accessibility of buildings or structures to persons with disabilities.

3

- (g) The material falseness of any statement, warranty, or representation when given or made by Borrower to Lender or any such statement, warranty, or representation becoming materially false at any time when any portion of the Loan remains outstanding.
- (h) Borrower or any surety (i) makes an assignment for the benefit of creditors, (ii) files a voluntary proceeding seeking protection from creditors under any bankruptcy or other law, (iii) becomes the subject of an involuntary proceeding under any bankruptcy or other similar law, or (iv) makes any admission of its inability to pay its debts generally as they become due.
 - (i) The appointment of a trustee, receiver, or liquidator for Borrower, any surety or the Collateral.
- (j) The occurrence of any of the following with respect to Borrower or the Collateral involving an amount in excess of \$100,000.00 (i) the imposition of any lien or other similar encumbrance (which does not include a voluntary grant or conveyance of a lien or encumbrance by Borrower against assets other than the Collateral), (ii) the issuance of any garnishment, attachment, levy, or any other form of execution or (iii) the entry of a material adverse judgment by a court having jurisdiction.
- (k) The occurrence of a change in Borrower's (a) ownership which results in (i) Liquidmetal Technologies, Inc. ("LTC") owning less than a majority of the outstanding membership interests of LMC or (ii) LMC owning less than 100% of LMCS, or (b) management, without Lender's prior written consent, which consent will not be unreasonably withheld or delayed.
- (l) Prior to payment in full of all obligations of Borrowers to C3 Capital Partners, LP and C3 Capital Partners II, L.P. (collectively "C3") under that certain Securities Purchase Agreement ("SPA") between and among Borrowers and C3 dated July 20, 2007, the failure of LMC to maintain (i) the allocation of two (2) board votes to be controlled by C3 and CEO Larry Buffington and (ii) a springing proxy control of a third board vote in favor of C3 to be triggered upon the occurrence of an event of default under the SPA.
 - (m) The loss or impairment of (i) Lender's lien or security interest or (ii) the priority of Lender's lien or security interest in the Collateral.
 - (n) A determination by Lender in good faith that the prospect of payment or performance under the Loan Documents is materially impaired.
 - "Excess Cash Flow" means, for any period, EBITDA less (i) federal and state income taxes, less (ii) capital expenditures, less (iii) Senior Debt Service, less (iv) Interest Charges arising from Borrowers' Debt to C3.

"Fixed Charge Coverage Ratio" means, with respect to any period, the ratio of (i) EBITDA minus federal and state income taxes minus capital expenditures to (ii) Total Debt Service.

4

"General Intangibles" means any and all existing or acquired educational licenses, certifications and other similar items of value as determined by Lender in good faith in accordance with Lender's reasonable and customary business practices.

"Interest Charges" means, for any period, the sum of: (a) all interest charges and related expenses payable with respect to that fiscal period to a lender in connection with borrowed money or the deferred purchase price of assets that are treated as interest in accordance with GAAP, <u>plus</u> (b) the portion of capitalized lease obligations with respect to that fiscal period that should be treated as interest in accordance with GAAP, <u>plus</u> (c) all charges paid or payable (without duplication) during that period with respect to any hedging agreements.

"Leverage Maximum" means (i) as of even date herewith through month six (6): 1.75 to 1.00; (ii) for months seven (7) through month twelve (12): 1.60 to 1.00; (iii) for months thirteen (13) to twenty-four (24): 1.50 to 1.00; (iv) for months twenty-five (25) through the Term Note Maturity Date: 1.25 to 1.00.

"Lien" means, with respect to any Person, any interest granted by such Person in any real or personal property, asset or other right owned or being purchased or acquired by such Person (including an interest in respect of any capital lease) which secures payment or performance of any obligation and shall include any mortgage, lien, encumbrance, title retention lien, charge or other security interest of any kind, whether arising by contract, as a matter of law, by judicial process or otherwise.

"Loans" shall mean all loans and advances made by Lender to or on behalf of Borrowers hereunder.

"Loan Documents" shall mean all agreements instruments and documents, including, without limitation, this Agreement, the Term Note, the Revolving Note, the LCM Security Agreement, the LMCS Security Agreement and all other writings heretofore, now or from time to time hereafter executed by or on behalf of Borrower or any other Person delivered to Lender in connection with the Loans or the transactions contemplated hereby, as each of the same may be amended, modified or supplemented from time to time.

"Net Income" means, with respect to Borrowers for any period, the net income (or loss) of Borrowers for such period as determined in accordance with GAAP.

"Net Worth" means, as of any given date, a Person's equity calculated in conformance with GAAP by subtracting total liabilities from total Tangible Assets.

"Permitted Liens" means:

- (a) Liens pursuant to any Loan Document;
- (b) Liens existing on the date hereof and disclosed to Lender and any refinancing, refunding, renewal or extension thereof; provided, that (i) the property covered thereby is not changed, (ii) the amount secured or benefited thereby is not increased, and (iii) the direct or any contingent obligor with respect thereto is not changed;

5

- (c) Liens for taxes not yet due or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person in accordance with GAAP;
- (d) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business which are not overdue for a period of more than thirty (30) days or which are being contested in good faith and by appropriate proceedings diligently conducted, if adequate reserves with respect thereto are maintained on the books of the applicable Person;
 - (e) pledges or deposits in the ordinary course of business in connection with workers' compensation, unemployment insurance and other social security legislation, other than any Lien imposed by ERISA;
- (f) deposits to secure the performance of bids, trade contracts and leases (other than indebtedness), statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business;
- (g) Liens in connection with capital leases and purchase money obligations for fixed or capital assets; provided, that (i) such Liens do not at any time encumber any property other than the property financed by such indebtedness and (ii) the indebtedness secured thereby does not exceed the cost or fair market value, whichever is lower, of the property being acquired on the date of acquisition;
- (h) Liens securing indebtedness (and any refinancing, refunding, renewal or extension thereof) on property in existence at the time such property is acquired by a borrower in connection with an acquisition; provided, that (i) such Liens do not at any time encumber any property other than the property so acquired, (ii) the amount secured or benefited thereby is not increased, and (iii) the direct or any contingent obligor with respect thereto is not changed;
 - (i) Liens not otherwise permitted by this definition; provided, that the aggregate amount of indebtedness secured by Liens permitted by this clause (j) shall not at any time exceed \$50,000; and

any and all other Liens specifically approved by Lender, including the Notes made by the Borrowers and held by C3, as such Notes are defined in the SPA.

"Person" means any natural person, partnership, limited liability company, corporation, trust, joint venture, joint stock company, association, unincorporated organization, government or agency or political subdivision thereof, or other entity, whether acting in an individual, fiduciary or other capacity.

"Security Agreements" means, collectively, the Security Agreement executed by LMC in favor of Lender dated as of even date herewith and the Security Agreement executed by LMCS in favor of Lender as of even date herewith.

6

"Senior Debt Service" means, with respect to any period, an amount equal to the sum of (i) Interest Charges arising from the Loans for such period, plus (ii) the scheduled amortization of any Loans during such period, plus (iii) Interest Charges arising from any capitalized lease obligations of Borrowers during such period.

"Senior Debt Service Coverage Ratio" means, with respect to any period, the ratio of (i) EBITDA minus federal and state income taxes minus capital expenditures to (ii) Senior Debt Service.

"Senior Funded Debt" means, at any time (i) the aggregate amount outstanding of all Loans, plus (ii) the aggregate amount of all capitalized lease obligations of Borrowers.

"Subsidiary" and "Subsidiaries" shall mean, respectively, with respect to any Person, each and all such corporations, partnerships, limited partnerships, limited liability companies, limited liability partnerships, joint ventures or other entities of which or in which such Person owns, directly or indirectly, such number of outstanding capital securities as have more than fifty percent (50%) of the ordinary voting power for the election of directors or other managers of such corporation, partnership, limited liability company or other entity. Unless the context otherwise requires, each reference to Subsidiaries herein shall be a reference to Subsidiaries of Borrower.

"Tangible Assets" means the total of all assets appearing on a balance sheet of Borrowers prepared in accordance with GAAP (with inventory being valued at the lower of cost or market), after deducting all proper reserves (including reserves for Depreciation) minus the sum of (i) goodwill, patents, trademarks, prepaid expenses, deposits, deferred charges and other personal property which is classified as intangible property in accordance with GAAP.

"Tangible Net Worth" means, as of any given date, Borrowers' consolidated Net Worth plus the outstanding amount of all Debt from Borrowers to C3.

"Total Debt Service" means, with respect to any period, an amount equal to the sum of (i) Interest Charges for such period plus (ii) the scheduled amortization of any Debt during such period.

Capitalized terms not defined herein shall have the same definition as set forth in the Security Agreements or the Uniform Commercial Code as adopted by the State of Missouri.

ARTICLE II ADVANCES

Section 1. Term Loan Advance. Subject to the terms and conditions of this Agreement and the Loan Documents, as of even date herewith, Lender shall make an advance to Borrower in an aggregate principal amount equal to Four Million and No/100 Dollars (\$4,000,000.00) (the "Term Loan"). Amounts repaid with respect to the Term Loan may not be reborrowed.

7

- Section 2. Revolving Loan Advance. Subject to the terms and conditions of this Agreement and the Loan Documents, as of even date herewith, Lender shall make revolving loans and advances to Borrower (the "Revolving Loan"), in an aggregate principal amount not to exceed One Million Five Hundred Thousand and 00/100 Dollars (the "Maximum Revolving Limit") as follows:
 - (a) Initial Advance. As of even date herewith, Lender shall make an initial advance of the loan amount in the sum of One Million and 00/100 Dollars (\$1,000,000.00) ("Initial Advance").
- (b) <u>Future Advance</u>. Upon satisfaction of the conditions referenced herein and provided no Event of Default has occurred and is continuing, Lender shall, at Borrower's request, from time to time, make advances to Borrower in the aggregate amount of up to the lesser of (i) eighty-five percent (85%) of the face amount (less maximum discounts, credits and allowances which may be taken by or granted to Account Debtors in connection therewith in the ordinary course of Borrower's business) of Borrower's Eligible Receivables, <u>plus</u> fifty percent (50%) of Eligible Inventory, each on a consolidated basis or (ii) the trailing twelve months EBITDA multiplied by the applicable Leverage Maximum [less amount of lender term loan outstanding] (as the case may be, the "Future Advance Amount"). Provided, however, the sum of all advances with respect to the Revolving Loans shall at no time exceed the Maximum Revolving Limit, except as such amount may be increased or, upon the occurrence and continuation of an Event of Default, decreased by Lender, in its sole discretion.

Section 3. Borrower Reporting. Each Borrower shall provide to Lender the following:

- (a) Within ninety (90) days after Borrower's fiscal year-end, Borrower shall provide to Lender, at Borrower's sole cost and expense, audited Financial Statements;
- (b) Within forty-five (45) days after Borrower's fiscal year-end, Borrower shall provide to Lender, at Borrower's sole cost and expense, a copy of Borrower's year-end tax return;
- (c) Within thirty (30) days after the end of Borrower's fiscal quarter, or at the request of Lender, and together with any request by Borrower for an Advance, Borrower shall deliver to Lender (i) a Borrowing Base Certificate in substantially the same form as the closing Borrowing Base Certificate attached hereto as Exhibit A, which shall be accompanied by copies of Borrower's sales journal, cash receipts journal and credit memo journal for the relevant period, and (ii) a certification of Borrower's financial covenants substantially in the same form as attached hereto as Exhibit B. Such Borrowing Base Certificate shall reflect activity of the Borrower with respect to Accounts for the immediately preceding period to the date of the last Advance made by Lender to Borrower, and shall be in a form and with such specificity as is satisfactory to Lender and shall contain additional information concerning Accounts, Inventory and Equipment as may be requested by Lender, including, without limitation, but only if specifically requested by Lender, copies of all invoices prepared in connection with such Accounts; and

8

- (d) Such other financial documents as Lender deems appropriate in the course of normal business with reasonable notice to Borrower.
- Section 4. As a condition to the Loan, Borrower shall at all times maintain all of its deposit and operating accounts with Lender and Lender will provide substantially competitive rates in its marketplace.

Notwithstanding the foregoing, Lender shall not be obligated to make advances ("Advances") of the Loan to Borrower, unless Lender is satisfied that the conditions precedent to the making of such Advance have been satisfied by Borrower.

ARTICLE III TERMS OF NOTES

Section 1. <u>Term Note</u>.

- (a) <u>Interest</u>. The "**Term Interest Rate**" prior to the occurrence of an Event of Default, will be a fixed rate of 8.48% per annum. All interest shall be calculated for the actual number of days elapsed over a year assumed to consist of 360 days.
- (b) Payments of Principal and Interest. Payments of principal and interest, at the Term Interest Rate, shall be due and payable on August 1, 2007, and on the first day of each calendar month thereafter ("**Term Note Period**"). Principal payments shall be (i) \$62,500.00 during months 1 through 12, (ii) \$83,333.33 during months 13 through 36 and (iii) \$104,166.67 during months 37 through 48.
 - (c) Payment of Principal and Interest on the Maturity Date. All remaining outstanding principal and interest shall be due and payable on July 20, 2011 ("Term Note Maturity Date").
 - (d) Fees. As of even date herewith, Borrower shall pay to Lender an initial loan fee in the amount of \$40,000.00.
- (e) Prepayment. If the Term Note is prepaid within the first twenty-four months ("Term Prepayment Penalty Period") following the Effective Date, Borrower must pay a prepayment penalty of one percent (1%) of the outstanding principal balance of the Term Loan, together with all accrued interest on the principal so prepaid. After the expiration of the Term Prepayment Penalty Period, Borrower shall have the right, at any time and from time to time, to prepay the Term Note in whole or in part without premium or penalty, provided that Borrower shall pay accrued interest on the principal so prepaid to the date of such prepayment.
 - (f) <u>Security</u>. The Term Note is secured by the Security Agreements which are the first lien security interest on all assets of Borrowers.

9

Section 2. Revolving Note

(a) <u>Interest.</u> The "**Revolving Interest Rate**", prior to the occurrence of an Event of Default, will be a fixed rate of 8.25% per annum until the first "**Rate Adjustment Date**" (as defined below). Thereafter, the Revolving Interest Rate, prior to the occurrence of an Event of Default, will be a variable rate per annum equal to the Index (as defined below). The Index is the "prime rate" of interest published from time to time by the Wall Street Journal in its listing of "**Money Rates**". The Revolving Interest Rate shall be adjusted daily (each a "**Rate Adjustment Date**"). If the Index is no longer available, Lender will choose a new index which is based upon

comparable information. All interest will be calculated for the actual number of days elapsed over a year assumed to be 360 days. Notwithstanding anything herein to the contrary, the Revolving Interest Rate shall not be less than 7% per annum.

- (b) Interest Only Payments. Payments of interest only, at the Revolving Interest Rate, shall be due and payable on August 1, 2007, and on the first day of each calendar month thereafter.
- (c) Payment of Principal and Interest on the Maturity Date. All remaining outstanding principal and interest shall be due and payable on July 20, 2008 ("Revolving Note Maturity Date").
- (d) Fees. As of even date therewith, Borrower shall pay to Lender an initial loan fee in the amount of \$15,000.00.
- (e) <u>Prepayment.</u> Borrower shall have the right, at any time and from time to time, to prepay the Revolving Note, in whole or in part without premium or penalty, provided that Borrower shall pay accrued interest on the principal so prepaid to the date of such prepayment.
 - (f) Security. The Revolving Note is secured by the Security Agreements which are the first lien security interest on all assets of Borrowers.
- Section 3. Default Interest Rate. At any time that an incurred Event of Default is outstanding, the Term Note will bear interest at a rate of interest which is 5.0 percentage points in excess of the Term Interest Rate in effect from time to time and the Revolving Note will bear interest at a rate of 5.0-percentage points in excess of the Revolving Interest Rate in effect from time to time (each a "Default Rate"). The Default Rate will be paid without prejudice to Lender's right to collect other amounts due hereunder or to declare a default under any Loan Documents. If the Event of Default is cured during the applicable cure period, if any, interest will begin to accrue at the Term Note Rate or Revolving Note Rate, as applicable, commencing on the date the Event of Default is cured.
- Section 4. Late Payment Charge. If Borrower fails to make any payment within 10 days after the due date, at Lender's option, a late charge of 5.0% of the payment amount due may be charged by Lender. A tender to Lender of any payment more than 10 days past due which does not include the late fee may be rejected as insufficient or Lender may accept the tendered amount as a partial payment without waiving Lender's right to receive the late payment charge.

10

The Term Note and the Revolving Note will remain in default and interest at the Default Rate will accrue until the late payment charge is paid. The late payment charge shall be paid without prejudice to Lender's rights to collect other amounts due hereunder or to declare a default under the Term Note, the Revolving Note or any other Loan Document.

- Section 5. Acceleration. Upon breach of any promise made in the Loan Documents, or in any other instrument given to secure the payment hereof after any applicable notice and cure period, at the option of Lender, the entire indebtedness hereby evidenced shall become due, payable and collectible, then or thereafter as Lender may elect, regardless of the date of maturity, subject to any applicable cure period in the Security Agreements.
 - Section 6. Fees. Borrower agrees to pay Lender's reasonable attorneys' fees if the Term Note or Revolving Note is placed in the hands of an attorney for collection.

ARTICLE IV FINANCIAL COVENANTS

- Section 1. Minimum Senior Debt Service Coverage Ratio. Borrowers shall have, on a consolidated basis at the end of each fiscal quarter of Borrowers, a Senior Debt Service Coverage Ratio of not less than 1.60:1.00.
- Section 2. Minimum Fixed Charge Coverage Ratio. Borrowers shall have, on a consolidated basis at the end of each fiscal quarter of Borrowers, a Fixed Charge Coverage Ratio of not less than 1.10:1.00.
- Section 3. Maximum Senior Funded Debt to EBITDA. Borrowers shall have, on a consolidated basis at the end of each fiscal quarter of Borrowers set forth below, a ratio of (i) Senior Funded Debt measured as of the last day of such fiscal quarter of Borrowers to (ii) EBITDA for such fiscal quarter of not more than the following:
 - 1.75:1.00 for the fiscal quarters ending September 30, 2007, and December 31, 2007;
 - 1.60:1.00 for the fiscal quarters ending March 31, 2008, and June 30, 2008;
 - 1.50:1.00 for the fiscal quarters ending September 30, 2008, December 31, 2008, March 31, 2009, and June 30, 2009; and
 - 1.25:1.00 for each fiscal quarter ending thereafter.
 - Section 4. Minimum Tangible Net Worth. Borrowers shall have, on a consolidated basis at the end of each fiscal quarter of Borrowers set forth below, a Tangible Net Worth of not less than the following:
 - \$7,000,000.00 for the fiscal quarters ending September 30, 2007, December 31, 2007, March 31, 2008, and June 30, 2008;

11

\$7,500,000.00 for the fiscal quarters ending September 30, 2008, December 31, 2008, March 31, 2009, and June 30, 2009;

\$8,500,000.00 for the fiscal quarters ending September 30, 2009, December 31, 2009, March 31, 2010, and June 30, 2010;

\$9,000,000.00 for the fiscal quarters ending September 30, 2010, December 31, 2010, March 31, 2011, and June 30, 2011;

9,500,000.00 for each fiscal quarter ending thereafter.

Section 5. <u>Excess Cash Flow Payments to Lender.</u>

- (a) Within ninety (90) days after the end of Borrowers' fiscal year ending December 31, 2007, Borrowers shall pay Lender the lesser of (i) Excess Cash Flow, provided Excess Cash Flow for purposes of this Section shall not be less than zero, or (ii) \$104,000.00.
- (b) Within ninety (90) days after the end of Borrowers' fiscal year ending December 31, 2008 and each fiscal year thereafter, Borrowers shall pay Lender the lesser of (i) Excess Cash Flow, provided Excess Cash Flow for purposes of this Section shall not be less than zero, or (ii) \$250,000.00.

ARTICLE V MISCELLANEOUS PROVISIONS

- Section 1. Waiver. Except as otherwise provided herein, Borrower, endorsers, guarantors, sureties and all other parties liable for the payment of any sum or sums due or to become due under the terms of the Loan Documents jointly and severally waive presentment, demand for payment, protest and notice of dishonor, and consent that the time of payment of the Term Note or Revolving Note, or any installment thereof, may be extended from time to time without notice to them.
 - Section 2. Notices. Any written notices required hereunder shall be sent in accordance with requirements of the Security Agreements
 - Section 3. Governing Law. This Agreement is to be construed and enforced according to and governed by the laws of the State of Missouri.
- Section 4. Ratification. Borrower hereby absolutely, unconditionally, irrevocably, and without limitation on amount, agrees that it shall be responsible for, and will perform all obligations contained in the Loan Documents. Borrower's obligations shall include, without limitation, the obligation to repay all amounts payable under the Term Note, the Revolving Note and all other Loan Documents and to observe and perform all other obligations contained in the Loan Documents, including without limitation the obligation to indemnify and hold Lender harmless from any breach of the agreements contained in Section 11 of the Security Agreements.

12

Section 5. Waiver of Jury Trial. EACH BORROWER HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, SUIT, PROCEEDING, OR COUNTERCLAIM THAT RELATES TO OR ARISES OUT OF ANY OF THE LOAN DOCUMENTS OR THE ACTS OR FAILURE TO ACT OF OR BY LENDER IN THE ENFORCEMENT OF ANY OF THE TERMS OR PROVISIONS OF THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS, OR BY ANY PARTY IN THE PERFORMANCE OF ANY OF THE TERMS OF THE PROVISIONS OF THIS AGREEMENT OR THE LOAN DOCUMENTS.

Section 6. Revival of Liability. To the extent any payment or payments made to the Lender, or any payment of proceeds from the sale of any property are received by Lender, in the reduction of the amounts due with respect to any of the obligations evidenced by this Agreement or under any of the obligations any Borrower owes the Lender are subsequently invalidated, declared to be fraudulent or preferential, set aside and/or required to be repaid to a trustee, to a surety, or any other person liable for any of the obligations evidenced and/or secured by this agreement or under any of the obligations are being a debtor-in-possession or to a receiver or any other person under any bankruptcy law, state or federal law, common law or equitable cause (collectively, the "Invalidated Payments"), then the portion of such indebtedness equal to the Invalidated Payments and the liens, if any, given to secure this agreement or any of the obligations each Borrowers owes the Lender will be revived and will continue in full force and effect as if such payment or proceeds had never been received by the Lender.

Section 7. Stay Waiver and/or Adequate Protection Payments. In the event either or both Borrowers for any reason becomes the subject of any Debtor Relief Proceeding Lender shall be entitled to the following: (i) the automatic and absolute lifting of any stay, injunction, restraining order or similar legal impediment (including, without limitation, the automatic stay imposed pursuant to the United States Bankruptcy Code ("Bankruptcy Code") § 362) to the enforcement of Lender's remedies under this Agreement and under all the obligations each Borrower owes to the Lender and, if applicable, (ii) adequate protection, pursuant to the Bankruptcy Code or other law, of Lender's interest in any property of either or both of the Borrowers. Moreover, each Borrower hereby consents to the immediate lifting of any such stay, injunction, restraining order or similar legal impediment in such Bankruptcy Proceeding against Lender's exercise of its remedies and hereby covenants, promises and agrees that each will not contest any effort by Lender to annul, lift, modify or extinguish such stay in such Bankruptcy Proceeding. Each Borrower also expressly acknowledges and agrees that: (a) "cause" shall exist to lift such stay pursuant to Bankruptcy Code § 362(d) or other applicable law preventing Lender from exercising its rights, and (b) neither Borrower has any equity in any property of any kind in which Lender has an interest and such property is not now and will never be necessary to any plan of reorganization of either or both Borrowers in such Bankruptcy Proceeding.

Mediation; Arbitration. Each Borrower agrees that any dispute under this or any other Loan Document shall be resolved only through mediation followed by arbitration, if necessary. If a dispute arises which cannot otherwise be resolved by the parties, Borrower and Lender shall mutually agree to a mediator within ten (10) days of either party's written notification to the other party that a dispute exists which requires mediation. If the parties cannot reach an agreement with respect to the selection of a mediator, then each party

13

shall nominate a mediator. The nominees shall jointly select a mediator who will undertake the mediation. The mediation must be completed within thirty (30) days of the mediator's selection. If any dispute is not resolved through such mediation, then Borrower agrees that any such dispute can only be resolved through binding arbitration. The selection of the arbitrator and the rules of arbitration shall be mutually agreed to by the parties. If, however, no agreement can be reached, then both the arbitrator selection process and the arbitration itself shall be governed by the rules of the American Arbitration Association in accordance with its Commercial Arbitration Rules, and shall be arbitrated by a single arbitrator. Such arbitrator's fees and any expenses shall be shared equally by Borrower and Lender.

 $\underline{\textbf{No Oral Agreements}}. \ \ \textbf{The following notice is given to comply with § 432.045 of the Revised Statutes of Missouri.}$ Section 9.

Oral agreements or commitments to loan money, extend credit or forbear from enforcing repayment of a debt, including promises to extend or renew such debt, are not enforceable. To protect Borrower and Lender from misunderstanding or disappointment, any agreements reached regarding such matters are contained in this Agreement, which is the complete and exclusive statement of the agreement between Borrower and Lender, except as they may later agree in writing to modify it.

[Remainder of Page Intentionally Left Blank]

14

IN WITNESS WHEREOF, Borrower and Lender have executed this Agreement as of the date first written above.

BORROWER:

LIQUIDMETAL COATINGS, LLC,

a Delaware limited liability company

/s/ Larry Buffington Name:

Larry Buffington President and Chief Executive Officer Its:

LIQUIDMETAL COATINGS SOLUTIONS, LLC.

a Delaware limited liability company

By: /s/ Larry Buffington Name

Larry Buffington President and Chief Executive Officer Its:

[Signature of Lender Appears on the Following Page]

15

LENDER:

BANK MIDWEST, N.A.

By: Name Judson Stanion Vice President Its:

16

Exhibit A

Borrowing Base Certificate



			_	1.00	
	Date:		Prepar	ed By:	
1	Total Accounts Receivable	\$ 1,510,000			
2	Less: Aged 90 Days or More	\$ _			
3	Less: Credits	\$ _			
4	Eligible Accounts Receivable	\$ 1,510,000			
5	Advance Rate	85.0%			
6	Loanable Value - Accounts Receivable		\$	1,283,500.00	
7	Total Inventory	\$ 682,000.00			
8	Less: Inventory Exclusions (if any)	\$ _			
9	Eligible Inventory	\$ 682,000.00			
10	Advance Rate	50.0%			
11	Loanable Value - Inventory		\$	341,000	
12	Total Loanable Value		\$	1,624,500	
13	Authorized Maximum of Revolver		\$	1,500,000	
14	Trailing 12 Months EBITDA	\$ 3,403,000			
15	Applicable Leverage Maximum	1.75			
16	Leverage Maximum	\$ 5,955,250			
.7	Beginning Loan Balance (RLOC)	\$ _			
18	Beginning Loan Balance (Term Loan)	\$ 4,000,000			
19	RLOC Loan Available	\$ 1,500,000			
19	Advance Amount	\$ 1,000,000			
20	Ending Loan Balance (RLOC plus Term)	\$ 5,000,000			
21	Remaining Availability		\$	500,000	

Cash Receipts Journ
Credit Memo Journa
Financial Covenants

Exhibit B

Disclosure Schedules are omitted pursuant to Item 601(b)(2) of Regulation S-K. The following sets forth a list of the omitted Schedules. The registrant agrees to furnish supplementally to the Commission a copy of any omitted Schedule, upon the Commission's request.

Liquidmetal / Quarterly Financial Covenants (based on annualized projections)

Exhibit 2.3

Page

SECURITIES PURCHASE AGREEMENT

DATED JULY 24, 2007

AMONG

C₃ CAPITAL PARTNERS, L.P.,

C₃ CAPITAL PARTNERS II, L.P.,

LIQUIDMETAL COATINGS, LLC

AND

LIQUIDMETAL COATINGS SOLUTIONS, LLC

Table of Contents

Section 1.	Definitions and Related Matters	1
1.1	<u>Definitions</u>	1
1.2	Accounting Principles	1
1.3	Other Interpretive Matters	1
		_
Section 2.	Authorization and Closing	2
2.1	Authorization of the Securities	2
2.2	Purchase and Sale of the Securities	2
2.3	The Closing	2
2.4	<u>Use of Proceeds</u>	2
Section 3.	Conditions of Purchasers' Obligation at the Closing	3
3.1	Representations, Warranties and Covenants; No Event of Default	3
3.2	Acptics industry, waitamine and Covenants, two Eventor Detains	3
3.3	Christia Agreement	3
3.4	Specialis Astronomia Secured Notes	3
3.5	Sale of the Membership Units to the Purchaser	1
3.6	Employment Agreement	4
3.7	Security Agreements	4
3.8	ectus, rigerments	4
3.9	Asset Purchase and Contribution Agreement	4
3.10	Financing	4
3.11	Proceedings	5
3.12	No Material Adverse Effect	5
3.13	Deliveries of the Company at Closing	5
Section 4.	Conditions of Company's Obligation at the Closing	6
4.1	Representations, Warranties and Covenants; No Event of Default	6
4.2	Asset Purchase and Contribution Agreement	7
4.3	<u>Financing</u>	7
4.4	Deliveries of C ₃ and C ₃ II at Closing	7
Section 5.	Covenants	7
5.1	Covenants Affirmative Covenants	7
5.1	Attendance at Board Meetings	9
5.3	Attenuance at Board Meetings Additional Affirmative Covenants	10
5.4	Negative Covenants Negative Covenants	12
5.5	Financial Covenants	15
5.6	SBIC Regulatory Provisions	15
5.7	Public Disclosures	16
5.8	Further Assurances	16
5.9	Post Closing Obligations	16
5.10	Annual Budget Approval	17
5.11	Put Provisions	17
	i	
	·	

Representations and Warranties of the Company and LMCS 19 19 20 20 21 21 21 21 22 22 22 23 23 23 23 23 23 6.1 6.2 Organization, Corporate Power and Licenses Capitalization and Related Matters 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10 6.11 Subsidiaries; Investments Authorization; No Breach Financial Statements Projections and Pro Forma Financial Statements Absence of Undisclosed Liabilities
No Material Adverse Change Assets Tax Matters Contracts and Commitments 6.12 Intellectual Property Rights 6.13 Litigation, Etc. Brokerage Governmental Consent, Etc Insurance 6.15 6.16 6.17 Employees ERISA 6.18

6.19	Compliance with Laws		24
6.20	Small Business Matters		24
6.21	Affiliated Transactions		25
6.22	Investment Company		25
6.23	Margin Regulations		25
6.24	Public Utility Holding Company Act		25
6.25	Customers and Suppliers		25
6.26	Disclosure		26
6.27	Closing Date		26
Section 7.	Representations and Warranties of the Purchasers		26
7.1	Organization, Corporate Power and Licenses		26
7.2	Due Authorizations		26
7.3	Purchaser's Investment Representations		27
7.4	Purchaser's Reliance		28
7.5	Permitted SBIC Investment		28
Section 8.	Events of Default		28
8.1	<u>Definition</u>		28
8.2	Consequences of Events of Default		31
Section 9.	Miscellaneous		32
9.1	<u>Expenses</u>		32
9.2	Remedies		33
9.3	Amendments and Waivers		33
9.4	Survival of Agreement		33
9.5	No Setoffs, Etc		34
9.6	Successors and Assigns		34
		ii	
9.7	Aggregation		34
9.8	<u>Severability</u>		34
9.9	<u>Counterparts</u>		34
9.10	Governing Law		34
9.11	<u>Notices</u>		35
9.12	Consideration for Purchased Units; Treatment of Fees		36
9.13	Construction		36
9.14	Complete Agreement		37
9.15	<u>Indemnification</u>		37
9.16	Payment Set Aside		38
0.17	Luisdistion and Vonus		20

List of Exhibits and Appendices List of Disclosure Schedules Appendix 1 – Definitions

9.18

iii

SECURITIES PURCHASE AGREEMENT

THIS SECURITIES PURCHASE AGREEMENT (this "Agreement") is made as of July 24, 2007, between Liquidmetal Coatings, LLC, a Delaware limited liability company (the "Company"), Liquidmetal Coatings Solutions, LLC, a Delaware limited liability company ("LMCS"), C₃ CAPITAL PARTNERS, L.P., a Delaware limited partnership ("G₃") and C₃ CAPITAL PARTNERS II, L.P., a Delaware limited partnership ("G₃ II"), and each of the other holders of Securities (as defined below) or any equity for which Securities are exchanged or converted, if any, who become a party hereto in accordance with the terms hereof (C₃, C₃ II and such other holders are collectively referred to herein as the "Purchasers").

The parties hereto agree as follows:

Waiver of Right to Jury Trial

Section 1. <u>Definitions and Related Matters.</u>

1.1 <u>Definitions</u>

Capitalized terms not otherwise defined in this Agreement shall have the meanings set forth on Appendix 1 attached hereto (such meanings are applicable to both the singular and plural forms of the terms defined).

1.2 Accounting Principles

The classification, character and amount of all assets, liabilities, capital accounts and reserves and of all items of income and expense to be determined, and any consolidation or other accounting computation to be made, and the interpretation of any definition containing any financial term, pursuant to this Agreement shall be determined and made in accordance with GAAP consistently applied.

1.3 Other Interpretive Matters

In each of the Investment Documents, unless a clear contrary intention appears: (a) the singular number includes the plural number and vice versa; (b) reference to any Person includes such Person's successors and assigns are permitted by such Investment Document, and reference to a Person in a particular capacity excludes such Person in any other capacity or individually; (c) reference to any agreement (including this Agreement and the Schedules and Exhibits and the Appendices hereto), document or instrument means such agreement, document or instrument as amended or modified; (d) reference to any applicable law, statute, rule or regulation means such applicable law, statute, rule or regulation as amended, modified, codified or reenacted, in whole or in part, and in effect from time to time, including rules and regulations promulgated thereunder; and (e) references to any Subsidiary of a Person shall be given effect only at such times as such Person has one or more Subsidiaries.

Section 2. <u>Authorization and Closing.</u>

- 2.1 <u>Authorization of the Securities.</u> The Company and LMCS have, or on or before the Closing will have, duly authorized the sale and issuance to:
- (a) C₃ of (i) a fourteen percent (14%) Subordinated Note in the principal amount of \$3,770,000 and containing the terms and conditions and in the form set forth on Exhibit A-1 attached hereto ("C₃ Note"), and (ii) 1,102 Membership Units, which shall represent 11.02% of the Membership Units of the Company (the "C₃ Membership Units") and shall have the terms and conditions set forth in the Company's Operating Agreement, attached hereto as Exhibit B-1; and
- (b) C_3 II of (i) a fourteen percent (14%) Subordinated Note in the principal amount of \$2,730,000 and containing the terms and conditions and in the form set forth on Exhibit A-2 attached hereto ("C₃ II Note", and together with the C₃ Note, and any other notes issued by the Company in respect of the purchase of the Securities are collectively referred to as the "Notes"), and (ii) 798 Membership Units, which shall represent 7.98% of the of Membership Units of the Company (the "C₃ II Membership Units") and together with the C₃ Membership Units, the "Purchased Membership Units") and shall have the terms and conditions set forth in the Company's Operating Agreement, attached hereto as Exhibit B-1. The Notes and the Purchased Membership Units are sometimes collectively referred to herein as the "Securities".

2.2 Purchase and Sale of the Securities.

At the Closing, the Company shall sell to C_3 and C_3 II and, subject to the terms and conditions set forth herein, C_3 and C_3 II shall purchase from the Company and LMCS, the Notes at an aggregate price equal to \$5,527,210 and the Membership Units at an aggregate price equal to \$972,790 (collectively, the "Purchase Price").

2.3 <u>The Closing</u>

The closing of the purchase and sale of the Note and the Purchased Membership Units (the "Closing") shall take place at the offices of Bryan Cave LLP, 3500 One Kansas City Place, 1200 Main Street, Kansas City, Missouri 64105 at 10:00 a.m. on July 24, 2007, subject however, to the satisfaction of the conditions set forth in Section 3 and Section 4. At the Closing, the Company shall deliver the Securities to C₃ and C₃ II and C₃ and C₃ II shall deposit the Purchase Price into an escrow account pursuant to the Escrow Agreement dated as of an even date herewith by and among C₃, C₃ II, Liquidmetal Technologies, Inc. ("LMT"), Bank Midwest, N.A. ("BMW"), Company and BMW as escrow agent (the "Escrow Agent").

2.4 Use of Proceeds

2

reasonable expenses related to the consummation of the transactions contemplated by the Investment Documents.

Section 3. <u>Conditions of Purchasers' Obligation at the Closing.</u> The obligation of Purchasers to purchase the Securities at the Closing is subject to the fulfillment as of the Closing of the following conditions, any one or more of which may be waived by Purchasers:

3.1 Representations, Warranties and Covenants; No Event of Default.

- (a) All of the representations and warranties contained in Section 6 hereof (considered collectively) and each of these representations and warranties (considered individually) shall be true, complete and correct in all material respects at and as of the Closing (without giving effect to any "material" or "materiality" qualification and both immediately prior to and immediately after giving effect to the transactions contemplated by the Investment Documents) as though then made, except for representations and warranties which refer to facts existing on a specific date, which shall be true and correct as of such date, and except where the failure of any such representation or warranty to be so true and correct, whether individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect and the Company shall have performed in all material respects all of the covenants required to be performed by it under the Investment Documents that are to be complied with or performed by the Company on or prior to the Closing (unless the same shall have been waived, in writing, by Company).
- (b) Notwithstanding Section 3.1(a), each of the representations and warranties in Sections 7.1, and 7.2 shall be true, complete and correct in all respects at and as of the Closing (both immediately prior to and immediately after giving effect to the transactions contemplated by the Investment Documents) as though then made.

3.2 Certificate of Formation

The Certificate of Formation of each of the Company and LMCS in the form attached as Exhibits C-1 and C-2 shall be in full force and effect under the laws of the Delaware.

3.3 Operating Agreement.

The Operating Agreement of each of the Company and LMCS shall be in the form of Exhibit B-1 and Exhibit B-2, respectively and such Operating Agreements shall be in full force and effect as of the Closing.

3.4 14% Subordinated Secured Notes

The Company shall have issued and delivered the Notes to C₃ and C₃ II and the Notes shall be in full force and effect as of the Closing

3

3.5 Sale of the Membership Units to the Purchaser.

The Company shall have issued and delivered to C3 and C3 II the Purchased Membership Units to be purchased hereunder at the Closing.

3.6 Employment Agreement

The confidentiality, nonsolicitation and noncompetition agreement between the Company and Larry Buffington attached hereto as Exhibit D and (the "Employment Agreement"), shall be in full force and effect as of the Closing and shall not have been modified or amended.

Security Agreements.

The Company and LMCS shall each have entered into definitive security agreements in the forms attached hereto as Exhibit E-1 and E-2, respectively (the "Security Agreements") for the purpose of securing the obligations arising under the Notes, and such Security Agreements shall be in full force and effect as of the Closing.

3.8 <u>Pledge Agreements</u>.

The Company and LMT shall each have entered into a Pledge Agreement for the purpose of securing the obligations arising under the Notes in the form attached hereto as Exhibit F (the "Pledge Agreements") and such Pledge Agreements shall be in full force and effect as of the Closing.

3.9 <u>Asset Purchase and Contribution Agreement.</u>

The Asset Purchase and Contribution Agreement shall be in full force and effect as of the Closing. The conditions in Articles 8 and 9 of the Asset Purchase and Contribution Agreement shall have been satisfied in full (without reliance on any waiver by the parties thereto). The transactions contemplated by the Asset Purchase and Contribution Agreement shall have been consummated simultaneously with the Closing hereunder in accordance with the terms of the Asset Purchase and Contribution Agreement. The Company shall have also executed a Collateral Assignment in form attached hereto as Exhibit G ("Collateral Assignment"), and such Collateral Assignment shall be in full force and effect as of the Closing.

3.10 <u>Financing</u>.

BMW shall have advanced or otherwise made available to the Company an aggregate of no less than \$5,000,000 and no more than \$5,500,000 pursuant to a secured credit facility on customary terms and conditions reasonably satisfactory to the Purchasers and Company (the "Senior Credit Facility"), and the Purchasers will agree to subordinate its Lien in the Company's assets to the Lien of such Senior Credit Facility pursuant to an intercreditor agreement with the provider of the Senior Credit Facility ("Senior Lender") on customary and reasonable terms acceptable to the Purchaser.

4

3.11 <u>Proceedings</u>.

All proceedings required to be taken by the Company in connection with the transactions contemplated hereby shall have been taken and all documents incident thereto shall be reasonably satisfactory in form and substance to C_3 and C_3 Π and their counsel.

3.12 No Material Adverse Effect

Since December 31, 2006, there shall have been no Material Adverse Effect.

3.13 Deliveries of the Company at Closing.

At the Closing, the Company shall deliver to the Purchasers, except as otherwise indicated below, all of the following:

- (a) the Notes in the principal amount of \$6,500,000, duly completed and executed by the Company;
- (b) the Purchased Membership Units, duly issued by the Company to C_3 and C_3 II, respectively, and which, in the aggregate, represent 19.00% of the Company's fully diluted equity;
- (c) an Officer's Certificate of the Company, dated the date of the Closing, stating that the conditions specified in Sections 3.1 through 3.13, inclusive, have been fully satisfied;
- (d) certified copies of the resolutions duly adopted by the Board of Directors of LMT, the board of managers of the Company and the board of managers of the LMCS authorizing the execution, delivery and performance of each of the Investment Documents to which it is a party, the issuance and sale of the Securities and the consummation of all other transactions contemplated by the Investment Documents;
 - (e) certified copies of the Company's and LMCS's Governing Documents, each as in effect at the Closing;
- (f) certificates of the Scoretary of the Company and LMCS certifying the names and signatures of the officers of the Company and LMCS, respectively, authorized to sign each of the Investment Documents to which the Company and LMCS is a party;
- (g) certificates of good standing, dated not more than ten (10) days prior to the date of the Closing, of the Company and LMCS issued by its jurisdiction of organization and from each jurisdiction in which the Company or LMCS is qualified to conduct business;
 - (h) copies of third party and governmental consents, approvals and filings as set forth on Schedule 3.13(h);

- (i) duly completed and executed SBA Forms 480, 652 and 1031 from the Company in the forms attached as Exhibit H-1, H-2 and H-3, respectively;
- (j) the Company's financial projections (including balance sheets and income and cash flow statements) through December 31, 2009;
- (k) an operating budget for the balance of 2007 reflecting anticipated Capital Expenditures for the Company and operating expenses approved in advance by C₃ and C₃ II, which approval shall not be unreasonably withheld or delayed (the "2007 Operating Budget");
 - (l) the Security Agreements;
 - (m) the Employment Agreement;
 - (n) the Pledge Agreements;
 - (o) payment of closing fees to C₃ and C₃ II in the aggregate amount of \$130,000 and the reimbursement for the fees and expenses described in Section 9.1 hereof;
 - (p) a solvency certificate in the form of Exhibit I executed by the Company's chief executive officer or secretary;
 - (q) the Asset Purchase and Contribution Agreement;
 - (r) a Use of Proceeds Schedule from the Company regarding the intended use of the proceeds from sale of the Securities;
 - (s) an irrevocable springing proxy in the form of Exhibit J executed by LMT and accepted and acknowledged, in writing, by C₃ and C₃ II (the "Irrevocable Springing Proxy"); and
 - (t) such other documents related to the transactions contemplated by the Investment Documents as C3 and C3 II or their counsel may reasonably request.

Section 4. <u>Conditions of Company's Obligation at the Closing.</u> The obligation of the Company to issue and sell the Securities at the Closing is subject to the fulfillment as of the Closing of the following conditions, any one or more of which may be waived by Company:

4.1 Representations, Warranties and Covenants; No Event of Default

All of the representations and warranties contained in Section 7 hereof (considered collectively) and each of these representations and warranties (considered individually) shall be true, complete and correct in all material respects at and as of the Closing (without giving effect to any "material" or "materiality" qualification) (both immediately prior to and immediately after giving effect to the transactions contemplated by the Investment Documents) as though then made, except for representations and warranties which refer to facts

6

existing on a specific date, which shall be true and correct as of such date, and the Purchaser shall have performed in all material respects all of the covenants required to be performed by it under the Investment Documents that are to be complied with or performed by the Purchaser on or prior to the Closing (unless the same shall have been waived, in writing, by Company, and there shall not exist any Event of Default or Potential Event of Default.

4.2 <u>Asset Purchase and Contribution Agreement</u>

The Asset Purchase and Contribution Agreement shall be in full force and effect as of the Closing. The conditions in Articles 8 and 9 of the Asset Purchase and Contribution Agreement shall have been satisfied in full (without reliance on any waiver by the parties thereto). The transactions contemplated by the Asset Purchase and Contribution Agreement shall have been consummated simultaneously with the Closing hereunder in accordance with the terms of the Asset Purchase and Contribution Agreement.

4.3 Financing.

BMW shall have advanced or otherwise made available to the Company an aggregate of no less than \$5,000,000 and no more than \$5,500,000 pursuant to a secured credit facility on customary terms and conditions reasonably satisfactory to each Purchaser and Company, and each Purchaser shall have agreed to subordinate its Lien in the Company's assets to the Lien of such Senior Credit Facility pursuant to an intercreditor agreement with the Senior Lender on customary and reasonable terms acceptable to the Purchasers.

4.4 Deliveries of C₃ and C₃ II at Closing.

C₃ and C₃ II shall have delivered to BMW an amount equal to \$6,500,000, net of the amounts specified in Section 3.13(0).

Section 5. <u>Covenants</u>

5.1 Affirmative Covenants.

So long as the principal amount under the Notes remain outstanding, prior to the indefeasible payment in full of all amounts due and owing under the Notes, the Company shall deliver to the Purchaser:

- (a) Notice of Litigation. Promptly after receipt of service of process and in any event within ten (10) Business Days after receipt of service of process, notice of all actions, suits, and proceedings before any governmental authority, court or arbitrator affecting the Company which, if determined adversely to the Company, could have a Material Adverse Effect;
- (b) Notice of Default. As soon as possible (but in any event within ten (10) Business Days) after (i) the discovery or receipt of notice of any Event of Default or Potential Event of Default, (ii) any default under any Investment Document or any other material agreement to which the Company is a party, (iii) any material investigation, notice, proceeding or adverse determination from any governmental or regulatory

7

authority or agency, or (iv) immediately (notwithstanding the reference to ten (10) Business Days stated above) after the receipt of notice (oral or written) of the acceleration of any material Indebtedness, an Officer's Certificate specifying the nature and period of existence thereof and what actions the Company has taken and proposes to take with respect thereto;

- (c) ERISA Reports. Within ten (10) Business Days after the filing or receipt thereof, copies of all reports, including annual reports, and notices which the Company or LMCS files with or receives from the PBGC or the U.S. Department of Labor under ERISA; and as soon as possible and in any event within ten (10) Business Days after the Company knows that any ERISA Event or Prohibited Transaction has occurred with respect to any Pension Plan or that the PBGC has instituted or will institute proceedings under Title IV of ERISA to terminate any Pension Plan, a certificate of the chief financial officer of the Company setting forth the details as to such ERISA Event or Prohibited Transaction or Pension Plan termination and the action that the Company or LMCS proposes to take with respect thereto;
- (d) Reports to Other Creditors. Promptly after the furnishing thereof, copies of any statement or report furnished to any other party pursuant to the terms of any indenture, loan or credit or similar agreement and not otherwise required to be furnished to the Purchaser pursuant to this Section;
- (e) Notice of Material Adverse Change. As soon as possible and in any event within ten (10) Business Days after the occurrence thereof, written notice of any matter that could reasonably be expected to have a Material Adverse Effect;
- (f) <u>Management Letters</u>. Promptly upon receipt thereof, a copy of any management letter or written report submitted to the Company by independent certified public accountants with respect to the business, condition (financial or otherwise), operations, prospects, or properties of the Company and LMCS;
 - (g) General Information. Promptly, such other information concerning the Company as the Purchaser may from time to time reasonably request;
- (h) Annual Budget. At least thirty (30) days, but not more than ninety (90) days prior to the beginning of each fiscal year, an annual budget of the operating expenses and Capital Expenditures for the Purchaser's approval, which shall not be unreasonably withheld or delayed, provided there is sufficient debt coverage and the budget demonstrates the Company's and LMCS's ability to comply with the financial covenants herein and in all other debt instruments to which the Company is bound. Each such annual budget shall be prepared (displaying anticipated statements of income, cash flows, balance sheets and detailed projected Capital Expenditures on a monthly basis) (the "Annual Budget") and within thirty (30) days after any quarterly period in which expenditures, costs or obligations incurred result in a deviation in excess of twenty percent (20%) of such corresponding quarterly period described within an Approved Annual Budget (as defined in Section 5.1(h)), the Company shall deliver to the Purchaser

8

(j) Insurance Reports. As soon as possible after each insurance policy to be maintained by the Company pursuant to Section 5.3(f) has been obtained, a statement from each such insurance carrier, which shall provide that each such policy shall name each of C_3 and C_3 II as a loss payee and shall provide that such policy will not be cancelled or reduced without thirty (30) days prior written notice to C_3 and C_3 II. As soon as possible (but in any event within ten (10) days) after becoming Aware of any material change in any insurance maintained by the Company, written notice thereof which describes the same and the intended course of action of the Company with respect thereto shall be delivered to the Purchasers; and

(k) <u>Compliance Certificate</u>. Concurrently with the delivery of each of the financial statements referred to in <u>Section 5.3(a)</u> and (b), a duly completed Officer's Certificate in the form of <u>Exhibit K</u> attached hereto, with appropriate insertions, dated the date of such financial statements and signed by the Company's and LMCS's Chief Financial Officer containing a computation of each of the financial ratios and restrictions set forth in Section 5.5 and to the effect that such officer has not become Aware of any Event of Default or Potential Event of Default that has occurred and is continuing, if there is any such event, describing it and the steps, if any, being taken or proposed to be taken to cure it.

5.2 Attendance at Board Meetings.

So long as any Securities remain outstanding or notes issued in exchange for any Securities remain outstanding and prior to the indefeasible payment in full of all amounts due and owing under the Notes, the Purchasers shall have the right to attend and observe all the Company's board of managers meetings (the "Purchasers' Observation Rights"). The Company shall give the Purchasers written notice of each meeting of its board of managers and each committee thereof at least three (3) Business Days prior to the date of each such meeting, and the Company shall permit one (1) representative of C₃ and C₃ II (the "Purchaser Representative") to attend all meetings of its board of managers and all committees thereof; provided, that in the case of telephonic meetings conducted in accordance with the Company's Governing Documents and applicable law, the Purchaser Representative shall receive actual notice thereof at least 48 hours prior to any such meeting, and the Purchaser Representative shall be given the opportunity to listen to such telephonic meetings unless the Purchaser Representative shall have waived in writing such advance notice. The Purchaser Representative shall be entitled to receive all written materials and other information (including, without limitation, copies of meeting minutes) given to managers in connection with such meetings at the same time such materials and information are given to the managers. If the Company proposes

9

to take any action by written consent in lieu of a meeting of its board of managers or of any committee thereof, the Company shall give written notice thereof to the Purchaser Representative prior to the effective date of such consent describing in reasonable detail the nature and substance of such action. The Company shall pay the reasonable out-of-pocket expenses of the Purchaser Representative incurred in connection with attending all such meetings.

- 5.3 Additional Affirmative Covenants. So long as any of the Securities remain outstanding and prior to the indefeasible payment in full of all amounts due and owing under the Notes or any Put Notes issued pursuant to Section 5.11(c):
 - Annual Financial Statements and Tax Return Information. The Company, as soon as available, and, in any event, within one hundred and twenty (120) days after the end of each fiscal year of the (a) Company, beginning with the fiscal year ending December 31, 2007, a copy of the annual consolidated financial statements of the Company and LMCS for such fiscal year containing balance sheets, statements of income, retained earnings and cash flows at the end of such fiscal year, in each, except for the fiscal year ending December 31, 2007, setting forth in comparative form the figures for the preceding fiscal year (if any), all in reasonable detail and audited and certified by independent certified public accountants of recognized standing reasonably acceptable to C₃ and C₃ II, to the effect that such report has been prepared in accordance with GAAP. Within ninety (90) days after the end of each Tax year, Company shall provide sufficient information with respect to the Company for such Tax year necessary for C3 and C3 II to prepare their federal, state and local income Tax Returns. The Company shall provide copies of the federal, state and local income Tax Returns of the Company for each Tax year promptly after the filing thereof.
 - (b) Monthly Financial Statements. The Company, as soon as available and, in any event, within thirty (30) days after the end of each month, a copy of an unaudited consolidated financial report of the Company and LMCS as of the end of such month and for the portion of the fiscal year then ended, including balance sheets, statements of income, retained earnings and cash flows, setting forth in each case comparisons to the Company's and LMCS's Annual Budget and to the corresponding period in the preceding fiscal year and a brief narrative explaining the results of the Company's and LMCS's operations for such period; all such statements shall have been prepared in accordance with GAAP (absent footnotes and customary year-end adjustments) and fairly present the financial condition and results of operations of the Company and LMCS at the date and for the periods indicated therein.
 - Maintenance of Existence; Conduct of Business. The Company shall, as commercially reasonable, preserve and maintain its and LMCS's existence and all of their respective leases, privileges, certifications, licenses, permits, franchises, qualifications, and rights that are necessary or desirable in the ordinary conduct of their respective businesses. The Company and LMCS will conduct each of their businesses in an orderly and efficient manner in accordance with good business practices. Neither the Company nor LMCS will, without the written consent of the Purchasers, make any material change in their credit collection policies if such change would materially impair

10

the collectibility of any material account owing to the Company, nor will either the Company or LMCS rescind, cancel or modify any material account owing to the Company or LMCS except in the ordinary course of business.

- Maintenance of Properties and Intellectual Property Rights. The Company and LMCS shall (i) maintain, keep and preserve all of their properties (real, personal, tangible and intangible) necessary or useful in the proper conduct of their business in good working order and condition and (ii) possess and maintain all material Intellectual Property Rights necessary to the conduct of the Company's and LMCS's businesses and own all right, title and interest in and to, or have a valid license for, all such Intellectual Property Rights.
- Taxes and Claims. The Company and LMCS shall use their commercially reasonable best efforts to pay or discharge at or before maturity or before becoming delinquent (i) all taxes, levies, assessments and governmental charges imposed on the Company and LMCS or their income or profits or any of their property and (ii) all lawful claims for labor, material, and supplies, which, if unpaid, might become a Lien upon any of their property; provided, however, that the Company and LMCS shall not be required to pay or discharge any tax, levy, assessment, or governmental charge which is being contested in good faith by appropriate proceedings diligently pursued, and for which adequate reserves have been established and such contest operates to suspend collections of the same.
- Insurance. The Company and LMCS shall use their commercially reasonable best efforts to maintain insurance with financially sound and reputable insurance companies in such amounts and covering such risks as is usually carried by companies engaged in similar businesses and owning similar properties in the same general areas in which the Company and LMCS operate, including, but not limited to, director and officer liability insurance.
- Inspection Rights. At any reasonable time and from time to time, the Company shall permit representatives of C3 and C3 II to examine, copy and make extracts from its books and records, to visit and inspect the Company's and LMCS's properties, and to discuss Company's and LMCS's business, operations, and financial condition with Company's LMCS's officers, employees and independent certified public accountants, provided that Purchasers agree to keep such information confidential and to use such information solely in connection with its investment in the Securities.
- Keeping Books and Records. The Company and LMCS shall maintain proper books of record and account in which full, true, and correct entries in conformity with GAAP shall be made of all dealings (h) and transactions in relation to their business and activities
- Compliance with Laws. The Company and LMCS shall comply in all material respects with all applicable laws, rules, regulations, orders and decrees of any governmental authority or arbitrator the failure to comply with which could reasonably be expected to have a Material Adverse Effect.

11

- Compliance with Agreements. The Company and LMCS shall use their commercially reasonable best efforts to comply in all material respects with all agreements, contracts, and instruments binding on it or affecting its properties or business, the failure to comply with which could reasonably be expected to have a Material Adverse Effect.
- (k) ERISA. The Company and LMCS shall use their commercially reasonable best efforts to comply with all minimum funding requirements and all other material requirements of ERISA, if applicable, so as not to give rise to any liability thereunder.
- Employment Agreement. The Company shall neither (i) modify, amend or terminate the Employment Agreement without the prior written consent of C3 and C3 II, which consent shall not be unreasonably withheld or delayed nor (ii) provide any employee, who is a party to the Employment Agreement, with any compensation, benefit or reimbursement for any expenses related to the performance of such employee's duties as contemplated by such employee's Employment Agreement, except as provided by the Employment Agreement.

Negative Covenants

So long as any of the principal amount under the Notes remain outstanding, neither the Company nor LMCS shall do any of the following, without the prior written consent of the Purchasers, which consent shall not be unreasonably withheld or delayed:

- (a) Additional Indebtedness. Create, incur, assume or suffer to exist any Indebtedness, other than (i) a working capital line of credit, of which the Company shall not draw more than \$1,000,000 at Closing; (ii) Indebtedness incurred pursuant to an annual budget approved by Purchasers;
- Merger or Consolidation. Become a party to a merger or consolidation, or purchase or otherwise acquire all or any part of the assets of any Person or any shares or other evidence of beneficial ownership of any Person, or wind-up, dissolve, or liquidate;
- (c) <u>Distributions</u>. Except as contemplated by this Agreement or to pay distributions to pay Taxes of its Members as provided in Article 5 of the LMC Operating Agreement (which provision shall not be amended without the prior written consent of the Purchaser), declare or pay any Dividends or make any other payment or distribution (in cash, property, or obligations) on account of the Company's Membership Units, or redeem, purchase, retire, or otherwise acquire any Membership Units, or set apart any money for a sinking or other analogous fund for any Dividend or other distribution on the Company's Membership Units or for any redemption, purchase, retirement, or other acquisition of any of the Company's Membership Units,
- (d) Advances and Loans. Make any loans or advances to, Guarantees for the benefit of, or investments in, any Person except for (i) reasonable advances to employees in the ordinary course of business, (ii) investments having a stated maturity no

greater than 1 year from the date such Investment is made in (A) obligations of the United States government or any agency thereof or obligations guaranteed by the United States government, (B) certificates of deposit of commercial banks having combined capital and surplus of at least \$50 million or (C) commercial paper with a rating of at least "Prime-1" by Moody's Investors Service, Inc., and (iii) intercompany loans or advances to or guarantees by the Company or LMCS for the benefit of the Company or LMCS in the ordinary course of business, consistent with past practice;

- (e) Equity Issuances. Authorize, issue, sell, assign, or otherwise dispose of (i) any of its Membership Units, (ii) any securities exchangeable for or convertible into or carrying any rights to acquire any of its Membership Units, or (iii) any option, warrant, or other right to acquire any of its Membership Units other than options and Equity Purchase rights issued in connection with any equity incentive plan or employee equity ownership plan as in existence as of the Closing or any new equity incentive plan or employee equity ownership plan approved by the board of managers; provided, however, any new equity incentive plan, together with any existing equity plan, shall not reduce the Purchaser's interest in the capital and profits of the Company;
- (f) Affiliated Transactions. Enter into any transaction, including, without limitation, the purchase, sale, or exchange of property or the rendering of any service, with any Affiliate of the Company or LMCS, except in the ordinary course of and pursuant to the reasonable requirements of the Company's or LMCS's business and upon fair and reasonable terms no less favorable to the Company or LMCS than would be obtained in a comparable arm's-length transaction with a Person not an Affiliate of the Company or LMCS and except as otherwise set forth on Section 4.4(f) of the Disclosure Schedules;
- (g) <u>Transfer of Assets</u>. Directly or indirectly in one or more related transactions, sell, lease, exchange or otherwise dispose of more than 5% of the Total Assets in any 12-month period (other than sales of inventory in the ordinary course of business, consistent with past practice) or sell, exchange or permanently dispose of any of its Intellectual Property Rights;
- (h) Sale Lease Back. Enter into any arrangement with any Person pursuant to which it leases from such Person real or personal property that has been or is to be sold or transferred, directly or indirectly by it to such Person;
- (i) <u>Prepayment of Indebtedness</u>. Prepay, redeem, purchase, defease or otherwise satisfy in any manner any principal or interest on any Indebtedness other than prepayment of the Notes as permitted according to their terms and prepayment of the Senior Debt;
- (j) Other Business Ventures. Enter into or cause LMCS to enter into the ownership, active management or operation of any business not directly related to the Business conducted by the Company or LMCS as of Closing;

13

- (k) <u>Environmental Compliance.</u> Violate any applicable Environmental Laws with respect to the real property owned or operated by the Company (including soils, groundwater, surface water, buildings or other structures), except for violations that would not have a Material Adverse Effect on the Company or LMCS;
- (l) Accounting Changes. Change the Company's or LMCS's respective fiscal years or make any change (i) in accounting treatment or reporting practices, except as required by GAAP and disclosed to the Purchasers, or (ii) in tax reporting treatment, except as required by law and disclosed to the Purchasers;
- (m) <u>Certain Security Matters</u>. Enter into or permit to exist any arrangement or agreement, other than pursuant to this Agreement or any Investment Document, which directly or indirectly prohibits the Company or LMCS from creating or incurring a Lien on any of its assets;
 - (n) <u>Business Organization</u>. Convert to any other type of business entity or cause the conversion of any of the Company's Membership Units;
- (o) <u>Joint Ventures</u>. Enter into any joint venture, involving an aggregate consideration (including the assumption of liabilities whether direct or indirect) in any calendar year exceeding \$100,000 with all such transactions being aggregated to such limit;
- (p) Additional Agreements. Enter into, become subject to, amend, modify or waive any agreement or instrument (other than Senior Loan Documents) which by its terms would (under any circumstances) restrict (i) the right of the Company to make loans or advances or pay Dividends to, transfer property to, or repay any Indebtedness owed to the Purchasers or (ii) the Company's right to perform any of the provisions of any of the Investment Documents;
- (q) <u>Compensation.</u> Increase any compensation (including salary, bonuses and other forms of current and deferred compensation) payable, directly or indirectly, to any of its officers, managers, members or their Affiliates or representatives or any Affiliate of the Company or LMCS by more than 5% per annum; except as set forth in the Employment Agreement, and except as provided for in any planned or budgeted increases under the 2007 Operating Budgets, or any performance bonuses or distributions based on EBITDA targets as agreed to in writing by C₃ and C₃ II prior to Closing;
 - $\hbox{$(r)$} \qquad \underline{\hbox{Subsidiaries}}. \ \ \hbox{Establish or acquire any Subsidiary other than LMCS};$
- (s) <u>Capital Expenditures</u>. Make Capital Expenditures (i) during the fiscal year ended December 31, 2007 in excess of \$100,000 and (ii) during any subsequent fiscal year in excess of \$200,000 (including, without limitation, payments with respect to Capitalized Leases, as determined in accordance with GAAP) other than Capital Expenditures made pursuant to an Approved Annual Budget;

14

- (t) <u>Use of Proceeds</u>. Use the proceeds from the Financing other than as set forth in <u>Section 2.4</u> of this Agreement;
- (u) <u>Amendment to Governing Documents.</u> Make any amendment to its Governing Documents which would adversely affect the Purchaser, directly or indirectly, whether by merger, conversion, operation of law or otherwise, or file any resolution of its board of managers (or similar governing body) with its jurisdiction of incorporation, formation or organization (as applicable);
- (v) <u>Take or Pay Contracts</u>. Enter into or be a party to any contract or agreement for the purchase of materials, supplies or other property or services if such contract or agreement requires that a payment be made by the Company or LMCS regardless of whether delivery is ever made of such materials, supplies or other property or services;

5.5 Financial Covenants

- (a) <u>Fixed Charge Coverage Ratio</u>. The Company shall have, on a consolidated basis at the end of each fiscal quarter of the Company, a Fixed Charge Coverage Ratio of not less than 1.09:1.00.
- (b) Maximum Total Funded Debt to EBITDA Ratio. The Company shall have, on a consolidated basis at the end of each fiscal quarter of the Company set forth below, a ratio of (i) Total Funded Debt measured as of the last day of such fiscal quarter to (ii) EBITDA for such fiscal quarter of not more than the following:

3.90:1:00 for the fiscal quarters ending September 30, 2007, and December 31, 2007;

3.5:1.00 for the fiscal quarters ending March 31, 2008, June 30, 2008, September 30, 2008, and December 31, 2008

3.0:1.0 for the fiscal quarter ending March 31, 2009 and for each fiscal quarter ending thereafter.

5.6 <u>SBIC Regulatory Provisions</u>

- (a) Equity Holders. As long as any Purchasers hold any Securities or any securities issued by the Company with respect thereto, the Company shall notify each SBIC Holder (i) at least 15 days prior to taking any action after which the number of record holders of the Company's voting securities would be increased from fewer than 50 to 50 or more and (ii) of any other action or occurrence after which the number of record holders of the Company's voting securities was increased (or would increase) from fewer than 50 to 50 or more, as soon as practicable after the Company, as applicable, becomes Aware that such other action or occurrence has occurred or is proposed to occur.
 - (b) <u>Use of Proceeds</u>. At such times as any SBIC Holder reasonably requests, the Company shall deliver to each SBIC Holder a written statement on behalf of

15

the Company certified by the chief executive officer of Company describing in reasonable detail the use of the Proceeds of the Financing hereunder by the Company. In addition to any other rights granted hereunder, the Company shall, and shall cause LMCS to, grant such SBIC Holder and the SBA access to their books, and records for the purpose of verifying the use of such proceeds and verifying the certifications made by the Company in SBA Forms 480, 652 and 1031 delivered pursuant to Section 3.13(i) and for the purpose of determining whether the principal business activity of the Company continues to constitute an eligible business activity (within the meaning of the SBIC Regulations).

(c) Economic Impact Information. Promptly after the end of each calendar year (but in any event prior to February 28 of each year), the Company shall deliver to each SBIC Holder a written assessment of the economic impact of each SBIC Holder's investment in the Company, specifying the full-time equivalent jobs created or retained in connection with the Investment, the impact of the Investment on the businesses of the Company and on taxes paid by the Company and its employees.

The Company shall not disclose any Purchaser's name or identity as an investor in the Company in any press release or other public announcement, without the prior written consent of such Purchaser unless such disclosure is required by this Agreement, applicable securities law, applicable rules of any securities exchange, law, statute, rule or regulation or by order of a court of competent jurisdiction, in which case, prior to making such disclosure, the Company shall give written notice to such Purchaser describing in reasonable detail the proposed content of such disclosure and shall permit such Purchaser to review and comment upon the form and substance of such disclosure.

5.8 Further Assurances.

At any time and from time to time, upon the reasonable request of the Purchasers, the Company shall execute, deliver and acknowledge or cause to be executed, delivered and acknowledged, such further documents and instruments and do such other acts and things as so requested in order to fully effect the purposes of this Agreement, the other Investment Documents and any other agreements, instruments and documents delivered pursuant hereto or in connection with the Securities. In addition, if requested by the Purchasers, the Company shall obtain and promptly furnish to the Purchasers evidence of all governmental approvals as may be required to enable the Company to comply with its respective obligations under the Investment Documents and to continue its business as conducted on the date hereof without material interruption or interference.

5.9 Post Closing Obligations. No later than fifteen (15) days following the Closing, the Company shall have made all filings under all applicable federal and state securities laws necessary to consummate the issuance of the Securities pursuant to this Agreement in compliance with such laws.

16

5.10 Annual Budget Approval

So long as any of the principal amount under the Notes remains outstanding, the Company will not submit any budget to its board of managers (or similar governing body) for approval or implement any Annual Budget unless such budget is first presented to the Purchasers for approval as set forth in Section 5.1(h) and approved by the Purchasers (an "Approved Annual Budget"), which approval by Purchasers shall not be unreasonably withheld or delayed. Notwithstanding anything to the contrary contained herein, if there shall be an Event of Default in existence as of the commencement of any fiscal year, the Purchasers will have absolute approval rights with respect to such upcoming Annual Budget and the Company will not submit any budget to its board of managers (or similar governing body) for approval or implement any Annual Budget unless the Purchasers have approved said budget in writing.

5.11 Put Provisions

- (a) At any time during the period commencing on the earlier to occur of (i) the fifth anniversary of the Closing, (ii) the indefeasible payment in full of all amounts due under the Notes or (iii) the occurrence of an Event of Default, the Purchasers shall have the right to put at any time all or any portion of the Purchased Membership Units then held by the Purchasers ("Put Equity") by delivering a written notice (the "Put Notice") to the Company, at a purchase price equal to the Purchasers' pro rata portion of the greater of (x) the fair market value of the Company determined by an appraisal (with no minority or liquidity discounts) conducted by an independent valuation expert agreeable to the Purchasers and the Company and (y) six (6) times the Company's trailing 12 month EBITDA, less funded Indebtedness calculated as of the date of the Put Notice plus cash and cash equivalents in excess of the minimum cash level maintained as part of working capital to operate LMC's business in the ordinary course (the "Put Price"). If the Purchasers and the Company cannot agree as to an appraiser, then the Purchasers shall select one appraiser, the Company shall select one appraiser and a third shall be selected by the first two appraisers chosen. The decision of a majority of the three appraisers shall be binding on all the parties and the Company shall bear all appraisal costs.
- (b) Upon the receipt of any Put Notice by the Company, the Company will be obligated to purchase all Put Equity requested to be repurchased in the Put Notice, at a mutually agreeable time and place which will in no event be later than 90 days after the date of the Company's receipt of the Put Notice (the "Put Closing").
- (c) At the Put Closing, the Purchasers selling Put Equity shall deliver to the Company certificates representing the Put Equity held by such Purchasers and the Company shall deliver to such holder the product of (i) the Put Price multiplied by (ii) the number of units of Put Equity owned by such holder by cashier's or certified check or wire transfer of immediately available funds payable to such Purchasers. The Company will undertake Diligent Efforts (as defined below) during the 90-day period immediately following the receipt of the Put Notice to finance the payment of the Put Price in accordance with this Section so that the Put Price may be paid in full in cash, but only to the extent such financing can be obtained on commercially reasonable terms. Such

17

diligent efforts ("Diligent Efforts") shall include, but shall not be limited to, pursuing private or public offerings of equity or debt securities, restructuring of the Company's debt and other recapitalizations. In the event that, notwithstanding such Diligent Efforts, the Company is unable to purchase all of the Put Equity at a Put Closing in cash within such 90-day period, the Company will at such Put Closing pay the maximum portion of the Put Price which it is legally able to pay in cash and pay the remaining portion of the Put Price which it is not able to pay in cash by issuing to the holders of Put Equity promissory notes (the "Put Notes") accruing interest at an annual rate equal to 12% per annum (but in no event greater than the highest rate permitted under the SBIC Regulations) (payable quarterly) and payable on demand, and such Put Notes shall be subordinate in right of payment to the Senior Debt (if any Senior Debt is then outstanding) on terms and conditions no less favorable than the contained in the Note; provided, that the interest on a Put Note shall increase by one additional percentage point up to 19% per annum (but in no event greater than the highest rate permitted under the SBIC Regulations) for each 90-day period after the Put Closing at which such Put Note was issued. In no event shall the term of the Put Notes exceed twenty-four (24) months. After a Put Closing, the Company will continuously undertake Diligent Efforts to arrange debt and/or equity financing in order to retire Put Notes for cash and will provide to the holders of such Put Notes and in order to retire Put Notes for cash and will provide to the holders of such Put Notes are shall apply in all circumstances to all Put Notes. For the avoidance of doubt, Put Notes shall be deemed to be "notes issued in exchange for Securities" and as a result the provisions of this Agreement applicable to the same shall apply in all circumstances to all Put Notes.

- (d) The Company shall not be required to pay cash or property to repurchase Put Equity to the extent the sole reason for not doing so is that it is prohibited from doing so under applicable law due to a lack of surplus; provided, that the Company will take all reasonable steps necessary to make such payments under this Section, including without limitation reducing its capital and/or increasing its net assets (by re-appraisal or otherwise). The Company shall deliver an Officer's Certificate to the Purchaser that specifies the provision of law on account of which the Company is so precluded from making such payment and the action that the Company is taking to remedy the same.
- (e) Notwithstanding anything contained herein to the contrary, if the Company is not able to pay the Put Price in full in cash within 90 days after the delivery of a Put Notice, the Purchasers may rescind the exercise of the put option at such Purchaser's election by delivering written notice to the Company within 30 days after the Company notifies each such electing Purchaser that it will be unable to pay the Put Price in full in cash. If the Company fails to satisfy its obligations pursuant to this Section the Purchasers may pursue any and all rights and remedies at law or in equity.

18

Section 6. Representations and Warranties of the Company and LMCS. As a material inducement to the Purchasers to enter into this Agreement and purchase the Securities hereunder, the Company and LMCS hereby represent and warrant to the Purchasers as follows:

6.1 <u>Organization, Corporate Power and Licenses.</u>

The Company is a limited liability company duly organized, validly existing and in good standing under the laws of the state of its organization and is qualified to do business in every jurisdiction in which its ownership of property or conduct of business requires it to qualify. The Company possesses all requisite company power and authority and all material licenses, certifications, permits and authorizations necessary to own and operate its properties, to carry on its businesses as now conducted and presently proposed to be conducted and to carry out the transactions contemplated by the Investment Documents.

6.2 <u>Capitalization and Related Matters</u>.

- (a) The attached <u>Capitalization Schedule</u> accurately sets forth the following information with respect to the Company's capitalization as of the Closing and immediately thereafter (i) the authorized Membership Units of the Company, (ii) the number of each class of Membership Units of the Company issued and outstanding, (iii) the number and each class of Membership Units of the Company reserved for issuance upon exercise of any Convertible Securities and (iv) the name of each holder of Membership Units in the Company and the amount of Membership Units owned by each such holder of such Membership Units and the capital contribution to the Company of each holder of Membership Units. As of the Closing, the Company does not have outstanding any of its Membership Units, except for the Securities and except as set forth on the <u>Capitalization Schedule</u>. As of the Closing, the Company is not subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any of the Company's Membership Units, except as set forth on the <u>Capitalization Schedule</u> and except pursuant to the terms of the Investment Documents, the Securities and the LMC Operating Agreement. As of the Closing, all of the outstanding Membership Units of the Company shall be validly issued, fully paid and nonassessable.
- (b) There are no statutory or, to the Company's Knowledge, except as set forth in the LMC Operating Agreement, contractual equity holders' preemptive rights or rights of refusal with respect to the issuance of the Securities hereunder. The Company has not violated and will not knowingly violate any applicable federal or state securities laws in connection with the offer, sale or issuance of any of its Membership Units, and, assuming the truth and accuracy of the Purchaser's representations and warranties set forth in <u>Section 7.3</u> of this Agreement, there is an exemption available to the Company from the registration requirements of the Securities Act or any applicable state securities laws in connection with the offer, sale and issuance of the Securities. To the Company's Knowledge, there are no agreements between the holders of the Company's Membership Units or with respect to the voting or transfer of the Company's Membership Units or with respect to the Company's affairs, except for the LMC Operating Agreement.

19

6.3 <u>Subsidiaries; Investments</u>

The attached <u>Subsidiary Schedule</u> correctly sets forth the name of each Subsidiary of the Company, the jurisdiction of its incorporation, formation or organization and the Persons owning the outstanding Equity Interests of such Subsidiary. Except as set forth on the <u>Subsidiary Schedule</u>, the Company does not own or hold the right to acquire any Equity Interests in any other Person.

6.4 Authorization; No Breach.

The execution, delivery and performance of each of the Investment Documents and all other agreements and instruments contemplated hereby and thereby to which the Company is a party, including the Asset Purchase and Contribution Agreement, and all agreements and instruments referenced therein, have been duly authorized by the Company. Except as set forth on the attached Restrictions Schedule, the execution and delivery by the Company of each of the Investment Documents and all other agreements and instruments contemplated hereby and thereby to which it is a party, including the Asset Purchase and Contribution Agreement, and all agreements and instruments referenced

therein, the offering, sale and issuance of the Securities hereunder and the fulfillment of and compliance with the respective terms hereof and thereof by the Company, do not and shall not: (a) conflict with or result in a breach of the terms, conditions or provisions of; (b) constitute a default under; (c) result in the creation of any Lien upon the Company's Membership Units or assets pursuant to; (d) give any third party the right to modify, terminate or accelerate any obligation under; (e) result in a violation of; or (f) require any authorization, consent, approval, exemption or other action by or notice or declaration to, or filling with, any court or administrative or governmental body or agency pursuant to, the Governing Documents of the Company, or any law, statute, rule or regulation to which the Company is subject, or any material agreement, instrument, order, judgment or decree to which the Company is subject.

6.5 Financial Statements.

- (a) Attached hereto as the Financial Statements Schedule are the following financial statements:
- (i) the audited Special-Purpose Financial Statements of the Company as a division of LMT, as of December 31, 2005 and December 31, 2006 (the "Special-Purpose Financial Statements"), and the related statements of assets and liabilities, revenues and direct expenses, and notes for the respective 12-month periods then ended; and
 - (ii) the internally prepared profit and loss statement of the Company as a division of LMT, for the period January 1, 2007 to May 31, 1007.
- (b) To the Company's Knowledge, each of the foregoing financial statements (including in all cases the notes thereto, if any) (i) is consistent with the books and records of LMT (which, in turn, are accurate and complete in all material respects), and (ii) presents fairly the financial condition and results of operations of the Company as a division of LMT for the periods set forth therein.

20

6.6 Projections and Pro Forma Financial Statements.

- (a) The Company delivered to Purchasers on May 26, 2007 a correct and complete copy of the latest projections of the income and cash flows of the Company and LMCS for the current year and for two years following Closing. Such projections have been prepared on the basis of the assumptions set forth therein, which the Company in good faith believes are fair and reasonable in light of the historical financial performance of the Company as a division of LMT.
- (b) The pro forma balance sheet of the Company and LMCS as of the date of the Closing (the "**Pro Forma Balance Sheet**"), attached hereto as Exhibit L, is complete and correct in all material respects and presents fairly in all material respects the financial condition of the Company and LMCS as of June 30, 2007 as if the transactions contemplated by the Investment Documents had occurred immediately prior to such date, and such balance sheet contains all pro forma adjustments necessary in order to fairly reflect such assumption.

6.7 <u>Absence of Undisclosed Liabilities</u>

The Company does not have any material obligation or liability (whether accrued, absolute, contingent, unliquidated or otherwise) arising out of transactions entered into at or prior to the Closing, or any action or inaction at or prior to the Closing, or any state of facts existing at or prior to the Closing other than: (a) liabilities set forth on the Pro Forma Balance Sheet (including any notes thereto); (b) other liabilities and obligations expressly disclosed on the attached Liabilities Schedule: and (c) any liabilities incurred in connection with the transactions contemplated in any of the Investment Documents.

6.8 No Material Adverse Change

Since the date of the Special-Purpose Financial Statements, there has been no change in the operating results, assets, liabilities, operations, prospects, business, condition, employee relations, dealer relations, distributor relations, customer relations or supplier relations of the Company or LMCS, taken as a whole, which has had or could reasonably be expected to have, a Material Adverse Effect.

6.9 Assets

Except as set forth on the attached <u>Assets Schedule</u>, the Company, upon Closing, will have good and marketable title to, or a valid leasehold interest in the material properties and assets located on their premises or shown on the Pro Forma Balance Sheet, free and clear of all Liens, except for Permitted Liens. Except as described on the <u>Assets Schedule</u>, upon Closing the Company's, equipment and other tangible assets will be in good operating condition (ordinary wear and tear excepted) and are fit for use in the ordinary course of business.

21

6.10 <u>Tax Matters</u>.

Except as set forth on the attached $\underline{Taxes\ Schedule}$:

- (a) The Company and LMCS have filed all federal and other material Tax Returns which they are required to file under applicable laws and regulations; all such Tax Returns are complete and correct in all material respects and have been prepared in compliance with all applicable laws and regulations in all material respects; the Company has paid all material Taxes due and owing (whether or not such Taxes are required to be shown on a Tax Return) and has withheld and paid over to the appropriate taxing authority all material Taxes which it is required to withhold from amounts paid or owing to any employee, holder of Membership Units, creditor or other third party.
- (b) The Company has not made an election under Treas. Reg. § 301.7701-3 to be taxed as a corporation. The Company is not liable for the Taxes of another Person (i) under Treas. Reg. § 1.1502-6 (or comparable provisions of state, local or foreign law, statute, rule or regulation), (ii) as a transferee or successor or (iii) by contract, indemnity or otherwise. The Company is not a party to any Tax sharing. Tax allocation, Tax indemnitication or similar agreement. The Company has disclosed on its federal income Tax Returns any position taken for which substantial authority (within the meaning of Code §6662(d)(2)(B)(i)) did not exist at the time the return was filed. The Company has not made any payments, is not obligated to make payments and is not a party to an agreement that could obligate it to make any payments that would not be deductible under Code §280G (determined without regard to Code §280G(b)(5)).
 - (c) The Company has not been a member of an Affiliated Group or filed or been included in a combined, consolidated or unitary income Tax Return, except for the consolidated group of LMT.

6.11 Contracts and Commitments

- (a) Except for the Investment Documents and the Schedules attached thereto and those contracts and agreements listed on the attached <u>Contracts Schedule</u>, after giving effect to the transactions contemplated by this Agreement, as of Closing, neither the Company nor LMCS will be a party to, or bound by, any written or oral material contract, pension, profit sharing or equity option or put rights, lease, warranty or employment Agreement.
- (b) C_3 has been provided access to a correct and complete copy of each of the written contracts, agreements and instruments and an accurate description of each of the oral contracts and agreements and other items which are referred to on Contracts Schedule, together with all amendments, waivers or other changes thereto.

6.12 <u>Intellectual Property Rights</u>.

The attached Intellectual Property Schedule contains a complete and accurate list of all material (a) patented or registered Intellectual Property Rights owned or licensed by the Company or LMCS, (b) pending patent applications and applications for registrations of other

22

Intellectual Property Rights filed by the Company or LMCS, (c) registered or unregistered trade names and corporate names owned or licensed by the Company or LMCS and (d) registered or unregistered trademarks, service marks, internet domain names, copyrights, mask works and computer software owned or licensed by the Company or LMCS (collectively, the "Intellectual Property"). The Company or LMCS shall have the right to use all Intellectual Property free and clear from any liens and on the same terms and conditions as LMT after the Closing. There have been no claims made against LMT asserting the invalidity, misuse or unenforceability of any Intellectual Property and there is no basis for any such claim and LMT has not received any notices of, and has no knowledge of any facts which indicate a likelihood of, any misappropriation by, or conflict with any Person with respect to any Intellectual Property.

6.13 <u>Litigation, Etc.</u>

Except as set forth on the Litigation Schedule, there are no material actions, suits, proceedings, orders, investigations or claims pending or, to the Company's Knowledge, threatened against or affecting the Company or LMCS (or affecting any of the officers, directors or employees of the Company or LMCS with respect to their businesses or proposed business activities), or pending or threatened by the Company or LMCS against any third party, at law or in equity, or before or by any governmental department, commission, board, bureau, agency or instrumentality.

6.14 Brokerage.

Except as set forth on the <u>Brokerage Schedule</u>, there are no claims for brokerage commissions, finders' fees or similar compensation in connection with the transactions contemplated by the Investment Documents based on any arrangement or agreement binding upon the Company. The Company shall pay, and hold the Purchasers harmless against, any liability, loss or expense (including reasonable attorneys' fees and out-of-pocket expenses) arising in connection with any such claim.

6.15 Governmental Consent, Etc.

No material permit, consent, approval or authorization of, or declaration to or filing with, any governmental authority is required in connection with the execution, delivery and performance by the Company of the Investment Documents and the other agreements contemplated hereby and thereby, except as set forth on the attached Consents Schedule.

6.16 <u>Insurance</u>.

Effective as of the Closing, the insurance coverage of the Company will be customary for prudent companies of similar size engaged in similar lines of business

6.17 Employees

The Company is not Aware of any executive or key employee of the Company or LMCS or any group of employees of the Company or LMCS that has any plans to terminate employment with the Company or LMCS. The Company and LMCS have complied in all material respects with all laws, statutes, rules and regulations relating to the employment of labor

23

(including provisions thereof relating to wages, hours, equal opportunity, collective bargaining and the payment of social security and other taxes), and the Company is not Aware of any material labor relations problems (including any union organization activities, threatened or actual strikes or work stoppages or material grievances). To the Company's Knowledge and except as set forth in the Employment Agreement, none of its employees or the employees of LMCS are subject to any noncompete, nondisclosure, confidentiality, employment, consulting or similar agreements relating to, affecting or in conflict with the present or proposed business activities of the Company or LMCS.

6.18 <u>ERISA</u>.

- (a) Neither the Company nor LMCS have any obligation to contribute to (or any other liability, including current or potential withdrawal liability with respect to) any "multiemployer plan" (as defined in §3(37) of ERISA).
- (b) Neither the Company nor LMCS have any obligation to contribute to (or any other liability with respect to) any plan or arrangement whether or not terminated, which provides medical, health, life insurance or other welfare-type benefits for current or future retired or terminated employees (except for limited continued medical benefit coverage required to be provided under Section 4980B of the Code or as required under applicable state law).
- (c) Except as set forth on the attached ERISA Schedule, neither the Company nor LMCS maintain, contribute to or have any liability under (or with respect to), any employee plan which is a tax-qualified "defined benefit plan" (as defined in §3(35) of ERISA), whether or not terminated.
- (d) Except as set forth on the attached ERISA Schedule, neither the Company nor LMCS maintain, contribute to or have any liability under (or with respect to) any employee plan which is a tax-qualified "defined contribution plan" (as defined in §3(34) of ERISA), whether or not terminated.
- (e) Except as set forth on the attached ERISA Schedule, neither the Company nor LMCS maintains, contribute to or have any liability under (or with respect to) any plan or arrangement providing benefits to current or former employees, including any bonus plan, plan for deferred compensation, employee health or other welfare benefit plan or other arrangement, whether or not terminated.

6.19 Compliance with Laws.

To the Company's Knowledge, neither the Company nor LMCS have violated any law, statute, rule or regulation which violation has had or would reasonably be expected to have a Material Adverse Effect, and the Company has not received notice of any such violation.

6.20 Small Business Matters

The Company acknowledges that each of C_3 and C_3 II is a federally licensed SBIC under the SBIC Act. To the knowledge of the Company, the Company, together with its

24

"affiliates" (as that term is defined in 13 CFR \S 121.103), is a "small business concern" within the meaning of the SBIC Regulations, including 13 CFR \S 121.103. After giving effect to the transactions contemplated by the Investment Documents, the Company will have 500 or fewer full-time equivalent employees. The information regarding the Company and affiliates (as defined above) set forth in the Small Business Administration Form 480, Form 652 and Form 1031 delivered at the Closing is accurate and complete. Copies of such forms have been completed and executed by the Company and delivered to C_3 and C_3 II at the Closing.

6.21 Affiliated Transactions.

Except as set forth on the attached Affiliated Transactions Schedule, no officer, manager, director, employee, member, stockholder, partner, limited partner, owner, principal or Affiliate of the Company or LMCS or to Company's Knowledge, any Person related, by blood, marriage or adoption to any such Person in which any such Person owns any beneficial interest, is a party to any agreement, contract, commitment, transaction or arrangement with the Company or LMCS has any material interest in any material property used by the Company or LMCS, except for employment arrangements and compensation in the ordinary course of business, consistent with past practice, or as otherwise contemplated by this Agreement.

6.22 <u>Investment Company</u>.

The Company is not an "investment company" or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company," as such terms are defined in the Investment Company Act of 1940, as amended, nor is the Company, directly or indirectly, controlled by or acting on behalf of any Person which is an "investment company" within the meaning of such act.

6.23 Margin Regulations

The Company does not own any "margin stock," as the term is defined in Regulation U of the Federal Reserve Board, and the proceeds of the sale of the Securities will be used only for the purposes contemplated in Section 2.4 hereof.

6.24 <u>Public Utility Holding Company Act</u>

The Company is not a "holding company," or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," or an "affiliate" of a "subsidiary company" of a "holding company," within the meaning of the Public Utility Holding Company Act of 1935, as amended.

6.25 <u>Customers and Suppliers</u>

(a) The attached <u>Customer Schedule</u> lists the ten (10) largest customers of each of the Company, as a division of LMT, for the period beginning January 1, 2005 and ending December 31, 2006 and sets forth opposite the name of each such customer the percentage of consolidated net sales during such period attributable to such customer.

25

- (b) The attached Supplier Schedule lists the five (5) largest suppliers of each of the Company, as a division of LMT, for the period beginning January 1, 2005 and ending December 31, 2006.
- (c) To the Knowledge of the Company, no supplier listed on the Supplier Schedule has indicated that it shall stop, or materially decrease the rate of, supplying materials, products or services of such company, and no customer listed on the Sustomer Schedule has indicated that it shall stop, or materially decrease the rate of, buying materials, products or services from the Company once the transactions contemplated by the Investment Documents occur.

6.26 <u>Disclosure</u>

Neither this Agreement nor any of the schedules, attachments, written statements, documents, certificates or other items provided to the Purchasers by or on behalf of the Company with respect to the transactions contemplated hereby contain any untrue statement of a material fact or omit a material fact necessary to make each statement contained herein or therein not misleading.

6.27 Closing Date

The representations and warranties of the Company contained in this Agreement and elsewhere in the Investment Documents and all information contained in any exhibit, schedule or attachment hereto or thereto or in any certificate or other writing delivered by, or on behalf of, the Company to the Purchasers is and shall be complete and correct as of the Closing, except for those representations and warranties which refer to facts existing on a specific date, which shall be true and correct as of such date (both immediately prior to and immediately after giving effect to the transactions contemplated by the Investment Documents).

Section 7. <u>Representations and Warranties of the Purchasers.</u> As a material inducement to the Company to enter into this Agreement and issuance of the Securities hereunder, the Purchasers hereby represent and warrant to the Company as follows:

7.1 <u>Organization, Corporate Power and Licenses.</u>

Each Purchaser is a limited liability company or corporation duly organized, validly existing and in good standing under the laws of the state of its organization and is qualified to do business in every jurisdiction in which its ownership of property or conduct of business requires it to qualify. Each Purchaser possesses all requisite company power and authority and all licenses, certifications, permits and authorizations necessary to own and operate its properties, to carry on its businesses as now conducted and presently proposed to be conducted and to carry out the transactions contemplated by the Investment Documents.

7.2 <u>Due Authorizations</u>

26

Transactions have been duly and validly authorized by all necessary action on the part of each Purchaser. This Agreement has been, and each of the other Investment Documents when executed, will be, duly executed and delivered by Purchaser and constitute legal, valid and binding obligations of Purchaser enforceable in accordance with their respective terms, except as such enforcement may be limited by (i) bankruptcy, insolvency, reorganization, moratorium and other laws of general application affecting the rights and remedies of creditors, and (ii) general principles of equity (regardless of whether such enforcement is considered in a proceeding in equity or at law).

7.3 <u>Purchaser's Investment Representations</u>

Each Purchaser, severally and not jointly, hereby represents that:

- (a) <u>Purchase Entirely for Own Account</u>. This Agreement is made with such Purchaser in reliance upon such Purchaser's representation to the Company, which by such Purchaser's execution of this Agreement, such Purchaser hereby confirms, that the Securities purchased hereby will be acquired for investment for such Purchaser's own account, not as a nominee or agent, and not with a view to the resale or distribution of any part thereof, and that such Purchaser has no present intention of selling, granting any participation in, or otherwise distributing the same.
- (b) Reliance upon Purchaser's Representations. Such Purchaser understands that the Securities are not registered under the Securities Act on the ground that the sale provided for in this Agreement and the issuance of the Securities hereunder is exempt from registration under the Securities Act pursuant to Section 4(2) thereof, and that the Company's reliance on such exemption is predicated on such Purchaser's representations set forth herein.
- (c) <u>Investment Experience; Economic Risk.</u> Such Purchaser understands that the Company has a limited financial and operating history and that an investment in the Company involves substantial risks. Such Purchaser is experienced in evaluating and investing in private placement transactions of securities of companies in a similar stage of development to that of the Company and acknowledges that such Purchaser is able to fend for himself, herself or itself. Such Purchaser has such knowledge and experience in financial and business matters that such Purchaser is capable of evaluating the merits and risks of the investment in the Securities. Such Purchaser can bear the economic risk of such Purchaser's investment and is able, without impairing such Purchaser's financial condition, to hold the Securities for an indefinite period of time and to suffer a complete loss of such Purchaser's investment.
 - (d) Accredited Purchaser. Such Purchaser is an "accredited investor" as defined in Rule 501(a) of Regulation D promulgated under the Securities Act.
- (e) Restricted Securities. Such Purchaser understands that the Securities are characterized as "restricted securities" under the federal securities laws inasmuch as they are being acquired from the Company in a transaction not involving a public offering and that under such federal securities laws and applicable regulations such

27

Securities may be resold without registration under the Securities Act only in certain limited circumstances. In this connection, such Purchaser represents that it is aware of the provisions of Rule 144 promulgated under the Securities Act which permit limited resale of shares purchased in a private placement subject to the satisfaction of certain conditions, including, among other things, the existence of a public market for the shares, the availability of certain current public information about the Company, the resale occurring not less than one year after a party has purchased and paid for the security to be sold, the sale being effected through a "broker's transaction" or in transactions directly with a "market maker" and the number of shares being sold during any three-month period not exceeding specified limitations.

7.4 <u>Purchaser's Reliance</u>.

Each Purchaser acknowledges that, except as expressly set forth in the Company's and LMCS's representations and warranties in Section 6, it is consummating the Transaction without any representation or warranty, express or implied, at law or in equity, by the Company, or any of the Company's officers, directors, employees, Affiliates or advisors, including with respect to (i) merchantability or fitness for any particular purpose, (ii) the operation of the business of the Company and LMCS after the Closing in any manner, or (iii) the probable success or profitability of the business of the Company and LMCS after the Closing. With respect to any projection or forecast delivered by or on behalf of the Company and LMCS to Purchasers, Purchasers acknowledge that (x) there are uncertainties in interent in attempting to make such projections and forecasts, (y) the accuracy and correctness of such projections and forecasts may be affected by information which may become available through discovery or otherwise after the date of such projections and forecasts and (z) it is familiar with each of the foregoing. Purchasers have no knowledge of any facts and/or circumstances that would make any of the representations and warranties of the Company contained in Section 6 untrue or misleading.

7.5 Permitted SBIC Investment. To Purchasers' knowledge, the Purchasers' investment in the Securities is permitted by the SBIC Act and SBIC regulations.

Section 8. Events of Default.

8.1 <u>Definition</u>

An "Event of Default" shall be deemed to have occurred if, during the time that any of the Securities are outstanding until the Notes, or any Put Notes issued pursuant to Section 5.11(c), have been repaid in full:

(a) Failure to Make Payments. The Company fails to pay when due and payable (whether at maturity or otherwise, and such failure continues uncured for ten (10) Business Days following notice provided by any Purchaser via electronic mail to the Company's President the full amount of interest then accrued on the Notes, or the full amount of any principal payment (together with any applicable premium) on the Notes or any other amounts payable under the Securities or the Investment Documents;

28

(b) Failure to Observe Covenants. The Company:

- (i) breaches, fails to perform or observe any of the covenants contained in Sections 5.1 through 5.10. inclusive and such failure continues uncured for twenty (20) days following notice provided by any Purchaser via electronic mail to the Company's President, or as otherwise provided by Section 9.11. unless such period is otherwise extended by the Purchaser; or
- (ii) breaches, fails to perform or observe any other provision contained in the Investment Documents and (A) if such failure has had or could have a Material Adverse Effect and such failure continues uncured for twenty (20) days following notice provided by any Purchaser via electronic mail to the Company's President or as otherwise provided by Section 9.11, unless such period is otherwise extended by Purchaser, or (B) if such failure has not had and could not reasonably be expected to have a Material Adverse Effect and such failure continues uncured for thirty (30) days or the Company is not proceeding diligently to cure such failure;
- (c) Representations. Any representation, warranty or information contained herein or required to be furnished to any holder of the Securities pursuant to the Investment Documents, or any writing furnished by the Company to any holder of the Securities, is false or misleading in any material respect on the date made;
- (d) Insolvency. The Company makes an assignment for the benefit of creditors or admits in writing its inability to pay its debts generally as they become due, or an order, judgment, decree or injunction is entered adjudicating the Company bankrupt or insolvent or requiring the dissolution or split up of the Company or preventing the Company from conducting all or any part of its business; or any order for relief with respect to the Company is entered under the Federal Bankruptcy Code; or the Company petitions or applies to any tribunal for the appointment of a custodian, trustee, receiver or liquidator of the Company, or of any substantial part of the assets of the Company, or commences any proceeding (other than a proceeding for the voluntary liquidation and dissolution of LMCS) relating to the Company under any bankruptcy reorganization, arrangement, insolvency, readjustment of debt, dissolution or liquidation or similar laws of any jurisdiction now or hereafter in effect; or any such petition or application is filed, or any such proceeding is commenced, against the Company and either (i) the Company by any act indicates its approval thereof, consent thereto or acquiescence therein or (ii) such petition, application or proceeding is not dismissed within 60 days;
- (e) <u>Payments on Indebtedness</u>. The Company shall fail to pay when due any principal of or interest on any Indebtedness or the maturity of any such Indebtedness shall have been required to be prepaid prior to the stated maturity thereof, or any event shall have occurred and shall continue uncured for twenty (20) days following notice provided by the Purchaser via electronic mail to the Company's President or as otherwise provided by <u>Section 9.11</u>, that permits (or, with the giving of notice or lapse of time or both,

29

would permit) any holder or holders of such Indebtedness or any Person acting on behalf of such holder or holders to accelerate the maturity thereof or require any such prepayment;

- (f) <u>Impairment of Security</u>. This Agreement or any other Investment Documents (excluding the Employment Agreement) shall cease to be in full force and effect or shall be declared null and void or the validity or enforceability thereof shall be contested or challenged by the Company, or the Company shall deny that it has any further liability or obligation under any of the Investment Documents (excluding the Employment Agreement);
- (g) ERISA Matters. (i) With respect to any Pension Plan, a Prohibited Transaction occurs which in the reasonable determination of C₃ and C₃ II could result in material liability to the Company, (ii) with respect to any Title IV Plan, the filing of a notice to voluntarily terminate any such plan in a distress termination, (iii) with respect to any Multiemployer Plan, the Company or any ERISA Affiliate shall incur any Withdrawal Liability, (iv) with respect to any Qualified Plan, the Company or any ERISA Affiliate shall incur an accumulated funding deficiency or request a funding waiver from the IRS or (v) with respect to any Title IV Plan or Multiemployer Plan which has an ERISA Event not described in clauses (ii) through (iv) hereof, in the reasonable determination of C₃ there is a reasonable likelihood for termination of any such plan by the PBGC; provided, that the events listed in clauses (i) through (iv) hereof shall constitute Events of Default only if the liability, deficiency or waiver request of the Company or any ERISA Affiliate, whether or not assessed, could, in the opinion of C₃ reasonably be expected to have a Material Adverse Effect;

- (h) Commencement of Proceedings. The Company shall fail to discharge within a period of thirty (30) days after the commencement thereof any attachment, sequestration, or similar proceeding or proceedings involving an aggregate amount in excess of Fifty Thousand Dollars (\$50,000) against any of its assets or properties, except and to the extent the Company has obtained surety bonds;
- (i) Entry of Judgment. A final judgment or judgments for the payment of money in excess of Fifty Thousand Dollars (\$50,000) in the aggregate shall be rendered by a court or courts against the Company and the same shall not be discharged (or provision shall not be made for such discharge), or a stay of execution thereof shall not be procured, within forty-five (45) days from the date of entry thereof and the Company shall not, within said period of forty-five (45) days, or such longer period during which execution of the same shall have been stayed, appeal therefrom and cause the execution thereof to be stayed during such appeal;
- (j) Failure to Observe Other Obligations. The Company defaults in the payment when due, or in performance or observance of, any material obligation of, or condition agreed to by, the Company with respect to any material purchase or lease of goods or services where such default, singly or in the aggregate with all other such defaults, could reasonably be expected to have a Material Adverse Effect;

30

- (k) Receivership. The Company's assets are attached, seized, subjected to a writ or distress warrant, or are levied upon, or come within the possession of any receiver, trustee, custodian or assignee for the benefit of creditors in connection with any obligations or liabilities of the Company and such attachment, seizure, warrant, levy or possession could reasonably be expected to have a Material Adverse Effect;
- (I) <u>Curtailment of Business Conduct</u>. The occurrence of any event that could cause the cessation or substantial curtailment of the conduct of business by the Company, including the Company being enjoined, retrained or prevented by any court or administrative agency from conducting any material part of its business;
 - (m) Change in Control. A Change in Control shall occur; or
- (n) Regulatory Default. Any (i) diversion by the Company of the proceeds from the sale of the Securities from the purposes set forth in Section 2.4 or (ii) change in the Company's business activity within the twelve (12) month period following the Closing that renders the Securities an ineligible investment for the Purchaser pursuant to Section 107.760 of Title 13 of the Code of Federal Regulations.

The foregoing shall constitute "Events of Default" whatever the reason or cause for any such Event of Default and whether it is voluntary or involuntary or is effected by operation of law or pursuant to any judgment, decree or order of any court or any order, rule or regulation of any administrative or governmental body.

8.2 Consequences of Events of Default.

- (a) If any Event of Default has occurred, the interest rate of the Note shall increase by the lesser of three percent (3%) per annum until paid.
- (b) If an Event of Default of the type described in Section 8.1(d) has occurred, then the aggregate outstanding principal amount of the Note and any notes issued in exchange for any Securities (together with all accrued interest thereon and all other amounts due and payable with respect thereto) shall become immediately due and payable without any action on the part of the holders thereof, and the Company shall immediately pay to the holders of the Note all amounts due and payable with respect thereto.
- (c) If an Event of Default (other than under Section 8.1(d) has occurred and is continuing, then the holder of the Note and the holders of any notes issued in exchange for any Securities then outstanding may declare all or any portion of the outstanding principal amount of the Note and any notes issued in exchange for any Securities (together with all accrued interest thereon and all other amounts due and payable with respect thereto) to be immediately due and payable and may demand immediate payment of all or any portion of the outstanding principal amount of the Note and any notes issued in exchange for any Securities (together with all such other amounts then due and payable) owned by such holder or holders. The Company shall give prompt written notice of any such demand to the other holders of any notes issued in exchange

31

for any Securities, each of which may demand immediate payment of all or any portion of such holder's Note and any notes issued in exchange for any Securities. If the holder of the Note and the holders of any notes issued in exchange for any Securities, the Company shall immediately pay to such holder or holders all amounts due and payable with respect thereto.

- (d) If, as a result of an Event of Default of the type described in Section 8.1(n), then the aggregate outstanding principal amount of the Note and any notes issued in exchange for any Securities (together with all accrued interest thereon and all other amounts due and payable with respect thereto) shall become immediately due and payable without any action on the part of the holders thereof, and the Company shall immediately (i) pay to the holder of such Note and the holders of any notes issued in exchange for any Securities all amounts due and payable with respect thereto and (ii) purchase all of the outstanding Purchased Membership Units pursuant to the terms of Article 12 of the LMC Operating Agreement.
- (e) If, as a result of an Event of Default of the type described in Section 8.1(i), or 8.1(d), the Company fails to pay when due and payable (whether at maturity or otherwise) the full amount of interest then accrued on any Note, or the full amount of any principal payment (together with any applicable premium) on any Note, and such failure continues for six (6) months, in addition to the consequences set forth in Section 8.2(a)-(c), C₃ and C₃ II, pursuant to the Irrevocable Springing Proxy, shall be permitted to vote the Membership Units held by LMT to elect the board of managers of the Company.

Section 9. <u>Miscellaneous.</u>

9.1 <u>Expenses</u>

The Company shall pay, and hold the Purchasers harmless against liability for the payment of, and reimburse on demand as and when incurred from and against: (a) all reasonable costs and expenses incurred by each of them in connection with their due diligence review of the Company, the preparation, negotiation, execution and interpretation of the Investment Documents, and the consummation of all of the transactions contemplated hereby and thereby discounsel, environmental consultants and accountants), which costs and expenses shall be payable at the Closing and which shall not exceed \$155,000 in the aggregate or, if the Closing does not occur, payable upon demand; (b) all reasonable fees and expenses incurred with respect to any amendments or waivers (whether or not the same become effective) under or in respect of each of the Investment Documents, the Governing Documents of the Company and the other agreements and instruments contemplated hereby and thereby (including all expenses incurred in connection with any proposed merger, sale or recapitalization of the Company); (c) all reasonable fees, and expenses incurred with respect of the Investment Documents or the issuance, delivery or acquisition of any Securities or any other Membership Units for which Securities are exchanged or converted; and (d) the reasonable fees and expenses incurred with respect to the enforcement of the rights granted under the Investment Documents, the Securities,

32

any other Membership Units for which Securities are exchanged or converted. If the Company fails to pay when due any amounts due to the Purchasers or fails to comply with any obligations pursuant to this Agreement or any other agreement, document or instrument executed or delivered in connection herewith, the Company shall, upon demand by the Purchasers, pay to the Purchasers such further amounts as shall be sufficient to cover the cost and expense (including, but not limited to attorneys' fees) incurred by or on behalf of the Purchasers in collecting all such amounts due or in otherwise enforcing the Purchasers' rights and remedies hereunder. To the extent any Event of Default, after investigation, is acknowledged and agreed to by the Company or otherwise determined, the Company agrees to pay to the Purchasers all reasonable costs and expenses incurred by them, including reasonable compensation to their attorneys for all services rendered, in connection with the investigation of any such Event of Default and enforcement of their rights hereunder or under the other Investment Documents.

9.2 <u>Remedies</u>

Each holder of Securities shall have all rights and remedies set forth in the Investment Documents and the Governing Documents of the Company and all rights and remedies which such holders have been granted at any time under any other agreement or contract and all of the rights which such holders have under any law, statute, rule or regulation. No remedy hereunder or thereunder conferred is intended to be exclusive of any other remedy, and each and every such remedy shall be cumulative and shall be in addition to every other remedy given hereunder or now or hereafter existing at law or in equity or by statute or otherwise. Any Purchaser having any rights under any provision of the Investment Documents shall be entitled to enforce such rights specifically (without posting a bond or other security), to recover damages by reason of any breach of any provision of the Investment Documents and to exercise all other rights granted by law, statute, rule or regulation.

9.3 Amendments and Waivers

Except as otherwise expressly provided herein, the provisions of this Agreement may be amended and the Company may take any action herein prohibited, or omit to perform any act herein required to be performed by it, only if the Company has obtained the written consent of the holders of a majority-in-interest of the Securities provided, that if there are no Securities outstanding, the provisions of this Agreement may be amended or waived by the Company and the Company may take any action herein prohibited. No other course of dealing between the Company and the holder of any Securities or any other equity for which Securities are exchanged or converted or any delay in exercising any rights hereunder or under the Governing Documents of the Company shall operate as waiver of any rights of any such holders. For purposes of this Agreement, the Securities or any equity for which Securities are exchanged or converted which are held by the Company shall not be deemed to be outstanding.

9.4 <u>Survival of Agreement.</u>

All covenants, representations and warranties contained in this Agreement or made in writing by the Company in connection herewith shall survive the execution and delivery of this Agreement until such time as the Notes, together with all unpaid interest thereon, are paid

33

All payments hereunder and under the Securities and any notes issued in exchange for any Securities shall be made by the Company without setoff, offset, deduction or counterclaim, free and clear of all taxes, levies, imports, duties, fees and charges, and without any withholding, restriction or conditions imposed by any governmental authority.

9.6

All covenants and agreements contained in this Agreement by or on behalf of any of the parties hereto shall bind and inure to the benefit of the respective successors and assigns of the parties hereto whether or not so expressed; provided, that the Company shall not be permitted to assign or delegate its rights or obligations under this Agreement, the Securities or any notes issued in exchange for any Securities. The Purchasers may assign, by sale, participation or otherwise all, or any, of their rights and obligations under this Agreement to any other venture capital or institutional investor as it may choose, subject to the approval of the Company, and such approval shall not be unreasonably withheld.

Aggregation.

For purposes of the Investment Documents, all holdings of Securities and any equity for which Securities are exchanged or converted by Persons who are Affiliates of each other shall be aggregated for purposes of meeting any threshold tests under the Investment Documents.

9.8

Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be prohibited by or invalid under applicable law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of this Agreement and shall be reformed and enforced to the maximum extent permitted under applicable law.

9.9 Counterparts

This Agreement may be executed in two (2) or more counterparts, any one (1) of which need not contain the signatures of more than one (1) party, but all such counterparts taken together shall constitute one (1) and the same Agreement.

9.10

All other issues and questions concerning the construction, validity, enforcement and interpretation of this Agreement and the schedules hereto and (except as otherwise expressly provided therein) the exhibits hereto shall be governed by, and construed in accordance with, the

34

laws of the State of Missouri, without giving effect to any choice of law or conflict of law rules or provisions (whether of the State of Missouri or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Missouri. In furtherance of the foregoing, the internal law of the State of Missouri shall control the interpretation and construction of this Agreement (and all schedules and exhibits hereto), even though under that jurisdiction's choice of law or conflict of law analysis, the substantive law of some other jurisdiction would ordinarily apply.

Notices

9.11

All notices, demands or other communications to be given or delivered under or by reason of the provisions of this Agreement shall be in writing and shall be deemed to have been given when delivered personally to the recipient, sent to the recipient by reputable overnight courier service (charges prepaid), mailed to the recipient by certified or registered mail, return receipt requested and postage prepaid or sent via facsimile to the number set forth below with a copy mailed to the recipient as set forth above. Such notices, demands and other communications shall be sent to the Company and Purchasers at the addresses indicated below:

To the Company

Liquidmetal Coatings, LLC 30452 Esperanza Rancho Santa Margarita, California 92688 Attn: Legal Department Facsimile: (813) 314-0270

With a copy (which shall not constitute notice) to:

Foley & Lardner L.L.P. 100 North Tampa Street Suite 2700 Tampa, Florida 33602 Attn: Curt P. Creely Facsimile: (813) 221-4210

To LMCS

Liquidmetal Coatings Solutions, LLC 30452 Esperanza Rancho Santa Margarita, California 92688 Attn: Legal Departmen Facsimile: (813) 314-0270

35

To the Purchasers:

C3 Capital, LLC 4520 Main Street Suite 1600 Kansas City, Missouri 64111 Robert L. Smith Facsimile: (816) 756-5552

with a copy (which shall not constitute notice) to:

Brvan Cave LLP 3500 One Kansas City Place 1200 Main Street Kansas City, Missouri 64105 Attn: Thomas W. Van Dyke Facsimile: (816) 374-3300

or to such other address or to the attention of such other person as the recipient party has specified by prior written notice to the sending party.

9.12 Consideration for Purchased Units: Treatment of Fees.

 C_3 , C_3 II and the Company acknowledge and agree that (i) the Notes and Purchased Membership Units constitute an investment unit for purposes of Section 1273(c)(2) of the Code, (ii) the fair market value of the Notes to be purchased by C_3 and C_3 II hereunder is \$5,527,210 and the fair market value of the Purchased Membership Units to be issued hereunder is \$972,790, and (iii) for purposes of Treasury Regulation section 1.1273-2(h), the issue price allocable to the Notes is \$5,527,210 and the issue price allocable to the Purchased Membership Units is \$972,790. C₃, C₃ II and the Company shall file their respective federal, state and local Tax Returns in a manner which is consistent with such valuation and allocation and shall not take any action or position (whether in preparation of Tax Returns, financial statements or otherwise) which is inconsistent or contrary with any of the above.

9.13 Construction

The parties hereto have participated jointly in the negotiation and drafting of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement. The parties intend that each representation, warranty and covenant contained herein shall have independent significance

9.14 Complete Agreement

This Agreement, those documents expressly referred to herein, and the other documents of even date herewith delivered or executed in connection with the transactions contemplated hereby embody the complete agreement and understanding among the parties and supersede any prior agreements or representations by or among the parties, written or oral, which may have related to the subject matter hereof in any way.

9.15 Indemnification

In consideration of the Purchasers' execution and delivery of this Agreement and the purchase of the Securities hereunder and in addition to all of the Company's other obligations under this Agreement and in addition to all other rights and remedies available at law or in equity, the Company shall defend, protect and indemify the Purchasers and each other holder of Securities or any other equity for which Securities are exchanged or converted and all of their officers, managers, directors, stockholders, members, partners, Imited partners, Imited partners, Imited partners, Imited partners, and save and hold each of them harmless from and against, and pay on behalf of or reimburse such part), on demand as and when incurred, any and all actions, causes of action, suits, claims, losses (including diminutions in value and consequential damages), costs, penalties, fees, liabilities and damages and expenses in connection therewith (irrespective of whether any such Indemnitee is a party to the action for which indemnification hereunder is sought), including reasonable attorneys' fees and disbursements, interest and penalties and all amounts paid in investigation, defense or settlement of any of the foregoing (the "Liabilities"), incurred by the Indemnitees or any of them as a result of, or relating to: (a) any transaction financed or to be financed in whole or in part, directly or indirectly, with the proceeds of the issuance of the Securities; (b) the execution, delivery, performance or enforcement of the Investment Documents, any Governing Documents of the Company and any other instrument, document or agreement executed pursuant hereto or thereto of the Company by any of the Indemnitees, except to the extent any such Liabilities are caused by Purchaser's breach of an Investment Document or by the particular Indemnitee's gross negligence or willful misconduct; and (c) the past, present or future environmental condition of any property owned, operated or used by the Company, its predecessors or successors or of any off

37

9.16 Payment Set Aside.

To the extent that any payment or payments is made to any Purchaser hereunder or under the Securities, any notes issued in exchange for any Securities or any Membership Units for which Securities are exchanged or converted or such Purchaser enforces its rights or exercises its right of setoff hereunder or thereunder, and such payment or payments or the proceeds of such enforcement or setoff or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside, recovered from, disgorged by or are required to be refunded, repaid or otherwise restored to such payor, a trustee, receiver or any other Person under any law, statute, rule or regulation (including, without limitation, any bankruptcy law, state or federal law, common law or equitable cause of action), then to the extent of any such restoration the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or setoff had not occurred.

9.17 Jurisdiction and Venue

Each of the parties: (a) submits to the jurisdiction of any state or federal court sitting in Kansas City, Missouri in any legal suit, action or proceeding arising out of or relating to this Agreement, the Securities or any Membership Units for which Securities are exchanged or converted; (b) agrees that all claims in respect of the action or proceeding may be heard or determined in any such court; and (c) agrees not to bring any action or proceeding arising out of or relating to this Agreement, the Securities, any notes issued in exchange for any Securities or any Membership Units for which Securities are exchanged or converted in any other court. Each of the parties waives any defense of inconvenient forum to the maintenance of any action or proceeding so brought and waives any bond, surety or other security that might be required of any other party with respect thereto.

9.18 Waiver of Right to Jury Trial

THE COMPANY AND EACH HOLDER OF SECURITIES AND MEMBERSHIP UNITS FOR WHICH SECURITIES ARE EXCHANGED OR CONVERTED HEREBY WAIVES, TO THE EXTENT PERMITTED BY APPLICABLE LAW, TRIAL BY JURY IN ANY LITIGATION IN ANY COURT WITH RESPECT TO, IN CONNECTION WITH, OR ARISING OUT OF THIS AGREEMENT, THE SECURITIES, ANY NOTES ISSUED IN EXCHANGE FOR ANY SECURITIES OR ANY MEMBERSHIP UNITS FOR WHICH SECURITIES ARE EXCHANGED OR CONVERTED OR THE VALIDITY, PROTECTION, INTERPRETATION, COLLECTION OR ENFORCEMENT THEREOF. THE COMPANY AGREES THAT THIS SECTION IS A SPECIFIC AND MATERIAL ASPECT OF THIS AGREEMENT AND ACKNOWLEDGES THAT PURCHASER WOULD NOT PURCHASE THE SECURITIES HEREUNDER IF THIS SECTION WERE NOT PART OF THIS AGREEMENT.

38

Signature Page to Securities Purchase Agreement

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on the date first written above.

COMPANY:

LIQUIDMETAL COATINGS, LLC

By: /s/ Larry Buffington
Name: Larry Buffington
Its: President and Chief Executive Officer

LMCS:

LIQUIDMETAL COATINGS SOLUTIONS, LLC

By: /s/ Larry Buffington
Name: Larry Buffington
Its: President and Chief Executive Officer

C₃:

C₃ CAPITAL PARTNERS, L.P.

By: C₃ Partners, LLC Its: General Partner

By: /s/ Robert L. Smith
Name: Robert L. Smith
Its: Manager

C₃ II:

C₃ CAPITAL PARTNERS II, L.P.

By: C₃ Partners II, LLC Its: General Partner

By: /s/ Robert L. Smith
Name: Robert L. Smith
Its: Manager

39

Exhibit A-1	C ₃ Note	2.1(a)
Exhibit A-2	C ₃ II Note	2.1(b)
Exhibit B-1	Company Operating Agreement	2.1(a)
Exhibit B-2	LMCS Operating Agreement	3.3
Exhibit C-1	Company Certificate of Formation	3.2
Exhibit C-2	LMCS Certificate of Formation	3.2
Exhibit D	Employment Agreement	3.6
Exhibit E-1	Security Agreement	3.7
Exhibit E-2	Patent and Trademark Security Agreement	3.7
Exhibit F	Pledge Agreement	3.8
Exhibit G	Collateral Assignment	3.9
Exhibit H-1	SBA Form 480	3.13(i)
Exhibit H-2	SBA Form 652	3.13(i)
Exhibit H-3	SBA Form 1031	3.13(i)
Exhibit I	Solvency Certificate	3.13(p)
Exhibit J	Irrevocable Springing Proxy	3.13(s)
Exhibit K	Compliance Certificate	5.1(k)
Exhibit L	Pro Forma Balance Sheet	6.6(b)
Appendix 1	Definitions	

LIST OF DISCLOSURE SCHEDULES

Section
3.13(h)
3.15(r)
6.2(a)
6.3
6.4
6.5
6.7
6.9
6.10
6.11
6.12
6.13
6.14
6.15
6.18
6.21
6.25
6.25

Exhibit A-1

THE SECURITIES REPRESENTED BY THIS INSTRUMENT WERE ISSUED ON JULY 24, 2007, AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THE TRANSFER OF THE SECURITIES REPRESENTED BY THIS INSTRUMENT IS SUBJECT TO THE CONDITIONS SPECIFIED IN THE SECURITIES PURCHASE AGREEMENT, DATED JULY 24, 2007 AND AS AMENDED AND MODIFIED FROM TIME TO TIME, AMONG LIQUIDMETAL COATINGS, LLC, AND C₃ CAPITAL PARTNERS, L.P. AND C₃ CAPITAL PARTNERS II, L.P. UPON WRITTEN REQUEST, A COPY OF SUCH CONDITIONS SHALL BE FURNISHED BY THE MAKER TO THE HOLDER HEREOF WITHOUT CHARGE.

14% SUBORDINATED NOTE

July 24, 2007 \$3,770,000

Liquidmetal Coatings, LLC, a Delaware limited liability company ("LMC"), and Liquidmetal Coatings Solutions, LLC, a Delaware limited liability company ("LMCS," hereinafter jointly and severally with LMC, "Borrowers" and each individually, "Borrowers") hereby jointly and severally promise to pay to the order of C₃ Capital Partners, L.P., a Delaware limited partnership ("C₃ Capital"), the principal amount of \$3,770,000 ("Principal Amount"), together with interest thereon calculated from the date hereof in accordance with the provisions of this Note.

This Note is issued pursuant to a Securities Purchase Agreement, dated as of July 24, 2007 (as amended and modified from time to time, the "Purchase Agreement"), among Borrowers, C₃ Capital and C₃ Capital Partners II, L.P. This Note is one of the "Notes" referred to in the Purchase Agreement. The Purchase Agreement contains terms governing the rights of the holder of this Note, and all provisions of the Purchase Agreement are hereby incorporated herein in full by reference. Unless otherwise indicated herein, capitalized terms used in this Note have the same meanings set forth in the Purchase Agreement.

- 1. <u>Payment of Interest</u>. Except as otherwise expressly provided in the Purchase Agreement,
- (a) In-Kind Interest. Interest shall accrue at a rate of 2% per annum (the "In-Kind Interest," which shall be computed on the basis of a 360-day year for the actual number of days elapsed in any year) on the unpaid Principal Amount, applied on a monthly basis to the Principal Amount, and paid, together with the Principal Amount in accordance with Section 2 or Section 4(c), as the case may be.
- (b) Paid Interest. Interest shall accrue at a rate of 12% per annum ("Paid Interest," which shall be computed on the basis of a 360-day year for the actual number of days elapsed in any year) on the unpaid Principal Amount of this Note and shall be paid on the first day of each calendar month beginning September 1, 2007. Unless prohibited under applicable law, any amount of accrued interest which is not paid pursuant to this Section 1(b) on the date on which it is due and payable shall bear interest at a rate of 12% per annum, or as otherwise specified in the Purchase Agreement, until such interest is paid to the holder of this Note. Any accrued interest which for any reason has not theretofore been paid during the term of this Note shall be paid in full on the date on which the final payment of the Principal Amount is made. (In-Kind Interest and Paid Interest, collectively, shall be referred to herein as "Combined Interest".)
 - (c) Form of Interest Payment. Payments of Combined Interest shall be made in lawful money of the United States of America and in immediately available

1

funds by automatic bank draft pursuant to the instructions attached hereto as Schedule A. or at such other place as the holder of this Note (the "Holder") may specify in writing.

- 2. Payment of the Principal Amount on the Note.
- (a) Schedule of Principal Payments. Borrowers shall pay the unpaid Principal Amount to the Holder on July 20, 2012 ("Maturity Date"), together with all accrued and unpaid Combined Interest on the Principal Amount being repaid.
- (b) Form of Principal Payments. Payments of principal and interest shall be made in lawful money of the United States of America and in immediately available funds by automatic bank draft pursuant to the instructions attached hereto as Schedule A, or at such other place as the Holder may specify in writing.
 - 3. <u>Default</u>. The occurrence of any one or more of the Events of Default (as defined in the Purchase Agreement) shall constitute an Event of Default hereunder.
- (a) <u>Default Rate</u>. Upon the occurrence and during the continuance of a default or an Event of Default by Borrowers under this Note or the Purchase Agreement, all amounts due and owing under the Note shall bear interest at a rate equal to three percent (3%) per annum greater than the then existing applicable Combined Interest rate on this Note ("Default Rate").
- (b) Insolvency or Regulatory Default. If any Event of Default contemplated by Section 8.1(d) or 8.1(n) of the Purchase Agreement shall occur for any reason, then in any such event, in addition to all rights and remedies of the Holder under the Purchase Agreement, applicable law or otherwise, all such rights and remedies being cumulative, not exclusive and enforceable alternatively, successively and concurrently, the Holder may, at its option, declare any or all of Borrowers' obligations, liabilities and indebtedness owing to the Holder under the Purchase Agreement and all amounts owing under this Note, to be due and payable, whereupon the then unpaid balance hereof, together with all Combined Interest accrued thereon, shall forthwith become due and payable, together with interest accruing thereafter at the then applicable interest rate stated above until the indebtedness evidenced by this Note is paid in full, plus the costs and expenses of collection hereof, including, but not limited to, reasonable attorneys' fees and legal expenses.
- (c) Waiver. Each Borrower (i) waives diligence, demand, presentment, protest and notice of any kind, (ii) agrees that it will not be necessary for the Holder to first institute suit in order to enforce payment of this Note and (iii) consents to any one or more extensions or postponements of time of payment, release, surrender or substitution of collateral security, or forbearance or other indulgence, without notice or consent. The pleading of any statute of limitations as a defense to any demand against any Borrower is hereby expressly waived by such Borrower.

endorsees hereof in such order and manner as the Holder may choose. None of the rights of the Holder shall be waived or diminished by any failure or delay in the exercise thereof.

- Covenants. Borrowers covenant and agree that, so long as this Note is outstanding and unpaid, Borrowers will punctually pay or cause to be paid the Principal Amount, premium, if any, and Combined Interest on this Note at the dates and places and in the manner specified herein. Any sums required to be withheld from any payment of principal, premium, if any, or Combined Interest on this Note by operation of law or pursuant to any order, judgment, execution, treaty, rule or regulation may be withheld by Borrowers and paid over in accordance therewith.
- Usury Laws. It is the intention of Borrowers and the Holder to conform strictly to all applicable usury laws now or hereafter in force, and any interest payable under this Note shall be subject to reduction to the amount not in excess of the maximum legal amount allowed under the applicable usury laws as now or hereafter construed by the courts having jurisdiction over such matters. If the maturity of this Note is accelerated by reason of an election by the Holder resulting from an Event of Default, voluntary prepayment by Borrowers or otherwise, then earned interest may never include more than the maximum amount permitted by law, statute, rule or regulation, computed from the date hereof until payment, and any interest in excess of the maximum amount permitted by law, statute, rule or regulation shall be canceled automatically and, if theretofore paid, shall at the option of the Holder hereof either Borrowers or credited on the principal amount of this Note, or if this Note has been paid, then the excess shall be rebated to Borrowers. The aggregate amount of all interest (whether designated as In-Kind Interest, Paid Interest, service charges, points or otherwise) contracted for, chargeable, or receivable under this Note shall under no circumstances exceed the maximum legal rate upon the unpaid Principal Amount of this Note remaining unpaid from time to time.
 - Prepayment.
 - Optional Prepayment. All or any portion of the outstanding Note may be prepaid by Borrowers in accordance with the terms hereof ("Optional Prepayment"). (a)
- (b) Mandatory Prepayment. Upon the occurrence of a Triggering Event (as defined below), the Holder, in its sole discretion, may require mandatory prepayment of the Note in full including the Principal Amount and unpaid interest at the Default Rate ("Mandatory Prepayment"). For purposes of this Note, a "Triggering Event" shall be defined as (i) a Change in Control of LMC (as defined in the Purchase Agreement); (ii) the sale of all or substantially all of LMC or its assets; (iii) an underwritten public offering of LMC's securities registered under the Securities Act of 1933, as amended, pursuant to which the aggregate gross proceeds to LMC are at least \$15,000,000; or (iv) the occurrence and continuation of an Event of Default as specifically provided for the Purchase Agreement.

3

- Prepayment. Borrowers may prepay the Principal Amount and unpaid In-Kind Interest at any time and from time to time, without premium or penalty; provided however, that (i) prior to the first anniversary of the execution and delivery of this Note, the prepayment price shall equal the outstanding Principal Amount being repaid, together with accrued but unpaid Paid Interest, plus a prepayment penalty of three percent (3%) of the outstanding Principal Amount, or (ii) prior to the second anniversary of the execution and delivery of this Note, the prepayment price shall equal the outstanding Principal Amount being repaid, together with accrued but unpaid Paid Interest, plus a prepayment penalty of two percent (2%) of the outstanding Principal Amount (the "Prepayment Price").
- Prepayment Notice. At least 30 days but not more than 60 days before a prepayment date, (ii) the prepayment date, (ii) the Prepayment Price, (iii) that the Note must be surrendered to Borrowers to collect the Prepayment Price and (iv) that interest on the Note called for prepayment ceases to accrue on and after the prepayment date. Once notice of prepayment is mailed, the Note called for prepayment becomes due and payable on the prepayment date.
 - Maturity.

To the extent this Note is not prepaid by Borrowers as provided in Section 6 hereof, the Principal Amount, together with accrued but unpaid Combined Interest, shall be due and payable on demand on the Maturity Date.

- 8 Other Provisions Relating to Rights of the Holder of this Note.
- Rights of the Holder of this Note. Except as otherwise provided by Sections 2.1 and 4.2 of the Purchase Agreement, this Note shall not entitle the Holder to any of the rights of a Member or Manager of any Borrower, including, without limitation, the right to vote, to receive distributions (except as specifically provided herein), or to receive any notice of, or to attend, meetings of Members of Managers or any other proceedings of any
- Lost, Stolen, Mutilated or Destroyed Note. If this Note shall be mutilated, lost, stolen, or destroyed, Borrowers shall execute and deliver, in exchange and substitution for and upon cancellation of a mutilated Note, or in lieu of or in substitution for a lost, stolen, or destroyed Note, a new Note for the principal amount of this Note so mutilated, lost, stolen, or destroyed but only upon receipt of evidence (which may consist of a signed affidavit of the Holder), of such loss, theft, or destruction of such Note, and of the ownership thereof, and indemnity, if requested, all reasonably satisfactory to Borrowers.
 - 9 Other Matters.
 - Binding Effect; Assignment. The provisions of this Note shall be binding upon and inure to the benefit of the Holder, Borrowers and their respective successors and assigns. (a)

- (b) <u>Further Actions</u>. At any time and from time to time, Borrowers and the Holder agree, without further consideration, to take such actions and to execute and deliver such documents as the other may reasonably mmate the transactions contemplated in this Note, including without limitation, the due execution of the Purchase Agreement and the execution and filing of financing statements with the appropriate filing offices resulting in perfected security interests
- Modification; Waiver. This Note and the Purchase Agreement set forth the entire understanding of Borrowers and the Holder with respect to the subject matter hereof and supersede all existing agreements between them concerning such subject matter. This Note may be amended, modified, superseded, canceled, renewed or extended, and the terms hereof may be waived, only by a written instrument signed by Borrowers and the Holder. Any waiver by Borrowers or the Holder of a breach of any provision of this Note shall not operate as or be construed to be a waiver of any other breach of such provision or of any breach of any other provision of this Note. The failure of Borrowers or the Holder to insist upon strict adherence to any term of this Note on one or more occasions shall not be considered a waiver or deprive that party of the right thereafter to insist upon strict adherence to that term or any other term of this Note. No delay on the part of any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof or hereof, nor shall any waiver on the part of any party of any right, power or privilege hereunder preclude any other or further exercise hereof or the exercise of any other right, power or privilege hereunder. Any waiver must be in writing. The rights and remedies provided herein are cumulative and are not exclusive of any rights or remedies which any party may otherwise have at law or in equity.
- Notices. All notices, demands or other communications to be given or delivered under or by reason of the provisions of this Note shall be in writing and shall be deemed to have been given when delivered personally to the recipient, sent to the recipient by credible of the recipient by credible to the reci

To I MC

Liquidmetal Coatings, LLC 30452 Esperanza Rancho Santa Margarita, California 92688 Attn: Legal Department Facsimile: (813) 314-0270

Liquidmetal Coatings Solutions, LLC Rancho Santa Margarita, California 92688

Attn: Legal Department Facsimile: (813) 314-0270

With a copy to:

Foley & Lardner LLP 100 North Tampa Street Suite 2700 Tampa, Florida 33602 Attn: Curt P. Creely Facsimile: (813) 221-4210 To the Holder:

 C_3 Capital Partners, L.P. C_3 Capital Partners II, L.P. 4520 Main Street, Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: (816) 756-5552

With a copy to:

Bryan Cave LLP 3500 One Kansas City Place 1200 Main Street Kansas City, Missouri 64105 Attn: Thomas W. Van Dyke Facsimile: (816) 374-3300

or to such other address or to the attention of such other person as the recipient party has specified by prior written notice to the sending party.

- (e) <u>Severability.</u> If any provision of this Note is invalid, illegal, or unenforceable, the balance of this Note shall remain in effect, and if any provision is inapplicable to any person or circumstance, it shall nevertheless remain applicable to all other persons and circumstances. The rate of interest on this Note is subject to any limitations imposed by applicable usury laws.
 - (f) Headings. The headings in this Note are solely for convenience of reference and shall be given no effect in the construction or interpretation of this Note.
- (g) Governing Law. This Agreement shall be governed by and construed in all respects under the laws of the State of Missouri, without reference to its conflict of laws rules or principles. Any suit, action, proceeding or litigation arising out of or relating to this Agreement shall be brought and prosecuted in such federal or state court or

6

courts located within the State of Missouri as provided by law. The parties hereby irrevocably and unconditionally consent to the jurisdiction of each such court or courts located within the State of Missouri and to service of process by registered or certified mail, return receipt requested, or by any other manner provided by applicable law, and hereby irrevocably and unconditionally waive any right to claim that any suit, action, proceeding or litigation so commenced has been commenced in an inconvenient forum.

(h) <u>Due Authorization</u>. The execution and delivery of this Note and the consummation of the transactions contemplated herein have been authorized by the Board of Managers of each Borrower and by any necessary vote or with the consent of the Members of each Borrower.

[Signatures on following page]

7

IN WITNESS WHEREOF, each Borrower has caused this Note to be executed on this 24th day of July, 2007.

LIOUIDMETAL COATINGS, LLC

By: Name: Title:

LIQUIDMETAL COATINGS SOLUTIONS, LLC

By: Name Title:

8

Exhibit A-2

THE SECURITIES REPRESENTED BY THIS INSTRUMENT WERE ISSUED ON JULY 24, 2007, AND HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THE TRANSFER OF THE SECURITIES REPRESENTED BY THIS INSTRUMENT IS SUBJECT TO THE CONDITIONS SPECIFIED IN THE SECURITIES PURCHASE AGREEMENT, DATED JULY 24, 2007 AND AS AMENDED AND MODIFIED FROM TIME TO TIME, AMONG LIQUIDMETAL COATINGS, LLC, AND C₃ CAPITAL PARTNERS, L.P. AND C₃ CAPITAL PARTNERS II, L.P. UPON WRITTEN REQUEST, A COPY OF SUCH CONDITIONS SHALL BE FURNISHED BY THE MAKER TO THE HOLDER HEREOF WITHOUT CHARGE.

14% SUBORDINATED NOTE

July 24, 2007 \$2,730,000

Liquidmetal Coatings, LLC, a Delaware limited liability company ("LMC"), and Liquidmetal Coatings Solutions, LLC ("LMCS." hereinafter jointly and severally with LMC, "Borrowers" and each individually, "Borrower"), hereby jointly and severally promise to pay to the order of C₃ Capital Partners II, L.P., a Delaware limited partnership ("C₃ Capital"), the principal amount of \$2,730,000 ("Principal Amount"), together with interest thereon calculated from the date hereof in accordance with the provisions of this Note.

This Note is issued pursuant to a Securities Purchase Agreement, dated as of July 24, 2007 (as amended and modified from time to time, the "Purchase Agreement"), among Borrowers, C₃ Capital and C₃ Capital Partners, L.P. This Note is one of the "Notes" referred to in the Purchase Agreement. The Purchase Agreement contains terms governing the rights of the holder of this Note, and all provisions of the Purchase Agreement are hereby incorporated herein in full by reference. Unless otherwise indicated herein, capitalized terms used in this Note have the same meanings set forth in the Purchase Agreement.

- 1. <u>Payment of Interest</u>. Except as otherwise expressly provided in the Purchase Agreement,
- (a) In-Kind Interest. Interest shall accrue at a rate of 2% per annum (the "In-Kind Interest," which shall be computed on the basis of a 360-day year for the actual number of days elapsed in any year) on the unpaid Principal Amount, applied on a monthly basis to the Principal Amount, and paid, together with the Principal Amount in accordance with Section 2 or Section 4(c), as the case may be.
- (b) Paid Interest. Interest shall accrue at a rate of 12% per annum ("Paid Interest," which shall be computed on the basis of a 360-day year for the actual number of days elapsed in any year) on the unpaid Principal Amount of this Note and shall be paid on the first day of each calendar month beginning September 1, 2007. Unless prohibited under applicable law, any amount of accrued interest which is not paid pursuant to this Section 1(b) on the date on which it is due and payable shall bear interest at a rate of 12% per annum, or as otherwise specified in the Purchase Agreement, until such interest is paid to the holder of this Note. Any accrued interest which for any reason has not theretofore been paid during the term of this Note shall be paid in full on the date on which the final payment of the Principal Amount is made. (In-Kind Interest and Paid Interest, collectively, shall be referred to herein as "Combined Interest".)
 - (c) Form of Interest Payment. Payments of Combined Interest shall be made in lawful money of the United States of America and in immediately available

1

- Payment of the Principal Amount on the Note.
- (a) Schedule of Principal Payments. Borrowers shall pay the unpaid Principal Amount to the Holder on July 20, 2012 ("Maturity Date"), together with all accrued and unpaid Combined Interest on the Principal Amount being repaid.
- (b) Form of Principal Payments. Payments of principal and interest shall be made in lawful money of the United States of America and in immediately available funds by automatic bank draft pursuant to the instructions attached hereto as Schedule A, or at such other place as the Holder may specify in writing.
 - 3. Default. The occurrence of any one or more of the Events of Default (as defined in the Purchase Agreement) shall constitute an Event of Default hereunder.
- (a) <u>Default Rate</u>. Upon the occurrence and during the continuance of a default or an Event of Default by Borrowers under this Note or the Purchase Agreement, all amounts due and owing under the Note shall bear interest at a rate equal to three percent (3%) per annum greater than the then existing applicable Combined Interest rate on this Note ("Default Rate").
- (b) Insolvency or Regulatory Default. If any Event of Default contemplated by Section 8.1(d) or 8.1(n) of the Purchase Agreement shall occur for any reason, then in any such event, in addition to all rights and remedies of the Holder under the Purchase Agreement, applicable law or otherwise, all such rights and remedies being cumulative, not exclusive and enforceable alternatively, successively and concurrently, the Holder may, at its option, declare any or all of the Borrowers' obligations, liabilities and indebtedness owing to the Holder under the Purchase Agreement and all amounts owing under this Note, to be due and payable, whereupon the then unpaid balance hereof, together with all Combined Interest accrued thereon, shall forthwith become due and payable, together with interest accruing thereafter at the then applicable interest rate stated above until the indebtedness evidenced by this Note is paid in full, plus the costs and expenses of collection hereof, including, but not limited to, reasonable attorneys' fees and legal expenses.
- (c) <u>Waiver</u>. Each Borrower (i) waives diligence, demand, presentment, protest and notice of any kind, (ii) agrees that it will not be necessary for the Holder to first institute suit in order to enforce payment of this Note and (iii) consents to any one or more extensions or postponements of time of payment, release, surrender or substitution of collateral security, or forbearance or other indulgence, without notice or consent. The pleading of any statute of limitations as a defense to any demand against any Borrower is hereby expressly waived by such Borrower.
 - (d) Certain Obligors. The Holder shall not be required to resort to any Collateral for payment, but may proceed against any Borrower and any guarantors or

2

endorsees hereof in such order and manner as the Holder may choose. None of the rights of the Holder shall be waived or diminished by any failure or delay in the exercise thereof.

- 4. <u>Covenants.</u> Borrowers covenant and agree that, so long as this Note is outstanding and unpaid, Borrowers will punctually pay or cause to be paid the Principal Amount, premium, if any, and Combined Interest on this Note at the dates and places and in the manner specified herein. Any sums required to be withheld from any payment of principal, premium, if any, or Combined Interest on this Note by operation of law or pursuant to any order, judgment, execution, treaty, rule or regulation may be withheld by Borrowers and paid over in accordance therewith.
- 5. <u>Usury Laws</u>. It is the intention of Borrowers and the Holder to conform strictly to all applicable usury laws now or hereafter in force, and any interest payable under this Note shall be subject to reduction to the amount not in excess of the maximum legal amount allowed under the applicable usury laws as now or hereafter construed by the courts having jurisdiction over such matters. If the maturity of this Note is accelerated by reason of an election by the Holder resulting from an Event of Default, voluntary prepayment by Borrowers or otherwise, then earned interest may never include more than the maximum amount permitted by law, statute, rule or regulation, computed from the date hereof until payment, and any interest in excess of the maximum amount permitted by law, statute, rule or regulation shall be canceled automatically and, if theretofore paid, shall at the option of the Holder hereof either be rebated to the Company or credited on the principal amount of this Note, or if this Note has been paid, then the excess shall be rebated to Borrowers. The aggregate amount of all interest (whether designated as In-Kind Interest, service charges, points or otherwise) contracted for, chargeable, or receivable under this Note shall under no circumstances exceed the maximum legal rate upon the unpaid Principal Amount of this Note remaining unpaid from time to time.
 - 6. Prepayment.
 - (a) Optional Prepayment. All or any portion of the outstanding Note may be prepaid by Borrowers in accordance with the terms hereof ("Optional Prepayment").
- (b) Mandatory Prepayment. Upon the occurrence of a Triggering Event (as defined below), the Holder, in its sole discretion, may require mandatory prepayment of the Note in full including the Principal Amount and unpaid interest at the Default Rate ("Mandatory Prepayment"). For purposes of this Note, a "Triggering Event" shall be defined as (i) a Change in Control of LMC (as defined in the Purchase Agreement); (ii) the sale of all or substantially all of LMC or its assets; (iii) an underwritten public offering of the LMC's securities registered under the Securities Act of 1933, as amended, pursuant to which the aggregate gross proceeds to LMC are at least \$15,000,000; or (iv) the occurrence and continuation of an Event of Default as specifically provided for the Purchase Agreement.

3

- (c) Prepayment. Borrowers may prepay the Principal Amount and unpaid In-Kind Interest at any time and from time to time, without premium or penalty; provided however, that (i) prior to the first anniversary of the execution and delivery of this Note, the prepayment price shall equal the outstanding Principal Amount being repaid, together with accrued but unpaid Paid Interest, plus a prepayment penalty of three percent (3%) of the outstanding Principal Amount, or (ii) prior to the second anniversary of the execution and delivery of this Note, the prepayment price shall equal the outstanding Principal Amount being repaid, together with accrued but unpaid Paid Interest, plus a prepayment penalty of two percent (2%) of the outstanding Principal Amount (the "Prepayment Price").
- (d) <u>Prepayment Notice</u>. At least 30 days but not more than 60 days before a prepayment date, Borrowers shall mail a notice of prepayment to the Holder. The notice shall state (i) the prepayment date, (ii) the Prepayment Price, (iii) that the Note must be surrendered to Borrowers to collect the Prepayment Price and (iv) that interest on the Note called for prepayment ceases to accrue on and after the prepayment date. Once notice of prepayment is mailed, the Note called for prepayment becomes due and payable on the prepayment date.
 - 7. <u>Maturity</u>

To the extent this Note is not prepaid by Borrowers as provided in Section 6 hereof, the Principal Amount, together with accrued but unpaid Combined Interest, shall be due and payable on demand on the Maturity Date.

- 8. Other Provisions Relating to Rights of the Holder of this Note.
- (a) Rights of the Holder of this Note. Except as otherwise provided by Sections 2.1 and 4.2 of the Purchase Agreement, this Note shall not entitle the Holder to any of the rights of a Member or Manager of any Borrower, including, without limitation, the right to vote, to receive distributions (except as specifically provided herein), or to receive any notice of, or to attend, meetings of Members of Managers or any other proceedings of any
- (b) <u>Lost, Stolen, Mutilated or Destroyed Note</u>. If this Note shall be mutilated, lost, stolen, or destroyed, Borrowers shall execute and deliver, in exchange and substitution for and upon cancellation of a mutilated Note, or in lieu of or in substitution for a lost, stolen, or destroyed Note, a new Note for the principal amount of this Note so mutilated, lost, stolen, or destroyed but only upon receipt of evidence (which may consist of a signed affidavit of the Holder), of such loss, theft, or destruction of such Note, and of the ownership thereof, and indemnity, if requested, all reasonably satisfactory to the Borrower.
 - 9. Other Matters.
 - (a) Binding Effect; Assignment. The provisions of this Note shall be binding upon and inure to the benefit of the Holder, Borrowers and their respective successors and assigns.

4

- (b) Further Actions. At any time and from time to time, Borrowers and the Holder agree, without further consideration, to take such actions and to execute and deliver such documents as the other may reasonably request to consummate the transactions contemplated in this Note, including without limitation, the due execution of the Purchase Agreement and the execution and filing of financing statements with the appropriate filing offices resulting in perfected security interests.
- (c) Modification: Waiver. This Note and the Purchase Agreement set forth the entire understanding of Borrowers and the Holder with respect to the subject matter hereof and supersede all existing agreements between them concerning such subject matter. This Note may be amended, modified, superseded, canceled, renewed or extended, and the terms hereof may be waived, only by a written instrument signed by Borrowers and the Holder. Any waiver by Borrowers or the Holder of a breach of any provision of this Note shall not operate as or be construed to be a waiver of any other breach of such provision or of any other provision of this Note. The failure of Borrowers or the Holder to insist upon strict adherence to any term of this Note on one or more occasions shall not be considered a waiver or deprive that party of the right thereafter to insist upon strict adherence to that term or any other term of this Note. No delay on the part of any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof or hereof, nor shall any waiver on the part of any party of any right, power or privilege hereunder preclude any other or further exercise hereof or the exercise of any other right, power or privilege hereunder. Any waiver must be in writing. The rights and remedies provided herein are cumulative and are not exclusive of any rights or remedies which any party may otherwise have at law or in equity.
- (d) Notices. All notices, demands or other communications to be given or delivered under or by reason of the provisions of this Note shall be in writing and shall be deemed to have been given when delivered personally to the recipient, sent to the recipient by reputable overnight courier service (charges prepaid), mailed to the recipient by certified or registered mail, return receipt requested and postage prepaid or sent via facsimile to the number set forth below with a copy mailed to the recipient as set forth above. Such notices, demands and other communications shall be sent to the addresses indicated below:

To LMC:

5

Attn: Legal Department Facsimile: (813) 314-0270

With a copy to:

Foley & Lardner LLP 100 North Tampa Street Suite 2700 Tampa, Florida 33602 Attn: Curt P. Creely Facsimile: (813) 221-4210

To the Holder:

 C_3 Capital Partners, L.P. C_3 Capital Partners II, L.P. 4520 Main Street, Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: (816) 756-5552

with a copy to:

Bryan Cave LLP 3500 One Kansas City Place 1200 Main Street Kansas City, Missouri 64105 Attn: Thomas W. Van Dyke Facsimile: (816) 374-3300

or to such other address or to the attention of such other person as the recipient party has specified by prior written notice to the sending party.

- (e) Severability. If any provision of this Note is invalid, illegal, or unenforceable, the balance of this Note shall remain in effect, and if any provision is inapplicable to any person or circumstance, it shall nevertheless remain applicable to all other persons and circumstances. The rate of interest on this Note is subject to any limitations imposed by applicable usury laws.
 - (f) Headings. The headings in this Note are solely for convenience of reference and shall be given no effect in the construction or interpretation of this Note.
- (g) Governing Law. This Agreement shall be governed by and construed in all respects under the laws of the State of Missouri, without reference to its conflict of laws rules or principles. Any suit, action, proceeding or litigation arising out of or relating to this Agreement shall be brought and prosecuted in such federal or state court or courts located within the State of Missouri as provided by law. The parties hereby irrevocably and unconditionally consent to the jurisdiction of each such court or

6

courts located within the State of Missouri and to service of process by registered or certified mail, return receipt requested, or by any other manner provided by applicable law, and hereby irrevocably and unconditionally waive any right to claim that any suit, action, proceeding or litigation so commenced has been commenced in an inconvenient forum.

(h) <u>Due Authorization.</u> The execution and delivery of this Note and the consummation of the transactions contemplated herein have been authorized by the Board of Managers of each Borrower and by any necessary vote or with the consent of the Members of each Borrower.

[Signatures on following page]

7

IN WITNESS WHEREOF, each Borrower has caused this Note to be executed on this 24th day of July, 2007.

LIQUIDMETAL COATINGS, LLC

By: Name: Title:

LIOUIDMETAL COATINGS

SOLUTIONS, LLC

By:

Name: Title:

8

Exhibit B-1

OPERATING AGREEMENT

OF

LIQUIDMETAL COATINGS, LLC

A Delaware Limited Liability Company

Adopted as of July 24, 2007

TABLE OF CONTENTS

ARTICLE I Section 1.1

Section 1.2 Section 1.3	PRINCIPAL PLACE OF BUSINESS REGISTERED OFFICE AND REGISTERED AGENT	1 1
Section 1.4	TERM	1
Section 1.5	NO STATE LAW PARTNERSHIP	1
ARTICLE II	PURPOSE AND POWERS OF THE COMPANY	2
Section 2.1	PURPOSE	2
Section 2.2	POWERS OF THE COMPANY	2
ARTICLE III	CAPITAL STRUCTURE, CONTRIBUTIONS TO CAPITAL AND CAPITAL ACCOUNTS	2
Section 3.1	INITIAL CONTRIBUTIONS	2
Section 3.2	ADDITIONAL CAPITAL CONTRIBUTIONS	2
Section 3.3 Section 3.4	RETURN OF CONTRIBUTIONS CAPITAL ACCOUNTS	2 2
Section 3.5	CAPITAL STRUCTURES AND PERCENTAGE INTERESTS	3
Section 3.6	LOANS BY MEMBERS	4
Section 3.7	PROFITS INTEREST OF CLASS C HOLDERS	4
ARTICLE IV	ALLOCATIONS OF TAXABLE PROFITS AND LOSSES	4
Section 4.1 Section 4.2	DETERMINATION OF PROFIT OR LOSS	4
Section 4.2 Section 4.3	COSTS AND EXPENSES ALLOCATION	5
Section 4.4	SPECIAL ALLOCATIONS	5
Section 4.5 Section 4.6	BUILT-IN GAIN OR LOSS INCOME CHARACTERIZATION	7 7
Section 4.7	CREDITS	7
Section 4.8	CHANGE IN INTERESTS	7
Section 4.9	CREDITING ACCOUNTS	8
ARTICLE V	DISTRIBUTIONS	8
Section 5.1	DISTRIBUTABLE AMOUNTS	8
Section 5.2 Section 5.3	ALLOCATION LIQUIDATING DISTRIBUTIONS	8
Section 5.4	MANDATORY TAX DISTRIBUTIONS	9
ADTICLE VI	NUMBER OF THE PROPERTY OF THE	
ARTICLE VI Section 6.1	DURATION OF BUSINESS; TERMINATION DURATION OF COMPANY	9 9
Section 6.2	DEATH, ETC., OF MEMBER	9
Section 6.3	LIQUIDATION	9
ARTICLE VII	RIGHTS AND DUTIES OF MEMBERS	10
Section 7.1	LIABILITIES OF MEMBERS	10
	i	
Section 7.2	LIMITATIONS ON POWERS OF MEMBERS	10
Section 7.3	PROHIBITION AGAINST PARTITION	10
ARTICLE VIII	MEETINGS OF MEMBERS	10
Section 8.1	ANNUAL MEETING ANNUAL MEETING	10
Section 8.2	SPECIAL MEETING	11
Section 8.3 Section 8.4	PLACE OF MEETING NOTICE OF MEETING	11 11
Section 8.5	WAIVER OF NOTICE	11
Section 8.6	FIXING OF RECORD DATE	11
Section 8.7 Section 8.8	QUORUM PROXIES	11 11
Section 8.9	VOTING OF MEMBERSHIP UNITS	12
Section 8.10	VOTING OF MEMBERSHIP UNITS BY CERTAIN HOLDERS	12
Section 8.11 Section 8.12	ACTION BY WRITTEN CONSENT OR TELEPHONE CONFERENCE ANNUAL REPORTS, ETC.	12 13
Section 0.12	AND ALTONIO, ETC.	15
ARTICLE IX	BOARD OF MANAGERS	13
Section 9.1 Section 9.2	MANAGEMENT AND CONTROL NUMBER AND ELECTION	13 13
Section 9.3	RESIGNATION	13
Section 9.4	REMOVAL OF MANAGER	14
Section 9.5 Section 9.6	EXPRESSLY AUTHORIZED RIGHTS AND POWERS CERTAIN LIMITATIONS	14 14
Section 9.7	MEETINGS OF MANAGERS	15
Section 9.8	COMPENSATION	16
ARTICLE X	OFFICERS	16
Section 10.1	NUMBER	16
Section 10.2	ELECTION AND TERM OF OFFICE	17
Section 10.3 Section 10.4	REMOVAL AND RESIGNATION VACANCIES	17 17
Section 10.5	CHAIR	17
Section 10.6	PRESIDENT WAGE INDESTRUCTOR	17
Section 10.7 Section 10.8	VICE PRESIDENTS SECRETARY	17 17
Section 10.9	TREASURER	18
Section 10.10	SALARIES	18
ARTICLE XI	INDEMNIFICATION OF MEMBERS, MANAGERS AND DIRECTORS	18
Section 11.1	RIGHT TO INDEMNIFICATION	18
Section 11.2 Section 11.3	ADVANCES SAVINGS CLAUSE	18 18
11.0		10
	ii	
ARTICLE XII	ASSIGNMENT OF MEMBERSHIP UNITS	19
Section 12.1 Section 12.2	GENERAL RESTRICTION ON ASSIGNMENT RIGHT OF FIRST REFUSAL	19 20
Section 12.3	INVOLUNTARY TRANSFER	21
Section 12.4 Section 12.5	TERMINATION OF EMPLOYMENT WITHDRAWALS	22 23
Section 12.5 Section 12.6	RIGHTS OF ASSIGNEE OF MEMBERSHIP UNITS	23
Section 12.7	SUBSTITUTED MEMBER	23
Section 12.8 Section 12.9	SECURITIES LAWS INVALID TRANSFER	24 24
Section 12.10	DISTRIBUTIONS AND ALLOCATIONS IN RESPECT OF A TRANSFERRED INTEREST	24
Section 12.11	DRAG-ALONG RIGHTS	24
Section 12.12	APPLICABILITY OF PROVISIONS TO CLASS A HOLDERS	25

Section 12.8 Section 12.9 Section 12.10 Section 12.11 Section 12.12 SECURITIES LAWS
INVALID TRANSFER
DISTRIBUTIONS AND ALLOCATIONS IN RESPECT OF A TRANSFERRED INTEREST
DRAG-ALONG RIGHTS
APPLICABILITY OF PROVISIONS TO CLASS A HOLDERS ARTICLE XIII CONVERSION TO CORPORATE SOLUTION

26

ARTICLE XIV	BOOKS AND RECORDS	26
Section 14.1	BOOKS AND RECORDS	26
Section 14.2	CUSTODY OF MEMBER FUNDS; BANK ACCOUNTS	27
Section 14.3	ACCOUNTANTS	27
Section 14.4	SECTION 754 ELECTION	27
Section 14.5	FISCAL YEAR	27
Section 14.6	TAX MATTERS PARTNER	27
Section 14.7	ANNUAL FINANCIAL STATEMENTS AND TAX RETURN INFORMATION	27
Section 14.8	MONTHLY FINANCIAL STATEMENTS	28
Section 14.9	INSPECTION RIGHTS	28
ARTICLE XV	EMPLOYMENT AGREEMENTS	28
ARTICLE XVI Section 16.1	DEFINITIONS DEFINITIONS	28 28
ARTICLE XVII	AGREEMENT PREPARED BY ATTORNEY FOR COMPANY	33
ARTICLE XVIII	AMENDMENTS	33
ARTICLE XIX	MANAGER AND MEMBER RELATIONS TO THE COMPANY	33
ARTICLE XX Section 20.1 Section 20.2	CHOICE OF LAW; SUBMISSION TO JURISDICTION; AND WAIVER OF JURY TRIAL LAW WAIVER OF JURY TRIAL	34 34 34

iii

ARTICLE XXI	MISCELLANEOUS	34
Section 21.1	DELAY OR OMISSIONS	34
Section 21.2	WAIVER	34
Section 21.3	WAIVER OF PARTITION	34
Section 21.4	SUCCESSORS AND ASSIGNS	34
Section 21.5	NOTICES	35
Section 21.6	ENTIRE AGREEMENT	35
Section 21.7	PRONOUNS	35
Section 21.8	TITLES AND SUBTITLES	35
Section 21.9	SEVERABILITY	36
Section 21.10	BINDING EFFECT	30
Section 21.11	CREDITORS	30
Section 21.12	EXECUTION OF ADDITIONAL INSTRUMENTS	30
Section 21.13	RIGHTS AND REMEDIES CUMULATIVE	30
Section 21.14	COUNTERPARTS	30

OPERATING AGREEMENT

įν

OF

LIQUIDMETAL COATINGS, LLC

THIS OPERATING AGREEMENT ("Agreement") is hereby entered into effective as of the 24th day of July, 2007, by and between the persons identified as Members on Exhibit "A" attached hereto and executing this Agreement on the signature pages hereof (each of whom is sometimes hereinafter referred to individually as a "Member" and collectively as "Members") and LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company (the "Company").

RECITALS

The Company was organized in accordance with the Delaware Limited Liability Company Act upon the filing of a Certificate of Formation with the Secretary of State of the State of Delaware effective July 9, 2007. The parties hereby provide for the Company's capitalization and management and its affairs and the conduct of its business.

NOW, THEREFORE, in consideration of the mutual promises set forth herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

ARTICLE I FORMATION AND ORGANIZATION

- Section 1.1 NAME AND FORMATION. The name of the Company is: LIQUIDMETAL COATINGS, LLC. All Company business shall be conducted in the name of "LIQUIDMETAL COATINGS, LLC" or such other names that comply with applicable law as the Board of Managers may select from time to time.
- **Section 1.2** PRINCIPAL PLACE OF BUSINESS. The initial principal office of the Company will be located at 30452 Esperanza, Rancho Santa Margarita, California 92688, and may be changed to such other place within or without the State of Delaware as may be determined from time to time by the Board of Managers. It is contemplated that the initial principal office will move to the Houston, Texas metropolitan area when a location for such office is leased in that area.
- Section 1.3 REGISTERED OFFICE AND REGISTERED AGENT. The Company's initial registered agent and office shall be CorpDirect Agents, Inc., 615 South DuPont Highway, Kent County, Dover, Delaware 19901. The Company may change its registered agent or registered office to any other place or places in the State of Delaware as may be determined from time to time by the Board of Managers.
 - Section 1.4 TERM. The term of the Company shall be perpetual, unless the Company is dissolved in accordance with the provisions of this Agreement.
 - Section 1.5 NO STATE LAW PARTNERSHIP. The Members intend that the Company (i) shall be taxed as a partnership for all applicable federal, state and local income tax

purposes and (ii) shall not be a partnership or joint venture for any other purpose, and that no Member or any Manager shall, by virtue of this Agreement, be a partner or joint venturer of any other Member or Manager.

ARTICLE II PURPOSE AND POWERS OF THE COMPANY

- Section 2.1 PURPOSE. The Company is formed for the purpose of manufacturing and applying metallic coating to various products and engaging in any other business activity permitted under the Act as the Board of Managers may determine from time to time.
- Section 2.2 POWERS OF THE COMPANY. Subject to the provisions of this Agreement, the Company shall have the power and authority to take any and all actions necessary, appropriate, proper, advisable, incidental or convenient to or for the furtherance of the purposes set forth in Section 2.1, and all powers reasonably connected with such activities and businesses that may be legally exercised by limited liability companies under the Act, and may engage in all activities necessary, customary, convenient, or incident to any of the foregoing.

ARTICLE III <u>CAPITAL STRUCTURE, CONTRIBUTIONS TO CAPITAL</u> <u>AND CAPITAL ACCOUNTS</u>

- Section 3.1 INITIAL CONTRIBUTIONS. The Members have previously contributed, or shall contribute simultaneously with the execution hereof, capital to the Company in accordance with the schedule attached hereto as Exhibit "A" (the "Initial Capital Contributions") in exchange for their respective Membership Units.
- Section 3.2 <u>ADDITIONAL CAPITAL CONTRIBUTIONS</u>. The Member may make additional contributions of capital to the Company as the Member determines are necessary, appropriate or desirable; provided, however, that the Member shall have no obligation to contribute any additional capital to the Company, and except as set forth in the Act, the Member shall have no personal liability for any obligations of the Company.
- Section 3.3 <u>RETURN OF CONTRIBUTIONS</u>. Until the dissolution and liquidation of the Company, no Member shall have the right to demand or receive any part of the Member's capital contribution and there is no right given to any Member to demand and receive property other than cash in return for the Member's capital contribution.

Section 3.4 CAPITAL ACCOUNTS. An individual Capital Account shall be maintained for each Member in accordance with Section 704(b) of the Code, paragraph 1.704-1(b)(2)(iv) of the accompanying Treasury Regulations, and the following rules:

(a) Computation of Capital Account Balance. The Capital Account of a Member shall consist of the amount of money and the fair market value of any property (other than money) comprising the Member's proportionate share of the Initial Capital Contribution pursuant to Section 3.1 hereof, as increased by: (i) the amount of money and the fair market value of any property (other than money) comprising any additional capital contributions made by the Member, (ii) any amount credited to the Capital Account of a Member pursuant to Section 4.9 hereof as a result of any Company income, profits or gains

2

allocated to the Member (and as adjusted pursuant to Section 1.704-1(b)(2)(iv) of the Treasury Regulations), and (iii) the amount of any Company liabilities assumed by the Member or that are secured by any Company property distributed to that Member, and decreased by: (i) the amount of money and the fair market value of any property (other than money) comprising any distributions to the Member pursuant to Article V hereof, (ii) any amount debited to the Capital Account of a Member gursuant to Section 4.9 hereof as a result of any Company expenses, deductions, losses and credits allocated to the Member (and as adjusted pursuant to Section 1.704-1(b)(2)(iv) of the Treasury Regulations), and (iii) the amount of any liabilities of such Member that are assumed by the Company or that are secured by any property contributed by that Member to the Company.

- (b) Built-In Gain or Loss. The Capital Account of a Member shall not be increased or decreased, as the case may be, with regard to any built-in gain or loss allocated to the Member pursuant to Section 4.5 hereof.
- (c) Transferee's Capital Account. In the event of a transfer of any Membership Units, the transferee shall assume the Capital Account balance of the transferor.
- (d) <u>Interest</u>. No interest shall be paid on any present or future Capital Account balance.
- (e) Conformance with Regulations. The provisions of this Section 3.4 are intended to comply with Treasury Regulation Section 1.704-1(b) regarding the maintenance of the Capital Accounts of the Members and this Section 3.4 shall be interpreted and applied in a manner consistent with such Treasury Regulations. In the event that the Board of Managers shall determine that it is prudent to modify the manner in which the Capital Accounts, or any debits or credits thereto, are computed in order to comply with such Treasury Regulations, the Board of Managers may make such modifications, provided that it is not likely to have a material effect on any amounts distributable to any Member upon the dissolution of the Company. The Board of Managers shall also make any appropriate modifications in the event unanticipated events might otherwise cause this Agreement not to comply with Treasury Regulation Section 1.704-1(b).

Section 3.5 CAPITAL STRUCTURES AND PERCENTAGE INTERESTS

- (a) <u>Capital Structure</u>. Ownership rights in the Company are divided into and represented by membership units ("<u>Membership Units</u>"). The Membership Units are further divided into three classes: (i) Class A Units; (ii) Class B Units; and (iii) Class C Units.
- (b) Initial Percentage Interest. Upon their admission to the Company, the Members shall own the number and class of Membership Units and the corresponding Percentage Interests in the Company as set forth on Exhibit "A" attached hereto.
- (c) Adjustments to Percentage Interests. In the event of the issuance of any additional Membership Units to an existing Member(s), including the admission of an additional member(s), each Member's Percentage Interest in the Company shall be deemed to be equal to a fraction, the numerator of which is the number of Membership Units issued and outstanding to the Member, and the denominator of which is the number of Membership Units then issued and outstanding to all Members. All adjustments to a Member's

3

Percentage Interest in the Company shall be permanent and shall not otherwise be altered except as provided in this Agreement

- (d) Effect of Percentage Interests. The Percentage Interests of the Members, as determined and adjusted pursuant to this Section 3.5, are maintained solely for the purposes of determining the amount of certain allocations of taxable profits and losses and cash distributions allocable to the Members pursuant to Articles IV and V hereof, and for other purposes set forth in this Agreement. The Percentage Interest of a Member shall not reflect that Member's proportionate interest in the capital of the Company at any time.
- Section 3.6 LOANS BY MEMBERS. A Member may at any time lend funds to the Company as may be agreed upon by the Board of Managers. Such funds shall represent a debt, payable on demand, unless otherwise specifically provided, from the Company to the Member making the loan, and interest, at a rate agreed upon by the Board of Managers, shall be paid thereon and charged as an expense to the Company.
- Section 3.7 PROFITS INTEREST OF CLASS C HOLDERS. The Class C Holders shall own a percentage of the profits of the Company equal to the right to receive their respective percentage interest as shown on Exhibit "A", attached hereto, of the available cash distributed among the Members pursuant to Section 5.2 and Section 5.3. In addition, taxable income shall be allocated to the Class C Holders equal to their respective percentage interest as shown on Exhibit "A", attached hereto, of the Net Profits allocated among the Members pursuant to Section 4.3(a)(ii).

It is the agreement of the Members that the rights of the Class C Holders under this Section 3.7 shall be limited to receiving allocations of taxable income and distributions of cash as described herein, but that the Class C Holders shall not receive any other rights in and shall not participate in any other allocations of taxable income, loss, credit, deduction, gain, or other tax items or distributions of cash from the Company except as specifically provided in this Section 3.7.

ARTICLE IV ALLOCATIONS OF TAXABLE PROFITS AND LOSSES

- Section 4.1 DETERMINATION OF PROFIT OR LOSS. The items of income, gains, expenses, deductions, losses and credits generated by the Company for federal income tax purposes shall be determined in accordance with a generally accepted method of accounting as soon as practicable after the close of the fiscal year of the Company.
- Section 4.2 COSTS AND EXPENSES. The Company shall pay all expenses (which expenses shall be billed directly to the Company) of the Company which may include but are not limited to: (i) legal, audit, accounting, and other fees; (ii) expenses and taxes incurred in connection with the issuance, distribution and transfer of documents evidencing ownership of Membership Units in the Company or in connection with the business of the Company; (iii) expenses of organizing, revising, amending, converting, modifying or terminating the Company; (iv) expenses in connection with distributions made by the Company to, and communications and bookkeeping work necessary in maintaining relations with, the Members; and (v) costs of any accounting, statistical or bookkeeping equipment necessary for the maintenance of the books and records of the Company.

4

- Section 4.3 ALLOCATION. The net profits, net gains and net losses generated by the Company for federal income tax purposes for a year shall be allocated among the Members as follows:
 - (a) <u>Profits</u>. Any net taxable income and gain (collectively, "Net Profits") of the Company for the year shall be allocated among the Members as follows:
 - (i) First, 100% of the Net Profits shall be allocated pro rata (relative to the entire amount to be allocated to all Members in that year pursuant to this Section 4.3(a)(i)) among the Members in an amount equal to the excess, if any, of (a) the cumulative Losses allocated pursuant to Section 4.3(a)(i) for all prior years, until such time as each Member has been allocated an aggregate amount of the Net Profits in all years pursuant to this Section 4.3(a)(i) equal to the aggregate amount of Losses allocated to the Member pursuant to Section 4.3(b) hereof in all prior years; and
 - (ii) Any remaining Net Profits shall be allocated among the Members according to their respective Percentage Interests in the Company at that time, as set forth in Section 3.5 hereof.
 - (b) Losses. Any net taxable losses and deductions (collectively, "Losses") of the Company for the year shall be allocated among the Members in the following order of priority:
 - (i) First, all Losses shall be allocated among the Members in proportion to the positive balances in their respective Capital Accounts at that time, after a reduction for any cash distributed from the Company to the Members pursuant to Article V in that year, until the balance of all Members' Capital Accounts have been reduced to zero (0); and
 - (ii) Any remaining Losses shall be allocated among the Members according to their respective Percentage Interests in the Company at that time, as set forth in Section 3.5 hereof.
- Section 4.4 SPECIAL ALLOCATIONS. Notwithstanding any other provision of this Agreement to the contrary, the following special allocations shall be made in the following order:
- (a) Minimum Gain Chargeback. Notwithstanding any other provision of this Section 4.4, if there is a net decrease in the Partnership Minimum Gain (as described in Article XVI hereof) during any Company fiscal year, each Member will be specially allocated, before any other allocation is made, items of income and gain for such year (and, if necessary, subsequent years) in an amount equal to that Member's share of the net decrease in Partnership Minimum Gain, determined as provided in Section 1.704-2(g)(2) of the Regulations.

Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be allocated shall be determined in accordance with Regulations Section 1.704-2(f), and this Section

(b) Partner Minimum Gain Chargeback. Notwithstanding any other provision of this Article IV except Section 4.4(a), above, if there is a net decrease in Partner Minimum Gain (as described in Article XVI hereof) attributable to a Partner Nonrecourse Debt (as described in Article XVI hereof) during any Company fiscal year, each Member who has a share of the Partner Minimum Gain attributable to such Partner Nonrecourse Debt (determined as provided in Section 1.704-2(i)(5) of the Regulations) as of the beginning of the year shall be specially allocated items of Company income and gain for such year (and, if necessary, subsequent years) in an amount equal to that Member's share of the net decrease in Partner Minimum Gain attributable to such Partner Nonrecourse Debt, as provided in Section 1.704-2(i)(4) of the Regulations.

Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with Section 1.704-2(i)(4) of the Regulations. This Section 4.4(b) is intended to comply with the minimum gain chargeback requirement in that Section of the Regulations and shall be interpreted consistently therewith.

- (c) Qualified Income Offset. In the event any Member unexpectedly receives an adjustment, allocation or distribution of a nature described in Treasury Regulations Sections 1.704-1(b)(2)(ii)(d)(4), (5) or (6), then each such Member will be specially allocated items of Company taxable income and gain in an amount and manner sufficient to eliminate, to the extent required by the Regulations, that Member's Deficit Capital Account Balance as quickly as possible; provided that an allocation pursuant to this Section 4.4(c) shall be made if and only to the extent that such Member would have a Deficit Capital Account Balance after all other allocations provided for in this Article IV have been tentatively made as if this Section 4.4(c) were not in this Agreement.
- (d) Partner Nonrecourse Deductions. Any Partner Nonrecourse Deductions (as described in Article XVI hereof) for any fiscal year or other period shall be specially allocated to the Member who bears the economic risk of loss with respect to the Partner Nonrecourse Debt (as described in Article XVI hereof) to which such Partner Nonrecourse Deductions are attributable in accordance with Section 1.704-2(i)(2) of the Regulations.
- (e) Curative Allocations. The allocations set forth in this Section 4.4 (the "Regulatory Allocations") are intended to comply with certain requirements of Regulations Sections 1.704-1(b) and 1.704-2.

 Notwithstanding any other provision of this Article IV (other than the Regulatory Allocations), the Regulatory Allocations shall be taken into account in allocating other items of profits, losses, income, gains and deductions among the Members so that, to the extent possible, the net amount of such allocations of other taxable items and the Regulatory Allocations to each Member shall be equal to the net amount which would have been allocated to each such Member if the Regulatory Allocations had not occurred. For purposes of applying the previous sentence:
 - (i) No allocations of Nonrecourse Deductions shall be made pursuant to this Section 4.4(e) with respect to Regulatory Allocations required pursuant to Section 4.4(a) hereof prior to the Company fiscal year during which there is a net

6

decrease in Partnership Minimum Gain, and then only to the extent necessary to avoid any potential economic distortions caused by such net decrease in Partnership Minimum Gain;

- (ii) Allocations pursuant to this Section 4.4(e) shall be deferred with respect to allocations of Nonrecourse Deductions to the extent the Board of Managers determine that such allocations are likely to be offset by subsequent allocations pursuant to subsection (a) of Section 4.4 hereof;
- (iii) No allocations of Partner Nonrecourse Deductions shall be made pursuant to this Section 4.4(e) with respect to Regulatory Allocations required under subsection (b) or (d) of this Section 4.4 prior to the Company fiscal year during which there is a net decrease in Partner Minimum Gain attributable to such Partner Nonrecourse Debt, and then only to the extent necessary to avoid any potential economic distortions caused by such net decrease in Partner Minimum Gain; and
- (iv) Allocations pursuant to this Section 4.4(e) shall be deferred with respect to allocations pursuant to subsection (d) of this Section 4.4 relating to a particular Partner Nonrecourse Debt to the extent the Board of Managers reasonably determine that such allocations are likely to be offset by subsequent allocations pursuant to subsection (b) of this Section 4.4.
- Section 4.5 BUILT-IN GAIN OR LOSS. Notwithstanding any other provision of this Article IV, in accordance with Code Section 704(c) and the accompanying Treasury Regulations thereunder, income, gain, loss and deduction with respect to any property other than cash contributed to the capital of the Company shall, solely for tax purposes, be allocated among the Members so as to take into account any variation between (i) the adjusted basis of such property to the Company for federal income tax purposes on the date of contribution and (ii) the fair market value of such property on the date contributed to the Company, as determined by the Board of Managers.
- Section 4.6 INCOME CHARACTERIZATION. For purposes of determining the character (as ordinary income or capital gain) of any taxable income of the Company allocated to the Members pursuant to this Article IV, such portion of the taxable income of the Company allocated pursuant to this Article IV which is treated as ordinary income attributable to the recapture of depreciation shall, to the extent possible, be allocated among the Members in the proportion which (i) the amount of depreciation previously allocated to each Member bears to (ii) the total of such depreciation allocated to all Members. This Section 4.6 shall not alter the amount of allocations among the Members pursuant to this Article IV, but merely the character of income so allocated.
 - Section 4.7 CREDITS. Tax credits shall be allocated among the Members in accordance with Section 4.3 hereof.
- Section 4.8 CHANGE IN INTERESTS. Notwithstanding the foregoing, in the event of a change in the Members' Percentage Interests in the Company during a year, whether occasioned by admission of a new Member, additional contributions, assignments of Membership Units or otherwise, the allocation of items of income and expense shall be made so as to reflect the Members' varying Percentage Interests in the Company during the year. Profits and losses for the year shall be prorated on a daily basis and allocated among the Members based upon the period of time during which they held their respective Percentage Interests.

7

Section 4.9 CREDITING ACCOUNTS. Items of income, gains, expenses, deductions, losses and credits shall be credited or debited, as the case may be, to each Member's Capital Account created pursuant to Section 3.4 as provided in this Article.

ARTICLE V DISTRIBUTIONS

- Section 5.1 <u>DISTRIBUTABLE AMOUNTS</u>. The Company may make distributions of any amounts in excess of its reasonable operating requirements as determined by the Board of Managers. The amounts available for distribution may be generated by operations of the Company through sale, condemnation, financing or refinancing of assets of the Company, by collection of amounts owed to the Company or by any other transaction. In the case of amounts attributable to a Capital Transaction, the net proceeds to be distributed hereunder shall be the net cash proceeds received by the Company after payment of, or provision for, all Company debts, obligations and reserves required or permitted to be paid upon, or incurred or established in connection with, the receipt by the Company of such proceeds (said reserves to be established in the reasonable discretion of the Board of Managers), and all expenses incurred by the Company in connection with the Capital Transaction giving rise to such proceeds. Notwithstanding the foregoing, no distribution shall be made unless after the distribution the Company retains assets sufficient to pay all its debts as they become due and such distribution, if made, would not cause the Company to otherwise become insolvent.
 - Section 5.2 ALLOCATION. Distributions of available cash pursuant to Section 5.1 hereof shall be made to the Members according to their respective Percentage Interests in the Company at that time.
- Section 5.3 <u>LIQUIDATING DISTRIBUTIONS</u>. After giving effect to all Member contributions, distributions, allocations, and capital account adjustments for all taxable years of the Company (including the taxable year during which the Company is liquidated), the Company shall distribute the liquidation proceeds derived from the dissolution, winding-up, liquidation, and termination of the Company (plus any remaining assets of the Company that are to be distributed in kind to the Members) in the following priority and proportions:
 - (a) To pay in the order of priority required by the Act all the liabilities and obligations of the Company that are then due and payable to creditors (including Members), including all costs and expenses directly relating to the liquidation, winding-up, and termination of the Company, but excluding Capital Contributions and distributions payable to the Members; and then
 - (b) To establish any reasonable cash reserve that the Members may determine unanimously by resolution to be appropriate for any contingent, conditional, or unmatured liabilities or obligations of the Company and as an allowance for doubtful accounts receivable or the unrealized portion of any promissory note, installment obligation, or other deferred revenue payable to the Company; and then
 - (c) To the Members, *pro rata*, in accordance with the sum of their Initial Capital Contributions plus any additional Capital Contributions which have not yet been returned to the respective Member (the "<u>Unreturned Capital Contributions</u>"), in an amount equal to each Member's Unreturned Capital Contributions; and then

В

(d) To the Members, pro rata, in accordance with the Members' respective Percentage Interests at the time of such liquidation..

The Board of Managers shall use reasonable efforts to cause the proceeds from a liquidation of the Company to be distributed in the same fiscal year in which the sale of the Company's assets occurs.

Section 5.4 MANDATORY TAX DISTRIBUTIONS. In order to permit the Members to pay taxes on Company taxable income, the Company shall, to the extent of Available Cash, distribute cash at least annually and no later than 75 days after the end of a Fiscal Year, an amount which, at the time of such distribution, when added to all prior distributions pursuant to this Section 5.4, equals the cumulative maximum tax payable by an individual taxpayer at the highest combined marginal tax rates of federal and state income taxes imposed on resident individuals in the United States on the taxable income of the Company. Such distribution shall be made ratably to the Members in proportion to their respective Member Percentages.

ARTICLE VI DURATION OF BUSINESS; TERMINATION

- Section 6.1 <u>DURATION OF COMPANY</u>. The Company shall continue until the earlier of
 - (a) The decision of the Board of Managers to dissolve the Company;
 - (b) The expiration of the term of the Company set forth in Section 1.4 hereof; or
 - (c) The sale or other disposition of all Company Property and the distribution of all sales proceeds resulting therefrom to the Members in accordance with this Agreement.

Section 6.2 <u>DEATH, ETC., OF MEMBER</u>. The Company shall not be dissolved upon the death, insanity, total disability, bankruptcy, dissolution or withdrawal of any Member, or by the assignment by any Member of all of the Membership Units or Economic Interest in the Company.

Section 6.3 <u>LIQUIDATION</u>. In the event of termination of the Company:

- (a) The Board of Managers shall wind up the affairs of the Company, shall sell all the Company assets as promptly as is consistent with obtaining the fair value thereof, and shall apply and distribute the proceeds of liquidation in the following order of priority:
 - i) To the payment of debts and liabilities of the Company (including to Members to the extent otherwise permitted by law) and the expenses of liquidation; then
 - (ii) To the setting up of such reserves as the Board of Managers winding up the Company's affairs may reasonably deem necessary or appropriate for any dispute, contingent or unforeseen liabilities or obligations of the Company; then
 - (iii) The remainder to the Members in accordance with Section 5.3 above.

9

- (b) Each Member shall look solely to the assets of the Company for the return of the Member's capital contribution, and if the Company Property remaining after payment or discharge of the debts and liabilities of the Company is insufficient to return the contributions of each Member, a Member shall have no recourse against any other Member(s).
 - (c) The proceeds of liquidation shall be distributed to the Members pursuant to Section 5.3 hereof
- (d) Upon the liquidation of the Company, if any Member has a Deficit Balance in the Member's Capital Account (after giving effect to all contributions, distributions and allocations for all taxable years, including the year in which such liquidation occurs), that Member shall have no obligation to make any contributions to the capital of the Company with respect to such deficit and such deficit shall not be considered a debt owed to the Company or any other person or entity for any purpose whatsoever.

ARTICLE VII RIGHTS AND DUTIES OF MEMBERS

- Section 7.1 LIABILITIES OF MEMBERS. No Member shall be obligated to make capital contributions to the Company except as provided in Article III. No Member shall have any personal liability with respect to the liabilities or obligations of the Company except as provided in Article III. No Member shall have any personal liability with respect to the liabilities or obligations of the Company except as provided in Article III. No Member shall have any personal liability with respect to the Romany except as provided in Article III. Failure of the Company to observe any formalities or requirements relating to the exercise of its powers or the management of its business or affairs under this Agreement or the Act shall not be grounds for imposing personal liability on the Members for liabilities or obligations of the Company.
- Section 7.2 LIMITATIONS ON POWERS OF MEMBERS. Except as expressly authorized by this Agreement, no Member may, directly or indirectly, (a) resign, retire or withdraw from the Company, (b) dissolve, terminate or liquidate the Company, (c) petition a court for the dissolution, termination or liquidation of the Company, or (d) cause any property of the Company to be subject to the authority of any court, trustee or receiver (including suits for partition and bankruptcy, insolvency, and similar proceedings).
 - Section 7.3 PROHIBITION AGAINST PARTITION. Each Member irrevocably waives any and all rights the Member may have to maintain an action for partition with respect to any property of the Company.

ARTICLE VIII MEETINGS OF MEMBERS

Section 8.1 ANNUAL MEETING. The annual meeting of the Members shall be held between January 1st and December 31st of each year, on such date and at such hour as may be specified in the Notice of Meeting or in a duly executed Waiver of Notice thereof, for the purpose of electing Managers to serve for the ensuing year and the transaction of such other business as may come before the meeting. If the day fixed for the annual meeting shall be a legal holiday in the State of Texas, such meeting shall be held on the next succeeding business day.

10

- Section 8.2 SPECIAL MEETING. Special meetings of the Members, for any purpose or purposes, may be called by the President or by a majority of the Board of Managers or by any Member owning Membership Units in the Company.
- Section 8.3 PLACE OF MEETING. The Board of Managers may designate any place, either within or without the State of Delaware, unless otherwise prescribed by statute, as the place of meeting for any annual or special meeting of Members. If no designation is made by the Board of Managers, the place of meeting shall be the principal office of the Company in the State of Texas.
- Section 8.4 NOTICE OF MEETING. Written or printed notice stating the place, day and hour of the meeting and, in the case of a special meeting, the purpose or purposes for which the meeting is called, shall be delivered to each Member of record entitled to vote at such meeting not less than ten (10) nor more than sixty (60) days before the date of the meeting, either personally or by first-class mail, by or at the direction of the President or the Secretary or by the Members calling the meeting. If mailed, such notice shall be deemed to be delivered at the earliest date of the following: (a) when received; (b) five days after its deposit in the United States mail addressed to the Member as as it appears on the records of the Company, with the postage thereon prepaid; or (c) on the date shown on the return receipt if sent by registered or certified mail, return receipt requested.
- Section 8.5 WAIVER OF NOTICE. Members may waive notice of a meeting before or after the date and time specified in the written notice of meeting. All waivers of notice must be in writing, be signed by the Member entitled to the notice and be delivered to the Company for inclusion in the appropriate records. None of the business to be transacted at, nor the purpose of, a Members' meeting must be specified in a written waiver of notice. Attendance of a Member at a meeting shall constitute a waiver of notice of the meeting, unless the Member at the beginning of the meeting objects to holding the meeting or transacting business at the meeting.
- Section 8.6 FIXING OF RECORD DATE. The Board of Managers may fix a date, not less than ten (10) nor more than sixty (60) days before the date set for any meeting of the Members, as the record date as of which the Members of record entitled to notice of and to vote at such meeting and any adjournment thereof shall be determined.
- Section 8.7 QUORUM. A majority of the Membership Units of the Company, represented in person or by proxy, shall constitute a quorum at a meeting of the Members. When a meeting is adjourned to another time or place, it shall not be necessary to give any notice of the adjourned meeting if the time and place to which the meeting is adjourned are announced at the meeting at which the adjournement is taken, and any business may be transacted at the adjourned meeting that might have been transacted at the original date of the meeting. If, however, after the adjournment, the Board of Managers fix a new record date for the adjourned meeting, a notice of the adjourned meeting shall be given in compliance with Section 8.4 to each Member of record on the new record date entitled to vote at such meeting. After a quorum has been established at a Members' meeting, the subsequent withdrawal of Members, so as to reduce the Membership Units entitled to vote at the meeting below the percentage required for a quorum, shall not affect the validity of any action taken at the meeting or any adjournment thereof.
- Section 8.8 PROXIES. Every Member entitled to vote at a meeting of Members or to express consent or dissent without a meeting, or his duly authorized attorney-in-fact, may authorize another person or persons to act for him by proxy. The proxy must be executed in writing by the Member or his duly authorized attorney-in-fact. Such proxy shall be filed with the Company before

11

or at the time of such meeting or at the time of expressing such consent or dissent without a meeting. No proxy shall be valid after the expiration of eleven (11) months after the date thereof unless provided otherwise in the proxy.

- Section 8.9 VOTING OF MEMBERSHIP UNITS. Except as provided in Section 9.6, each Member shall be entitled to one (1) vote per one (1) Membership Unit (and a corresponding fractional vote for each fraction of a Membership Unit) upon each matter submitted to a vote at a meeting of the Members. If a quorum is present, the affirmative vote of the Members owning a majority of the Membership Units represented at the meeting and entitled to vote on the subject matter shall be the act of the Members unless a greater percentage is required by the Delaware Statutes, this Agreement or the Company's Certificate of Formation.
- Section 8.10 <u>VOTING OF MEMBERSHIP UNITS BY CERTAIN HOLDERS</u>. Membership Units standing in the name of a corporation may be voted by the officer, agent or proxy designated by the bylaws of the corporate Member or, in the absence of any applicable bylaws, by such person as the managers of the corporate Member may designate. Proof of such designation may be made by presentation of a certified copy of the bylaws or other instrument of the corporate Member. In the absence of any such designation or, in case of conflicting designation by the corporate Member, the chairman of the board, the president, any vice president, the secretary, and the treasurer of the corporate Member shall be presumed to possess, in that order, authority to vote such Membership Units.

Membership Units held by an administrator, executor, guardian or conservator may be voted by him, either in person or by proxy, without a transfer of such Membership Units into his name,

- Membership Units standing in the name of a trustee may be voted by him, either in person or by proxy, but no trustee shall be entitled to vote Membership Units held by him without a transfer of such Membership Units into his
- Membership Units standing in the name of a receiver may be voted by such receiver, and Membership Units held by or under the control of a receiver may be voted by such receiver without the transfer thereof into his name, if authority so to do be contained in an appropriate order of the court by which such receiver was appointed.

A Member whose Membership Units are pledged shall be entitled to vote such Membership Units until the Membership Units have been transferred into the name of the pledgee, and thereafter the pledgee or his nominee shall be entitled to vote the Membership Units so transferred only in the event the pledgee becomes a Substituted Member (as defined in Section 12.3 hereinbelow).

Section 8.11 ACTION BY WRITTEN CONSENT OR TELEPHONE CONFERENCE

(a) Any action required or permitted to be taken at any meeting of Members may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by Members having not less than the minimum number of Membership Units necessary to authorize or take such action at a meeting at which all Members entitled to vote were present and voted. A telegram,

Members without a meeting by less than unanimous written consent shall be given to those Members who did not consent in writing to the action.

- (b) The record date for determining Members entitled to consent to action in writing without a meeting shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Company by delivery to its registered office, its principal place of business or the Board of Managers.
- (c) Members may participate in and hold a meeting by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can speak to and hear each other. Participation in such meeting shall constitute attendance and presence in person at such meeting, except where a person participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened.
- Section 8.12 ANNUAL REPORTS, ETC.. The Company shall file any necessary reports with the State of Delaware and pay the annual tax for Limited Liability Companies to the State of Delaware.

ARTICLE IX BOARD OF MANAGERS

- Section 9.1 MANAGEMENT AND CONTROL. The management and control of the Company shall be vested in the Board of Managers. The Board of Managers shall have, except as specifically limited in this Agreement, full and exclusive authority in the management and control of the Company, and shall have all the rights and powers which are otherwise conferred by law or are necessary or advisable for the discharge of their duties and the management of the business and affairs of the Company.
- Section 9.2 NUMBER AND ELECTION. The Board of Managers shall consist of three Managers (unless it has only two Managers as a result of the last two sentences of this paragraph). Managers shall be elected as follows: Members holding Class A Units shall have the right to elect one Manager, Members holding Class B Units shall have the right to elect one Manager, and Larry Buffington shall serve as a Manager. In the event of the death, resignation or removal of a Manager, the Member who elected such Manager may elect a successor Manager. Notwithstanding the foregoing, in the event that Larry Buffington ceases to be actively employed by the Company on a full-time basis as the Company's principal executive officer, then as of the date of such cessation, he shall cease to serve as a Manager, and the size of the Board of Managers will thereupon be reduced to two Managers. In addition, Larry Buffington shall not have the right to serve as a Manager during any period of time during which he is serving as an officer or employee of Liquidmetal Technologies, Inc. ("LMT")). The Member holding Class A Units hereby elects John Kang to serve as an initial Manager, and the Members holding Class B Units hereby elect Robert Smith to serve as an initial Manager.
- Section 9.3 RESIGNATION. A Manager may resign at any time by delivering written notice to the Board of Managers or to the Company. A resignation is effective when notice is delivered unless the notice specifies a later effective date.

13

- Section 9.4 REMOVAL OF MANAGER. Each Member may remove its elected Manager, with or without cause, at any time.
- Section 9.5 EXPRESSLY AUTHORIZED RIGHTS AND POWERS. Without limiting the generality of Section 9.1, but subject to the provisions of Section 9.6, the Board of Managers are expressly authorized on behalf of the Company to:
 - (a) elect officers to manage the day-to-day operations of the Company;
 - (b) procure and maintain with responsible companies such insurance as may be advisable in such amounts and covering such risks as are deemed appropriate by the Board of Managers;
 - (c) take and hold any assets of the Company in the Company name, or in the name of a nominee of the Company;
 - (d) execute and deliver on behalf of and in the name of the Company, or in the name of a nominee of the Company, all instruments necessary or incidental to the conduct of the Company's business;
 - (e) protect and preserve the assets of the Company and incur indebtedness;
 - (f) sell, dispose of, trade, exchange, convey, quitclaim, surrender, release or abandon, upon terms and conditions which the Board of Managers may negotiate and deem appropriate, personal property of the Company;
 - (g) execute and deliver documents and instruments on behalf of the Company in connection with the acquisition and disposition of its assets, and to execute, terminate, modify, enforce, continue or otherwise deal with any Company indebtedness and security interests, to sell Company assets, and to take any other action with respect to agreements made between the Company and a lender or any affiliate thereof, all subject to the limitations of Section 9.6:
 - (h) open Company bank accounts in which all Company funds shall be deposited and from which payments shall be made; and
 - (i) invest Company funds and working capital reserves.
- Section 9.6 CERTAIN LIMITATIONS. Notwithstanding the generality of the foregoing, and in addition to other acts expressly prohibited by this Agreement or by law, the Board of Managers, without the prior written consent or approval of the Members holding a Supermajority (as defined in Article XVI) of the issued and outstanding Membership Units, may not cause the Company to do any of the following:
 - (a) amend or restate the Certificate of Formation
 - $\begin{tabular}{ll} \textbf{(b)} & \textbf{sell, lease, exchange or otherwise dispose of all or substantially all the Company's property and assets;} \end{tabular}$

14

- (c) be a party to (i) a merger, (ii) interest exchange or (iii) other transaction affecting the ownership or structure of the Company;
- (d) do any act in contravention of this Agreement;
- (e) do any act which would make it impossible to carry on the ordinary business of the Company, except as expressly provided in this Agreement;
- (f) confess a judgment against the Company;
- (g) file a bankruptcy petition on behalf of the Company;
- (h) execute or deliver any general assignment for the benefit of the creditors of the Company;
- (i) Convert the Company to a Corporation pursuant to Article XIII;
- (j) assign rights in specific Company property for other than a Company purpose; or
- (k) knowingly or willingly do any act (except an act expressly required by this Agreement) which would cause the Company to become an association taxable as a corporation.

Section 9.7 MEETINGS OF MANAGERS.

- (a) Following Annual Meeting of Members. In connection with any annual meeting of Members at which Managers were appointed, the Board of Managers may, if a quorum is present, hold its first meeting for the transaction of business immediately after and at the same place as such annual meeting of the Members. Notice of such meeting at such time and place shall not be required.
- (b) Regular Meetings. Regular meetings of the Board of Managers shall be held at such times and places as shall be designated from time to time by resolution of the Board of Managers. Notice of such regular meetings shall not be required after the initial notice of the schedule of meetings.
- (c) Special Meetings. Special meetings of the Board of Managers may be called by the President or by any Manager on at least forty-eight (48) hours notice to each Manager. Such notice shall state the purpose or purposes of, or the business to be transacted at, such meeting.
- (d) Attendance and Place of Meeting. Meetings of the Board of Managers may be held at such place or places as shall be determined from time to time by resolution of the Board of Managers. The chair of the Board of Managers, if one has been designated by the Board of Managers, shall preside when present at meetings of the Board of Managers. Attendance of a Manager at a meeting shall be presumed to constitute a waiver of notice of such meeting, except where a Manager attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened or unless he or she shall file a written objection to such action with the Person

acting as secretary of the meeting before the adjournment thereof on the ground that the meeting is not lawfully called or convened.

- (e) Quorum and Action. A quorum for the transaction of business of the Board of Managers shall require the presence of a majority of the number of Managers fixed pursuant to Section 9.2 of this Article. At all meetings of the Board of Managers, business shall be transacted in such order as shall from time to time be determined by resolution of the Board of Managers. The act of the majority of the Managers present at a meeting at which a quorum is present shall be the act of the Board of Managers. A Manager who is present at a meeting of the Board of Managers at which action on any Company matter is taken shall be presumed to have assented to the action unless he or she shall either voice his or her dissent at the meeting or unless he or she shall file a written dissent to such action with the person acting as secretary of the meeting before the adjournment thereof. Such right to dissent shall not apply to a Manager who voted in favor of such action.
- (f) Action by Written Consent or Telephone Conference. Any action permitted or required to be taken at a meeting of the Board of Managers or of any committee designated by the Board of Managers may be taken without a meeting, without prior notice and without a vote if a consent or consents in writing, setting forth the action so taken, is signed by a quorum of the Managers as set forth in Section 9.7(e). Such consent shall have the same force and effect as a vote taken at a meeting and may be stated as such in any document or instrument filed with the Secretary of State of Delaware, and the execution of such consent or consents shall constitute attendance or presence in person at a meeting of the Board of Managers or any such committee, as the case may be. A telegram, telex, cablegram or similar transmission by a person, or a photographic, photostatic, facsimile or similar reproduction of a writing signed by a person, shall be regarded as signed by that person for the purposes of this Section 9.7(f).

Subject to the requirements of the Act, the Certificate of Formation or this Agreement, the Board of Managers, or members of any committee designated by the Board of Managers, may participate in and hold a meeting of the Managers or any committee of Managers, as the case may be, by means of a conference telephone or similar communications equipment by means of which all persons participating in the meeting can speak to and hear each other, and participation in such meeting shall constitute attendance and presence in person at such meeting, except where a person participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened.

Section 9.8 COMPENSATION. If approved by the Board of Managers and the Members, one or more Managers, or members of any committee designated by the Board of Managers, may be reasonably compensated for services rendered. Each Manager and members of any committee designated by the Board of Managers shall be reimbursed for actual expenses incurred in attending meetings.

ARTICLE X OFFICERS

Section 10.1 NUMBER. The Officers of the Company shall be the President, a Secretary and a Treasurer, each of whom shall be elected by the Board of Managers. Such other Officers and

16

assistant Officers and agents as may be deemed necessary may be elected or appointed by the Board of Managers. Any two (2) or more offices may be held by the same person. The Board of Managers hereby elects Larry Buffington as the initial President and Chief Executive Officer of the Company to serve until his successor his duly elected or until his earlier resignation or removal.

- Section 10.2 ELECTION AND TERM OF OFFICE. The Officers of the Company shall be elected annually by the Board of Managers at the regular meeting of the Managers held after each annual meeting of the Members. If the election of Officers shall not be held at such meeting, such election shall be held as soon thereafter as conveniently may be. Each Officer shall hold office until his successor shall have been duly elected and shall have qualified or until his earlier resignation, removal from office or death.
- Section 10.3 <u>REMOVAL AND RESIGNATION</u>. Any Officer or agent elected or appointed by the Board of Managers may be removed by the Board of Managers with or without cause, but such removal shall be without prejudice to the contract rights, if any, of the person so removed. Election or appointment of an Officer or agent shall not of itself create contract rights. Any Officer of the Company may resign at any time by giving written notice to the Board of Managers. Any such resignation shall take effect after time specified therein, or, if the time is not specified therein, upon its acceptance by the Board of Managers.
 - Section 10.4 VACANCIES. A vacancy, however occurring, in any office may be filled by the Board of Managers for the unexpired portion of the term.
- Section 10.5 CHAIR. The Chair of the Board of Managers, if one is elected or appointed, shall preside at all meetings of the Board of Managers and shall have such other powers and duties as may from time to time be prescribed by the Board of Managers, upon written directions given to him or her pursuant to resolutions duly adopted by the Board of Managers.
- Section 10.6 PRESIDENT. The President shall be the principal executive Officer of the Company and, subject to the control of the Board of Managers, shall in general supervise and control all of the business affairs of the Company. The President shall, when present, preside at all meetings of the Members and of the Board of Managers, unless the Board of Managers has elected a Chair of the Board of Managers and the Chair is present at such meeting. The President may sign deeds, mortgages, bonds, contracts, or other instruments which the Board of Managers has authorized to be executed, except in cases where the signing and execution thereof shall be expressly delegated by the Board of Managers or by this Agreement to some other Officer or agent of the Company, or shall be required by law to be otherwise signed or executed; and in general shall perform all duties as from time to time may be assigned to him by the Board of Managers.
- Section 10.7 VICE PRESIDENTS. If a Vice-President is elected or appointed, in the absence of the President or in the event of his death, inability or refusal to act, the Vice-President shall have the duties of the President, and when so acting, shall have all the powers of, and be subject to all the restrictions upon, the President. The Vice-President shall perform such other duties as from time to time may be assigned to him by the President or the Board of Managers.
- Section 10.8 SECRETARY. The Secretary shall: (a) keep the minutes of all the meetings of the Members and the Board of Managers in one or more books provided for that purpose; (b) see that all notices are duly given in accordance with the provisions of this Agreement or as required by law; (c) be custodian of the Company records; (d) keep a register of the post office address of each

17

Member; (e) have general charge of the Membership Unit transfer books of the Company, if the Company issues certificates representing such Membership Units; and (f) in general perform all duties incident to the office of Secretary and such other duties as from time to time may be assigned to him by the President or by the Board of Managers.

Section 10.9 TREASURER. The Treasurer shall: (a) have charge and custody of and be responsible for all funds and securities of the Company; receive and give receipts for moneys due and payable to the Company from any source whatsoever, and deposit all such moneys in the name of the Company in such banks, trust companies or other depositories as shall be selected by the Board of Managers; and (b) in general perform all of the duties incident to the office of Treasurer and such other duties as from time to time may be assigned to him by the President or by the Board of Managers, the Treasurer shall give a bond for the faithful discharge of his duties in such sum and with such surety or sureties as the Board of Managers shall determine.

Section 10.10 SALARIES. The salaries of the Officers shall be fixed from time to time by the Board of Managers and no Officer shall be prevented from receiving such salary by reason of the fact that he is also a Manager of the Company.

ARTICLE XI INDEMNIFICATION OF MEMBERS, MANAGERS AND DIRECTORS

- Section 11.1 RIGHT TO INDEMNIFICATION. The Company shall indemnify the Members, Managers and Officers to the fullest extent allowed by law from all claims brought by third parties relating to or arising out of the Company's business. The Company shall indemnify and hold harmless the Members, Managers and Officers to the fullest extent permitted or authorized by the Act or future legislation or by current or future judicial or administrative decision (but, in the case of future legislation or decision, only to the extent that it permits the Company to provide broader indemnification rights than permitted prior to the legislation or decision, against all fines, liabilities, settlements, Manager and/or Officer or incurred by any of them in their capacity as Member, Manager and/or Officer or arising out of their status as Member, Manager and/or Officer, as the case may be. The foregoing right of indemnification shall not be exclusive of other rights to which those seeking indemnification may be entitled. The Company may maintain insurance, at its expense, to protect itself and the indemnified persons against all fines, liabilities, costs and expenses, including attorneys' fees, whether or not the Company would have the legal power to indemnify him directly against such liability.
- Section 11.2 <u>ADVANCES.</u> Costs, charges and expenses (including attorneys' fees) incurred by a person referred to in Section 11.1 in defending a civil or criminal suit, action or proceeding shall be paid by the Company in advance of the final disposition thereof upon receipt of an undertaking to repay all amounts advanced if it is ultimately determined that the person is not entitled to be indemnified by the Company as authorized by this Article.
- Section 11.3 SAVINGS CLAUSE. If this Article XI or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Company shall nevertheless indemnify and hold harmless the Members, Managers, Officers or any other person indemnified pursuant to this Article XI as to costs, charges and expenses (including attorneys' fee), judgments, fines and amounts paid in settlement with respect to any action, suit or proceeding,

18

whether civil, criminal, administrative or investigative to the full extent permitted by any applicable portion of this Article XI that shall not have been invalidated and to the fullest extent permitted by applicable law.

ARTICLE XII ASSIGNMENT OF MEMBERSHIP UNITS

Section 12.1 GENERAL RESTRICTION ON ASSIGNMENT. Except as expressly permitted hereunder, no Member or Economic Interest Owner may Transfer all or any portion of, or any interest or rights in, their Membership Units or Economic Interest without the prior written consent of the Board of Managers (excluding Managers who are Affiliates of the transferring Member) (the "Non-Affiliated Managers") or, if there are no Non-Affiliated Managers, then the consent of all of the Members other than the transferring Member (the "Non-Affiliated Members"). Each Member hereby acknowledges the reasonableness of this prohibition in view of the purposes of the Company and the relationship of the Members.

The Transfer of any Membership Units or Economic Interests in violation of the prohibition contained in this Section 12.1 shall be deemed invalid, null and void, and of no force and effect. Any Person to whom Membership Units are attempted to be Transferred in violation of this Section 12.1 shall not be entitled to vote on matters coming before the Members, participate in the management of the Company, act as an agent of the Company, receive distributions from the Company or have any other rights in or with respect to the Economic Interest.

For purposes of this Agreement, the term "<u>Transferr</u>", or "<u>Transferred</u>", when used in this Agreement with respect to the Membership Units or Economic Interests, includes a sale, assignment, gift, pledge, encumbrance or any other disposition, whether voluntary, by operation of law or otherwise, and "<u>transferee</u>" and "<u>transferer</u>" have corresponding meanings. Further, any such proposed Transfer must be accomplished by written instrument satisfactory in form and content to the Non-Affiliated Managers or Non-Affiliated Members (as the case may be), accompanied by such assurances of genuineness and effectiveness of signatures and the obtaining of any governmental approvals or legal opinions as the Non-Affiliated Managers or Non-Affiliated Members (as the case may be) may reasonably require and payment of any reasonable costs of transfer as the Non-Affiliated Managers or Non-Affiliated Members (as the case may be) may require and all transfer taxes as may be imposed.

In the event of a "change of control" (as defined hereinbelow) of the Class A Holder, the Class C Holder shall be deemed to have made an offer to sell all of his Membership Units and Economic Interests to the Company immediately prior to the effectuation of the change of control in accordance with the terms of Section 12.2 hereinbelow; provided, however, that for purposes of this provision, the per Membership Unit purchase price for the Offered Interest shall be the Fair Market Value (as defined in Article XVI hereinbelow) of the Offered Interest and the terms of payment shall be the payment terms set forth in Section 12.2(d) below. For purposes of this provision, the term (change of control)" shall mean a transaction or series of transactions (related or unrelated) which result in the Class A Holder no longer owning, immediately after such transaction or series of transactions, in excess of fifty percent (50%) (determined by vote and value) of the issued and outstanding equity securities or beneficial interests of the Company.

19

Section 12.2 RIGHT OF FIRST REFUSAL. In addition to the general restrictions contained in Section 12.1 above, no Member or Economic Interest Owner may voluntarily Transfer any right, title or interest in its Membership Units or Economic Interest, or any part thereof, to any Person, unless the Member or Economic Interest Owner desiring to make the transfer, hereinafter referred to as the "Transferor", shall (a) have received a bona fide written offer to purchase such interest in the Company by an unrelated third party (the "Proposed Transferore") and (b) have made an offer to tender to the Company for redemption all of the Transferor interest in the Company which the Transferor desires to sell, give or otherwise transfer to the Proposed Transferore (the "Offered Interest") in the manner hereinafter described in this Section 12.2, and the offer shall not have been accepted.

- (a) Offer by Transferor. The offer by a Transferor shall be given in writing to the Company and shall be accompanied by a copy of the written offer, proposal or contract, between the Transferor and the Proposed Transferee (hereinafter referred to as the "Written Offer"), and shall set forth the nature of the transaction (whether sale, gift, or other transfer), the name and address of the Proposed Transferee and the terms of the transaction, including an identification of the interest involved, the number of Membership Units involved, the purchase price, and the payment terms (hereinafter referred to as the "Proposed Transaction"). The offer by the Transferor shall consist of an offer to tender to the Company for redemption all of the Offered Interest at the price and upon the terms set forth in this Section 12.2 below.
- (b) Exercise of Option. Within fifteen (15) days after the receipt of the offer, the Company shall have the option, but not the duty, to redeem all or any portion of the Offered Interest; provided, however, that the Company's decision to accept or reject the offer shall be made by a vote of a majority of the Non-Affiliated Managers or, if there are no Non-Affiliated Managers, then by a Majority Vote of the Non-Affiliated Members. The election to redeem shall be exercised by the giving of notice thereof to the Transferor.

The notice of exercise of option shall specify a date for the closing of the redemption of the Offered Interest (hereinafter referred to as the "Closing" or the "Closing Date"), which shall not be less than thirty (30) days nor more than sixty (60) days after the expiration of the time within which the Company may exercise its option.

- (c) Redemption/Purchase Price. If the Company elects to redeem all of the Offered Interest, the price for each Membership Unit of the Offered Interest shall be the price per Membership Unit set forth in the Written Offer.
 - (d) Payment of Redemption/Purchase Price. Upon any redemption under this Section 12.2, the redemption price shall be paid in accordance with the terms provided in the Written Offer.
- (e) Closing. The Closing of a purchase under this Section 12.2 shall take place at the principal office of the Company on a date specified in writing in the written acceptance by the Company to the Transferor, unless the Transferor and the Company otherwise mutually agree on another place or date. At the Closing, the Transferor shall deliver, in exchange for the total purchase price, whether in cash or partially in cash and partially by promissory notes, as the case may be, the certificate, if any, representing the Offered Interest

20

being transferred, duly endorsed, and such other documents as shall be necessary and reasonably required to conclude the transfer.

- (f) Release from Restriction. If the Transferor's offer to tender or sell is not accepted by the Company as to the entire interest of the Transferor, the Transferor may make a transfer to the prospective bona fide transferee of the entire Offered Interest, the transfer to be made only in strict accordance with the terms of the Written Offer. If the Transferor shall fail to make the transfer within thirty (30) days following the expiration of the time hereinabove provided for the exercise of the election to purchase by the Company, the Membership Units of the Transferor shall again be subject to all the restrictions of this Agreement.
- (g) Continuance of Restrictions Upon Subsequent Owners. In the event the Transferor makes a bona fide transfer of the Offered Interest under the provisions of this Section 12.2, then the Offered Interest transferred to the Proposed Transferee shall be subject to all the provisions of this Agreement. No Membership Units or Economic Interest shall be transferred on the books of the Company and no certificate evidencing such Membership Units or Economic Interest shall be issued to the Proposed Transferee unless and until the Proposed Transferee has executed a counterpart to this Agreement, the original of which shall be retained as part of the Company's records. Failure of the Proposed Transfere to execute a counterpart of this Agreement, to the Offered Interest, it being the intention of each Member and the Company that any and all subsequent owners of Membership Units and Economic Interests voluntarily transferred shall only receive and own the Membership Unit or Economic Interest subject to the same restrictions upon transfer and encumbrance as set forth in this Agreement, to which the Transferor was subject, including, without limitation, all of the provisions of this Article XII.
- Section 12.3 INVOLUNTARY TRANSFER. In the event any Membership Units or Economic Interest are the subject of an involuntary Transfer, whether due to bankruptcy, assignment for benefit of creditors, judicial order, legal process, divorce, execution, attachment, enforcement of a pledge or other encumbrance, or otherwise (hereinafter referred to as the "Affected Interest"), the Member or Economic Interest Owner owning the Affected Interest shall be deemed to have made, immediately prior to such involuntary transfer, an offer first to tender to the Company for redemption all of the Affected Interest in the manner hereinafter described in this Section 12.3. There shall be no obligation or requirement that the Company redeem any of the Affected Interest under this Section 12.3, any redemption of the Affected Interest being solely upon election to do so. The Company may redeem all or any portion of the Affected Interest.
 - (a) Exercise of Option. If the Company elects to redeem all or any portion of the Affected Interest in accordance with this Section 12.3, the Company shall serve notice in writing of its election upon the Member owning the Affected Interest and the creditor(s) of the Member, spouse (in the event of a divorce) or other person or entity who is to be the recipient of the Affected Interest (hereinafter referred to collectively as the "<u>Transferor</u>") within ninety (90) days after the Board of Managers shall have received actual notice of the involuntary transfer. The Company's decision to accept or reject the offer shall be made by a majority vote of the Non-Affiliated Managers, then by a Majority Vote of the Non-Affiliated Members.

21

The notice of exercise of option shall specify a date for the closing of the redemption of the Affected Interest (hereinafter referred to as the "Closing" or the "Closing Date"), which shall not be less than thirty (30) days nor more than ninety (90) days after the expiration of the time within which the Company may exercise its option.

- (b) Redemption/Purchase Price. If the Company elects to redeem all or any portion of the Affected Interest, the price for each Membership Unit of the Affected Interest shall be the lesser of: (i) the Fair Market Value (as defined in Article XVI hereinbelow) of the Affected Interest, or (ii) the total amount, including acquisition costs, if any, which had been due to the creditor of the Member who was to be the recipient of the Affected Interest.
 - (c) Payment of Redemption/Purchase Price. Upon any redemption under this Section 12.3, the redemption price shall be paid in the same manner provided in Section 12.2(d) above.
- (d) Closing. The closing of a purchase under this Section 12.3 shall take place at the principal office of the Company on a date specified in writing in the written acceptance by the Company to the Transferor, unless the Transferor and the Company otherwise mutually agree on another place or date. At the Closing, the Transferor shall deliver, in exchange for the total purchase price, whether in cash or partially in cash and partially by promissory notes, as the case may be, the certificate, if any, representing the Affected Interest being transferred, duly endorsed, and such other documents as shall be necessary and reasonably required to conclude the transfer.
- (e) Continuance of Restrictions Upon Transferor. In the event the "deemed offer" of the Transferor is not accepted by the Company as to all of the Affected Interest, then the portion of the Affected Interest not redeemed under this Section 12.3 may be transferred to the transferee/creditor subject to all of the provisions of this Agreement. The transferee/creditor shall execute a counterpart of this Agreement, the original of which shall be retained as part of the Company's records. The failure of the transferee/creditor to execute a counterpart to this Agreement shall not affect the applicability of this Agreement to the Affected Interest, it being the intention of each Member and the Company that any and all subsequent owners of Membership Units or Economic Interests acquired pursuant to an involuntary transfer shall only receive and own the Membership Units or Economic Interests subject to the restrictions upon transfer and encumbrance as set forth in this Agreement to which the Original Member was subject.

Section 12.4 TERMINATION OF EMPLOYMENT. In the event that Larry Buffington's ("Buffington") employment with the Company is terminated for any reason (including but not limited to death, voluntary severance or permanent disability) (the "Terminating Event"), then the Company may redeem, within six (6) months after the Terminating Event, all of the Class C Units in the Company owned by Buffington on the following terms:

(a) For twenty-percent (20%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within one year of the effective date of this Agreement;

- (b) For forty-percent (40%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within the second year after the effective date of this Agreement;
- (c) For sixty-percent (60%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within the third year after the effective date of this Agreement;
- (d) For eighty-percent (80%) of the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within the fourth year after the effective date of this Agreement.
- (e) For the Fair Market Value (as defined in Article XVI hereinbelow) of the Class C Units owned by Buffington if the Terminating Event occurs within or after the fifth year after the effective date of this Agreement.

The redemption price shall be paid within ninety (90) days after the Company notifies Buffington or his successor of the election to redeem his Class C Units. The redemption price shall be paid in cash to the extent it will not impair the operations of the Company, and the balance shall be paid by a promissory note payable on the first anniversary of the Termination Event with interest at Prime Rate published in the Wall Street Journal, adjusted quarterly.

- Section 12.5 WITHDRAWALS. No Member may withdraw from the Company, except upon the prior written consent of the Board of Managers.
- Section 12.6 RIGHTS OF ASSIGNEE OF MEMBERSHIP UNITS. Unless the assignee of a Membership Unit is admitted as a Substituted Member as provided in Section 12.7, the assignee will be merely an Economic Interest Owner, and the assignee's rights shall be limited to sharing in the profits to which the assignor would otherwise have been entitled and to receiving the assignor's share of any proceeds and an accounting upon dissolution. The assignee shall have no right to vote on Company matters, exercise any purchase rights granted to Members hereunder, inspect Company books and records or otherwise participate in Company affairs and the interest of the assignee shall be disregarded for purposes of determining whether Members owning the required Membership Units have voted on any matter requiring a vote of the Members or in determining the total of the Membership Units outstanding for voting purposes. For example, in the event a judgment creditor obtains a charge against a Member's Membership Unit, pursuant to Section 608.433(4) of the Act, or any successor provision, then unless the judgment creditor is admitted as a Substituted Member, the judgment creditor will be merely an Economic Interest Owner and will not acquire any other rights of a Member. All remaining rights and interest in the Membership Units which were owned by the Transferring Member to the Transfer and that were associated with the assigned Economic Interest (including, without limitation, the rights of the Transferring Member to participate in the management and affairs of the Company) shall immediately lapse until the Managers, in their sole discretion, reinstate such rights to the Economic Interest Owner or to a successor or transfere of such Economic Interest Owner.
- Section 12.7 SUBSTITUTED MEMBER. An assignee of the whole or any portion of a Member's Membership Interests in the Company, validly assigned under this Article XII, may become a Substituted Member in the place of his assignor(s), to the extent of the Membership Interests validly assigned, if all of the following conditions are satisfied:

23

- (a) A fully executed and acknowledged written instrument of assignment has been filed with the Company which sets forth the intention of the assignor(s) that the assignee become a Substituted Member in his/their place, to the extent of the Membership Interests assigned.
- (b) The assignee executes, acknowledges and delivers to the Managers a written acceptance and adoption of the provisions of this Agreement, in form and substance acceptable to the Managers in their sole discretion.
 - (c) The assignee pays a transfer fee to the Company in an amount sufficient to cover all reasonable expenses connected with the admission of such person as a Substituted Member.
- Section 12.8 SECURITIES LAWS. The Membership Units have not been registered under the Federal or state securities laws of any state and, therefore, may not be resold unless appropriate Federal and state securities laws, as well as the other provisions of this Article XII have been complied with.
- Section 12.9 INVALID TRANSFER. No Transfer of a Membership Unit or Economic Interest that is in violation of this Article XII shall be valid or effective, and the Company shall not recognize any improper transfer for the purposes of making allocations, payments of profits, return of capital contributions or other distributions with respect to such Membership Unit or Economic Interest. The Company may enforce the provisions of this Article XII either directly or indirectly or through its agents by entering an appropriate stop transfer order on its books or otherwise refusing to register or transfer or permit the registration or transfer on its books of any proposed transfers not in accordance with this Article XII.
- Section 12.10 <u>DISTRIBUTIONS AND ALLOCATIONS IN RESPECT OF A TRANSFERRED INTEREST</u>. If any Member Transfers any part of a Membership Unit or Economic Interest in the Company during any accounting period in compliance with the provisions of this Article XII, Company income, gain, deductions and losses attributable to such interest for the respective period shall be divided and allocated between the transferor and the transferee by taking into account their varying interests during the applicable accounting period in accordance with Code section 706(d), using the daily proration method. All Company distributions on or before the effective date of such transfer shall be made to the transferor and all such Company distributions, the Company of such transferes. Solely for purposes of making Company tax allocations and distributions, the Company shall recognize a transfer on the day following the day of transfer. Neither the Company nor any Member shall incur any liability for making Company allocations and distributions in accordance with the provisions of this Section 12.10.
 - Section 12.11 <u>DRAG-ALONG RIGHTS</u>(a) . Notwithstanding anything in this Agreement to the contrary, in the event that either the Class A Holders or the Class B Holders elect to sell all of their Membership Units in a single transaction to a <u>bona fide</u> third party purchaser, either the Class A Holders or the Class B Holders, as the case may be, may require all of the other Members (the "<u>Affected Members</u>") to sell all of their Membership Units for the same price and on the same terms as those to be received by either the Class B Holders, as the case may be. In such case, the provisions of Sections 12.1 and 12.2 hereof shall not apply to such Transfer and the Affected Members shall be deemed to have made an offer to sell all of their Membership Units on such terms and conditions as those under which

24

either the Class A Holder or the Class B Holder, as the case may be, will sell its Membership Units; provided, however, that the Affected Members shall have no liability regarding any representations made by either the Class A Holder or the Class B Holder, as the case may be, except representations relating to the Affected Members' ownership On the Membership Units, their ability to deliver marketable title to such Membership Units and similar representations.

- (b) Notwithstanding anything in this Agreement to the contrary, in the event that either the Class A Holders or the Class B Holders elect to sell all of their Membership Units in a single transaction to a bona fide third party purchase, either the Class A Holders or the Class B Holders, as the case may be, shall require the bona fide third party purchaser to offer to the Affected Members to purchase all their Membership Units for the same price and on the same terms as those to be received by either the Class B Holders, as the case may be. If the third party purchaser will not make such offer to the Affected Members, then the Class A Holders or the Class B Holders, as the case may be, shall not be allowed to sell their Membership Units to such third party purchaser.
- Section 12.12 APPLICABILITY OF PROVISIONS TO CLASS A HOLDERS(a) The provisions of Sections 12.1 and 12.2 shall not apply to the Membership Units owned by any Class A Holder, and any Class A Holder may transfer any or all of its Membership Units without the consent of the Board of Managers or the Members, subject to the provisions of Section 12.12(b) below.
- (b) At any time a Class A Holder reasonably anticipates that it has a good faith desire to sell some or all of its Membership Units, such Class A Holder shall notify the Class B Holders of that desire and determine whether the Class B Holders have an interest in purchasing all the Membership Units owned by such Holders (the "Offered Units"). If the Class B Holders, or any of them, have such an interest, they shall make an offer to purchase the Offered Units, and the parties will negotiate in good faith to enter into a definitive purchase agreement with respect thereto. If the parties have not entered into a definitive purchase greement with innerty (90) days after the original notice from the Class A Holder, the Class A Holder may offer and sell the Offered Units to a third party purchaser (the "Third Party Purchaser"). If the Class A Holder shall fail to locate a Third Party Purchaser to purchase the Offered Units within one hundred eighty (180) days following the expiration of the time hereinabove provided for, the Offered Units shall again be subject to all the restrictions of this Agreement.
- (c) Continuance of Restrictions Upon Subsequent Owners. In the event a Class A Holder sells the Offered Units to a Third Party Purchaser under the provisions of this Section 12.12, then the Offered Units transferred to the Third Party Purchaser shall be subject to all the provisions of this Agreement. No Membership Units or Economic Interest shall be transferred on the books of the Company and no certificate evidencing such Membership Units or Economic Interest shall be issued to the Third Party Purchaser under the original of which shall be retained as part of the Company's records. Failure of the Third Party Purchaser to execute a counterpart of this Agreement, however, shall not affect the applicability of this Agreement to the Offered Units, it being the intention of each Member and the Company that any and all subsequent owners of Membership Units and Economic Interests voluntarily transferred shall only receive and own the Membership Units or Economic Interests subject

25

to the same restrictions upon transfer and encumbrance as set forth in this Agreement, to which the Class A Holder was subject, including the applicable provisions of this Article XII.

(d) With the exception of the application of Sections 12.1 and 12.2 to any Class A Holder, any transfer by any Class A Holder of any or all of its Membership Units shall be subject to the remaining Sections of this Article XII.

ARTICLE XIII CONVERSION TO CORPORATE SOLUTION

Subject to Section 9.6, the Board of Managers shall have the power and authority to effect the conversion of the Company's business form from a limited liability company to a corporation (for any reason whatsoever, including without limitation, a public offering of the Company's Capital Securities) or the merger of the Company with or into a new or previously-established but dormant corporation having no assets or liabilities, debts or other obligations of any kind whatsoever other than those associated with its formation and initial capitalization (such a conversion or merger is referred to as a "Conversion" and such corporation is referred to as a "Newco").

Upon the consummation of a Conversion, the Units held by each holder thereof shall thereupon be converted into a number of shares of Newco's Capital Securities containing the economic and other terms and rights relative to each other holder of Units as the Board of Managers shall determine to be as nearly as practicable in all material respects the same as such holder's Units as provided herein. The Board of Managers' determination of the class (and the terms thereof and rights associated therewith) and number of shares of Newco Capital Securities that each Member receives upon a Conversion shall be final and binding on the holders of Units absent manifest arithmetic error.

In connection with a Conversion, each Member hereby covenants and agrees to take any and all such action and execute and deliver any and all such instruments and other documents as the Board of Managers may reasonably request in order to effect or evidence such Conversion. Without limiting the generality of the foregoing, no Member shall have or be entitled to exercise any dissenter's rights, appraisal rights or other similar rights in connection with such

ARTICLE XIV BOOKS AND RECORDS

Section 14.1 BOOKS AND RECORDS. The Board of Managers shall keep or cause to be kept complete books and records of the Company and supporting documentation of the transactions with respect to the conduct of the Company's business. These and all other records of the Company, including information relating to the Company's activities, information with respect to the sale by a Member or any Affiliate of goods or services to the Company, and a list of the names and business addresses of all Members shall be kept at the offices of the Company, or at such other location as may be determined by the Board of Managers, and shall be available for examination there by any Member, or his duly authorized representative, at reasonable times upon reasonable notice. Any Member, or his duly authorized representative, upon paying the costs of collection, duplication and mailing, shall be entitled to a copy of the list of names and addresses of the Members. The books and records shall be maintained in accordance with sound accounting practices.

26

The Company will also cause Liquidmetal Coatings Solutions, LLC, a Delaware limited liability company ("LMCS"), its wholly owned subsidiary, to keep complete books and records in the same manner as required above.

Section 14.2 CUSTODY OF MEMBER FUNDS; BANK ACCOUNTS

- (a) The Board of Managers shall have fiduciary responsibility for the safekeeping and use of all funds and assets of the Company, whether or not in the immediate possession or control of the Board of Managers. The funds of the Company shall not be commingled with the funds of any other person and the Board of Managers shall not employ, or permit any other person to employ, such funds in any manner except for the benefit of the Company.
- (b) All funds of the Company not otherwise invested shall be deposited in one or more accounts maintained in such banking institutions as the Board of Managers shall determine, and withdrawals shall be made only in the regular course of Company business on such signature or signatures as the Board of Managers may, from time to time, determine.
- Section 14.3 ACCOUNTANTS. The accountants for the Company shall be such certified public accountants as shall be selected by the Board of Managers.
- Section 14.4 SECTION 754 ELECTION. In the event of a distribution of Company Property (other than money) to a Member or upon a transfer of all or any part of the Membership Units of a Member, the Board of Managers may in their sole and absolute discretion, upon the written request of the Member receiving the distribution or the transferee of the Membership Units, as the case may be (the "Electing Member"), elect pursuant to Section 754 of the Code, to adjust the basis of the Company 's property in the manner provided in Sections 734 and 743 thereof, respectively. Each Member agrees to furnish the Company with all information necessary to give effect to such election. The election will be filed with the Company information tax return for the first taxable year to which the election applies. If the Board of Managers decide to make the Section 754 election, the Electing Member will be responsible for all additional accounting costs incurred by the Company as a result of the Electing Member's request to make the election under Code Section 754.
 - Section 14.5 FISCAL YEAR. The fiscal year of the Company shall be the calendar year.
- Section 14.6 TAX MATTERS PARTNER. LMT shall be the "Tax Matters Partner" of the Company for purposes of Section 6231(a)(7) of the Code. The Tax Matters Partner shall have the power and authority, subject to the review and control of the Board of Managers, to manage and control, on behalf of the Company, any administrative proceeding at the Company level with the Internal Revenue Service relating to the determination of any item of Company income, gain, loss, deduction or credit for federal income tax purposes. The Tax Matters Partner may be removed, and a new Tax Matters Partner appointed, by the Board of Managers in accordance with the Code and the Treasury Regulations.
- Section 14.7 ANNUAL FINANCIAL STATEMENTS AND TAX RETURN INFORMATION. The Company shall provide to C³ Capital Partners, L.P. ("C³"), and C³ Capital Partners II, L.P. ("C³"II"), as soon as available, and, in any event, within one hundred and twenty (120) days after the end of each fiscal year of the Company, beginning with the fiscal year ending December 31, 2007, a copy of the annual consolidated financial statements of the Company and

27

LMCS for such fiscal year containing balance sheets, statements of income, retained earnings and cash flows at the end of such fiscal year, in each, except for the fiscal year ending December 31, 2007, setting forth in comparative form the figures for the preceding fiscal year (if any), all in reasonable detail and audited and certified by independent certified public accountants of recognized standing reasonably acceptable to C³ and C³ II, to the effect that such report has been prepared in accordance with GAAP. Within ninety (90) days after the end of each Tax year, Company shall provide sufficient information with respect to the Company for such Tax year necessary for C³ and C³ II to prepare their federal, state and local income Tax Returns. The Company shall provide copies of the federal, state and local income Tax Returns of the Company for each Tax year promptly after the filing thereof.

- Section 14.8 MONTHLY FINANCIAL STATEMENTS. The Company shall provide to C³ and C³ II, as soon as available and, in any event, within thirty (30) days after the end of each month, a copy of an unaudited consolidated financial report of the Company and LMCS as of the end of such month and for the portion of the fiscal year then ended, including balance sheets, statements of income, retained earnings and cash flows, setting forth in each case comparisons to the Company's and LMCS's Annual Budget and to the corresponding period in the preceding fiscal year and a brief narrative explaining the results of the Company's and LMCS's operations for such period; all such statements shall have been prepared in accordance with GAAP (absent footnotes and customary year-end adjustments) and fairly present the financial condition and results of operations of the Company and LMCS at the date and for the periods indicated therein.
- Section 14.9 INSPECTION RIGHTS. At any reasonable time and from time to time, the Company shall permit representatives of C³ and C³ II to examine, copy and make extracts from its books and records, to visit and inspect the Company's and LMCS's properties, and to discuss Company's and LMCS's business, operations, and financial condition with Company's or LMCS's officers, employees and independent certified public accountants, provided that C³ and C³ II agree to keep such information confidential and to use such information solely in connection with its investment in the Membership Units.

ARTICLE XV EMPLOYMENT AGREEMENTS

The Company shall neither (i) modify, amend or terminate the confidentiality, nonsolicitation and noncompetition agreements between the Company and Larry Buffington (the "Employment Agreements") without the prior written consent of C³ and C³ II, which consent shall not be unreasonably withheld or delayed nor (ii) provide any employee, who is a party to the Employment Agreements, with any compensation, benefit or reimbursement for any expenses related to the performance of such employee's duties as contemplated by such employee's Employment Agreements, except as provided by the Employment Agreements.

ARTICLE XVI DEFINITIONS

- Section 16.1 DEFINITIONS. Unless the context otherwise requires, the terms defined in this Article XVI shall, for the purposes of this Agreement, have the meanings herein specified.
- "Act" means the Delaware Limited Liability Company Act, as amended.

28

"Affiliate" or "Affiliated Party." means, with respect to any Member, a partner of a Member; any member of the immediate family of any Member; any shareholder, officer or director of a Member or any member of their respective immediate families; any person, firm or entity which, directly or indirectly, controls, is controlled by, or is under common control with a Member, any partner of any Member or any shareholder, officer or director of a partner of any Member or their respective families; or any person, firm or entity which is associated with a Member, any partner of a Member or any member or any member or their respective immediate families in a joint venture, partnership or other form of business association. In this definition, the term "control" shall mean the ownership of ten percent (10%) or more of the beneficial interest in the firm or entity referred to, and the term "immediate family" shall mean the spouse, ancestors, lineal descendants, brothers and sisters of the person in question, including those adopted. To the extent the term "Affiliate" or "Affiliated Party." is used in the context of an affiliation with a Person (the "Subject") other than a Member then such term shall have the same meaning above; however, the term "Member" within such definition shall be replaced with the Subject.

- "Agreement" means this Limited Liability Company Operating Agreement, as amended, modified, supplemented or restated from time to time in accordance with the terms hereof.
- "Business Day," means any day other than a Saturday, a Sunday, or a holiday on which national banking associations in the State of Delaware are closed.
- "Capital Account" means, with respect to any Member, the account maintained for such Member in accordance with the provisions of Section 3.4.
- "Capital Contribution" means, with respect to any Member, the aggregate amount of money and the initial Gross Asset Value of any property (other than money) contributed to the Company pursuant to Section 3.2.
- "Capital Securities" means as to any Person that is a corporation, the authorized shares of such Person's capital stock, including all classes of common, preferred, voting and nonvoting capital stock, and, as to any Person that is not a corporation or an individual, the ownership or membership interests in such Person, including, without limitation, the right to share in profits and losses, the right to receive distributions of cash and property, and the right to receive allocations of items of income, gain, loss, deduction and credit and similar items from such Person, whether or not such interests include voting or similar rights entitling the holder thereof to exercise control over such Person.
- "Capital Transaction" means any of the following events: (A) any sale or other disposition of all or any part of any capital assets of the Company, (B) any loans secured by all or any part of the capital assets of the Company, (C) the refinancing of any Company indebtedness, (D) the condemnation of all or any part of the capital assets of the Company, or (E) any insurance recovery relating to any capital assets owned by the Company.
 - "Class A Holder" means a Member holding Class A Units.
 - "Class B Holder" means a Member holding Class B Units.
 - "Class C Holder" means a Member holding Class C Units.

"Class A Percentage Interest" means, with respect to a Class A Holder at any time and from time to time, a percentage equal to a fraction, the numerator of which is the number of Class A Units owned by such Class A Holder and the denominator of which is the aggregate number of Common Units owned by all Members.

"Class B Percentage Interest" means, with respect to a Class B Holder at any time and from time to time, a percentage equal to a fraction, the numerator of which is the number of Class B Units owned by such Class B Holder and the denominator of which is the aggregate number of Common Units owned by all Members.

"Class C Percentage Interest" means, with respect to a Class C Holder at any time and from time to time, a percentage equal to a fraction, the numerator of which is the number of Class C Units owned by such Class C Holder and the denominator of which is the aggregate number of Common Units owned by all Members;

"Class A Unit" means a Unit representing a fractional part of the Member Interests of the Members and having the rights and obligations specified with respect to the Class A Units in this Agreement.

"Class B Unit" means a Unit representing a fractional part of the Member Interests of the Members and having the rights and obligations specified with respect to the Class B Units in this Agreement.

"Class C Unit" means a Unit representing a fractional part of the Member Interests of the Members and having the rights and obligations specified with respect to the Class C Units in this Agreement.

"Code" means the U.S. Internal Revenue Code of 1986, as amended, and the Treasury Regulations promulgated thereunder.

"Company" means the Delaware limited liability company that is the subject of this Agreement.

"Company Property" means all real and personal property acquired by the Company and any improvements thereto, including, without limitation, any tangible or intangible property of the Company.

"Conversion" has the meaning specified in Article XIII of this Agreement.

"Deficit Capital Account Balance" means, with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant fiscal year, after giving effect to the following adjustments:

- (a) A credit to such Capital Account of any amount which such Member is obligated to restore or is deemed to be obligated to restore pursuant to Treasury Regulations Sections 1.704-2(g) and 1.704-2(j)(5); and
- (b) A debit to such Capital Account of the items described in Sections 1.704-1(b)(2)(ii)(d)(4), (5) and (6) of the Treasury Regulations.

30

"Economic Interest" shall mean a Member's or Economic Interest Owner's pro rata share of the Company's profits and distributions of the Company's assets pursuant to this Operating Agreement and the Act, but shall not include any right to participate in the management of affairs of the Company, including, the right to vote on, consent to or otherwise participate in any decision of the Members or Managers.

"Economic Interest Owner" shall mean the owner of an Economic Interest who is not a Member.

"Fair Market Value" of a Membership Interest shall mean the amount the Selling Member or Transferor would receive with respect to its Membership Interest upon the dissolution and termination of the Company assuming (A) such dissolution or termination occurred on the date of the Notice, and (B) the assets of the Company were sold for their fair market value without compulsion for the Company to sell such assets. The fair market value of the Company is assets shall be agreed upon by the Selling Member or Transferor, as the case may be, and the Company. If they are unable to agree upon an appraiser within five (5) working days after either party serves written demand on the other, then each shall select one appraiser and the two appraisers of selected shall select at hird appraiser whose determination shall be conclusive and binding for this purpose. In the event that the Company or the Selling Member or Transferor, as the case may be, fails to designate an appraiser pursuant to the preceding sentence within five (5) working days after either party serves written demand on the other, then the appraisal shall be divided equally between the Company and the Selling Member or Transferor, as the case may be.

"Fiscal Year" means the fiscal year of the Company which shall be the calendar year unless otherwise required by the Code.

"Lien" means (a) any encumbrance, mortgage, pledge, lien, charge or other security interest of any kind upon any property or assets of any character, or upon the income or profits therefrom; (b) any acquisition of or agreement to have an option to acquire any property or assets upon conditional sale or other title retention agreement, device or arrangement (including a capitalized lease); or (c) any sale, assignment, pledge or other transfer for security of any accounts, general intangibles or chattel paper, with or without recourse; excluding in each instance the lien of this Agreement.

"Manager" has the meaning specified in Article IX of this Agreement.

"Member" means the persons listed on Exhibit "A" hereto, and includes any Person admitted as an additional Member or a substitute Member pursuant to the provisions of this Agreement, in such Person's capacity as a member of the Company, and "Members" means two (2) or more of such Persons when acting in their capacities as members of the Company.

"Membership Rights" means all legal and beneficial ownership interests in, and rights and duties as a Member of, the Company, including, without limitation, the right to share in Profits and Losses, the right to receive distributions of cash and other property from the Company, and the right to receive allocations of items of income, gain, loss, deduction and credit and similar items from the Company.

31

"Membership Units" shall mean the units into which the ownership interests of the Members in the Company are divided, including such Member's Economic Interest and the right of such Member to any and all benefits to which such Member may be entitled as provided in this Agreement or under the Act, together with the obligation of such Member to comply with all of the provisions of this Agreement and of the Act.

"Newco" has the meaning specified in Article XIII of this Agreement.

"Nonrecourse Deductions" has the meaning set forth in Sections 1.704-2(b) and 1.704-2(c) of the Regulations. The amount of Nonrecourse Deductions for a Company fiscal year equals the net increase in Partnership Minimum Gain during that fiscal year determined pursuant to Section 1.704-2(d) of the Regulations reduced (but not below zero) by the aggregate distributions during that fiscal year of proceeds of a Nonrecourse Liability that are allocable to an increase in Partnership Minimum Gain, determined according to the provisions of Sections 1.704-2(h) of the Regulations.

"Nonrecourse Liability" has the meaning set forth in Section 1.752-1(a)(2) of the Regulations.

"Partner Minimum Gain" means an amount, with respect to each Partner Nonrecourse Debt, equal to the Partnership Minimum Gain that would result if such Partner Nonrecourse Debt were treated as a Nonrecourse Liability, determined in accordance with Section 1.704-2(i) of the Regulations.

"Partner Nonrecourse Debt" has the meaning set forth in Section 1.704-2(b)(4) of the Regulations.

"Partner Nonrecourse Deductions" has the meaning set forth in Section 1.704-2(i)(2) of the Regulations. The amount of Partner Nonrecourse Deductions with respect to a Partner Nonrecourse Debt for a Company fiscal year equals the net increase during the year in Partner Minimum Gain attributable to such Partner Nonrecourse Debt reduced (but not below zero) by proceeds of the liability distributed during that fiscal year to the Member bearing the economic risk of loss for the liability that are both attributable to the liability and allocable to an increase in Partner Minimum Gain attributable, determined according to the provisions of Section 1.704-2(h) of the Regulations.

"Partnership Minimum Gain" has the meaning set forth in Regulations Sections 1.704-2(b)(2) and 1.704-2(d).

"Percentage Interest" means, with respect to a holder of Common Units at any time and from time to time, the Class A Percentage Interest, Class B Percentage and/or Class C Percentage held by such holder

"Person" includes any individual, corporation, association, partnership (general or limited), joint venture, trust, estate, limited liability company, or other legal entity or organization.

"Requisite Majority" means the approval or vote of the holders owning at least a majority of the Company's issued and outstanding Membership Units at any given time

"Supermajority" means the approval or vote of the holders owning at least eighty-five percent (85%) of the Company's issued and outstanding Membership Units at any given time.

32

"Tax Matters Partner" has the meaning specified in Section 14.6.

"Transfer" means any sale, assignment, encumbrance, hypothecation, pledge, conveyance in trust, gift, transfer by bequest, devise or descent, or other transfer or disposition of any kind, including, but not limited to, transfers to receivers, levying creditors, trustees or receivers in bankruptcy proceedings or general assignees for the benefit of creditors, whether voluntary or by operation of law, directly or indirectly.

ARTICLE XVII AGREEMENT PREPARED BY ATTORNEY FOR COMPANY

The parties acknowledge that the counsel for the Company, Foley & Lardner, prepared this Agreement on behalf of and in the course of its representation of the Company, and that:

- (a) Each Member and Manager has been advised that a conflict of interest may exist among the Members, the Board of Managers and the Company; and
- (b) Each Member and Manager has been urged and has had the opportunity to seek the advice of independent legal counsel.

ARTICLE XVIII AMENDMENTS

Except as otherwise expressly provided herein, the written approval of the Members holding a Supermajority, as that term is defined herein, of the Membership Units in the Company shall be required to alter, modify or amend this Agreement; provided, however, that no alteration, modification or amendment of Articles III, IV, V, VI or VII hereof or this Article XVIII which would materially and adversely affect the economic interest of one or more Members, or their successors or assigns, may be made without the unanimous written consent of all such Members so adversely affected. Notwithstanding the above, this Agreement may be amended from time to time by the Board of Managers, without the consent of any of the Members, (i) to add to the representations, duties or obligations of the members of the Board of Managers, (ii) to cure any ambiguity, to correct or supplement any provision herein which may be inconsistent with any other provisions with respect to matters or questions arising under this Agreement and (iii) to delete or add any provision of this Agreement required to be so deleted or added by the staff of the Securities and Exchange Commission or by any state securities commission or similar official, which addition or deletion is deemed by such commission or official to be for the benefit or protection of the Members.

ARTICLE XIX MANAGER AND MEMBER RELATIONS TO THE COMPANY

No Manager or Member shall be restricted in any way from engaging in any other business venture or activity and no Manager or Member shall be accountable to the Company or to any other Manager or Member because of any activity or venture which does not directly involve the Company. Neither the Company nor the Members shall have any right under this Agreement in and

33

to the other activities of any Manager or Member or to their income or profits from such business venture or activity.

ARTICLE XX CHOICE OF LAW; SUBMISSION TO JURISDICTION; AND WAIVER OF JURY TRIAL

Section 20.1 LAW. This Agreement shall be governed by and construed under the laws of the State of Delaware. The parties agree that any action brought by any party under or in relation to this Agreement, including without limitation to interpret or enforce any provision of this Agreement, shall be brought in, and each party agrees to and does hereby submit to the jurisdiction and venue of, any state or federal court located in Delaware.

Section 20.2 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY VOLUNTARILY AND IRREVOCABLY WAIVES TRIAL BY JURY IN ANY ACTION OR OTHER PROCEEDING BROUGHT IN CONNECTION WITH THIS AGREEMENT, ANY OF THE RELATED AGREEMENTS, DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY.

ARTICLE XXI MISCELLANEOUS

Section 21.1 DELAY OR OMISSIONS. It is agreed that no delay or omission to exercise any right, power or remedy accruing to any party, upon any breach, default or noncompliance by another party under this Agreement shall impair any such right, power or remedy, nor shall it be construed to be a waiver of any such breach, default or noncompliance, or any acquiescence therein, or of or in any similar breach, default or noncompliance thereafter occurring. It is further agreed that any waiver, permit, consent or approval of any kind or character on any party's part of any breach, default or noncompliance under this Agreement or any waiver on such party's part of any provisions or conditions of the Agreement must be in writing and shall be effective only to the extent specifically set forth in such writing. All remedies, either under this Agreement, by law, or otherwise afforded to any party, shall be cumulative and not alternative.

Section 21.2 WAIVER. No waivers of any breach of this Agreement extended by any party hereto to any other party shall be construed as a waiver of any rights or remedies of any other party hereto or with respect to any subsequent breach.

Section 21.3 WAIVER OF PARTITION. The Members hereby agree that no Member, nor any successor in interest to any Member, shall have the right, while this Agreement remains in effect, to have any Company property partitioned, or to file a complaint or institute any proceeding at law or in equity to have such property partitioned, and all Members, on behalf of themselves and their heirs, successors and assigns, hereby waive any such right.

Section 21.4 SUCCESSORS AND ASSIGNS. Except as otherwise expressly provided herein, the provisions hereof shall inure to the benefit of, and be binding upon the parties hereto and their respective successors, assigns, heirs, executors and administrators and shall inure to the benefit of and be enforceable by each Person who shall be a holder of the Capital Securities of the Company from time to time; provided, however, that prior to the receipt by the Company of adequate written notice of the transfer of any Capital Securities of the Company specifying the full name and address

34

of the transferee and such transferee's satisfaction of all requirements to be a Permitted Transferee hereunder, the Company may deem and treat the person listed as the holder of such Capital Securities of the Company in its records as the absolute owner and holder of such Capital Securities of the Company for all purposes.

Section 21.5 NOTICES. All notices required in connection with this Agreement shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient; if not, then on the next business day, (c) five (5) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written notification of receipt. All communications shall be sent to the Company at the address below and to each Member at the address as set forth on Exhibit "A" hereto or at such other address as such party may designate by ten (10) days advance written notice to the other parties hereto:

If to the Company, addressed to:

LIQUIDMETAL COATINGS, LLC 30452 Esperanza Rancho Santa Margarita, California 92688 Attention: Larry Buffington, Chief Executive Officer

Fax No.: 813.314.0270

with a copy to:

Foley & Lardner LLP 100 North Tampa Street, Suite 2700 Tampa, FL 33602-5804 Tel: (813) 225-4122

Facsimile: (813) 221-4210| Attention: Curt P. Creely, Esq.

Section 21.6 ENTIRE AGREEMENT. This Agreement and the Exhibits hereto, along with the other documents delivered pursuant thereto, including but not limited to that certain Securities Purchase Agreement of even date herewith, constitute the full and entire understanding and agreement between the parties with regard to the subjects hereof and hereof and no party shall be liable or bound to any other in any manner by any oral or written representations, warranties, covenants and agreements except as specifically set forth herein and therein. Each party expressly represents and warrants that it is not relying on any oral or written representations, warranties, covenants or agreements except as specifically set forth herein and therein.

Section 21.7 PRONOUNS. All pronouns contained herein, and any variations thereof, shall be deemed to refer to the masculine, feminine or neutral, singular or plural, as to the identity of the parties hereto may require.

Section 21.8 TITLES AND SUBTITLES. The titles of the sections and subsections of the Agreement are for convenience of reference only and are not to be considered in construing this Agreement.

35

- Section 21.9 SEVERABILITY. In the event one or more of the provisions of this Agreement should, for any reason, be held by any court of competent jurisdiction to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceable provision had never been contained herein.
- Section 21.10 BINDING EFFECT. Except as provided to the contrary, the terms and provisions of this Agreement shall be binding upon and shall inure solely to the benefit of all the Members, their personal representatives, heirs, successors and assigns.
 - Section 21.11 CREDITORS. The provisions of this Agreement are not for the benefit of and may not be specifically enforced by any creditors of the Company.

Section 21.12 EXECUTION OF ADDITIONAL INSTRUMENTS. Each Member hereby agrees to execute such other and further statements of interest and holdings, designations, powers of attorney, and other instruments necessary to comply with any applicable laws, rules, or regulations.

Section 21.13 RIGHTS AND REMEDIES CUMULATIVE. The rights and remedies provided by this Agreement are cumulative and the use of any one right or remedy by any party shall not preclude or waive the right to use any or all other remedies. Said rights and remedies are given in addition to any other rights the parties may have by law, statute, ordinance, or otherwise.

Section 21.14 COUNTERPARTS. This Agreement may be executed in one or more counterparts, each of which will be deemed an original, but all of which together shall constitute one instrument. Counterparts of this Agreement (or applicable signature pages hereof) that are manually signed and delivered by facsimile transmission shall be deemed to constitute signed original counterparts hereof and shall bind the parties signing and delivering in such

IN WITNESS WHEREOF, the parties have executed this Limited Liability Company Operating Agreement as of the date first set forth hereinabove.

	COMPANY:
	LIQUIDMETAL COATINGS, LLC
	Ву:
	Name: Title:
	MEMBERS:
	CLASS A HOLDER:
	LIQUIDMETAL TECHNOLOGIES, INC.
	By:
	,
36	5
Tit	le:
	ASS B HOLDERS:
	CAPITAL PARTNERS, L.P.
Ву	: Its General Partner
	C ₃ Capital Partners, LLC, a Delaware limited liability company
Ву	: Print
Na	me:
	Title:
	A A A A A A A A A A A A A A A A A A A
33	7
	C₃ CAPITAL PARTNERS II, L.P.
	By: Its General Partner
	C_3 Capital Partners II, LLC, a Delaware limited liability
	company
	Ву:
	Print Name:
	Title:
	CLASS C HOLDERS:
1	LARRY BUFFINGTON
ī	Larry Buffington
	Global Strategy & Capital Group, Inc. d.b.a CRESO Capital Partners
,	Зу:
	Print Name:
	Піде:
<u>-</u>	
38	3
Exhibi	ı A

Initial Capital

Number and Class

Members	C	ontribution	of Units	Percentage Interest
Liquidmetal Technologies, Inc., a			6,986	·
Delaware corporation	\$	3,577,210	Class A Units*	69.25%
30452 Esperanza				
Rancho Santa Margarita, California				
92688				
Attention: Legal Department				
Fax No.: 813.314.0270				
C ₃ Capital Partners, LP, a Delaware			1,102	
limited partnership	\$	564,218	Class B Units**	11.02%

C ₃ Capital, LLC 4520 Main Street Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: 816-756-5552			
C ₃ Capital Partners II, LP, a Delaware limited partnership C ₃ Capital, LLC 4520 Main Street Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: 816-756-5552	\$ 408,572	798 Class B Units**	7.98%
Larry Buffington 2205 Riva Row, Unite 2410 The Woodlands, TX 77380 Fax No. (832) 585-0341	\$ 0.00	1,000 Class C Units***	10.00%

<u>Members</u>	 Initial Capital Contribution	Number and Class of Units	Percentage Interest
Global Strategy & Capital Group, Inc. d.b.a CRESO Capital Partners, a corporation 660 Newport Center Drive, Suite 800 Newport Beach, CA 92660 Fax No. (949) 209-5441 Attn: Thomas Papa	\$ 0.00	114 Class C Units****	1.75%
TOTALS	\$ 4,550,000	10,000 Units	100 %

^{*}Issued pursuant to that certain Asset Purchase and Contribution Agreement of even date herewith between the Company and Liquidmetal Technologies, Inc.

2

Exhibit B-2

OPERATING AGREEMENT

0

${\bf LIQUIDMETAL\ COATINGS\ SOLUTIONS, LLC,}$

A Delaware Limited Liability Company

Adopted as of July 24, 2007

TABLE OF CONTENTS

		PAGE
ARTICLE I ORGAN	NIZATION	1
1.1	FORMATION	1
1.2	CERTIFICATE OF FORMATION	1
1.3 1.4	NAME TERM	1
1.4	PRINCIPAL OFFICE	2
1.6	REGISTERED AGENT/REGISTERED OFFICE	2
1.7	MEMBER	2
	OSE AND BUSINESS OF THE COMPANY	2
2.1 2.2	PURPOSE AUTHORITY OF THE COMPANY	2
2.2	AUTHORITY OF THE COMPANY	2
ARTICLE III CONT	TRIBUTIONS TO CAPITAL AND CAPITAL ACCOUNTS	2
3.1	INITIAL CONTRIBUTIONS	2
3.2	ADDITIONAL CAPITAL CONTRIBUTIONS	2
3.3	CAPITAL ACCOUNTS	2
3.4	LOANS.	2
ARTICLE IV TAX	CHARACTERIZATION	3
ARTICLE V COSTS	TS AND EXPENSES	3
ARTICLE VI DISTI	TRIBUTIONS	3
6.1	DISTRIBUTABLE AMOUNTS	3
6.2	LIQUIDATING DISTRIBUTIONS	4
7.1	RATION, LIQUIDATION, AND TERMINATION DURATION OF COMPANY	4
7.1	DURATION OF COMPANY LIQUIDATION	4
7.3	ARTICLES OF DISSOLUTION	4
ARTICLE VIII MA	NAGEMENT	4
ARTICI E IX I IMIT	ITATION OF LIABILITY AND INDEMNIFICATION	5
9.1	LIMITATION OF LIABILITY	5
9.2	INDEMNIFICATION	5
9.3	NON-EXCLUSIVE RIGHT	5
ADTICLE V TRANS	NSFER OF INTEREST AND ADDITIONAL MEMBERS	Ē
10.1	SSFER OF INTEREST AND ADDITIONAL MEMBERS TRANSFERS	6
10.2	RIGHTS OF ASSIGNEE	6

^{**}Issued pursuant to that certain Securities Purchase Agreement of even date herewith between the Company, C3 Capital Partners, L.P., and C3 Capital Partners II, L.P.

^{***}Issued pursuant to that certain Employment Agreement of even date herewith between the Company and Larry Buffington.

^{****}Issued as consideration for placement agent services provided by Cresco Capital Partners.

10.3 ADDITIONAL MEMBERS

6

RTICLE XI BOOK	KS, RECORDS AND ACCOUNTING
11.1	BOOKS AND RECORDS
11.2	CUSTODY OF FUNDS
11.3	FISCAL YEAR
11.4	ANNUAL REPORT
ARTICLE XII DEFI	
12.1	"Act" and "Delaware Limited Liability Company Act"
12.2	"Agreement"
12.3	"Capital Contributions"
12.4	"Code"
12.5	"Company"
12.6	"MANAGER"
12.7	"Member"
12.8	"Membership Interest"
12.9	"Membership Rights"
12.10	"Person"
ARTICLE XIII AM	ENDMENT
ARTICLE XIV GEN	NERAL PROVISIONS
14.1	CAPTIONS
14.2	VARIATIONS OF PRONOUNS
1/12	CONSTRUCTION

ii

OPERATING AGREEMENT

OF

LIQUIDMETAL COATINGS SOLUTIONS, LLC

THIS OPERATING AGREEMENT (the "Agreement") is hereby entered into as of July 24, 2007, by and between LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company (hereinafter sometimes referred to as the "Manager"), and LIQUIDMETAL COATINGS SOLUTIONS, LLC, a Delaware limited liability company (hereinafter referred to as the "Company").

RECITAL

The Member has formed the Company pursuant to the provisions of the Delaware Limited Liability Company Act, Chapter 18 of the Delaware Code, for the purposes set forth herein, and, accordingly, desires to enter into this Agreement in order to set forth the terms and conditions of the business and affairs of the Company and to determine the rights and obligations of its Member.

NOW, THEREFORE, the parties, intending to be legally bound by this Agreement, hereby agree as follows:

ARTICLE I ORGANIZATION

- **1.1 FORMATION**. The undersigned Member has formed the Company as a single-member limited liability company pursuant to the Act and the terms and conditions of this Agreement.
- 1.2 CERTIFICATE OF FORMATION. The Member has caused a Certificate of Formation to be prepared, executed and filed with the Delaware Secretary of State for the formation of the Company effective July 9, 2007. Any and all amendments to the Certificate required by law to be filed and recorded hereafter for any reason shall be filed by the Company in such office or offices as are required under the laws of the State of Delaware or elsewhere. The Company shall do all other acts and things that may now or hereafter be required for the perfection and continuation of the Company as a limited liability company under the laws of the State of Delaware or necessary in order to protect the limited liability of the Member under the laws of the State of Delaware or elsewhere.
- 1.3 NAME. The name of the Company is LIQUIDMETAL COATINGS SOLUTIONS, LLC. All Company business must be conducted in such name or other names that comply with applicable law as the Manager may, in the Manager's sole discretion, select from time to time. If the Company does business under a name other than that set forth in its Certificate of Formation, then the Company shall file a fictitious name registration as required by law.
- 1.4 TERM. The term of the Company commenced on the filing of the Certificate of Formation with the Delaware Secretary of State and shall continue until terminated in accordance with the provisions of this Agreement or by operation of law.
- 1.5 PRINCIPAL OFFICE. The principal office of the Company shall be maintained at 30452 Esperanza, Rancho Santa Margarita, California 92688, or at such other place which the Manager, in the Manager's sole discretion, determines.
- 1.6 REGISTERED AGENT/REGISTERED OFFICE. The name of the registered agent of the Company is CorpDirect Agents, Inc. The street address of the registered office of the Company is 615 South DuPont Highway, Kent County, Dover, Delaware 19901.
- 1.7 MEMBER. The name of the sole Member is LIQUIDMETAL COATINGS, LLC. The present mailing address of the sole Member of the Company is 30452 Esperanza, Rancho Santa Margarita, California 92688. The Member shall own 100% of the Membership Rights in the Company, including 100% of the Membership Interests.

ARTICLE II PURPOSE AND BUSINESS OF THE COMPANY

- 2.1 PURPOSE. The Company is organized for the purpose of transacting all lawful activities and businesses that may be conducted by a limited liability company under the laws of Delaware.
- 2.2 AUTHORITY OF THE COMPANY. This Company shall have the powers and authority to do all things necessary to carry out its business and affairs as authorized by the Act.

ARTICLE III CONTRIBUTIONS TO CAPITAL AND CAPITAL ACCOUNTS

- 3.1 INITIAL CONTRIBUTIONS. Upon the execution of this Agreement, the Member shall contribute to the Company the cash and property (other than cash) set forth on Schedule "A" attached hereto.
- 3.2 <u>ADDITIONAL CAPITAL CONTRIBUTIONS</u>. The Member may make additional contributions of capital to the Company as the Member determines are necessary, appropriate or desirable; provided, however, that the Member shall have no obligation to contribute any additional capital to the Company, and except as set forth in the Act, the Member shall have no personal liability for any obligations of the Company.
 - 3.3 CAPITAL ACCOUNTS. A capital account shall be maintained by the Company for the Member.
 - 3.4 LOANS.

14.4

145

BINDING EFFECT

COUNTERPART SIGNATURES

SEVERABILITY

(a) Loan Terms. The Member may, at any time, make or cause a loan to be made to the Company in any amount and on those terms upon which the Company and the

(b) Repayment of Loans. Distributions of cash to the Member in repayment of loans made by the Member shall be made pursuant to the terms of such loans, but all distributions shall be subject to maintaining the Company in a sound financial condition, including the establishment of reserves reasonably required in the judgment of the Manager for the proper operation of the business of the Company.

ARTICLE IV TAX CHARACTERIZATION

The Member acknowledges that at all times that two or more persons or entities hold equity positions in the Company for federal income tax purposes (i) it is the intention of the Company to be treated as a "partnership" for federal and all relevant state tax purposes, and (ii) the Company will be treated as a "partnership" for federal and all relevant state tax purposes and shall make all available elections to be so treated. Until such time, however, it is the intention of the Member that the Company be disregarded for federal and all relevant state tax purposes. Accordingly, until such time, and notwithstanding anything contained herein to the contrary, pursuant to Treasury Regulation Section 301.7701-3(b) of the Code, the Company shall be disregarded as an entity separate from the Member for federal and state income tax purposes, unless and until the Member causes the Company to file an election pursuant to Treasury Regulation Section 301.7701-3(c) of the Code. All provisions of the Company's Certificate of Formation and this Agreement are to be so construed so as to preserve that tax status under these circumstances.

ARTICLE V COSTS AND EXPENSES

The Company shall pay all expenses of the Company (which expenses shall be billed directly to the Company) which may include but are not limited to: (i) legal, audit, accounting and other fees; (ii) expenses and taxes incurred in connection with the issuance, distribution and transfer of documents evidencing ownership of an interest in the Company or in connection with the business of the Company; (iii) expenses of organizing, revising, amending, converting, modifying or terminating the Company; (iv) expenses in connection with distributions made by the Company to, and communications and bookkeeping work necessary in maintaining relations with, the Member; and (v) costs of any accounting, statistical or bookkeeping equipment necessary for the maintenance of the books and records of the Company.

ARTICLE VI DISTRIBUTIONS

6.1 <u>DISTRIBUTABLE AMOUNTS</u>. The Company may make distributions to the Member of any amount in excess of its reasonable operating requirements as determined by the Manager. Notwithstanding the foregoing, no distribution shall be made unless after the distribution the Company retains assets sufficient to pay all its debts as they become due and such distribution, if made, would not cause the Company to otherwise become insolvent.

3

6.2 <u>LIQUIDATING DISTRIBUTIONS</u>. In the event of liquidation of the Company, the assets of the Company shall be distributed to the Member in accordance with Section 7.2 hereinafter.

ARTICLE VII DURATION, LIQUIDATION, AND TERMINATION

- 7.1 DURATION OF COMPANY. The Company shall continue in existence until the Manager, in the Member's sole discretion, determines to dissolve the Company. In the event there should at any time be no Members of the Company, the Company shall not be dissolved and the successor-in-interest (or the personal or other legal representative of the last remaining Member) shall have the option to either dissolve the Company or agree in writing to continue the Company and to the admission of the successor-in-interest (or personal or other legal representative of the last remaining Member) or its nominee or designee to the Company as a Member, effective as of the occurrence of the event that terminated the continued membership of the last remaining Member.
 - 7.2 LIQUIDATION. In the event of dissolution of the Company, the Manager shall wind up the affairs of the Company and shall distribute the money and other property of the Company in the following order of priority:
- (a) To creditors, including a Member who is a creditor, to the extent permitted by law in satisfaction of the Company's debts and liabilities whether by payment or establishment of reserves, other than liabilities for distributions to the Member under Sections 18-601 and 18-604 of the Act; then
 - (b) The remainder, if any, to the Member or to the Member's successors, heirs or assigns.
 - 7.3 ARTICLES OF DISSOLUTION. In the event the Company is dissolved, Articles of Dissolution shall be promptly filed with the Delaware Secretary of State

ARTICLE VIII MANAGEMENT

The management and control of the Company shall be vested solely in the Manager. The Manager shall have full and exclusive authority in the management and control of the Company, and shall have all the rights and powers to make all decisions with respect thereto and to do or cause to be done any and all acts or things deemed by the Manager to be necessary, appropriate or desirable to carry out or further the business and affairs of the Company.

The Manager may appoint such officers as the Manager may desire from time to time and delegate to such officers certain duties and obligations with regard to the day-to-day operations of the Company. The Manager shall appoint such officers in writing signed by the Manager, and any officer so appointed may be removed by the Manager, with or without cause, at any time immediately upon notice to such officer. Any officer may resign upon giving written notice to the Manager. Any officers of the Company shall have the duties, authorities and powers

4

customarily granted to such officer positions. The Manager hereby appoints Larry Buffington to serve as the initial President and Chief Executive Officer of the Company.

ARTICLE IX LIMITATION OF LIABILITY AND INDEMNIFICATION

- 9.1 LIMITATION OF LIABILITY. Except as otherwise provided by the Act, the debts, obligations and liabilities of the Company, whether arising in contract, tort or otherwise, shall be solely the debts, obligations and liabilities of the Company, and neither the Member nor the Manager shall be obligated personally for any such debt, obligation or liability of the Company, solely by reason of being a Member and/or Manager. The failure of the Company to observe any formalities or requirements relating to the exercise of its powers or the management of its business or affairs under this Agreement or the Act shall not be grounds for imposing personal liability on the Member or Manager for any debts, liabilities or obligations of the Company. Except as otherwise expressly required by law, the Member, in the Member's capacity as such, shall have no liability in excess of (a) the amount of such Member's Capital Contributions, (b) such Member's share of any assets and undistributed profits of the Company, and (c) the amount of any distributions required to be returned pursuant to Section 18-607(b) or Section 18-804(c) of the Act.
- 9.2 INDEMNIFICATION. The Company (including any receiver or trustee of the Company), shall, to the fullest extent provided or allowed by law, indemnify, save harmless and pay all judgments and claims against the Member and/or Manager and each of the Member's and/or Managers agents, affiliates, heirs, legal representatives, successors and assigns (each hereinafter individually referred to as an "Indemnified Party.") from, against and in respect of any and all liability, loss, claim, damage, obligation, penalty, action, judgment, suit, proceeding, cost, disbursement and expense of any kind or nature whatsoever incurred or sustained by the Indemnified Party in connection with the business of the Company or by reason of any act performed or omitted to be performed in connection with the activities of the Company or in dealing with third parties on behalf of the Company, including costs and attorneys' fees before and at trial and at all appellate levels, whether or not suit is instituted (which attorneys' fees may be paid as incurred), and any amounts expended in the settlement of any claims of liability, loss or damage, provided that the act or omission of the Indemnified Party does not constitute fraud or willful misconduct by such Indemnified Party. The Company shall not pay for any insurance covering liability of the Member and/or Manager or the Member's and/or Manager's agents, affiliates, heirs, legal representatives, successors and assigns for actions or omissions for which indemnification is not permitted hereunder; provided, however, that nothing contained herein shall preclude the Company from purchasing and paying for such types of insurance, including extended coverage liability and casualty and worker's compensation, as would be customary for any Person owning, managing and/or operating comparable property and engaged in a similar business or from naming the Member and/or Manager and any of the Member's and/or Manager's agents, affiliates, heirs, legal representatives, successors or assigns, or any Inde
- **9.3 NON-EXCLUSIVE RIGHT**. The provisions of this Article IX shall be in addition to and not in limitation of any other rights of indemnification or reimbursement or limitations of liability to which Indemnified Party may be entitled under the Act, common law,

5

or otherwise. Notwithstanding any repeal of this Article IX or other amendment hereof, its provisions shall be binding upon the Company (subject only to the exceptions above set forth) as to any claim, loss, expense, liability, action or damage due to or arising out of matters which occur during or relate to the period prior to any such repeal or amendment of this Article IX.

ARTICLE X TRANSFER OF INTEREST AND ADDITIONAL MEMBERS

- 10.1 TRANSFERS. The Member may sell, hypothecate, pledge, assign, or otherwise transfer all, or any portion, of its Membership Rights to one or more Persons.
- 10.2 RIGHTS OF ASSIGNEE. In the event of any transfer of all or any part of the Member's Membership Rights to a successor, the successor shall thereupon become a Member and the Company shall continue in existence. In the event a judgment creditor obtains a charge against the Member's Membership Interest pursuant to Section 18-703 of the Act, the judgment creditor shall have only the rights of an assignee and shall not become a Member and shall not have any other Membership Rights.
 - **10.3** ADDITIONAL MEMBERS. The Manager may, in the Manager's sole discretion, determine to admit additional Members.

- 11.1 BOOKS AND RECORDS. The books and records of the Company, if any, shall be maintained on a cash or accrual basis as determined by the Manager, in the Manager's sole discretion, in accordance with a generally accepted method of accounting, consistently applied. These and all other records of the Company required to be kept pursuant to the Act shall be kept at the registered office of the Company.
- 11.2 CUSTODY OF FUNDS. The Manager shall have fiduciary responsibility for the safekeeping and use of all funds and assets of the Company, whether or not in the immediate possession or control of the Manager. The funds of the Company shall not be commingled with the funds of any other Person and the Manager shall not employ, or permit any other Person to employ, such funds in any manner except for the benefit of the Company. All funds of the Company not otherwise invested shall be deposited in one or more accounts maintained in such banking institutions as the Manager shall determine, and withdrawals shall be made only in the regular course of Company business.
 - 11.3 **FISCAL YEAR**. The fiscal year of the Company shall be the calendar year.
 - 11.4 ANNUAL REPORT. The Company shall file an annual report with the Delaware Secretary of State each year in the form provided by the Secretary of State.

6

ARTICLE XII DEFINITIONS

- 12.1 "ACT" AND "DELAWARE LIMITED LIABILITY COMPANY ACT" shall mean the Delaware Limited Liability Company Act, Chapter 18 of the Delaware Code, as amended from time to time and any successor statute
 - 12.2 "AGREEMENT" shall mean this Operating Agreement, as amended, modified, or supplemented from time to time.
- 12.3 "CAPITAL CONTRIBUTIONS" shall mean the amount of cash and the agreed value of the property, the services, or the promissory note or other obligation to contribute cash or property or to perform services contributed by the Member for such Member's interest in the Company, equal to the sum of the Member's initial capital contributions plus the Member's additional capital contributions, if any, made pursuant to Sections 3.1 and 3.2, respectively.
 - 12.4 "CODE" shall mean the Internal Revenue Code of 1986 and the regulations promulgated thereunder, as amended, or any corresponding provision of any succeeding law.
 - 12.5 "COMPANY" shall mean the limited liability company organized in accordance with the Act and this Agreement.
 - 12.6 "MANAGER" shall mean LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company and any Person who is subsequently admitted as a member of the Company.
 - 12.7 "MEMBER" shall mean LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company and any successor Manager of the Company.
 - 12.8 "MEMBERSHIP INTEREST" shall mean a Person's share of the profits and losses of, and the right to receive distributions from, the Company.
- 12.9 "MEMBERSHIP RIGHTS" shall mean all of the rights of a Member in the Company, including a Member's: (i) Membership Interest; (ii) right to inspect the Company's books and records; (iii) right to participate in the management of the Company and vote on matters coming before the Company; and (iv) unless this Agreement or the Certificate of Formation provide to the contrary, right to act as an agent of the Company.
 - 12.10 "PERSON" shall mean and includes an individual, corporation, partnership, association, limited liability company, trust, estate, or other entity.

ARTICLE XIII AMENDMENT

This Agreement may not be altered or modified except by the written consent of the Member and the Company.

7

ARTICLE XIV GENERAL PROVISIONS

- 14.1 CAPTIONS. Section titles or captions contained in this Agreement are inserted only as a matter of convenience and for reference and in no way define, limit, extend or describe the scope of this Agreement, or the intent of any provision hereof.
 - 14.2 VARIATIONS OF PRONOUNS. All pronouns and any variations thereof shall be deemed to refer to the masculine, feminine, neuter, singular or plural as the identity of the Person may in the context require.
 - 14.3 **CONSTRUCTION**. This Agreement shall be interpreted in accordance with the laws of the State of Delaware.
 - 14.4 BINDING EFFECT. This Agreement shall be binding upon and inure to the benefit of the undersigned, its legal representatives, heirs, successors and assigns.
- 14.5 <u>SEVERABILITY</u>. Every provision of this Agreement is intended to be severable. If any term or provision is illegal or invalid for any reason, such illegality or invalidity will not affect the validity of the remainder of this Agreement.
- 14.6 COUNTERPART SIGNATURES. This Agreement may be executed in one or more counterparts (and by facsimile signatures) and, notwithstanding that all of the parties did not execute the same counterpart, each of such counterparts shall, for all purposes, be deemed to be an original, and all of such counterparts shall constitute one and the same instrument, binding on all of the parties hereto.

IN WITNESS WHEREOF, the parties have executed this Agreement, effective as of the date first set forth above.

	9
MEMBER AND MANAGER:	COMPANY:
LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company	LIQUIDMETAL COATINGS SOLUTIONS, LLC, a Delaware limited liability company
	By Its Manager:
	LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company
By:	Ву:
Name:	Name:
Title:	Title:

8

SCHEDULE "A"

INITIAL CAPITAL CONTRIBUTION OF MEMBER

Description of Item Contributed	Agreed Valu	e of Contribution
Cash	\$	10.00



The First State

I, HARRIET SMITH WINDSOR, SECRETARY OF STATE OF THE STATE OF DELAWARE, DO HEREBY CERTIFY THE ATTACHED IS A TRUE AND CORRECT COPY OF THE CERTIFICATE OF FORMATION OF "LIQUIDMETAL COATINGS, LLC", FILED IN THIS OFFICE ON THE NINTH DAY OF JULY, A.D. 2007, AT 11:49 O' CLOCK A.M.

[SEAL]

/s/ Harriet Smith Windsor

Harriet Smith Windsor, Secretary of State

DATE: 07-09-07

AUTHENTICATION: 5826271

070791788

1

CERTIFICATE OF FORMATION

OF

LIQUIDMETAL COATINGS, LL C

The undersigned hereby executes this Certificate of Formation of Liquidmetal Coatings, LLC (the "Company") for the purpose of forming a limited liability company pursuant to Delaware Limited Liability Company Act, 6 Del. C. § 18.101 et seq.

The name of the Company is Liquidmetal Coatings, LLC.

The address of the registered office of the Company is 615 South DuPont Highway, Kent County, Dover, Delaware 19901.

The Company's registered agent at such registered office address is CorpDirect Agents, Inc.

IN WITNESS WHEREOF, the undersigned has duly executed this Certificate of Formation on the 9th day

of July, 2007.

4385035 8100

/s/ Curt P. Creely

Curt P. Creely, Esq. Authorized Person

Address: c/o Foley & Lardner LLP 100 North Tampa Street

Suite 2700 Tampa, Florida 33602

Exhibit C-2



The First State

I, HARRIET SMITH WINDSOR, SECRETARY OF STATE OF THE STATE OF DELAWARE, DO HEREBY CERTIFY THE ATTACHED IS A TRUE AND CORRECT COPY OF THE CERTIFICATE OF FORMATION OF "LIQUIDMETAL COATINGS SOLUTIONS, LLC", FILED IN THIS OFFICE ON THE NINTH DAY OF JULY, A.D. 2007, AT 11:52 O' CLOCK A.M.

[SEAL]

/s/ Harriet Smith Windsor Harriet Smith Windsor, Secretary of State

AUTHENTICATION: 5826272 4385038 8100

070791815

DATE: 07-09-07

CERTIFICATE OF FORMATION

OF

LIQUIDMETAL COATINGS SOLUTIONS, LLC

The undersigned hereby executes this Certificate of Formation of Liquidmetal Coatings Solutions, LLC (the "Company") for the purpose of forming a limited liability company pursuant to Delaware Limited Liability Company Act, 6 Del. C. §§ 18.101 et seq.

The name of the Company is Liquidmetal Coatings Solutions, LLC.

The address of the registered office of the Company is 615 South DuPont Highway, Kent County, Dover, Delaware 19901.

The Company's registered agent at such registered office address is CorpDirect Agents, Inc.

IN WITNESS WHEREOF, the undersigned has duly executed this Certificate of Formation on the 9th day of July, 2007.

/s/ Curt P.Creely Curt P. Creely, Esq. Authorized Person

Address: c/o Foley & Lardner LLP 100 North Tampa Street Suite 2700 Tampa, Florida 33602

EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT (this "Agreement") is made and entered into effective as of July 24, 2007 (the "Effective Date"), by and between LIQUIDMETAL COATINGS, LLC, a Delaware limited liability company (the "Company"), and LARRY BUFFINGTON, an individual residing in the State of Texas (the "Employee").

RECITAL

WHEREAS, the Employee desires to be employed by the Company upon the terms and conditions set forth in this Agreement; and

WHEREAS, the Company desires to assure itself of the Employee's continued employment in the capacities set forth herein.

NOW, THEREFORE, in consideration of the foregoing recitals and for other good and valuable consideration, the parties hereto covenant and agree as follows:

- 1. Employment. The Company hereby employs Employee, and the Employee hereby accepts such employment, upon the terms and conditions set forth in this Agreement.
- 2. Term. Subject to the terms and conditions of this Agreement, including, but not limited to, the provisions for termination set forth in Section 5 hereof, the employment of the Employee under this Agreement shall commence on the Effective Date and shall continue through the close of business on fifth (5th) anniversary of the Effective Date (the "Initial Term"). Upon the expiration of the Initial Term, the Employee's employment with the Company will continue on an "at-will" basis and may be terminated by Employee or the Company for any reason and at any time, provided that the terminating party shall provide at least ninety (90) days prior written notice of the termination to the other party (unless the termination is for Cause as defined in this Agreement, in which case the Employee's employment may be terminated immediately). Notwithstanding the expiration or termination of Employee's employment, the provisions of this Agreement other than those of Sections 2, 4, and 5 shall remain in full force and effect. All other provisions of this Agreement, including but without limitation, Sections 1, 6, and 7, shall survive the expiration of the Initial Term. Notwithstanding the expiration of the this Agreement or the termination of employment by any party, Sections 2, 6, and 7 of this Agreement shall survive and remain fully enforceable.
- 3. <u>Duties.</u> Employee will initially serve as President and Chief Executive Officer of the Company and report to the Board of Managers of the Company. Employee will be based at the Company's principal executive offices located in or around Houston, Texas. Employee will devote the necessary business time, attention, skill, and energy to the business of the Company, will use the Employee's best efforts to promote the success of the Company's business, and will cooperate fully with the Board of Managers in the advancement of the best interests of the Company. Furthermore, the Employee shall assume and competently perform such reasonable responsibilities and duties as may be assigned to the Employee from time to time by the Board of Managers of the Company or their designee. To the extent that the Company shall have any parent company, subsidiaries, affiliated corporations, partnerships, or joint ventures other than

Liquidmetal Technologies, Inc. (collectively "Related Entities"), the Employee shall perform such duties to promote these entities and to promote and protect their respective interests to the same extent as the interests of the Company without additional compensation. In addition, Employee may, upon the agreement of Liquidmetal Technologies, Inc., continue to serve as President and Chief Executive Officer of Liquidmetal Technologies, Inc.

4. Compensation.

- (a) Annual Base Salary. As compensation for Employee's services and in consideration for the Employee's covenants contained in this Agreement, the Company shall pay the Employee an annual base salary of \$250,000.00. The annual base salary may be adjusted upward (but not downward) in the sole discretion of the Board of Managers of the Company. For purposes of this Agreement, the term "Salary Year" means the one year, 365-day period (or 366 day period for a leap year) that begins on the Effective Date and each successive one year period thereafter.
- (b) Bonuses. In addition to the Employee's annual base compensation, during each fiscal year during the term of the Employee's employment hereunder, the Employee shall be entitled to an annual performance bonus upon terms and conditions more particularly established by the Board of Managers of the Company. The annual performance bonus will be in an amount up to 50% of Employee's base salary (the "Maximum Bonus Amount"), payable as follows: (i) 25% of the Maximum Bonus Amount will be paid if the Company meets its EBITDA target (as established by the Board of Managers) for the applicable fiscal year, (ii) an additional 25% of the Maximum Bonus Amount will be paid if the Company meets its debt repayment target (as established by the Board of Managers) for the applicable fiscal year, and (iii) the remaining Maximum Bonus Amount (or any portion thereof) will be paid only at the discretion of the Board of Managers based on any factor or consideration that the Board of Managers, in its sole discretion, deems relevant.
- (c) Reimbursement of Expenses. The Employee shall be reimbursed for all reasonable and customary travel and other business expenses incurred by Employee in the performance of Employee's duties hereunder, provided that such reimbursement shall be subject to, and in accordance with, any expense reimbursement policies and/or expense documentation requirements of the Company that may be in effect from time to time.
- (d) Other Benefits. During the term of the Employee's employment hereunder, the Employee shall be eligible to participate in a health insurance plan selected by the Company. The Company shall pay 100 % of the premium of such health insurance plan. Also, Employee will, during his employment hereunder, be entitled to an automobile allowance in the amount of \$800.00 per month.
- (e) <u>Vacation</u>. The Employee shall be entitled to four (4) weeks paid vacation during each Salary Year during the term of the Employee's employment hereunder. Vacation shall be taken at such times and with such notice so as to not disrupt or interfere with the business of the Company. Unused vacation from a particular Salary Year will carry over to succeeding Salary Years, but only up to a maximum of four (4) weeks may be carried into any succeeding Salary Year.

2

5. Termination.

- (a) Death. The Employee's employment under this Agreement shall terminate immediately upon Employee's death. In the event of a termination pursuant to this Section 5(a), the Employee's estate shall be entitled to receive any unpaid base salary, bonuses or benefits owing to Employee up through and including the date of the Employee's death.
- (b) <u>Disability.</u> If, during the term of the Employee's employment hereunder, the Employee becomes physically or mentally disabled in the determination of a physician appointed or selected by the Company, or, if due to any physician or mental condition, the Employee becomes unable for a period of more than sixty (60) days during any six-month period to perform Employee's duties hereunder on substantially a full-time basis as determined by a physician selected by the Company, the Company may, at its option, terminate the Employee's employment upon not less than thirty (30) days written notice. In the event of a termination pursuant to this Section 5(b), the Employee shall be entitled to receive any unpaid base salary, bonuses or benefits owing to Employee up through and including the effective date of termination.
- (c) Termination By Company With Cause. The Company may terminate the Employee's employment at any time with Cause. As used in this Agreement, "Cause" shall include the following: (1) the Employee's failure or inability to perform Employee's duties under this Agreement; (2) dishonesty or other serious misconduct, (3) the commission of an unlawful act material to Employee's employment, (4) a material violation of the Company's policies or practices which reasonably justifies immediate termination; (5) committing, pleading guilty, nolo contender or no contest (or their equivalent) to, entering into a pretrial intervention or diversion program regarding, or conviction of, a felony or any crime or act involving moral turpitude, fraud, dishonesty, or misrepresentation; (6) the commission by the Employee of any act which could reasonably affect or impact to a material degree the interests of the Company or Related Entities or in some manner injure the reputation, business, or business relationships of the Company or Related Entities; (7) the Employee's inability to perform an essential function of Employee's position; or (8) any material breach by Employee of this Agreement. The Company may terminate this Agreement for Cause at any time without notice. In the event of a termination for Cause, the Company shall be relieved of all its obligations to the Employee provided for by this Agreement as of the effective date of termination, and all payments to the Employee hereunder shall immediately cease and terminate as of such date, except that Employee shall be entitled to receive any unpaid base salary, bonuses or benefits owing to Employee up to and including the effective date of termination, provided, however, that the Employee's obligations under Sections 6 and 7 shall survive such a termination for Cause and any liabilities or obligations which have accrued and are owed by the Employee to the Company shall not be extinguished or released thereby.
- (d) <u>Termination By Company Without Cause</u>. In addition to the other termination provisions of this Agreement, the Company may terminate the Employee's employment at any time and for any reason and without cause (a "Termination Without Cause"). In the event of a Termination Without Cause, the Employee shall continue to receive the Employee's base salary (as then in effect) through the later of one year after the date of the

3

Termination Without Cause or the end of the Initial Term (the "Severance Period"). In addition to the severance pay described in the preceding sentence, the Employee shall continue to receive, during the Severance Period, all employee health and welfare benefits that Employee would have received during the Severance Period in the absence of such termination, but only to the extent that the Company is permitted to continue providing such benefits after termination in accordance with the applicable employee benefit plans. Employee agrees and acknowledges, however, that Employee will forfeit the right to receive base salary and benefits during the Severance Period immediately upon the Employee's breach of any covenant set forth in Section 6 of this Agreement. The Employee will also forfeit the right to salary and benefits during the Severance Period upon accepting employment with another employer with comparable salary and benefits hereunder shall be forfeited and shall cease upon the Employee becoming eligible for benefits from the Employee's new employer. Notwithstanding the foregoing, the termination of the Employee's employment pursuant to the second sentence of Section 2 of this Agreement shall not constitute a Termination Without Cause and shall not give rise to any severance payment or other benefits pursuant to this Section 5(d).

6. Noncompetition, Nonsolicitation, and Nondisclosure Covenants.

(a) Rationale for Restrictions. Employee acknowledges that Employee's services hereunder are of a special, unique, and extraordinary character, and Employee's position with the Company places Employee in a position of confidence and trust with customers, suppliers, and other persons and entities with whom the Company and its Related Entities have a business relationship. The Employee further acknowledges that the rendering of services under this Agreement will require the disclosure to Employee of Confidential Information (as defined below) including Trade Secrets of the Company relating to the Company and/or Related Entities, and the Company hereby agrees to provide such Confidential Information. As a consequence, the Employee agrees that it is reasonable and necessary for the protection of the goodwill and legitimate business interests of the Company and Related Entities that the Employee make the covenants contained in this Section 6, that such covenants are a material inducement for the Company to employ the Employee and to enter into this Agreement, and that the covenants are given as an integral part of and incident to this Agreement.

(b) Noncompetition and Nonsolicitation Covenants. As used herein, the term "Restrictive Period" means the time period commencing on the Effective Date of this Agreement and ending on the second (2nd) anniversary of the date on which the Employee's employment by the Company (or any Related Entity) expires or is terminated for any reason, including both a termination by the Company for Cause and a Termination Without Cause. In

addition, the term "Covered Business" means any business which is the same as, or similar to, any business conducted by the Company or any of the Related Entities at any time during the Restrictive Period. The Employee agrees that the Employee will not engage in any of the following acts anywhere in the world during the Restrictive Period:

(i) directly or indirectly engage or invest in; own, manage, operate, finance, control, or participate in the ownership, management, operation, financing, or control of; be employed by, associated with, or in any manner connected with; lend the Employee's name or any similar name to; lend

4

Employee's credit to; or render services or advice to, any business which competes with, is engaged in, or carries on any aspect of a Covered Business;

- (ii) directly or indirectly assist, promote or encourage any existing or potential employees, customers, clients, or vendors of the Company or any Related Entity, as well as any other parties which have a business relationship with the Company or a Related Entity, to terminate, discontinue, or reduce the extent of their relationship with the Company or a Related Entity;
- (iii) directly or indirectly solicit business of the same or similar type as a Covered Business, from any person or entity known by the Employee to be a customer or client of the Company, whether or not the Employee had contact with such person or entity during the Employee's employment with the Company;
- (iv) disparage the Company, any Related Entities, and/or any shareholder, director, officer, employee, or agent of the Company or any Related Entity; and/or
- (v) engage in any practice the purpose of which is to evade the provisions of this Section 6 or commit any act which adversely affects the Company, any Related Entity, or their respective businesses.

Employee acknowledges that Employee's services hereunder are of a special, unique, and extraordinary character, and Employee's position with the Company places Employee in a position of confidence and trust with customers, suppliers, and other persons and entities with whom the Company and its Related Entities have a business relationship. The Employee further acknowledges that the rendering of services under this Agreement will likely require the disclosure to Employee of Confidential Information (as defined below) and Trade Secrets (as defined below) of the Company relating to the Company and/or Related Entities. As a consequence, the Employee agrees that it is reasonable and necessary for the protection of the goodwill and legitimate business interests of the Company and Related Entities that the Employee make the covenants contained in this Section 6, that such covenants are a material inducement for the Company to employ the Employee agrees that it he geographic scope of the above covenants is a reasonable means of protecting the Company's (and the Related Entities') legitimate business interests. Notwithstanding the foregoing covenants, nothing set forth in this Agreement shall prohibit the Employee from owning the securities of (i) corporations which are listed on a national securities exchange or traded in the national over-the-counter market in an amount which shall not exceed 5% of the outstanding shares of any such corporation or (ii) any corporation, partnership, firm or other form of business organization which does not compete with, is not engaged in, and does not carry on any aspect of, either directly or indirectly through a subsidiary or otherwise, any Covered Business.

5

- (c) <u>Disclosure of Confidential Information</u>. The Employee acknowledges that the inventions, innovations, software, Trade Secrets, business plans, financial strategies, finances, and all other confidential or proprietary information with respect to the business and operations of the Company and Related Entities are valuable, special, and unique assets of the Company. Accordingly, the Employee agrees not to, at any time whatsoever either during or after the Employee's term of employment with the Company, disclose, directly or indirectly, to any person or entity, or use or authorize any person or entity to use, any confidential or proprietary information with respect to the Company or Related Entities without the prior written consent of the Company, including, without limitation, information as to the financial condition, results of operations, identities of clients or prospective clients, products under development, acquisition strategies or acquisitions under consideration, pricing or cost information, marketing strategies, passwords or codes or any other information relating to the Company or any of the Related Entities which could be reasonably regarded as confidential (collectively referred to as "Confidential Information"). However, the term "Confidential Information" does not include any information which is or shall become generally available to the public other than as a result of disclosure by the Employee or by any person or entity which the Employee reasonably should know) has a duty of confidentiality to the Company or a Related Entities which to all of the protections and benefits afforded by the any applicable law. "Trade Secret" shall mean information, including a formula, pattern, compliation, program, device, method technique, or process that derives independent economic value, actual or potential, from being not generally known to, and not being readily ascertainable by proper means by, other persons who can derive economic value from its disclosure or use, including but not limited to the pat
- (d) Removal and Return of Proprietary Items. The Employee will not remove from the Company's premises (except to the extent such removal is for purposes of the performance of the Employee's duties at home or while traveling, and under such conditions and restrictions as are specifically authorized and/or required by the Company) or transmit by any means, electronic or otherwise, any document, record, notebook, plan, model, component, device, computer software or code, or Confidential Information or Trade Secret whether embodied in a disk or in any other form, including electronic form (collectively, the "Proprietary Items"). The Employee recognizes that, as between the Company and the Employee, all of the Proprietary Items, whether or not developed by the Employee, are the exclusive property of the Company. Upon termination of Employee's employment with the Company by either party (regardless of the reason for termination), or upon the request of the Company during the term of employment, the Employee will return to the Company all of the Proprietary Items in the Employee's possession or subject to the Employee's control, and the Employee shall not retain any copies, abstracts, sketches, or other physical embodiment of any of the Proprietary Items, Confidential Information, Trade Secret or any part thereof.
- (e) <u>Enforcement and Remedies.</u> In the event of any breach of any of the covenants set forth in this Section 6, the Employee recognizes that the remedies at law will be inadequate and that in addition to any relief at law which may be available to the Company for

6

such violation or breach and regardless of any other provision contained in this Agreement, the Company shall be entitled to equitable remedies (including an injunction) and such other relief as a court may grant after considering the intent of this Section 6. Additionally, the period of time applicable to any covenant set forth in this Section 6 will be extended by the duration of any violation by Employee of such covenant. In the event a court of competent jurisdiction determines that any of the covenants set forth in this Section 6 are excessively broad as to duration, geographic scope, prohibited activities or otherwise, the parties agree that this covenant shall be reduced or curtailed to the extent, but only to the extent, necessary to render it enforceable.

Employee Inventions.

- (a) <u>Definition</u>. For purposes of this Agreement, "Employee Invention" means any idea, invention, technique, modification, process, or improvement (whether patentable or not), any industrial design (whether registerable or not), any mask work, however fixed or encoded, that is suitable to be fixed, embedded or programmed in a semiconductor product (whether recordable or not), and any work of authorship (whether or not copyright protection may be obtained for it) created, conceived, or developed by the Employee, either solely or in conjunction with others, during the Employee's employment with the Company or during the ninety (90) day period following such employment, that relates in any way to, or is useful in any manner in, the businesses then being conducted or proposed to be conducted by the Company or any Related Entity.
- (b) Ownership of Employee Inventions. Employee agrees and acknowledges that all Employee Inventions will belong exclusively to the Company and that all Employee Inventions are works made for hire and the property of the Company, including any copyrights, patents, semiconductor mask protection, or other intellectual property rights pertaining thereto. If it is determined that any such works are not works made for hire, the Employee hereby assigns to the Company all of the Company's right, title, and interest, including all rights of copyright, patent, semiconductor mask protection, and other intellectual property rights, to or in such Employee Inventions. The Employee covenants that the Employee will promptly:
 - (i) disclose to the Company in writing any Employee Invention;
 - (ii) assign to the Company or to a party designated by the Company, at the Company's request and without additional compensation, all of the Employee's right to the Employee Invention for the United States and all foreign jurisdictions;
 - (iii) execute and deliver to the Company such applications, assignments, and other documents as the Company may request in order to apply for and obtain patents or other registrations with respect to any Employee Invention in the United States and any foreign jurisdictions;
 - (iv) sign all other papers necessary to carry out the above obligations; and

7

- (v) give testimony and render any other assistance in support of the Company's rights to any Employee Invention.
- 8. <u>Essential and Independent Covenants</u>. The Employee's covenants in Sections 6 and 7 of this Agreement are independent covenants, and the existence of any claim by the Employee against the Company under this Agreement or otherwise will not excuse the Employee's breach of any covenant in Section 6 or 7. The covenants of Sections 6 and 7 shall survive the termination, extinguishment, or lapse of this Agreement under any circumstances, even if this Agreement is terminated by either party, whether for Cause or Not for Cause.
- 9. Representations and Warranties by The Employee. The Employee. The Employee represents and warrants to the Company that the execution and delivery by the Employee of this Agreement do not, and the performance by the Employee of the Employee's obligations hereunder will not, with or without the giving of notice or the passage of time, or both: (a) violate any judgment, writ, injunction, or order of any court, arbitrator, or governmental agency applicable to the Employee, or (b) conflict with, result in the breach of any provisions of or the termination of, or constitute a default under, any agreement to which the Employee is a party or by which the Employee is or may be bound, including, without limitation, any noncompetition agreement or similar agreement. Employee further represents and warrants that he fully and completely understands this Agreement and that he has engaged in negotiations with the Company and has either consulted with an attorney of his choice or has had ample opportunity to do so and is fully satisfied with the opportunity he has had.
- 10. Notices. For purposes of this Agreement, notices and all other communications provided for herein shall be in writing and shall be deemed to have been duly given when hand-delivered, sent by facsimile transmission (as long as receipt is acknowledged), or mailed by United States certified or registered mail, return receipt requested, postage prepaid, addressed to the address or facsimile number for each party set forth on the signature page hereto, or to such other address or facsimile number as either party may have furnished to the other in writing in accordance herewith, except that a notice of change of address shall be effective only upon receipt.

by any other party hereto shall be deemed a waiver of any similar or dissimilar term or condition at the same or at any Employee's employment by the Company, and there are no agreements or representations, oral or otherwise, expressed This Agreement shall be binding upon, and inure to the benefit of, the Company, its respective successors and assigns,	or implied, with respect to or related to the employment of the Employee which are not set forth in this Agreement. and the Employee and Employee's heirs, executors, administrators and legal representatives. The duties and covenants prior written consent of the Company, and any attempted delegation or assignment without such prior written consent
	8
	nce of this Agreement shall be governed by the laws of the State of Texas without regard to principles of choice of law
or conflicts of law thereunder. Any action or proceeding seeking to enforce any provision of, or based on any right ari Harris or Montgomery, or, if it has or can acquire jurisdiction, in the federal courts located in, Harris County or Montg courts) in any such action or proceeding and waives any objection to venue laid therein. Process in any action or proce agree that having venue and jurisdiction solely in Texas is reasonable in that the headquarters for the Company will be PARTIES HEREBY WAIVE A JURY TRIAL IN ANY LITIGATION ARISING UNDER OR RELATING TO THIS I be construed against either party but shall be construed without regard to the participation of either party in the drafting	omery County, Texas, and each of the parties consents to the jurisdiction of such courts (and of the appropriate appellate eding referred to in the preceding sentence may be served on either party anywhere in the world. The parties hereto in Harris County or Montgomery County, Texas and that site for litigation is the most central for such matters. THE AGREEMENT OR THE EMPLOYMENT OF THE EMPLOYEE WITH THE COMPANY. This Agreement shall not
13. <u>Counterparts; Facsimile Signatures</u> . This Agreement may be executed in one or more counterparts: Agreement may be effective upon the execution and delivery by any party hereto of facsimile copies of signature covenants and agrees to deliver promptly after the date hereof two (2) original copies to the other party hereto.	rts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. pages hereto duly executed by such party; <u>provided</u> , <u>however</u> , that any party delivering a facsimile signature page
Company agree that the unenforceable provision or Section should not cause the entire Agreement to become unenforceal	ny law of the state of Texas or is for some other reason unenforceable as written in the state of Texas, the Employee and the ble unless it is caused to fail in its essential purpose. In the event that any provision or Section of this Agreement violates set that the provision should be reduced in scope or length or otherwise modified by the Court, if possible under the law, to num protection available to it under the law.
[signatu	res follow]
	9
IN WITNESS WHEREOF, the parties have executed this Agreement as of the day and year first above write	ten.
	LIQUIDMETAL COATINGS, LLC
	By:
	Liquidmetal Coatings, LLC 30452 Esperanza Rancho Santa Margarita, CA 92688 Attention: Chief Financial Officer Fax No.: (949) 635-2108 (initial address until Houston office is opened)
	EMPLOYEE
	By:
	Printed Name: Larry Buffington
	Address and Facsimile Number:
	Larry Buffington 2205 Riva Row, Unit 2410 The Woodlands, TX 77380 Fax: 832-585-0341
	10
	Exhibit E-
	AGREEMENT Coatings, LLC)

1. <u>Grant of Security Interest.</u> Liquidmetal Coatings, LLC, a Delaware limited liability company (**"Debtor"**), in order to induce C₃ Capital Partners, L.P., a Delaware limited partnership ("C₃") and C₃ Capital Partners II, L.P., a Delaware limited partnership ("C₃"), together with C₃ shall hereinafter be referred to as the "**Lender**"), to extend certain financial accommodations to Debtor and Liquidmetal Coatings Solutions, LLC for the purpose, inter alia, of acquiring certain assets of Liquidmetal Technologies, Inc. and in consideration thereof and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, hereby transfers, assigns, and grants to Lender a continuing and irrevocable security interest and general lien in and to all of the following property and rights of Debtor:

- (a) All now owned or hereafter acquired Accounts, accounts receivable, other receivables, any right to payment of a monetary obligation, whether or not earned by performance, leases and lease payments, contract rights, any other obligations or indebtedness owed to Debtor from whatever source arising; all other rights of Debtor to receive performance or any payments in money or in kind, whether or not earned by performance, all guaranties, security interests and Supporting Obligations of any of the foregoing and insurance policies and proceeds relating thereto, and all rights of Debtor as an unpaid seller of Goods and services, including, but not limited to, the rights to stoppage in transit, replevin, reclamation, and resale, and rights to payment for money or funds advanced or sold. The rights and property described in this Section 1(a) are referred to herein collectively as the "Accounts Collateral."
- All now owned or hereafter acquired Inventory, merchandise, raw materials, goods in process, work in progress, materials used or consumed in a business, finished goods, findings or component materials, and all supplies, incidentals, office supplies, packaging materials, and any and all property or items used or consumed in the operation of the business of Debtor or which contribute to the finished products or to the sale, promotion and shipment thereof, As-Extracted collateral, all property leased by Debtor, held by Debtor for sale or lease or to be furnished under a contract for service and all Documents evidencing any part of any of the foregoing. The rights and property described in this Section 1(b) are referred to herein collectively as the "Inventory Collateral."
- All now owned or hereafter acquired Equipment, goods other than Inventory Collateral, parts, computers, including data, hardware and software, machinery, fixtures, furniture, furnishings, tools, dies, aircraft, vessels and vehicles of every kind and description, whether or not titled, and all parts and accessories for or relating to any of the foregoing. The rights and property described in this Section 1(c) are referred to herein collectively as the "Equipment Collateral."
- All now owned or hereafter acquired General Intangibles, all claims and causes of action, and all other intangible personal property of Debtor of every kind and nature, whether registered or unregistered, Payment Intangibles, corporate or other business records, all books, mailing and customer lists, ledgers, books of account,

records, writings, data bases, software, information and data however stored or embedded, inventions, designs, blueprints, plans specifications, permits, leases, contracts, governmental approvals, franchises, applications and renewals of any of the foregoing, privileges, rights, tax refunds and tax claims, any swap, hedging or derivatives agreements, insurance proceeds, pension and insurance surpluses. The rights and property described in this Section 1(d) are referred to herein as the "General Intangibles Collateral."

All now owned or hereafter acquired, Chattel Paper, Instruments, Notes, Promissory Notes, Deposit Accounts, Investment Property, Securities described in SPA, letters of credit, Letter-of-Credit Rights, Documents, Payment Intangibles, Financial Assets, all Supporting Obligations for any of the foregoing ("Other Property Collateral").

(f) All proceeds including proceeds and products of all of the foregoing and all additions and accessions to, replacements and substitutions of, insurance policies and payments, condemnation proceeds of, and documents covering all of the foregoing, all property received wholly or partly in trade or exchange for all of the foregoing, and all income, rents, revenues, distributions, issues, profits, cash or non-cash proceeds and accessions arising from the sale, lease, license, encumbrance, collection, or any other temporary or permanent disposition of any of the foregoing or any interest therein (the "**Proceeds**").

Capitalized terms used and not defined in this Section 1 shall have the meanings given to them in the Uniform Commercial Code as adopted and in force in the State of Missouri, as from time to time amended.

The Accounts Collateral, Inventory Collateral, Equipment Collateral, General Intangibles Collateral, Other Property Collateral and Proceeds are collectively referred to herein as the "Collateral."

- 2. <u>Proceeds</u>. The security interests granted to Lender in any proceeds or other property arising out of the disposition of the Collateral and anything contained herein or in any financing statement shall not be deemed permission or assent by Lender to any sale of disposition of the Collateral except to the extent expressly provided herein.
- 3. Indebtedness Secured. The security interest granted hereby is to secure payment in full of (a) any and all sums from time to time due from Debtor to Lender, any instruments evidencing the indebtedness of Debtor to Lender and the full and complete performance of all agreements and documents executed or delivered pursuant to any indebtedness due from Debtor to Lender, all as same may be amended, modified or extended from time to time, (b) any other indebtedness of Debtor, whether evidenced by instruments executed by Debtors or not, payable and owing to Lender as provided by the terms of any such instrument, (c) all advances made by Lender to discharge taxes or levies on, or made for repairs to, maintenance of, or insurance on, the Collateral, (d) all money or other credit heretofore and hereafter advanced by Lender to or for the account of Debtor, (e) all other present or future, direct or contingent, liabilities of Debtor to Lender of any nature whatsoever, and (f) all costs and expenses incurred in the collection of the

2

foregoing, including representation in any bankruptcy proceedings, including reasonable attorney's fees (all of the above being referred to, collectively, as the "Obligations")

It is the true, clear, and express intention of Debtor that the continuing grant of this security interest remain as security for payment and performance of the Obligations, whether now existing, or which may hereinafter be incurred, or whether or not contemplated by the parties at the time of the granting of this security interest. The notice of the continuing grant of this security interest, therefore, shall not be required to be stated on the face of any document representing any Obligations, nor otherwise identify it as being secured hereby. Any Obligations shall be deemed to have been made pursuant to Section 400.9-204 of the Uniform Commercial Code of Missouri. This Security Agreement shall terminate, and the security interest and liens granted hereunder shall terminate, upon Debtor's repayment in full to Lender of the Obligations.

4. <u>Debtor's Name, Place of Business and Location of Collateral</u>. (a) Debtor's chief executive office is located in Rancho Santa Margarita, California and all Collateral locations are in Montgomery County, Texas and Houston County, Alabama (b) Debtor's State of organization is Delaware and Debtor shall not change its respective State of organization until such time as all outstanding Obligations to Lender have been satisfied in full; and (c) Debtor's exact legal name is as first provided above.

Collateral shall not be attached to any real estate ("Real Property"). Debtor agrees to notify Lender in writing of any intended sale, mortgage or conveyance of any Real Property on which the Collateral is located and to give written notice of the terms and conditions of this Security Agreement to any prospective purchaser, mortgagee or grantee of said Real Property and a copy of such notice to Lender.

When any Collateral is in the possession of a third party, Debtor will join with Lender in notifying the third party of Lender's security interest and obtaining an acknowledgement from the third party that it is holding the Collateral for the benefit of Lender. Debtor will obtain control agreements in form satisfactory to Lender as deemed necessary by Lender for purposes of further perfecting or enforcing the security interests of Lender hereunder. Debtor shall not create any Chattel Paper or certificated Collateral without delivering same to Lender or placing a legend on the Chattel Paper acceptable to Lender indicating that Lender has a security interest in the Chattel Paper.

- 5. <u>Collateral Use</u>. The Collateral shall be kept in good order and repair and Debtor will not permit waste or do anything to impair the value of the Collateral or any part thereof, reasonable wear and tear excepted, or use or permit others to use the Collateral in violation of any insurance policy covering the Collateral or any statute, ordinance or state or federal regulation. Debtor shall give Lender immediate written notice of any damage, destruction, theft, loss or the occurrence of any event which materially impairs the value of the Collateral.
- 6. Adverse Security Interests and Liens. Debtor is, or, to the extent that the Collateral will be acquired after the date hereof, will be, the owner of the Collateral free from any and all liens, security interests or encumbrances other than the lien proposed to be granted pursuant to the Senior Revolving Credit Facility described in Section 3.10 of the Securities Purchase Agreement between Debtor and Lender entered into simultaneously herewith (the "SPA");

3

Debtor shall not transfer or assign any interest in this Security Agreement or the Collateral; and Debtor, at Debtor's expense, will defend the Collateral against all claims and demands of all other persons at any time claiming the same or an interest therein. There is no financing statement now on file in any public office covering the Collateral, or intended so to be, or in which Debtor is named or signed as debtor, and Debtor will not execute and there will not be on file in any public office any financing statement or statements covering the Collateral except (a) the financing statement to be filed in respect of and for the security interest in Lender hereby granted or provided for and (b) any financing statement in favor of Senior Lender (as defined in the SPA). With respect to the security interest of the Senior Lender described herein, Lender has agreed to subordinate its interest to those of the Senior Lender and shall enter into an Intercreditor Agreement on customary and reasonable terms acceptable to Lender.

- 7. Insurance. Debtor, at Debtor's sole cost, shall at all times keep the Collateral insured against physical loss or damage with coverage to be in special coverage form, plus earthquake, if the Collateral is located in California, flood, if the Collateral is located in a 100 year flood zone as determined by FEMA and other hazards in an amount not less than the greater of (a) the full replacement cost or (b) such other amount as may reasonably from time to time be required by Lender, with no co-insurance clauses or deductibles in excess of \$25,000.00 or \$50,000.00 in the case of a discontinued operations product liability policy in the policies of insurance unless Lender shall consent thereto in writing, in such form, for such periods and written by such companies as may be satisfactory to Lender, payable to and protecting Lender for not less than the total insured amount. Debtor shall maintain combined form business interruption and extra expense coverage. In addition, Debtor shall maintain commercial general liability insurance in occurrence form with coverage limits of at least \$2,000,000.00 annual aggregate, \$1,000,000.00 per occurrence or as otherwise acceptable to Lender. All such insurance shall be carried by companies authorized to insure in the state in which the Collateral is located and which have an AM Best rating of A or better and are otherwise reasonably acceptable to Lender, and all such policies shall be in form acceptable to Lender. All property policies of insurance shall provide that Lender be co-loss payees and that the proceeds shall be paid to Lender and Debtor and that Lender shall be protected against loss from any act or neglect of Debtor or third parties, and such other endorsements as Lender may from time to time request. Debtor will promptly provide with evidence of such insurance. Such insurance shall require a minimum of thirty (30) days prior written notice to Lender of any cancellation thereof or any cancellation thereof or any cancellation thereof or any cancellation thereof or
 - 8. Records. The records concerning the Collateral will be kept at the address of the chief executive office indicated in Section 4 hereof. Lender may inspect such records or the

4

Collateral at any time at any address during reasonable business hours upon request. Debtor will not remove any part of such records from said location without the prior written consent of Lender.

9. Financing Statement and Others Acts. Debtor irrevocably authorizes Lender at any time and from time to time to file financing or continuation statements and/or amendments thereto, without the signature of Debtor, and Debtor shall execute and deliver such other instruments and documents as may be requested by Lender to perfect, confirm and further evidence the security interest and assignments hereby granted and shall pay the fees incurred in filing all such financing statements or other instruments or documents. If any applicable law requires the registration of the Collateral or the issuance of a certificate of title therefor or both, Debtor agrees to promptly comply with such law(s) and shall cause notice of the security interest of Lender to be shown on any such certificate of title and will join in executing such application for the title forms as Lender shall require.

Upon request of Lender, Debtor will promptly do all other acts and things, and will execute and file all other instruments deemed necessary by Lender under applicable law to establish, maintain and continue Lender's perfected security interest in the Collateral and to effectuate the intent of this Security Agreement and will pay all reasonable costs and expenses of filing and recording or promptly reimburse Lender therefor if such costs and expenses are incurred by Lender, including the costs of any searches deemed necessary by Lender to establish, determine or maintain the validity and the priority of the security interest of Lender, and pay or otherwise satisfy all other claims and charges which in the opinion of Lender might prejudice, imperil or otherwise affect the Collateral or Lender's security interest therein. A photocopy of this Agreement shall be deemed an original for purposes of filing or recording.

- 10. Taxes and Assessments. Debtor will pay promptly when due all taxes, assessments and other charges levied or assessed upon the Collateral or for its use or operation or upon this Security Agreement or upon any or other documents evidencing the Obligations secured hereby.
- 11. Collateral Certificates and Schedules. Debtor shall furnish to Lender from time to time, upon request, written statements, certificates and schedules identifying and describing the Collateral and any additions thereto and substitutions therefor in such detail as Lender may require and certified as to accuracy by the Board of Managers of Debtor.
 - 12. <u>Collateral Disposition</u>. Until default hereunder:
 - (a) Debtor may have possession of the Collateral and use it in any lawful manner not inconsistent with this Agreement or with any policy of insurance thereon;
 - (b) Debtor may sell the Inventory Collateral in the ordinary course of Debtor's business (excluding, however, transfers or dispositions on satisfaction of debt), and Debtor may use and consume raw materials or supplies, the use and consumption of which is necessary in order to carry on Debtor's business in the ordinary course; and
 - (c) Debtor will, at its own expense, collect, as and when due, all amounts due under the Accounts Collateral, including the taking of such action with respect to such collection as Lender may reasonably request or, in the absence of such request, as Debtor

may deem advisable, and may grant, in the ordinary course of Debtor's business, to any party obligated on any of the Accounts Collateral, any rebate, refund or adjustment to which such party may be lawfully entitled, and may accept, in connection therewith, the lawful return of goods, the sale or lease of which shall have given rise to such Accounts Collateral. Lender may, however, at any time and at Debtor's expense, notify any parties obligated on any of the Accounts Collateral to make payment directly to Lender of any amounts due or to become due thereunder and enforce collection of any of the Accounts Collateral by suit or otherwise and surrender, release or exchange all or any part thereof, or compromise, extend or renew the same for any period.

- 13. <u>Undertakings by Lender</u>. Lender may from time to time, at its sole option, and without notice to Debtor, perform any undertaking of Debtor hereunder which Debtor shall fail to perform and take any other action which Lender deems necessary for the maintenance or preservation of any of the Collateral or the interest of Lender therein (including, without limitation, the discharge of taxes or liens of any kind against the Collateral or the procurement of insurance) and Debtor agrees to forthwith reimburse Lender, on demand, for all expenses of Lender in connection with the foregoing, together with interest thereon at a per annum rate equal to the highest rate of interest applicable to any of the Obligations secured hereby, until reimbursed by Debtor and all amounts not so reimbursed shall be added to and become a part of the Obligations secured hereby. Lender may, for the foregoing purposes, act in its own name or that of Debtor and may also act for the purpose of adjusting or settling any policy of insurance on the Collateral, or endorsing any draft received in connection therewith. For all of the foregoing purposes, Debtor hereby grants to any officer of Lender its power of attorney, irrevocable so long as any of the Obligations secured hereby shall be outstanding.
- 14. Warranties Correct. Debtor hereby warrants and represents that all financial statements, certificates and schedules heretofore and hereafter delivered to Lender by or on behalf of Debtor, and any statement and data submitted in writing to Lender in connection with this Security Agreement or any Obligations of Debtor to Lender fairly present the financial condition of Debtor for the periods involved.
- 15. Identification of Collateral. Upon request of Lender, Debtor will stamp on its records concerning the Collateral, a notation, in form satisfactory to Lender, of the security interest of Lender hereunder, and when requested by Lender, Debtor shall further affits to the Collateral such signs or labels as shall be satisfactory to Lender to indicate the security interest of Lender in the Collateral. Upon request of Lender at any time, Debtor will deliver to Lender lists or copies of all Collateral promptly and will deliver to Lender, promptly upon receipt, all proceeds of Collateral received by Debtor, including proceeds of the Accounts Collateral referred to above, in the exact form in which they are received. To protect Lender's rights hereunder, Debtor will assign or endorse proceeds to Lender as Lender may request, and hereby constitute any officer or employee of Lender its true and lawful attorney-in-fact, with full power to endorse the name of Debtor upon any invoice, freight or express bill or bill of lading relating to any such accounts, upon drafts against account debtors and assignments and verifications of accounts and notices to account debtors, upon any and every remittance or instrument of payment, including checks, drafts and money orders, and in whatever form received, and to do and perform all other acts and things reasonably necessary, proper and requisite to carry out the intent of this Security Agreement. The power herein granted shall be deemed to be coupled with an interest and shall

6

not be revoked by Debtor until Lender has been paid all sums due it, including all proper expenses, with interest. All such items received by Lender for the Collateral shall be deposited to the credit of Debtor in an account designated and controlled by Lender, as security for the payment of the Obligations. Lender may, from time to time, in its discretion, (a) apply all of the then existing balance representing collected funds in such deposit account, toward payment of all of any part of the Obligations secured hereby, whether or not then due, in such order of application as Lender, in its sole discretion, may determine, or (b) permit Debtor to use all or part of said account in the normal course of Debtor's business.

- 16. Account Debtors. With respect to the Accounts Collateral, Lender may, following an Event of Default (as defined below) and after giving prior written notice to Debtor, notify account debtors that the accounts have been assigned to Lender and shall be paid to Lender. Upon request of Lender, following and during the continuance of an Event of Default, Debtor will so notify such account debtors and will indicate on all invoices to such account debtors that the accounts are payable directly to Lender.
 - 17. Default. Debtor shall be in default under this Security Agreement upon the occurrence of any one or more of the following events or conditions (each of which is an "Event of Default"):
 - (a) Failure to Make Payments. Debtor fails to pay when due and payable (whether at maturity or otherwise, and such failure continues uncured for 10 Business Days following notice by Lender via electronic mail to Debtor's President) the full amount of interest then accrued on the Notes due to Lender, or the full amount of any principal payment (together with any applicable premium) on any notes or any other amounts payable under the SPA with Lender or the Investment Documents with Senior Lender (the "Indebtedness");
 - (b) Failure to Observe Covenants. Debtor:
 - i. breaches, fails to perform or observe any of the covenants contained in Sections 5.1 through 5.11 inclusive of the SPA and such failure continues uncured for twenty (20) days following notice provided by Lender via electronic mail to Debtor's President, or as otherwise provided by Section 21(e), unless such period is otherwise extended by Lender; or
 - ii. breaches, fails to perform or observe any other provision contained in the Investment Documents and (1) if such failure has had or could have a Material Adverse Effect and such failure continues uncured for twenty (20) days following notice provided by Lender via electronic mail to Debtor's President, or as otherwise provided by Section 21(e) unless such period is otherwise extended by Lender, or (2) if such failure has not had and could not have a Material Adverse Effect and such failure continues uncured for thirty (30) days or such Debtor is not proceeding diligently to cure such failure. "Material Adverse Effect" means any matter or

7

matters which would, alone or in the aggregate, have a materially adverse effect on: (a) the operating results, prospects, assets, liabilities, operations, condition (financial or otherwise) or business of Debtor taken as a whole; or (b) the ability of such Debtor to perform any of its obligations under the Securities or the Investment Documents;

- (c) Representations. Any representation, warranty or information contained herein or required to be furnished to any holder of the Securities pursuant to the Investment Documents, or any writing furnished by either Debtor to any holder of the Securities, is false or misleading in any material respect on the date made;
- (d) Insolvency. Either Debtor makes an assignment for the benefit of creditors or admits in writing its inability to pay its debts generally as they become due, or an order, judgment, decree or injunction is entered adjudicating Debtor bankrupt or insolvent or requiring the dissolution or split up of Debtor or preventing Debtor from conducting all or any part of its business; or any order for relief with respect to Debtor is entered under the Federal Bankruptcy Code; or Debtor petitions or applies to any tribunal for the appointment of a custodian, trustee, receiver or liquidator of Debtor, or of any substantial part of the assets of Debtor, or commences any proceeding (other than a proceeding for the voluntary liquidation and dissolution of any of its Subsidiaries) relating to Debtor under any bankruptcy reorganization, arrangement, insolvency, readjustment of debt, dissolution or liquidation or similar laws of any jurisdiction now or hereafter in effect; or any such petition or application is filed, or any such proceeding is commenced, against Debtor and either (i) Debtor by any act indicates its approval thereof, consent thereto or acquiescence therein or (ii) such petition, application or proceeding is not dismissed within 60 days;
- (e) <u>Payments on Indebtedness</u>. Debtor shall fail to pay when due any principal of or interest on any Indebtedness or the maturity of any such Indebtedness shall have been accelerated, or any such Indebtedness shall have been required to be prepaid prior to the stated maturity thereof, or any event shall have occurred and shall continue uncured for twenty (20) days following notice provided by Cs via electronic mail to Debtor's President, or as otherwise provided by <u>Section 21(e)</u>, that permits (or, with the giving of notice or lapse of time or both, would permit) any holder or holders of such Indebtedness or any Person acting on behalf of such holder or holders to accelerate the maturity thereof or require any such prepayment;
- (f) Impairment of Security. This Agreement or any other Investment Documents (excluding the Employment Agreements) shall cease to be in full force and effect or shall be declared null and void or the validity or enforceability thereof shall be contested or challenged by Debtor, or Debtor shall deny that it has any further liability or obligation under any of the Investment Documents (excluding the Employment Agreements);

8

- (g) ERISA Matters. (i) With respect to any Pension Plan, a Prohibited Transaction occurs which in the reasonable determination of Lender could result in material liability to Debtor, (ii) with respect to any Title IV Plan, the filing of a notice to voluntarily terminate any such plan in a distress termination, (iii) with respect to any Multiemployer Plan, Debtor or any ERISA Affiliate shall incur any Withdrawal Liability, (iv) with respect to any Qualified Plan, Debtor or any ERISA Affiliate shall incur and accumulated funding deficiency or request a funding waiver from the IRS or (v) with respect to any Title IV Plan or Multiemployer Plan which has an ERISA Event not described in clauses (ii), through (iv), hereof, in the reasonable determination of C₃ there is a reasonable likelihood for termination of any such plan by the PBGC; provided, that the events listed in clauses (ji), through (iv), hereof shall constitute Events of Default only if the liability, deficiency or waiver request of Debtor or any ERISA Affiliate, whether or not assessed, could, in the opinion of Lender reasonably be expected to have a Material Adverse Effect:
- (h) Commencement of Proceedings. Debtor shall fail to discharge within a period of thirty (30) days after the commencement thereof any attachment, sequestration, or similar proceedings involving an aggregate amount in excess of Fifty Thousand Dollars (\$50,000) against any of its assets or properties, except and to the extent Debtor has obtained surety bonds;
- (i) Entry of Judgment. A final judgment or judgments for the payment of money in excess of Fifty Thousand Dollars (\$50,000) in the aggregate shall be rendered by a court or courts against Debtor and the same shall not be discharged (or provision shall not be made for such discharge), or a stay of execution thereof shall not be procured, within forty-five (45) days from the date of entry thereof and Debtor shall not, within said period of forty-five (45) days, or such longer period during which execution of the same shall have been stayed, appeal therefrom and cause the execution thereof to be stayed during such appeal;
- (j) Failure to Observe Other Obligations. Debtor defaults in the payment when due, or in performance or observance of, any material obligation of, or condition agreed to by, Debtor with respect to any material purchase or lease of goods or services where such default, singly or in the aggregate with all other such defaults, could reasonably be expected to have a Material Adverse Effect;
- (k) <u>Receivership.</u> Debtor's assets are attached, seized, subjected to a writ or distress warrant, or are levied upon, or come within the possession of any receiver, trustee, custodian or assignee for the benefit of creditors in connection with any obligations or liabilities of Debtor and such attachment, seizure, warrant, levy or possession could reasonably be expected to have a Material Adverse Effect;
- (l) <u>Curtailment of Business Conduct</u>. The occurrence of any event that could cause the cessation or substantial curtailment of the conduct of business by Debtor, including Debtor being enjoined, retrained or prevented by any court or administrative agency from conducting any material part of its business.

(n) Regulatory Default. Any (a) diversion by Debtor of the proceeds from the sale of the Notes and Membership Units from the purposes set forth in Section 2.4 of the SPA or (b) change in Debtor's business activity within the twelve (12) month period following the date of this Agreement that renders the investment by Lender into Debtor an ineligible investment for Debtor pursuant to Section 107.760 of Title 13 of the Code of

The foregoing shall constitute "Events of Default" whatever the reason or cause for any such Event of Default and whether it is voluntary or involuntary or is effected by operation of law or pursuant to any judgment, decree or order of any court or any order, rule or regulation of any administrative or governmental body.

18. Remedies. Upon the occurrence of any default under this Security Agreement, Lender may at its option, without notice or demand, declare all Obligations secured hereby immediately due and payable and Lender, upon the occurrence of any such default, may exercise any and all of the rights and remedies of a secured party under the Uniform Commercial Code of Missouri, then in effect, provided, however, that with regard to Section 17(f), Lender will provide written notice of the default and Debtor shall have fifteen (15) days to cure such default prior to Lender taking further action. Lender may take immediate possession of the Collateral or any part thereof wherever the same may be found, and for said purposes may, and is hereby appointed Debtor's agent and authorized by Debtor to, enter Debtor's premises for the purpose of removing, assembling or taking possession of the Collateral without liability for trespass or any other right of action by reason of taking possession of said Collateral. Whenever the Collateral is in Lender's possession, Lender may use and operate same as appropriate for the purpose of protecting Lender's interest with respect thereto. In addition, if any Collateral shall require rebuilding, repairing, maintenance, preparation, or is in process or other unfinished state, Lender shall have the right at its option to do such rebuilding, repairing preparation, processing or completion of manufacturing on or off Debtor's premises, for the purpose of putting the Collateral in such saleable form as Lender shall deem appropriate. Lender may require Debtor, at Debtor's expense, to assemble the Collateral and make it available to Lender at a place to be designated by Lender. Debtor agrees to pay all costs of Lender in the collection of the Obligations and enforcement of rights hereunder, including reasonable attorney's fees and legal expense and of any repairs to any realty or other property to which any of the Collateral may be affixed or be a part. Any notice of any sale, lease, or other disposition, or other intended action by Lender shall be deemed reasonable if it is in writing and deposited in the United States mail at least ten (10) days in advance of the intended disposition or other intended action or, with respect to a private sale, at least ten (10) days in advance of the date after which a private sale or sales shall occur, first class postage prepaid, addressed to Debtor at the address set forth in Section 22(e) hereof or to any other address of Debtor appearing on the records of Lender. At any sale, Lender may specifically disclaim any warranties including of title or the like. Lender may comply with any applicable state or federal law requirements in connection with a disposition of the Collateral and compliance will not be considered adversely to affect the commercial

10

reasonableness of any sale or disposition of the Collateral. Debtor waives all rights to require any marshalling of assets

Lender shall also have the right to apply for and have a receiver appointed by a court of competent jurisdiction to enforce its rights and remedies hereunder in order to manage, protect and preserve the Collateral, continue the peration of the business of Debtor, and to collect all revenues and profits thereof and apply the same to the payment of (a) all expenses and other charges of such receivership, including the compensation of the receiver, and (b) the Obligations secured hereby until a sale or other disposition of such Collateral shall be finally made and consummated.

Lender may notify any and all parties obligated on any of the Collateral that the Collateral has been assigned to Lender and that all payments thereon are to be made directly to Lender. Lender may settle, compromise or release, on terms acceptable to Lender, in whole or in part, any amounts owing on such Collateral; sue to enforce payments and prosecute any action or proceeding with respect to the Collateral in its own name or the name of Debtor; and extend the time of payment, make allowance and adjustments, and issue credits in its own name or the name of Debtor.

The proceeds of any sale shall be applied in the following order: first, to pay all costs and expenses of every kind for care, safekeeping, collection, sale, delivery or otherwise (including expenses incurred in the protection of Lender's title to or lien upon or right in any such property, expenses for legal services of any kind in connection therewith or in making any such sale or sales, insurance, commission for sale and guaranty), then to interest on all Obligations of Debtor to Lender; then to the principal thereof, whether or not such Obligations are due or accrued. Any remaining surplus shall be paid to whomever shall be legally entitled thereto. Application of proceeds as between particular Obligations to Lender shall be in the absolute and sole discretion of Lender. If the proceeds of any such sales are insufficient to pay all Obligations of Debtor to Lender, Debtor shall remain liable for the deficiency.

- 19. Inspection. Lender or its nominee shall have the privilege at any time, upon request, of inspecting during reasonable business hours any of the business properties or premises of Debtor and the books and records of Debtor relating not only to the Collateral, or the processing or collecting thereof, but also those relating to its general business affairs and financial condition of Debtor. Debtor further agrees from time to time to time to time to the tother reports, data and financial statements, in respect of its business and financial condition, as Lender may reasonably require.
- 20. Lender's Duties. The powers conferred on Lender hereunder are solely to protect its interest in the Collateral and shall not impose any duty upon it to exercise any such powers. Unless otherwise required by law, Debtor has the risk of loss of the Collateral, and Lender shall have no duty as to any Collateral (other than storing and disposing of the Collateral in a commercially reasonable manner) or as to the taking of any necessary steps to preserve rights against other parties or any other rights pertaining to any Collateral.

11

- 21. Miscellaneous. Debtor and Lender further agree as follows:
 - Governing Law. This Security Agreement shall be governed by and construed in accordance with the laws of the State of Missouri without regard to conflict of laws principles (a)
- Non-Waiver. Waiver of or acquiescence by Lender in any default by Debtor, or failure of Lender to insist upon strict performance by Debtor of any warranties, agreements or other obligations contained in this (h) Security Agreement shall not constitute a waiver of any subsequent or other default, failure or waiver of strict performance, whether similar or dissimilar.
- Modifications. No modification of any provision of this Security Agreement, no approvals required from Lender and no consent by Lender to any departure therefrom by Debtor shall be effective unless such modification, approval or consent shall be in writing and signed by a duly authorized officer of Lender, and the same shall then be effective only for the period and on the conditions and for the specific instances and purposes specified in such writing. No notice to or demand on Debtor in any case shall entitle Debtor to any other or further notice or demand in similar or other circumstances.
- Severability. Wherever possible, each provision of this Security Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Security Agreement shall be prohibited by or invalid under applicable law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Security Agreement.
- Notices. All notices and other communications provided for herein shall, unless otherwise stated herein, be in writing and shall be personally delivered or sent by certified mail, postage prepaid, by prepaid overnight nationally recognized courier, or by facsimile, to the intended party at the address or facsimile number of such party set forth as follows:

To Debtor:

Liquidmetal Coatings, LLC 30452 Esperanza Rancho Santa Margarita, California 92688 Attn: Legal Department Facsimile: (813) 314-0270

12

With a copy to: (which shall not constitute notice)

Foley & Lardner LLP 100 North Tampa Street Suite 2700 Tampa, Florida 33602 Attn: Curt P. Creely Facsimile: (813) 221-4210

To Lender

C3 Capital Partners, L.P. 4520 Main Street Suite 1600 Kansas City, Missouri 64111 Attn: Robert L. Smith Facsimile: (816) 756-5552

with a copy to:

Bryan Cave LLP 3500 One Kansas City Place 1200 Main Street Kansas City, Missouri 64105 Attn: Thomas W. Van Dyke Facsimile: (816) 374-3300

or at such other address or facsimile number as shall be designated by such party in a written notice to the other parties hereto. All such notices and communications shall be effective (i) if personally delivered, when delivered, (ii) if sent by certified mail, three (3) days after having been deposited in the mail, postage prepaid, (iii) if sent by overnight courier, one business day after having been given to such courier, or (iv) if transmitted by facsimile, when sent.

- (f) Rights and Remedies Cumulative. The rights and remedies of Lender under this Security Agreement are cumulative and are not in lieu of, but are in addition to any other rights or remedies which Lender shall have under this Security Agreement or any other instrument, or at law or in equity. No course of dealing between Lender and Debtor or any failure or delay on the part of Lender in exercising any rights or remedies hereunder shall operate as a waiver of any rights or remedies of Lender and no single or partial exercise of any rights or remedies hereunder.
 - (g) Security Interest and Pledge Absolute. All rights, including the security interest of Lender granted hereunder, and all obligations of Debtor hereunder, shall be absolute and unconditional irrespective of:

13

- (i) any lack of validity or enforceability of the Obligations or any other agreement or instrument relating thereto;
- (ii) any change in the time, manner or place of payment of, or in any other term of, all or any of the Obligations, or any other amendment or waiver of or any consent to any departure from the Obligations or any agreement or instrument relating thereto; or
- (iii) any exchange, release or non-perfection of any other collateral, or any release or amendment or waiver of or consent to departure from any guaranty, for any of the Obligations.
- (h) Costs of Enforcement. In the event that Lender shall retain or engage an attorney or attorneys to collect or enforce or protect its interests with respect to this Security Agreement or any instrument or document delivered pursuant to this Security Agreement, including the representation of Lender in connection with any bankruptcy, reorganization, receivership or any other action affecting creditor's rights, and regardless of whether a suit or action is commenced, Debtor shall pay all of the costs and expenses of such collection, enforcement or protection, including reasonable attorneys' fees, and Lender may take judgment for all such amounts.
 - (i) Successors and Assigns. This Security Agreement shall be binding upon and inure to the benefit of Lender and its successors and assigns and Debtor and its heirs, successors and permitted assigns
- (j) <u>Assignment; Sale of Interest</u>. Debtor hereby consents to Lender's participation, sale, assignment, transfer or other disposition, at any time or times hereafter, of this Security Agreement, or of any portion hereof or thereof, including, without limitation, Lender's rights, title, interests, remedies, powers and duties hereunder subject to the approval of Debtor, and such approval shall not be unreasonably withheld.
- (k) Fees and Expenses. Debtor shall pay all reasonable out-of-pocket costs and expenses, including reasonable attorneys' fees and expenses, incurred by Lender in connection with the preparation of this Security Agreement and any document or instrument delivered pursuant to or in connection with this Security Agreement and all related documentation, recording or filing fees. Debtor shall also pay all like reasonable costs and expenses incurred by Lender in connection with any amendments, waivers, renewals or modifications of or made pursuant to this Security Agreement or any document or instrument delivered pursuant to or in connection with this Security Agreement and all other related documentation.
- (l) Reinstatement of Obligations. Debtor expressly agrees that to the extent a payment or payments to Lender, or any part thereof, are subsequently invalidated, declared to be void or voidable, set aside and are required to be repaid to a trustee, custodian, receiver or any other party under any bankruptcy act, state or federal law, common law or equitable cause, then to the extent of such payment or repayment, the obligation or part thereof intended to be satisfied and any collateral given therefore

14

including this Agreement shall be revived and continued in full force and effect as if said payment had not been made.

- (m) <u>Financing Statement</u>. At the option of Lender, this Security Agreement, or a carbon, photographic or other reproduction of this Security Agreement or of any Uniform Commercial Code financing statement covering the Collateral or any portion thereof, shall be sufficient as a Uniform Commercial Code financing statement and may be filed as such.
 - (n) Capitalized Terms. Except as provided in Section 1, capitalized terms used and not defined herein shall have the meanings given to them in the SPA.
- (o) Controlling Provisions. If any item of Collateral hereunder also constitutes collateral granted to Lender under any other mortgage, deed of trust, agreement or instrument, in the event of any conflict between the provisions under this Security Agreement and those under such other mortgage, agreement or instrument relating to such Collateral, the provision or provisions selected by Lender shall control with respect to such Collateral.
- (p) Setoff. In addition to any rights now or hereafter granted under the provisions of any applicable law, rule or regulation and, not by way of limitation of any such rights, upon the occurrence of (i) any event of default after observing any applicable cure periods or (ii) any event after observing any applicable cure periods which with the lapse of time or the giving of notice, or both, would constitute an Event of Default, Lender is hereby authorized by Debtor, at any time or from time to time, without notice to Debtor or to any other person, any such notice being hereby expressly waived,
 - (i) to setoff and to appropriate and to apply any and all deposits (general or special, time or demand, including, but not limited to, indebtedness evidenced by certificates of deposit, in each case whether matured or unmatured) and any other indebtedness at any time held or owing by Lender to or for the credit or account of Debtor against and on account of the obligations and liabilities of Debtor to Lender, including, but not limited to, all claims of any nature or description arising out of or connected with this Security Agreement or any instrument or document delivered in connection with or pursuant to this Security Agreement, irrespective of whether or not (1) Lender shall have made any demand under this Security Agreement or any instrument or document delivered in connection with or pursuant to this Security Agreement, or (2) Lender shall have declared the principal of and interest on amounts under this Security Agreement or any instrument or document delivered in connection with or pursuant to this Security Agreement to be due and payable as permitted pursuant to this Security Agreement or any instrument or document delivered in connection with or pursuant to this Security Agreement, and the pursuant to this Security Agreement to be due and payable as permitted pursuant to this Security Agreement or any instrument or document delivered in connection with or pursuant to this Security Agreement, and the property of the prope

15

although said obligations and liabilities, or any of them, shall be contingent or unmatured, and

- (ii) pending any such setoff or appropriation or application, to hold the amounts of all deposits as collateral and to return as unpaid any or all checks drawn against such deposits that are presented for payment as Lender in its sole discretion shall decide.
- (q) Consent to Forum. DEBTOR HEREBY CONSENTS TO THE JURISDICTION OF ANY STATE COURT LOCATED WITHIN THE CITY OF KANSAS CITY OR JACKSON COUNTY, MISSOURI OR FEDERAL COURT IN THE WESTERN DISTRICT OF MISSOURI, WESTERN DIVISION. DEBTOR WAIVES ANY OBJECTION TO JURISDICTION AND VENUE OF ANY ACTION INSTITUTED AGAINST IT AS PROVIDED HEREIN AND AGREES NOT TO ASSERT ANY DEFENSE BASED ON LACK OF JURISDICTION OR VENUE. DEBTOR FURTHER AGREES NOT TO ASSERT AGAINST LENDER (EXCEPT BY WAY OF A DEFENSE OR COUNTERCLAIM IN A PROCEEDING INITIATED BY LENDER) ANY CLAIM OR OTHER ASSERTION OF LIABILITY WITH RESPECT TO THIS SECURITY AGREEMENT, LENDER'S CONDUCT OR OTHERWISE IN ANY JURISDICTION OTHER THAN THE FOREGOING JURISDICTIONS.
- (r) Waiver of Jury Trial. DEBTOR HEREBY WAIVES ANY RIGHT TO TRIAL BY JURY (WHICH LENDER ALSO WAIVES) IN ANY ACTION, SUIT, PROCEEDING OR COUNTERCLAIM OF ANY KIND ARISING OUT OF OR RELATING TO THIS SECURITY AGREEMENT, THE OBLIGATIONS OF DEBTOR HEREUNDER OR LENDER'S CONDUCT IN RESPECT OF ANY OF THE FOREGOING.
- (s) Mo. Rev. Stat. § 432.045 Statement. The following notice is given pursuant to Section 432.045 of the Missouri Revised Statutes; nothing contained in such notice shall be deemed to limit or modify the terms of this Agreement: "ORAL AGREEMENTS OR COMMITMENTS TO LOAN MONEY, EXTEND CREDIT OR TO FOREBEAR FROM ENFORCING REPAYMENT OF A DEBT INCLUDING PROMISES TO EXTEND OR RENEW SUCH DEBT ARE NOT ENFORCEABLE. TO PROTECT YOU (COMPANY) AND US (CREDITOR) FROM MISUNDERSTANDING OR DISAPPOINTMENT, ANY AGREEMENTS WE REACH COVERING SUCH MATTERS ARE CONTAINED IN THIS WRITING, WHICH IS THE COMPLETE AND EXCLUSIVE STATEMENT OF THE AGREEMENT BETWEEN US, EXCEPT AS WE MAY LATER AGREE IN WRITING TO MODIFY IT.

[Remainder of page intentionally left blank.]

16

Signature Page to Security Agreement

IN WITNESS WHEREOF, this Security Agreement has been executed and delivered by Debtor this 24th day of July, 2007.

LIQUIDMETAL COATINGS, LLC a Delaware limited liability company

By:	
By: Name:	
Title:	

PATENT AND TRADEMARK SECURITY AGREEMENT

(Liquidmetal Coatings, LLC)

This Patent and Trademark Security Agreement ("**Agreement**"), dated as of July 24, 2007, is made by and between Liquidmetal Coatings, LLC, a Delaware limited liability company having a business location at the address set forth below next to its signature (the "**Debtor**"), C₃ Capital Partners, L.P., a Delaware limited partnership ("C₃") and C₃ Capital Partners II, L.P., a Delaware limited partnership ("C₃ II," together with C₃ shall hereinafter be referred to as the "**Secured Partv**").

Recitals

The Debtor and the Secured Party are parties to a Securities Purchase Agreement dated of even date herewith (as the same may hereafter be amended, supplemented or restated from time to time, collectively the "Purchase Agreement") and Debtor has issued a 14% Subordinated Note dated of even date herewith to each of C₃ and C₃ II (as amended, supplemented, or restated from time to time, collectively, the "Notes") (the Purchase Agreement, Notes and any and all other related documents are collectively referred to as the "Loan Documents") both setting forth the terms on which the Secured Party may now or hereafter extend credit to or for the account of the Debtor.

The Debtor and Liquidmetal Technologies, Inc. ("LMT") have entered into an Asset Purchase and Contribution Agreement dated July 24, 2007 ("APA") wherein the Debtor is acquiring certain assets described in the APA from LMT.

As a condition to extending credit to or for the account of the Debtor, the Secured Party has required the execution and delivery of this Agreement by the Debtor.

ACCORDINGLY, in consideration of the mutual covenants contained in the Loan Documents and herein, the parties hereby agree as follows:

- 1. <u>Definitions.</u> All terms defined in the Recitals hereto or in the Purchase Agreement that are not otherwise defined herein shall have the meanings given to them therein. In addition, the following terms have the meanings set forth below:
 - "Obligations" means each and every debt, liability and obligation of every type and description arising under or in connection with any Loan Documents which the Debtor may now or at any time hereafter owe to the Secured Party, whether such debt, liability or obligation now exists or is hereafter created or incurred and whether it is or may be direct or indirect, due or to become due, absolute or contingent, primary or secondary, liquidated or unliquidated, independent, joint, several or joint and several, and including specifically, but not limited to, the Securities (as defined in the Purchase Agreement).
 - "Patents" means all of the Debtor's right, title and interest in and to patents or applications for patents, fees or royalties with respect to each, and including without limitation the right to sue for past infringement and damages therefor, and licenses

thereunder, all as presently existing or hereafter arising or acquired, including without limitation the patents listed on $\underline{\text{Exhibit A}}$.

"Security Interest" has the meaning given in Section 2.

- "Trademarks" means all of the Debtor's right, title and interest in and to: (i) trademarks, service marks, collective membership marks, registrations and applications for registration for each, and the respective goodwill associated with each, (ii) licenses, fees or royalties with respect to each, (iii) the right to sue for past, present and future infringement, dilution and damages therefor, (iv) and licenses thereunder, all as presently existing or hereafter arising or acquired, including, without limitation, the marks listed on Exhibit B.
- 2. <u>Security Interest</u>. The Debtor hereby irrevocably pledges and assigns to, and grants the Secured Party a security interest (the "Security Interest"), with power of sale to the extent permitted by law, in the Patents and in the Trademarks to secure payment of the Obligations. As set forth in the Purchase Agreement and the Notes, the Security Interest is coupled with a security interest in substantially all of the personal property of the Debtor.
 - Representations, Warranties and Agreements. The Debtor represents, warrants and agrees as follows:
 - (a) Existence; Authority. The Debtor is a limited liability company duly organized, validly existing and in good standing under the laws of its state of organization, and this Agreement has been duly and validly authorized by all necessary action on the part of the Debtor.
 - (b) **Patents.** Exhibit A accurately lists all Patents owned or controlled by the Debtor as of the date hereof, or to which the Debtor has a right as of the date hereof to have assigned to it, and accurately reflects the existence and status of applications and letters patent pertaining to the Patents as of the date hereof. If after the date hereof, the Debtor owns, controls or has a right to have assigned to it any Patents not listed on Exhibit A. or if Exhibit A cases to accurately reflect the existence and status of applications and letters patent pertaining to the Patents, then the Debtor shall within 60 days provide written notice to the Secured Party with a replacement Exhibit A, which upon acceptance by the Secured Party shall become part of this Agreement.
 - (c) **Trademarks**. Exhibit B accurately lists all Trademarks owned or controlled by the Debtor as of the date hereof and accurately reflects the existence and status of Trademarks and all applications and registrations pertaining thereto as of the date hereof; provided, however, that Exhibit B need not list common law marks (i.e., Trademarks for which there are no applications or registrations) which are not material to the Debtor's or any Affiliate's business(es). If after the date hereof, the Debtor owns or controls any Trademarks not listed on Exhibit B (other than common law marks which are not material to the Debtor's or any Affiliate's business(es)), or if Exhibit B ceases to accurately reflect the existence and status of applications and registrations pertaining to the Trademarks, then the Debtor shall within sixty (60) days promptly provide written

notice to the Secured Party with a replacement Exhibit B, which upon acceptance by the Secured Party shall become part of this Agreement.

- (d) Title. The Debtor has absolute title to each Patent and each Trademark listed on Exhibits A and B, free and clear of all Liens except Permitted Liens. The Debtor (i) will have, at the time the Debtor acquires any rights in Patents or Trademarks hereafter arising, absolute title to each such Patent or Trademark free and clear of all Liens except Permitted Liens, and (ii) will keep all Patents and Trademarks free and clear of all Liens except Permitted Liens.
- (e) No Sale. Except as permitted in the Purchase Agreement, the Debtor will not assign, transfer, encumber or otherwise dispose of the Patents or Trademarks, or any interest therein, without the Secured Party's prior written consent.
- (f) Defense. The Debtor will at its own expense and using commercially reasonable efforts, protect and defend the Patents and Trademarks against all claims or demands of all Persons other than those holding Permitted Liens.
- (g) Maintenance. The Debtor will at its own expense maintain the Patents and the Trademarks to the extent reasonably advisable in its business including, but not limited to, filing all applications to obtain letters patent or trademark registrations and all affidavits, maintenance fees, annuities, and renewals possible with respect to letters patent, trademark registrations and applications therefor. The Debtor covenants that it will not abandon nor fail to pay any maintenance fee or annuity due and payable on any Patent or Trademark, nor fail to file any required affidavit or renewal in support thereof, without first providing the Secured Party: (i) sufficient written notice, of at least 30 days, to allow the Secured Party to timely pay any such maintenance fees or annuities which may become due on any Patents or Trademarks, or to file any affidavit or renewal with respect thereto, and (ii) a separate written power of attorney or other authorization to pay such maintenance fees or annuities, or to file such affidavit or renewal, should such be necessary or desirable.
- (h) Secured Party's Right to Take Action. If the Debtor fails to perform or observe any of its covenants or agreements set forth in this Section 3. and if such failure continues for a period of ten (10) calendar days after the Secured Party gives the Debtor written notice thereof (or, in the case of the agreements contained in subsection (b), immediately upon the occurrence of such failure, without notice or lapse of time), or if the Debtor notifies the Secured Party that it intends to abandon a Patent or Trademark, the Secured Party may (but need not) perform or observe such covenant or agreement or take steps to prevent such intended abandonment on behalf and in the name, place and stead of the Debtor (or, at the Secured Party's option, in the Secured Party's own name) and may (but need not) take any and all other actions which the Secured Party may reasonably deem necessary to cure or correct such failure or prevent such intended abandonment.
 - (i) Costs and Expenses. Except to the extent that the effect of such payment would be to render any loan or forbearance of money usurious or otherwise

illegal under any applicable law, the Debtor shall pay the Secured Party on demand the amount of all moneys expended and all costs and expenses (including reasonable attorneys' fees and disbursements) incurred by the Secured Party in connection with or as a result of the Secured Party's taking action under <u>subsection (i)</u> or exercising its rights under <u>Section 6</u>, together with interest thereon from the date expended or incurred by the Secured Party at the

- (j) Power of Attorney. To facilitate the Secured Party's taking action under subsection (i) and exercising its rights under Section 6, the Debtor hereby irrevocably appoints (which appointment is coupled with an interest) the Secured Party, or its delegate, as the attorney in fact of the Debtor with the right (but not the duty) from time to time to create, prepare, complete, execute, deliver, endorse or file, in the name and on behalf of the Debtor, any and all instruments, documents, applications, financing statements, and other agreements and writings required to be obtained, executed, delivered or endorsed by the Debtor under this Section 3, or, necessary for the Secured Party, after an Event of Default, to enforce or use the Patents or Trademarks or to grant or issue any exclusive or non exclusive license under the Patents or Trademarks to any third party, or to sell, assign, transfer, pledge, encumber or otherwise transfer title in or dispose of the Patents or Trademarks to any third party. The Debtor hereby ratifies all that such attorney shall lawfully do or cause to be done by virtue hereof. The power of attorney granted herein shall terminate upon the termination of the Purchase Agreement as provided therein and the payment and performance of all Obligations.
- 4. <u>Debtor's Use of the Patents and Trademarks.</u> The Debtor shall be permitted to control and manage the Patents and Trademarks, including the right to exclude others from making, using or selling items covered by the Patents and Trademarks and any licenses thereunder, in the same manner and with the same effect as if this Agreement had not been entered into, so long as no Event of Default occurs and remains uncured.

5. Events of Default. Each of the following occurrences shall constitute an event of default under this Agreement (herein called "Event of Default"): (a) an Event of Default, as defined in the Purchase Agreement, shall occur; or (b) the Debtor shall fail promptly to observe or perform any covenant or agreement herein binding on it which is not remedied within fifteen (15) calendar days after Debtor receives written notice thereof; or (c) any of the representations or warranties contained in Section 3 shall prove to have been incorrect in any material respect when made. Remedies. Upon the occurrence of an Event of Default and at any time thereafter, the Secured Party may, at its option, take any or all of the following actions: (a) The Secured Party may exercise any or all remedies available under the Purchase Agreement. (b) The Secured Party may sell, assign, transfer, pledge, encumber or otherwise dispose of the Patents and Trademarks. The Secured Party may enforce the Patents and Trademarks and any licenses thereunder, and if Secured Party shall commence any suit for such enforcement, the Debtor shall, at the request of Secured (c) Party, do any and all lawful acts and execute any and all proper documents required by Secured Party in aid of such enforcement. Miscellaneous. This Agreement can be waived, modified, amended, terminated or discharged, and the Security Interest can be released, only explicitly in a writing signed by the Secured Party. A waiver signed by the Secured Party shall be effective only in the specific instance and for the specific jurisdance and for the specific jur the exercise or enforcement of any other. All notices to be given to Debtor under this Agreement shall be given in the manner and with the effect provided in the Purchase Agreement. The Secured Party shall not be obligated to preserve any rights the Debtor may have against prior parties, to realize on the Patents and Trademarks at all or in any particular manner or order, or to apply any cash proceeds of Patents and Trademarks in any particular order of application. This Agreement shall be binding upon and inure to the benefit of the Debtor and the Secured Party and their respective participants, successors and assigns and shall take effect when signed by the Debtor and delivered to the Secured Party, and the Debtor waives notice of the Secured Party is acceptance hereof. The Secured Party may execute this Agreement if appropriate for the purpose of filing, but the failure of the Secured Party to execute this Agreement shall not affect or impair the validity or effectiveness of this Agreement. A carbon, photographic or other reproduction of this Agreement or of any financing statement signed by the Debtor shall have the same force and effect as the original for all purposes of a financing statement. This Agreement shall be governed by the internal law of Missouri without regard to conflicts of law provisions. If any provision or application of this Agreement is held unlawful or unenforceable in any respect, such illegality or unenforceablity shall not affect other provisions or application had never been contained herein or prescribed hereby. All representations and warranties contained in this Agreement shall survive the execution, delivery and performance of this Agreement and the creation and payment of the Obligations. [The remainder of this page is intentionally left blank.] THE PARTIES WAIVE ANY RIGHT TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING BASED ON OR PERTAINING TO THIS AGREEMENT. IN WITNESS WHEREOF, the parties have executed this Patent and Trademark Security Agreement as of the date written above. Liquidmetal Coatings, LLC Liquidmetal Coatings, LLC 30452 Esperanza a Delaware limited liability company Rancho Santa Margarita, California 92688 Attn: Legal Department Facsimile: (813) 314-0270 By: Its: C3 Capital Partners, L.P. C₃ Capital Partners, L.P., 4520 Main Street, Suite 1600 a Delaware limited partnership Kansas City, Missouri 64111 Bv:C3 Partners, LLC Attention: Robert L. Smith Its:General Partner email: rsmith@c3cap.com By: Its: C3 Capital Partners II, L.P. 4520 Main Street, Suite 1600 C3 Capital Partners II, L.P., Kansas City, Missouri 64111 a Delaware limited partnership Attention: Robert L. Smith By:C3 Partners II, LLC email: rsmith@c3cap.com Its:General Partner By: Its: STATE OF SS. COUNTY OF , 2007, by The foregoing instrument was acknowledged before me this , the of Liquidmetal Coatings, LLC, a Delaware limited liability company, on behalf of the day of Notary Public My appointment expires: STATE OF) SS COUNTY OF The foregoing instrument was acknowledged before me this day of , 2007, by , the of C3 Capital Partners, LLC, the general partner of C3 Capital Partners, L.P., a Delaware limited partnership, on behalf of the partnership Notary Public My appointment expires: STATE OF) SS COUNTY OF The foregoing instrument was acknowledged before me this , 2007, by of C₃ Capital Partners II, LLC, the general partner of C₃ Capital Partners II, L.P., a day of , the Delaware limited partnership, on behalf of the partnership Notary Public My appointment expires:

EXHIBIT A

UNITED STATES ISSUED PATENTS

Title		Patent Number		Issue Date
		FOREIGN ISSUED PATENTS		
Title		Country	Patent Number	Issue Date
		EXHIBIT B		
	Ī	JNITED STATES ISSUED TRADEMARKS, SERVICE MARK	<u>S</u>	
		AND COLLECTIVE MEMBERSHIP MARKS		
		REGISTRATIONS		
Mark		Registration Number		Registration Date
		APPLICATIONS		
		COLLECTIVE MEMBERSHIP MARKS		
		UNREGISTERED MARKS		
				Execution Copy
		PLEDGE AGREEMENT		***
		(Liquidmetal Technologies, Inc.)		
		as of this 24th day of July, 2007, by and among Liquidmetal Tec $_{3}$ and their successors and assigns shall hereinafter be referred to		gor"), C ₃ Capital Partners, L.P., a
		PRELIMINARY STATEMENTS:		
	m time to time, the "SPA") dated as of the date hereof by and a	by that certain Securities Purchase Agreement, together with any imong the Lender, Liquidmetal Coatings, LLC, a Delaware lim issued and outstanding membership units ("Membership Units")	ited liability company (the "Company") and Lie	
(2)	The execution and delivery of this Agreement by Pledgor i	is a condition precedent to the SPA.		
(3)	Capitalized terms, unless otherwise defined herein, shall ha	ave the meanings set forth in the SPA.		
NOW	V, THEREFORE, in consideration of the premises and in order	to induce the Lender to enter into the SPA, the Pledgor hereby co	ovenants to the Lender as follows:	

SECTION 1. Pledge. The Pledgor hereby pledges to the Lender and grants to the Lender a security interest in all of the Pledgor's now owned or hereinafter acquired right, title and interest in and to the following (the "Pledged Collateral"):

- the Membership Units: (a)
- the proceeds, if and when received, of the Membership Units and other instruments and documents representing the Membership Units, and all cash, instruments and other property from time to time received, (b) receivable or otherwise distributed in respect of or in exchange for any or all of the Membership Units;
- the proceeds, if and when received, of all equity interests from time to time received or acquired by the Pledgor in any manner resulting from the acquisition or formation by the issuer of the Membership Units of a subsidiary, the certificates and other instruments and documents representing such equity interests and all cash, instruments and other property from time to time received, receivable or otherwise distributed in respect of or in exchange for any or all of such equity interests; and
- all other property hereafter delivered to the Lender in substitution for or in addition to any of the foregoing, all certificates, certificates of deposit, notes, instruments and documents representing or evidencing such property, and all cash, securities, interest, rights and other property at any time and from time to time received, receivable or otherwise distributed in respect of or in exchange for any or all thereof.
- SECTION 2. Security for Obligations. This Agreement secures the payment and performance of all the Company's obligations of every kind and nature under the SPA (the "Obligations"). For the purpose of this Agreement, the term "Event of Default" is defined as an Event of Default including any applicable cure periods under the SPA.
- SECTION 3. No Personal Liability of Pledgor. Notwithstanding anything in this Agreement, with the exception of the Pledgor's pledge of the Pledged Collateral, Pledgor shall have no personal liability for the Company's obligations to Lender and Pledgor does not guaranty the Company's obligations to Lender.
- Delivery of Pledged Collateral. All instruments representing or evidencing the Pledged Collateral shall be delivered to and held by the Lender pursuant hereto and shall be in suitable form for transfer by delivery, or shall be accompanied by duly executed instruments of transfer or assignment in blank, all in form and substance satisfactory to the Lender. The Lender shall have the right, at any time after the occurrence of an Event of Default, in its discretion, to transfer to or to register in the name of the Lender or any of its nominees any or all of the Pledged Collateral. In addition, the Lender shall have the right at any time to exchange instruments representing or evidencing Pledged Collateral for instruments of smaller or larger denominations
 - SECTION 5. Representations and Warranties. The Pledgor represents and warrants as follows:
 - The Membership Units have been duly authorized and validly issued and are fully paid and non-assessable. (a)
- The Pledgor is and will be at all times the legal and beneficial owner of the Membership Units and the Pledged Collateral free and clear of any lien, security interest, option or other charge or encumbrance, except (b) for security interests created in favor of the Lender, the Senior Lender and security interests otherwise consented to in advance and in writing by the Lender
 - (c) The pledge pursuant to this Agreement of any Pledged Collateral creates or will create a valid and perfected security interest in the Pledged Collateral, securing the payment of the Obligations.
- No authorization, approval or other action by, and no notice to or filing with, any governmental authority or regulatory body is required either (i) for the pledge by the Pledged Collateral pursuant to this Agreement or for the execution, delivery or performance of this Agreement by the Pledgor or (ii) for the exercise by the Lender of the rights

provided for in this Agreement or the remedies in respect of the Pledged Collateral pursuant to this Agreement (except as may be required in connection with such disposition by laws affecting the foreclosure of security interests generally).

SECTION 6. <u>Further Assurances</u>. The Pledgor agrees that at any time and from time to time, at the expense of the Pledgor, promptly execute and deliver all further instruments, certificates and documents and take all further action that may be reasonably necessary or desirable or that the Lender may reasonably request in order to perfect and protect any security interest granted or purported to be granted hereby or to enable the Lender to exercise and enforce its rights and remedies hereunder with respect to any Pledged Collateral.

SECTION 7. <u>Pledgor's Rights; Distributions; Etc.</u>

- (a) So long as no Event of Default shall have occurred and be continuing:
- (i) The Pledgor shall be entitled to exercise any rights pertaining to the Pledged Collateral or any part thereof for any purpose not inconsistent with the terms of this Agreement including the right to vote the Membership Units for any purpose allowed under the Company's Certificate of Formation, Operating Agreement or applicable law.
 - (ii) The Pledgor shall be entitled to receive and retain any and all interest and distributions in respect of the Pledged Collateral; provided, however, that any and all:
- (A) proceeds, interest and distributions paid or payable other than in cash in respect of, and instruments and other property received, receivable or otherwise distributed in respect of, or in exchange for, any Pledged Collateral;
 - (B) proceeds and distributions paid or payable in cash in respect of any Pledged Collateral in connection with a partial or total liquidation or dissolution; and
 - (C) cash paid, payable or otherwise distributed in respect of principal of, or in redemption of, or in exchange for the Membership Units or any Pledged Collateral,

shall be, and shall be forthwith delivered to the Lender to hold as, Pledged Collateral and shall, if and when received by the Pledgor, be received in trust for the benefit of the Lender, be segregated from the other property or funds of the Pledgor, and be forthwith delivered to the Lender as Pledged Collateral in the same form as so received (with any necessary endorsement).

- (b) Upon the occurrence and during the continuance of an Event of Default:
 - (i) All rights of the Pledgor to receive the interest and distributions on the Pledged Collateral which Pledgor would otherwise be authorized to receive and retain

3

pursuant to Section 6(a)(ii) shall cease, and all such rights shall thereupon become vested in the Lender which shall thereupon have the sole right to receive and hold as Pledged Collateral such interest, distributions and payments.

(ii) All interest, distributions and payments which are received by the Pledgor contrary to the provisions of paragraph (i) of this Section 6(b) shall be received in trust for the benefit of the Lender, shall be segregated from other funds of the Pledgor and shall be forthwith paid over to the Lender as Pledged Collateral in the same form as so received (with any necessary endorsement).

SECTION 8. <u>Transfers and Other Liens; Notice to Prior Secured Party.</u> The Pledgor agrees that Pledgor will not create or permit to exist any lien, security interest, or other charge or encumbrance upon or with respect to the Membership Units or any of the Pledged Collateral, except for the security interest under this Agreement.

SECTION 9. Lender Appointed Attorney-in-Fact. Upon the occurrence of an Event of Default, the Pledgor hereby irrevocably appoints the Lender the Pledgor's attorney-in-fact, with full authority in the place and stead of the Pledgor and in the name of the Pledgor or otherwise, from time to time in the Lender's discretion, to take any action and to execute any instrument which the Lender may deem necessary or advisable to accomplish the purposes of this Agreement, including, without limitation, to demand, receive, endorse, hold and collect all instruments made payable to the Pledgor representing any distribution in respect of the Pledged Collateral or any part thereof and to give full discharge for the same.

SECTION 10. Lender May Perform. If the Pledgor fails to perform any agreement contained herein, the Lender may itself perform, or cause performance of, such agreement, and the reasonable expenses of the Lender incurred in connection therewith shall be payable by the Pledgor in accordance with Section 13.

SECTION 11. Reasonable Care. The Lender shall be deemed to have exercised reasonable care in the custody and preservation of the Pledged Collateral in its possession if the Pledged Collateral is accorded treatment substantially equal to that which the Lender accords its own property.

SECTION 12. Remedies. If any Event of Default shall have occurred and be continuing:

(a) The Lender may exercise in respect of the Pledged Collateral, in addition to other rights and remedies provided for herein and as otherwise available to it at law or in equity, all the rights and remedies of a secured party on default under the Uniform Commercial Code in effect in the State of Missouri at that time, and the Lender may also, without notice as specified below, execute upon and apply (for itself or any designee it elects) the Pledged Collateral directly toward the payment of the Obligations, and sell the Pledged Collateral or any portion thereof at public or private sale, at any exchange, broker's board or at any of the Lender's offices or elsewhere, for cash, on credit or for future delivery, and at such price or prices and upon such other terms as the Lender may deem commercially reasonable. The

4

Pledgor agrees that, to the extent notice of sale shall be required by law, at least twenty-five (25) days' notice to the Pledgor of the time and place of any public sale and one publication in a local newspaper, or at least twenty-five (25) days' notice of the time after which any private sale is to be made, shall constitute reasonable notification. The Lender shall not be obligated to make any sale of Pledged Collateral regardless of notice of sale having been given. The Lender may adjourn any public or private sale from time to time by announcement at the time and place fixed therefore, and such sale may, without further notice, be made at the time and place to which it was so adjourned.

(b) Any cash held by the Lender as Pledged Collateral and all cash proceeds received by the Lender in respect of any sale of, collection from, or other realization upon all or any part of the Pledged Collateral may, in the discretion of the Lender, be held by the Lender as collateral for, and then or at any time thereafter applied (after payment of any amounts payable to the Lender pursuant to Section 13) in whole or in part by the Lender against, all or any part of the Obligations in such order as the Lender shell elect. Any surplus of such cash or cash proceeds held by the Lender and remaining after payment in full of all the Obligations shall be paid over to the Pledgor or to whomsoever may be lawfully entitled to receive such surplus.

SECTION 13. Indemnification. Neither the Lender, nor any partner, officer, agent or employee of the Lender, shall be liable for any action taken or omitted to be taken by it or them hereunder or in connection therewith, except for its or their own gross negligence or willful misconduct.

SECTION 14. Expenses. The Pledgor will, upon demand, pay to the Lender the amount of any and all reasonable expenses, including the reasonable fees and expenses of its counsel and of any experts and agents, which the Lender may incur in connection with (a) the custody or preservation of, or the sale of, collection from, or other realization upon, any of the Pledged Collateral, (b) the exercise or enforcement of any of the rights of the Lender hereunder, or (c) the failure by Pledgor to perform or observe any of the provisions hereof.

SECTION 15. Security Interest Absolute

- (a) All rights of the Lender and security interests hereunder, and all obligations of the Pledgor hereunder, shall be absolute and unconditional irrespective of:
 - (i) any lack of validity or enforceability of the SPA;
 - (ii) any change in the time, manner or place of payment of, or in any other term of, all or any of the Obligations, or any other amendment or waiver of or any consent to any departure from the SPA;
 - (iii) any exchange, release or non-perfection of any other collateral, or any release of any of the Obligations; or

5

- (iv) any other circumstance which might otherwise constitute a defense available to, or a discharge of, the Pledgor or any other pledgor or any guarantor or co maker in respect of the Obligations or the Pledgor in respect of this Agreement.
 - (b) The Lender may, from time to time, whether before or after any of the Obligations shall become due and payable, without notice to the Pledgor, take all or any of the following actions:
 - (i) retain or obtain a security interest in any property, in addition to the Pledged Collateral, to secure any of the Obligations,
 - (ii) retain or obtain the primary or secondary liability of any party or parties with respect to any of the Obligations,
- (iii) extend or renew for any period (whether or not longer than the original period) or exchange any of the Obligations or release or compromise any obligation of any nature of any party with respect thereto,
- (iv) surrender, release or exchange all or any part of any property, in addition to the Pledged Collateral, securing any of the Obligations, or compromise or extend or renew for any period (whether or not longer than the original period) any obligations of any nature of any party with respect to any such property, and

(v) resort to the Pledged Collateral for payment of any of the Obligations whether or not it shall have resorted to any other property securing the Obligations or shall have proceeded against any party primarily or secondarily liable on any of the Obligations.

SECTION 16. <u>Amendments; Etc.</u> No amendment or waiver of any provision of this Agreement nor consent to any departure by the Pledgor herefrom shall in any event be effective unless the same shall be in writing and signed by the Lender, and then such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given.

SECTION 17. Addresses for Notices. All notices and other communications provided for hereunder shall be in writing and mailed or delivered:

If to Pledgor: Liquidmetal Technologies, Inc.

30452 Esperanza

Rancho Santa Margarita, California 92688

Attn: Legal Department Facsimile: (813) 314-0270

6

With copy to: (which shall not constitute notice) Foley & Lardner LLP 100 North Tampa Street Suite 2700 Tampa, Florida 33602

Tampa, Florida 33602 Attn: Curt P. Creely Facsimile: (816) 221-4210

If to Lender:

 C_3 Capital Partners, L.P. C_3 Capital Partners II, L.P. 4520 Main Street, Suite 1600 Kansas City, Missouri 64111 Attention: Robert L. Smith

Bryan Cave LLP

With copy to: (which shall not constitute notice)

One Kansas City Place 1200 Main, Suite 3500 Kansas City, Missouri 64105 Attention: Thomas W. Van Dyke

or as to any party at such other address as shall be designated by such party in a written notice to each other party complying as to delivery with the terms of this Section. All such notices and other communications shall, when mailed or telegraphed, respectively, be effective when deposited in the mails, addressed as stated above.

SECTION 18. Continuing Security Interest. This Agreement shall create a continuing security interest in the Pledged Collateral and shall (a) remain in full force and effect until payment in full of the Obligations and the termination of any commitment by the Lender with respect thereto, (b) be binding upon the Pledgor, the Pledgor's successors, transferees and assigns, and (c) inure, together with the rights and remedies of the Lender hereunder, to the benefit of the Lender and its successors, transferees and assigns.

SECTION 19. Governing Law; Terms. This Agreement shall be effective upon delivery to the Lender, without further act, condition or acceptance by the Lender. Any invalidity or unenforceability of any provision or application of this Agreement shall not affect other lawful provisions and application thereof, and to this end the provisions of this Agreement are declared to be severable. This Agreement may not be waived, modified, amended, terminated, released or otherwise changed except by a writing signed by Pledgor and the Lender. This Agreement shall be governed by and construed in accordance with the substantive laws (other than conflict laws) of the State of Missouri. The Pledgor (a) consents to the personal jurisdiction of the state and federal courts located in the State of Missouri in connection with any controversy related to this Agreement; (b) waives any argument that venue in any such forum is not convenient, (c) agrees that any litigation initiated by the Lender or the undersigned in connection with this Agreement may be venued in either the state courts located in Jackson County, Missouri or the United States Federal Court in the Western District of Missouri; and (d) agrees that a final judgment in any

7

such suit, action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.

[Remainder of page intentionally left blank.]

8

Signature Page to Liquidmetal Technologies, Inc. Pledge Agreement

THE PLEDGOR WAIVES ANY RIGHT TO TRIAL BY JURY (WHICH THE LENDER ALSO WAIVES) IN ANY CLAIM, COUNTERCLAIM, ACTION OR PROCEEDING BASED ON OR PERTAINING TO THIS AGREEMENT.

 $IN\ WITNESS\ WHEREOF, the\ Pledgor\ has\ executed\ this\ Agreement\ as\ of\ the\ date\ first\ written\ above.$

LIQUIDMETAL TECHNOLOGIES, INC.

a Delaware corporation

By:
Printed:
Its:

S-1

Exhibit F

PLEDGE AGREEMENT (Liquidmetal Coatings, LLC)

This Membership Unit Pledge Agreement ("Agreement") is made as of this 24th day of July, 2007, by and among Liquidmetal Coatings, LLC a Delaware limited liability company, (Pledgor"), C₃ Capital Partners, L.P., a Delaware limited partnership ("C₃") and C₃ Capital Partners II, L.P., ("C₃ II" and C₃ and their successors and assigns shall hereinafter be referred to as the "Lender").

PRELIMINARY STATEMENTS:

- (1) Upon the consummation of the transactions contemplated by that certain Securities Purchase Agreement, together with any and all other documentation related thereto (all as amended, restated, extended or otherwise modified from time to time, the "SPA") dated as of the date hereof by and among the Lender, Pledgor and Liquidmetal Coatings Solutions, LLC, a Delaware limited liability company (the "Company"), Pledgor shall be the owner of all of the issued and outstanding membership units of the Company ("Membership Units").
 - (2) The execution and delivery of this Agreement by Pledgor is a condition precedent to the SPA.
 - (3) Capitalized terms, unless otherwise defined herein, shall have the meanings set forth in the SPA.

NOW, THEREFORE, in consideration of the premises and in order to induce the Lender to enter into the SPA, the Pledgor hereby covenants to the Lender as follows:

SECTION 1. Pledge. The Pledgor hereby pledges to the Lender and grants to the Lender a security interest in all of the Pledgor's now owned or hereinafter acquired right, title and interest in and to the following (the "Pledged Collateral"):

- (a) the Membership Units;
- (b) the proceeds, if and when received, of the Membership Units and other instruments and documents representing the Membership Units, and all cash, instruments and other property from time to time received, received, receivable or otherwise distributed in respect of or in exchange for any or all of the Membership Units;

- (c) the proceeds, if and when received, of all equity interests from time to time received or acquired by the Pledgor in any manner resulting from the acquisition or formation by the issuer of the Membership Units of a subsidiary, the certificates and other instruments and documents representing such equity interests and all cash, instruments and other property from time to time received, receivable or otherwise distributed in respect of or in exchange for any or all of such equity interests; and
- (d) all other property hereafter delivered to the Lender in substitution for or in addition to any of the foregoing, all certificates, certificates of deposit, notes, instruments and documents representing or evidencing such property, and all cash, securities, interest, rights and other property at any time and from time to time received, receivable or otherwise distributed in respect of or in exchange for any or all thereof.
- SECTION 2. Security for Obligations. This Agreement secures the payment and performance of all the Company's obligations of every kind and nature under the SPA (the "Obligations"). For the purpose of this Agreement, the term "Event of Default" is defined as an Event of Default including any applicable cure periods under the SPA.
- SECTION 3. <u>Delivery of Pledged Collateral</u>. All instruments representing or evidencing the Pledged Collateral shall be delivered to and held by the Lender pursuant hereto and shall be in suitable form for transfer by delivery, or shall be accompanied by duly executed instruments of transfer or assignment in blank, all in form and substance satisfactory to the Lender. The Lender shall have the right, at any time after the occurrence of an Event of Default, in its discretion, to transfer to or to register in the name of the Lender or any of its nominees any or all of the Pledged Collateral. In addition, the Lender shall have the right at any time to exchange instruments representing or evidencing Pledged Collateral for instruments of smaller or larger denominations.

SECTION 4. <u>Representations and Warranties</u>. The Pledgor represents and warrants as follows:

- (a) The Membership Units have been duly authorized and validly issued and are fully paid and non-assessable.
- (b) The Pledgor is and will be at all times the legal and beneficial owner of the Membership Units and the Pledged Collateral free and clear of any lien, security interest, option or other charge or encumbrance, except for security interests created in favor of the Lender, the Senior Lender and security interests otherwise consented to in advance and in writing by the Lender.
 - (c) The pledge pursuant to this Agreement of any Pledged Collateral creates or will create a valid and perfected security interest in the Pledged Collateral, securing the payment of the Obligations.
- (d) No authorization, approval or other action by, and no notice to or filing with, any governmental authority or regulatory body is required either (i) for the pledge by the Pledger of the Pledged Collateral pursuant to this Agreement or for the execution, delivery or performance of this Agreement by the Pledger or (ii) for the exercise by the Lender of the rights provided for in this Agreement or the remedies in respect of the Pledged Collateral pursuant to this Agreement (except as may be required in connection with such disposition by laws affecting the foreclosure of security interests generally).

2

SECTION 5. <u>Further Assurances</u>. The Pledgor agrees that at any time and from time to time, at the expense of the Pledgor, promptly execute and deliver all further instruments, certificates and documents and take all further action that may be reasonably necessary or desirable or that the Lender may reasonably request in order to perfect and protect any security interest granted or purported to be granted hereby or to enable the Lender to exercise and enforce its rights and remedies hereunder with respect to any Pledged Collateral.

SECTION 6. Pledgor's Rights; Distributions; Etc.

- (a) So long as no Event of Default shall have occurred and be continuing:
- (i) The Pledgor shall be entitled to exercise any rights pertaining to the Pledged Collateral or any part thereof for any purpose not inconsistent with the terms of this Agreement including the right to vote the Membership Units for any purpose allowed under the Company's Certificate of Formation, Operating Agreement or applicable law.
 - (ii) The Pledgor shall be entitled to receive and retain any and all interest and distributions in respect of the Pledged Collateral; provided, however, that any and all:
 - (A) proceeds, interest and distributions paid or payable other than in cash in respect of, and instruments and other property received, receivable or otherwise distributed in respect of, or in exchange for, any Pledged Collateral;
 - (B) proceeds and distributions paid or payable in cash in respect of any Pledged Collateral in connection with a partial or total liquidation or dissolution; and
 - (C) cash paid, payable or otherwise distributed in respect of principal of, or in redemption of, or in exchange for the Membership Units or any Pledged Collateral,

shall be, and shall be forthwith delivered to the Lender to hold as, Pledged Collateral and shall, if and when received by the Pledgor, be received in trust for the benefit of the Lender, be segregated from the other property or funds of the Pledgor, and be forthwith delivered to the Lender as Pledged Collateral in the same form as so received (with any necessary endorsement).

- (b) Upon the occurrence and during the continuance of an Event of Default:
- (i) All rights of the Pledgor to receive the interest and distributions on the Pledged Collateral which Pledgor would otherwise be authorized to receive and retain pursuant to Section 6(a)(ii) shall cease, and all such rights shall thereupon become vested in the Lender which shall thereupon have the sole right to receive and hold as Pledged Collateral such interest, distributions and payments.

3

- (ii) All interest, distributions and payments which are received by the Pledgor contrary to the provisions of paragraph (i) of this Section 6(b) shall be received in trust for the benefit of the Lender, shall be segregated from other funds of the Pledgor and shall be forthwith paid over to the Lender as Pledged Collateral in the same form as so received (with any necessary endorsement).
- SECTION 7. <u>Transfers and Other Liens; Notice to Prior Secured Party.</u> The Pledgor agrees that Pledgor will not create or permit to exist any lien, security interest, or other charge or encumbrance upon or with respect to the Membership Units or any of the Pledged Collateral, except for the security interest under this Agreement and the security interest of the Senior Lender.
- SECTION 8. <u>Lender Appointed Attorney-in-Fact</u>. Upon the occurrence of an Event of Default, the Pledgor hereby irrevocably appoints the Lender the Pledgor's attorney-in-fact, with full authority in the place and stead of the Pledgor and in the name of the Pledgor or otherwise, from time to time in the Lender's discretion, to take any action and to execute any instrument which the Lender may deem necessary or advisable to accomplish the purposes of this Agreement, including, without limitation, to demand, receive, endorse, hold and collect all instruments made payable to the Pledgor representing any distribution in respect of the Pledged Collateral or any part thereof and to give full discharge for the same.
- SECTION 9. <u>Lender May Perform</u>. If the Pledgor fails to perform any agreement contained herein, the Lender may itself perform, or cause performance of, such agreement, and the reasonable expenses of the Lender incurred in connection therewith shall be payable by the Pledgor in accordance with Section 13.
- SECTION 10. Reasonable Care. The Lender shall be deemed to have exercised reasonable care in the custody and preservation of the Pledged Collateral in its possession if the Pledged Collateral is accorded treatment substantially equal to that which the Lender accords its own property.
 - SECTION 11. Remedies. If any Event of Default shall have occurred and be continuing:
- (a) The Lender may exercise in respect of the Pledged Collateral, in addition to other rights and remedies provided for herein and as otherwise available to it at law or in equity, all the rights and remedies of a secured party on default under the Uniform Commercial Code in effect in the State of Missouri at that time, and the Lender may also, without notice as specified below, execute upon and apply (for itself or any designee it elects) the Pledged Collateral directly toward the payment of the Obligations, and sell the Pledged Collateral or any portion thereof at public or private sale, at any exchange, broker's board or at any of the Lender's offices or elsewhere, for cash, on credit or for future delivery, and at such price or prices and upon such other terms as the Lender may deem commercially reasonable. The Pledgor agrees that, to the extent notice of sale shall be required by law, at least twenty-five (25) days' notice to the Pledgor of the time and place of any public sale and one publication in a local

4

newspaper, or at least twenty-five (25) days' notice of the time after which any private sale is to be made, shall constitute reasonable notification. The Lender shall not be obligated to make any sale of Pledged Collateral regardless of notice of sale having been given. The Lender may adjourn any public or private sale from time to time by announcement at the time and place fixed therefore, and such sale may, without further notice, be made at the time and place to which it was so adjourned.

- (b) Any cash held by the Lender as Pledged Collateral and all cash proceeds received by the Lender in respect of any sale of, collection from, or other realization upon all or any part of the Pledged Collateral may, in the discretion of the Lender, be held by the Lender as collateral for, and then or at any time thereafter applied (after payment of any amounts payable to the Lender pursuant to Section 13) in whole or in part by the Lender against, all or any part of the Obligations in such order as the Lender such sale lelect. Any surplus of such cash or cash proceeds held by the Lender and remaining after payment in full of all the Obligations shall be paid over to the Pledgor or to whomsoever may be lawfully entitled to receive such surplus.
- SECTION 12. <u>Indemnification</u>. Neither the Lender, nor any partner, officer, agent or employee of the Lender, shall be liable for any action taken or omitted to be taken by it or them hereunder or in connection therewith, except for its or their own gross negligence or willful misconduct.
- SECTION 13. Expenses. The Pledgor will, upon demand, pay to the Lender the amount of any and all reasonable expenses, including the reasonable fees and expenses of its counsel and of any experts and agents, which the Lender may incur in connection with (a) the custody or preservation of, or the sale of, collection from, or other realization upon, any of the Pledged Collateral, (b) the exercise or enforcement of any of the rights of the Lender

SECTION 14. Security Interest Absolute

- All rights of the Lender and security interests hereunder, and all obligations of the Pledgor hereunder, shall be absolute and unconditional irrespective of:
 - any lack of validity or enforceability of the SPA; (i)
 - any change in the time, manner or place of payment of, or in any other term of, all or any of the Obligations, or any other amendment or waiver of or any consent to any departure from the SPA; (ii)
 - (iii) any exchange, release or non-perfection of any other collateral, or any release of any of the Obligations; or
- (iv) any other circumstance which might otherwise constitute a defense available to, or a discharge of, the Pledgor or any other pledgor or any guarantor or co maker in respect of the Obligations or the Pledgor in respect of this Agreement.

5

- The Lender may, from time to time, whether before or after any of the Obligations shall become due and payable, without notice to the Pledgor, take all or any of the following actions: (b)
 - retain or obtain a security interest in any property, in addition to the Pledged Collateral, to secure any of the Obligations, (i)
 - (ii) retain or obtain the primary or secondary liability of any party or parties with respect to any of the Obligations,
- (iii) extend or renew for any period (whether or not longer than the original period) or exchange any of the Obligations or release or compromise any obligation of any nature of any party with respect

- surrender, release or exchange all or any part of any property, in addition to the Pledged Collateral, securing any of the Obligations, or compromise or extend or renew for any period (whether or not (iv) longer than the original period) any obligations of any nature of any party with respect to any such property, and
- resort to the Pledged Collateral for payment of any of the Obligations whether or not it shall have resorted to any other property securing the Obligations or shall have proceeded against any party primarily or secondarily liable on any of the Obligations.
- SECTION 15. Amendments; Etc. No amendment or waiver of any provision of this Agreement nor consent to any departure by the Pledgor herefrom shall in any event be effective unless the same shall be in writing and signed by the Lender, and then such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given

SECTION 16. Addresses for Notices. All notices and other communications provided for hereunder shall be in writing and mailed or delivered:

If to Pledgor: Liquidmetal Coatings, LLC

30452 Esperanza

Rancho Santa Margarita, California 92688

Attn: Legal Department Facsimile: (813) 314-0270

Foley & Lardner LLP

With copy to: (which shall not constitute notice)

100 North Tampa Street Suite 2700 Tampa, Florida 33602 Attn: Curt P. Creely Facsimile: (813) 221-4210

If to Lender: C3 Capital Partners, L.P.

C₃ Capital Partners II, L.P. 4520 Main Street, Suite 1600 Kansas City, Missouri 64111 Attention: Robert L. Smith

With copy to: (which shall not constitute notice)

Bryan Cave LLP One Kansas City Place 1200 Main, Suite 3500 Kansas City, Missouri 64105 Attention: Thomas W. Van Dyke

or as to any party at such other address as shall be designated by such party in a written notice to each other party complying as to delivery with the terms of this Section. All such notices and other communications shall, when mailed or telegraphed, respectively, be effective when deposited in the mails, addressed as stated above.

Continuing Security Interest. This Agreement shall create a continuing security interest in the Pledged Collateral and shall (a) remain in full force and effect until payment in full of the Obligations and the termination of any commitment by the Lender with respect thereto, (b) be binding upon the Pledgor, the Pledgor's successors, transferees and assigns, and (c) inure, together with the rights and remedies of the Lender hereunder, to the benefit of the Lender and its successors, transferees and assigns.

Governing Law: Terms. This Agreement shall be effective upon delivery to the Lender, without further act, condition or acceptance by the Lender. Any invalidity or unenforceability of any provision or SECTION 18. application of this Agreement shall not affect other lawful provisions and application thereof, and to this end the provisions of this Agreement are declared to be severable. This Agreement may not be waived, modified, amended, terminated, released or otherwise changed except by a writing signed by Pledgor and the Lender. This Agreement shall be governed by and construed in accordance with the substantive laws (other than conflict laws) of the State of Missouri. The Pledgor (a) consents to the personal jurisdiction of the state and federal courts located in the State of Missouri in connection with any controversy related to this Agreement; (b) waives any argument that venue in any such forum is not convenient, (c) agrees that any litigation initiated by the Lender or the undersigned in connection with this Agreement may be venued in either the state courts located in Jackson County, Missouri or the United States Federal Court in the Western District of Missouri; and (d) agrees that a final judgment in any such suit, action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by

[Remainder of page intentionally left blank.]

Signature Page to Liquidmetal Coatings, LLC Pledge Agreement

THE PLEDGOR WAIVES ANY RIGHT TO TRIAL BY JURY (WHICH THE LENDER ALSO WAIVES) IN ANY CLAIM, COUNTERCLAIM, ACTION OR PROCEEDING BASED ON OR PERTAINING TO THIS AGREEMENT

IN WITNESS WHEREOF, the Pledgor has executed this Agreement as of the date first written above.

LIQUIDMETAL COATINGS, LLC

a Delaware limited liability company

Printed Its:

COLLATERAL ASSIGNMENT OF RIGHTS AND SUMS DUE UNDER THE ASSET PURCHASE AND CONTRIBUTION AGREEMENT

WHEREAS, LIQUIDMETAL COATINGS, LLC a Delaware limited liability company (the "Company") and LIQUIDMETAL TECHNOLOGIES, INC., a Delaware corporation ("Seller") are parties to a certain Asset Purchase and Contribution Agreement, dated as of July 24, 2007 (hereinafter, together with any and all amendments thereto, the "Agreement"), pursuant to which the Company shall acquire certain assets of Seller, as specifically identified therein.

WHEREAS, pursuant to the Agreement, Seller has made certain representations, warranties and covenants in favor of the Company and agreed to indemnify Company in certain respects; and

WHEREAS, the Company, **LIQUIDMETAL COATINGS SOLUTIONS, LLC**, a Delaware limited liability company, **C**₃ **CAPITAL PARTNERS, L.P.**, a Delaware limited partnership ("C₃ **Capital**") and **CAPITAL PARTNERS II, L.P.**, a Delaware limited partnership ("C₃ **Capital II**," together with C₃ Capital shall hereinafter be referred to as "C₃") are parties to a certain Securities Purchase Agreement dated as of July 24, 2007 (hereinafter, as at any time amended, modified or supplemented, the "SPA"); and

WHEREAS, C_3 has required, as a condition to Closing under the SPA, that the Company assign to C_3 all of the Company's rights and remedies under the Agreement and all of the Company's right, title and interest in and to any and all sums due from Seller under the Agreement in order to secure the Company's obligations to C_3 pursuant to the SPA; and

WHEREAS, capitalized terms, unless otherwise defined herein, shall have the meanings set forth in the SPA.

NOW, THEREFORE, in consideration of the sum of Ten Dollars (\$10.00), the premises set forth herein and for other good and valuable consideration, the Company agrees as follows:

- 1. As security for the payment and performance of the Company's obligations to C₃ arising under the SPA (the "Liabilities"), the Company hereby assigns, transfers and sets over to C₃ and hereby grants to C₃ as security interest in, all of the Company's rights and remedies and right, title and interest in and to any and all sums due from Seller under the Agreement, including, without limitation, its rights and remedies with respect to (a) any breach by Seller of Seller's representations, warranties and covenants under the Agreement and (b) any indemnification from Seller arising under or pursuant to the Agreement.
- 2. Prior to the occurrence of an Event of Default, insofar as the Company may have any right, privilege or claim against Seller under the Agreement, including, without limitation, all of the Company's rights and remedies with respect to indemnification, the Company will use prudent business judgment concerning the enforcement of such rights and, if in the exercise of such judgment the Company determines to enforce such rights or remedies, it will enforce the same diligently and in good faith.
- 3. Effective from and after the occurrence of an Event of Default, the Company hereby irrevocably authorizes and empowers C₃ or its agents, in C₃'s sole discretion, to assert, either directly or on behalf of the Company, any right, privilege or claim the Company may, from time to time, have against Seller under the Agreement, as C₃ may deem proper, and to receive and collect any and all damages, awards and other monies resulting therefrom and all other sums due from Seller under the Agreement and to apply the same on account of the Liabilities. In no event, however, shall C₃ be obligated to assert any such right, privilege or claim of the Company or to collect any sums and C₃'s failure to do so shall not give rise to any liability to the Company or any other Persons.
- 4. The Company hereby irrevocably makes, constitutes and appoints C_3 (and all officers, employees or agents designated by C_3) as its true and lawful attorney (and agent-in-fact) for the purposes of enabling C_3 or its respective agents to assert such rights, privileges and claims, to collect such sums related to the foregoing and to apply such monies in the manner set forth hereinabove.
- 5. The Company shall keep C₃ fully informed of all circumstances bearing upon the Company's rights, remedies and sums due under the Agreement, and the Company shall not waive, amend, alter or modify any of its rights or remedies under the Agreement in any material respect, without the prior written consent of C₃. The Company shall promptly advise C₃ in writing of any default by Seller in the observance or performance of any of Seller's obligations under the Agreement.
 - 6. This Agreement shall continue effective until the termination of the SPA and the payment in full of the Liabilities.
 - 7. This Agreement shall be governed by and construed according to the laws and decisions of the State of Missouri and shall be binding upon the Company and its successors and assigns.
- 8. Notwithstanding the foregoing, the Company expressly acknowledges and agrees that it shall remain liable under the Agreement to observe and perform all of the conditions and obligations therein contained to be observed and performed by it, and that neither this Agreement, nor any action taken pursuant hereto, shall cause C_3 to be under any obligation or liability in any respect whatsoever to any party to the Agreement for the observance or performance of any of the representations, warranties, conditions, covenants, agreements or terms therein contained.

2

Signature Page to Collateral Assignment of Rights and Sums Due Under the Asset Purchase and Contribution Agreement

IN WITNESS WHEREOF, this instrument has been duly executed and delivered as of July 24, 2007.

LIQUIDMETAL COATINGS, LLC

	Name:tts:		
Accepted	in Kansas City, Missouri:		
C ₃ CAPI	TAL PARTNERS, L.P.		
By: Its:	C ₃ Partners, LLC General Partner		
	By: Name:	Its:	Manager
C ₃ CAPI	TAL PARTNERS II, L.P.		
By: Its:	C ₃ Partners, LLC General Partner		
	By: Name:	Its:	Manager

Exhibit H-1

U.S. SMALL BUSINESS ADMINISTRATION SIZE STATUS DECLARATION

S-1

OMB Approval No. 3245-0009 Expiration Date: 08/31/2007

This form must be completed by a business concern ("Applicant") before it can receive financing or consulting and advisory services from a small business investment company licensed by SBA ("Licensee"). The Applicant should complete Part A and Part B (if necessary), sign the Applicant's certification, and return the form to the Licensee from whom it is seeking assistance. The Licensee should sign the Licensee's certification and retain the form in its files. Please do not send forms to SBA or to the Office of Management and Budget.

Name and address of Licensee Name and address of Applicant

Kansas City,	MO 64111					
Applicant's I	Form of Organization:	o Corporation o Proprietorship	o Partnership	o Lim	ited Liability Company	
			PART A			
1. 2.	Does Applicant (including affiliates) have ts \$18,000,000? (Tangible net worth = total ne Does Applicant (including affiliates) have a taxes (excluding any carry-over losses) for in excess of \$6,000,000?	et worth minus goodwill) verage net income after Federal income			Yes o o	No 0 0
			PART B			
		ther question in Part A was "Yes". Applicant mus gaged. Find the appropriate industry size standard			ined with its affiliates is primarily e	engaged, and (2)
1. Applicant co	Primary industry (include NAICS code): ombined with affiliates				Applicant Alor	ne
2.	Total annual receipts of Applicant (excluding	g affiliates) for each of its 3 most recently comple Year ended Year ended Year ended	eted fiscal years see 13 CFR 12 \$ \$ \$:1.104):		
3.		excluding affiliates) based on the number of person		art-time, temporary, or other basis during each of	the pay periods of the preceding 1	2 calendar months
	(see 13 CFR 121.106):					
4. 4.653	(A.);	T	21.		1 11	
	of Applicant d foreign)—Names and full	Total annual receipts of aff (excluding Applicant) for p completed fiscal years		Average no. of persons em affiliates (excluding Appli full-time, part-time, tempo basis during each of the pe the preceding 12 calendar	cant) on a orary, or other ny periods of	
a.		a. Yr. Yr.	\$ \$	a.		
		Yr. 3-year average	\$ \$			
b.		b. Yr. Yr.	\$ \$	b.		
		Yr. 3-year average	\$ \$			
C.		c. Yr. Yr. Yr.	\$ \$ \$	c.		
		3-year average	\$			
period of not	t less than 1 year from the date of the final di-	uthorized officer, hereby certifies that all informat sbursement of the funds involved in the subject fi peration for which the application for financing o	nancing and for a period of not	less than 1 year from the date of the commence		
				- Na	nme of Applicant	
Date		By: (Signature of Officer)		Title		
		n available to us, including all information and fac 58, as amended, and the Regulations of SBA there		vestigation, the Licensee has concluded that the	Applicant is a small business conc	ern within the
Date		By: (Signature of Officer)		Title		
comments co	oncerning this estimate or other aspects of thi	nis form is 10 minutes per response. You will not s information collection, please contact the U.S. S				
and Budget, Clearance Officer, Paperwork Reduction Project (3245-0009), Washington, D.C. 20503.			This form was	s electronically produced by Elite F	ederal Forms, Inc.	
						Exhibit H-2
).	ASSUI	U.S. SMALL BUSINESS. RANCE OF COMPLIANCE	ADMINISTRATION FOR NONDISCRIMINATION		
comply with		see/Recipient/Subrecipient, (hereinafter referred of FR parts 112, 113, and 117, of the Regulations iss			Business Administration, herewith	agrees that it will
13 CFR Parts	s 112, 113 and 117 require that no person sha	ll on the grounds of age, color, handicap, marital te applicant received Federal financial assistance	status, national origin, race, reli	, ,	be denied the benefits of or otherwi	se be subjected to
record keepii successors, ti	ng requirements have been approved under C ransferees and assignees as long as it/they re	rements of 13 CFR 112.9, 113.5, and 117.9 as set MB Number 3245-0076. Applicant further agree ceive or retain possession of any Federal financial nt or suspend any or all Federal financial assistan	s to obtain or require similar As I assistance from SBA. In the ev	ssurance of Compliance for Nondiscrimination f	rom subrecipients, contractors/subo	contractors,
Executed the		day of				
				Name Address & Phor	ne No. of Applicant	
			Ву	Typed Name & Title of Authorized Official		

Corporate Seal Signature of Authorized official

By		
	Typed Name & Title of Authorized Official	

Corporate Seal

Signature of Authorized Official

Exhibit H-3



U.S. Small Business Administration Portfolio Financing Report

OMB No. 3245-0078 Expiration Date: 09/30/2007

Name of Licensee C3 CAPITAL PARTNERS, LP Part A - Small Business Concern Data

1. Name of Small Concern

3. Street Address

6. Zip Code

9. Contact Person for FAX

11. Form of Business (select one)

12. NAICS Code

13. Percentage of Small Concern (if any) Owned by: Black or African American

American Indian or Alaska Native

14. Percentage of Small Concern Owned by Women (if any)

Part B - Prefinancing Information

Prefinancing Status (select one)

15. Fiscal Year End Immediately Prior to Date of Financing (Month/Day/Year)

16. Gross Revenue for Prior Fiscal year \$

18. Income Taxes for Prior Fiscal Year: Federal \$

19. Employee Payroll Tax Withholdings for Prior Fiscal Year: Federal \$

20. Total Assets \$

23. Number of Employees

Part C - Financing Information

24. Date of Financing 26. Purpose of Financing (enter appropriate number(s); 1-10)

1) Working Capital or Inventory Purchase

2) Plant Modernization or Leasehold Improvement

3) Acquisition of All or Part of an Existing Business

4) Consolidation of Obligations or Non-SBIC Debt Refunding

5) New Building or Plant Construction

27. Is this the First Financing of this Small Business by the Licensee?

28. Financing Instruments and Applicable Amounts (for participations, include Licensee's portion only):

Instrument Amount Loan Only Debt with Equity Features Equity Only \$ Total Financing

29. Comments

License Number 07/07-0111

2. Employer Identification Number

State

8 Small Business FAX

10. Date Business Established 1) Corporation 2) Partnership 3) Proprietor 4) Limited Liability Company

%Hispanic or Latino

Native Hawaiian/Other Pacific Islander

%White 14a. Percentage Owned by Veterans (if

any)

1) New Information 2) Previously Submitted 3) Acquired Business 4) New Business

17. After-Tax Profit (Loss) for Prior Fiscal Year \$

Local \$ State and Local \$ 22. Retained Earnings

21. Net Worth (Deficit) \$

(Deficit) \$

25. Date of Disbursement / 2) 1)

3) 6) Acquisition of Machinery and Equipment 7) Land Acquisition or Dwelling Construction on

Existing Land 8) Marketing Activities 9) Research and Development

> 10) Other (specify) o Yes o No

% Ownership Initial Interest Rate(s)

%

Part D - Financing Information

Transmission Date /

PLEASE NOTE: The estimated burden for the completion of this form is 12 minutes per response. You will not be required to respond to this information collection if a valid OMB approval number is not displayed. If you have questions or comments concerning this estimate or other aspects of this information collection, please contact the U.S. Small Business Administration, Chief, Administrative Information Branch, Washington, D.C. 20416 and/or Office of Management and Budget, Clearance Officer, Paperwork Reduction Project (3245-0078), Washington, D.C. 20503. PLEASE DO NOT SEND FORMS TO OMB.

7 County

or Industry (if NAICS Code is not known)

Exhibit I

SOLVENCY CERTIFICATE

Furnished pursuant to Section 3.13(p) of the Securities Purchase Agreement dated as of July 24, 2007 by and among Liquidmetal Coatings, LLC (the "Company"), Liquidmetal Coatings Solutions, LLC ("LMCS"), C₃ Capital Partners, L.P. ("C₃"), C₃ Capital Partners, L.P. II ("C₂,") together with C₃ II, shall collectively be referred to herein the "Lenders") (the "Purchase Agreement").

The undersigned hereby certifies that he is the of the Company and that, as such, he is authorized to execute this Solvency Certificate on behalf of the Company and the undersigned further certifies that, on the date hereof and based on the financial information delivered pursuant to Section 6.5 of the Purchase Agreement, after giving effect to the transactions contemplated by the Investment Documents

- the fair salable value of the assets of the Company is greater than the total amount of the liabilities of the Company (including contingent, subordinated, unmatured and unliquidated liabilities, whether or not includable on its balance sheet in accordance with GAAP):
- (b) to the undersigned's knowledge, the present fair salable value of the assets of the Company is greater than the amount that will be required to pay the probable liabilities of the Company as they become absolute and matured:
- to the undersigned's knowledge, the Company will be able to realize upon its assets, or will have sufficient cash flow, to enable the Company to pay its debts, other liabilities, contingent obligations and (c) other commitments as they mature in the ordinary course of business; and
 - to the undersigned's knowledge, the Company does not have an unreasonably small amount of capital with which to engage in its anticipated businesses.

Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to them in the Purchase Agreement. For purposes of the foregoing, the "fair salable value" of any asset or investment was determined on the basis of the amount which may be realized within a reasonable time, either through collection or sale of such asset or investment at its regular market value, conceiving the latter as the amount which could be obtained therefor within such period by a capable and diligent businessman from an interested buyer who is willing to purchase under ordinary selling conditions.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand on behalf of the Company on this 24th day of July, 2007,

LIQUIDMETAL COATINGS, LLC

By:

		Name: Title:
		Exhibit J
IRREVOC	ABLE SPRI	NGING PROXY
when payable (whether at maturity or otherwise) the full amount of interest then accrued on any Note, or the f months (the "Triggering Event of Default"), the undersigned holder (the "Holder") of the amount of Members Capital Partners, L.P., a Delaware limited partnership, and C ₃ Capital Partners, II L.P., a Delaware limited part other ways shall have complete discretion to act in its place and stead as to all Membership Units subject here	full amount of ship Units of the thership (collecto as long as the ers of the Comproxies hereto eement among	ctively, the "Proxy Holder"), as its attorney and proxy to call and attend meetings, vote, give consents, and in all his Irrevocable Proxy is in effect for the purpose of removing, designating, electing or appointing members to the pany. Upon the occurrence of the Triggering Event of Default and until cured in accordance with the Purchase offer given shall, at such time, then be revoked. All capitalized terms used in this Irrevocable Proxy and not g the Proxy Holder, the Company and Liquidmetal Coatings Solutions, LLC, dated July 24, 2007, as may be
$2. \qquad \text{Upon the cure of the Triggering Event of Default in accordance with the Purchase Agreement, all right Default.}$	hts of the Prox	ty Holder in this Irrevocable Proxy shall revert to the Holder until the re-occurrence of such Triggering Event of
3. In compliance with applicable law, this Irrevocable Proxy is irrevocable and coupled with an interest	and executed	in consideration of the purchase of the Notes and Membership Units pursuant to the Purchase Agreement.
4. Any additional Membership Units of the Company issued to the Holder shall be subject to this Irrevocable Proxy. Certificates representing the Membership Units indicated below and any additional Membership Company issued to the Holder shall be affixed with a legend indicating that those Membership Units are subject to this Irrevocable Proxy as follows:		
"The voting power of the securities represented by this Certificate has been previously granted to C_3 which proxy is irrevocable."	Capital Partne	ers, L.P. and C_3 Capital Partners II, L.P. pursuant to a written springing proxy effective December 20, 2007
5. This Proxy shall terminate upon the earlier of: (i) the indefeasible payment in full of all amounts due this Irrevocable Proxy shall terminate by	and owing un	der the Note; or (ii) the longest period permitted under the laws applicable to this Irrevocable Proxy. In the event
the execution and delivery of such new proxy is not prohibited by or in violation of applicable law. 6. In the event of a dispute or controversy arising out of or relating to this Irrevocable Proxy, or perform consents, and in all other ways to act in its place and stead as to all Membership Units subject hereto for the properties of the prevailing process. The prevailing party in any litigation or proceeding proceeding process of the p	ourpose of rem pertaining to th	is Irrevocable Proxy shall be entitled to reasonable attorney's fees actually incurred, together with costs of the

By: Name: Title: Membership Interests subject to this Irrevocable Proxy: 6,986		
AGREED AND ACKNOWLEDGED: C ₃ CAPITAL PARTNERS, L.P.	C ₂ CAPT	TAL PARTNERS II, L.P.
By: C ₃ Partners, LLC Its: General Partner By: Name:	By: Its:	C ₃ Partners, LLC General Partner By: Name:
Title: Manager	2	Title: Manager

Exhibit K

COMPLIANCE CERTIFICATE

Liquidmetal Coatings, LLC

Pursuant to Section 11.2(c) of the Asset Purchase Agreement among Liquidmetal Coatings, LLC, a Delaware limited liability company", and Liquidmetal Technologies, Inc., a Delaware corporation ("LMT"), dated as of July 24, 2007 (the "Asset Purchase Agreement"), the Company hereby certifies to LMT as follows:

- A. Each of the representations and warranties made by Company set forth in the Asset Purchase Agreement is true and correct as of the date hereof; and
- В. Company has performed and complied with all of Company's obligations which are to be performed or complied with under the Asset Purchase Agreement on or prior to the date hereof.

IN WITNESS WHEREOF, the undersigned has executed this Compliance Certificate as of July 24, 2007.

Larry Buffington Chief Executive Officer

Exhibit L

Liquidmetal Technologies Inc. Balance Sheet FY 2007 (Thousands of USD)

CORPORATE CONSOLIDATED

	DR After Tr Coatings	DRAFT After Transaction Coatings Application Tota	
Cash	120		120
Accounts receivable	1,687	_	1,687
Inventories	387	380	767
Prepaid expenses	_	_	_
Total Current Assets	2,194	380	2,574

Property, Plant & Equipment, Net	2	435	437
Intangible Assets, Net	21	_	21
Other Assets	474	_	474
Goodwill	9,978		9,978
Investment in Coatings			_
Investment in LSI	_	_	_
TOTAL ASSETS	\$ 12,669	\$ 815	\$ 13,484
Accounts Payable and Accrued Expenses	1,151	78	1,229
Deferred Revenue	81	_	81
Current Portion of Notes Payable	1,905	300	2,205
Total Current Liabilities	3,137	378	3,515
Capital lease & other L/T Curr. Portion	_	_	_
Discount			_
Long Term Notes Payable	7,543	150	7,693
Total Liabilities	10,680	528	11,208
Common Stock	_	_	_
Paid-In Capital		_	_
Intercompany LMT/LMG			_
Accumulated Deficit		_	_
Acc. Comprehensive Incom (Loss)	_	_	_
Total Shareholders' (Deficit) Equity	1,989	287	2,276
TOTAL LIABILITIES & EQUITY	\$ 12,669	\$ 815	\$ 13,484
	Ψ 12,000	. 010	10,101

Appendix 1

Definitions

- "2007 Operating Budget" has the meaning set forth in Section 3.13(k).
- "Affiliate" of any particular party means: (a) any other party which directly or indirectly, controls or is controlling by or is under common control with such party and (b) any officer, director, manager, managing partner or member of such party or any other party holding a similar position with respect to such party. A party shall be deemed to be "controlled by" any other party if such other party possesses, directly or indirectly, power to vote 20% or more of the securities (on a fully diluted basis) having ordinary voting power for the election of directors, managers or managing partners or power to direct or cause the direction of the management and policies of such party whether by contract or otherwise.
 - "Agreement" has the meaning set forth in the introductory paragraph.
 - "Annual Budget" has the meaning set forth in Section 5.1(h).
 - "Approved Annual Budget" has the meaning set forth in Section 5.10.
 - "Asset Purchase and Contribution Agreement" has the meaning set forth in Section 2.4.
 - "BMW" has the meaning set forth in Section 2.3.
- "Business Day" means any day other than a Saturday, Sunday or public holiday under the laws of the State of Missouri or other day on which banking institutions are authorized or obligated to close in Kansas City, Missouri or the location of LMC's headquarters.
 - "C3" has the meaning set forth in the introductory paragraph.
 - " C_3 II" has the meaning set forth in the introductory paragraph.
 - "C₃ Membership Units" has the meaning set forth in Section 2.1(a).
 - " C_3 II Membership Units" has the meaning set forth in Section 2.1(b).
 - " C_3 Note" has the meaning set forth in Section 2.1(a).
 - " C_3 II Note" has the meaning set forth in Section 2.1(b).
- "Capital Expenditures" means all expenditures which, in accordance with GAAP, would be required to be capitalized and shown on the balance sheet of the Company but excluding expenditures made in connection with the replacement, substitution or restoration of assets to the extent financed: (a) from insurance proceeds (or similar recoveries) paid on account of the loss of or damage to the assets being replaced or restored; (b) with awards of

compensation arising from the taking by eminent domain or condemnation of the assets being replaced; or (c) substantially concurrently with the proceeds from the sale of similar assets.

- "Capitalized Lease" means a lease under which the obligations of the lessee should, in accordance with GAAP, be included in determining total liabilities as shown on the liability side of a balance sheet of the lessee.
- "Capitalized Lease Obligations" means the amount of the liability reflecting the aggregate discounted amount of future payments under all Capitalized Leases calculated in accordance with GAAP and Statement of Financial Accounting Standards No. 13.
- "Change in Control" means: (a) the acquisition of the Company by another entity by means of any transaction or series of related transactions (including, without limitation, any reorganization, merger or consolidation, but excluding any merger effected primarily for the purpose of changing the domicile of the Company), unless the Company's members of record as constituted immediately prior to such transaction or series of related transactions will, immediately after such transaction or series of related transactions, hold at least a majority of the voting power (and if such entity is a partnership for income tax purposes, a majority interest in the capital and profits) of the surviving or acquiring entity or (b) a sale of all or substantially all of the assets of the Company.
 - " $\pmb{\text{Closing}}$ ' has the meaning set forth in Section 2.3.
- "Code" means the Internal Revenue Code of 1986, as amended, and any reference to any particular Code section shall be interpreted to include any revision of or successor to that section regardless of how numbered or classified.
 - "Collateral Assignment" has the meaning set forth in Section 3.9.
 - "Company" has the meaning set forth in the introductory paragraph.
- "Convertible Securities" of a Person means any securities (directly or indirectly) convertible into or exchangeable for any Membership Interest of such Person, including, without limitation, all warrants, options and other rights to acquire any Membership Units of such Person.
 - "Diligent Efforts" has the meaning set forth in Section 5.11(c).
- "Dividend" means any distribution by a Person with respect to its ownership interests whether in cash, securities (including common and preferred equity) or other property, including, without limitation, distributions upon any liquidation, dissolution or winding up of such Person.
 - $\begin{tabular}{ll} \bf ``EBITDA" means earnings before interest, taxes, depreciation and amortization according to GAAP. \\ \end{tabular}$
 - "Employment Agreement" has the meaning set forth in Section 3.6.

"Environmental Law" means any applicable federal, state or local law, regulation, order, decree, permit, authorization, opinion, or common law relating to: (1) the protection, investigation or restoration of the environment, health, safety, or natural resources, (2) the handling, use, presence, disposal, release or threatened release of any Hazardous Substance, (3) noise, odor, wetlands, pollution, or contamination or (4) standards of conduct concerning protection of human health (including, without limitation, employee health and safety), in each case as amended and as now or hereafter in effect, and the term "Hazardous Substance" means any substance that is: (A) oil or other petroleum products, (B) "hazardous wastes," as defined by the Resource Conservation and Recovery Act, as amended, (RCRA), 42 U.S.C. § 6901 et seq., or similar state or local law, ordinance, regulation or order, (C) "hazardous substances," as defined by the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, (CERCLA), 42 U.S.C. § 9601 et seq., or similar state or local law, ordinance, regulation or order, (D) "hazardous materials," as defined by the Hazardous Materials Transportation Act, as amended, (HMTA), 49 U.S.C. § 1802, or similar state or local law, ordinance, regulation or order, (E) radioactive materials subject to the Atomic Energy Act, as amended, (AEA), 42 U.S.C. § 2014 et seq., or similar state or local law, ordinance, regulation or order, (E) radioactive materials subject to the Atomic Energy Act, as amended, (AEA), 42 U.S.C. § 2014 et seq., or similar state or local law, ordinance, regulation or order, (E) radioactive materials subject to the Atomic Energy Act, as amended, (AEA), 42 U.S.C. § 2014 et seq., or similar state or local law, ordinance, regulation or order, (E) radioactive materials subject to the Atomic Energy Act, as amended, (AEA), 42 U.S.C. § 2014 et seq., or similar state or local law, ordinance, regulation or order, (E) radioactive materials subject to the Atomic Energy A

"ERISA" means the Employee Retirement Income Security Act of 1974 (or any successor legislation thereto), as amended, or any similar federal law then in force.

"ERISA Affiliate" means, with respect to the Company or any of its Affiliates, any trade or business (whether or not incorporated) under common control with such Person within the meaning of §414(b) or (c) of the Code (and §414(m) and (o) of the Code for purposes of provisions relating to §412 of the Code).

"ERISA Event" means, as to the Company or any of its Affiliates or any ERISA Affiliate: (a) a Reportable Event as defined in §4043 of ERISA and the regulations issued thereunder (other than a Reportable Event for which notice has been waived by regulation); (b) the withdrawal of the Company, any Subsidiary thereof or any ERISA Affiliate from a Pension Plan in which it was a "substantial employer" as defined in §4001(a)(2) of ERISA or was deemed a "substantial employee" under §4062(e) of ERISA; (c) the termination of a Pension Plan, the filing of notice of intent to terminate a Pension Plan or the treatment of a Pension Plan amendment as a termination under §4041 of ERISA; (d) the institution of proceedings to terminate a Pension Plan by the PBGC; (e) the partial or complete withdrawal of any Borrower or any ERISA Affiliate from a Multiemployer Plan to which any Borrower or any ERISA affiliate from a Multiemployer Plan to which any Borrower or any ERISA affiliate base any liability under §4241 or §4245 of ERISA, respectively; and (h) any event or condition which results in the termination of a Multiemployer Plan, or the institution by the PBGC of proceedings to terminate a Multiemployer Plan to which any

Borrower or any ERISA Affiliate has any liability under §4041A of ERISA or §4042 of ERISA, respectively.

- "Escrow Agent" has the meaning set forth in Section 2.3.
- "Event of Default" has the meaning set forth in Section 8.1.
- "Federal Bankruptcy Code" means Title 11 of the United States Code, as amended.
- "Financing" means the purchase of the Securities by the Purchaser hereunder.
- "Fixed Charge Coverage Ratio" means, with respect to any period, the ratio of EBITDA minus actual distributions of cash for federal and state income taxes minus capital expenditures to (ii) Total Debt Service.
- "GAAP" means generally accepted accounting principles as promulgated by the Financial Accounting Standards Board and/or any other governing body or boards having jurisdiction, authority or responsibility for promulgating accounting standards, as in effect from time to time. Except as otherwise expressly stated herein, all references to GAAP shall be deemed to mean GAAP as consistently applied.
 - "Governing Documents" of a Person means such Person's Articles of Organization, Certificate of Formation, By-laws and/or Operating Agreement.
- "Guarantee" means any guarantee of the payment or performance of any Indebtedness or other obligation and any other arrangement whereby credit is extended (or continued) to one obligor on the basis of any promise of another Person, whether that promise is expressed in terms of an obligation to: (a) pay the Indebtedness or other liabilities of such obligor; (b) purchase an obligation owed by such obligor; (c) purchase goods and services from such obligor pursuant to a take-or-pay contract; (d) maintain the capital, working capital, solvency or general financial condition of such obligor; or (e) otherwise assure any creditor of such obligor against loss (including by way of an agreement to repurchase or reimburse), whether or not any such arrangement is listed on the balance sheet of such other Person or referred to in a footnote thereto, but shall not include endorsements of items for collection in the ordinary course of business. The amount of any Guarantee shall be equal to the amount of the obligation so guaranteed or otherwise supported, or, if not a fixed or determined amount, the maximum amount guaranteed or supported.
- "Indebtedness" means at a particular time, without duplication: (a) any indebtedness for borrowed money or issued in substitution for or exchange of indebtedness for borrowed money; (b) any indebtedness evidenced by any note, bond, debenture or other debt instrument; (c) any indebtedness for the deferred purchase price of property or services with respect to which a Person is liable, contingently or otherwise, as obligor or otherwise (other than trade payables and other current liabilities incurred in the ordinary course of business, consistent with past practice which are not more than 90 days past due unless the same are being contested in good faith by appropriate proceedings and with respect to which a Person has set aside adequate reserves therefore in accordance with GAAP; (d) any commitment by which a Person

assures a creditor against loss (including, without limitation, contingent reimbursement obligations with respect to letters of credit); (e) any obligations for which a Person is obligated pursuant to a Guarantee; (f) any obligations under Capitalized Leases with respect to which a Person is liable, contingently or otherwise, as obligor, guarantor or otherwise, or with respect to which obligations a Person against loss; (g) any indebtedness secured by a Lien on a Person's assets; (h) any unsatisfied obligation for Withdrawal Liability to a Multiemployer Plan; (i) all indebtedness of any partnership of which such Person is a general partner or in which such Person may incur liability as if such Person was a general partner; and (j) all indebtedness of a Person for which such Person may become liable as a fiduciary or otherwise.

- "Indemnitees" has the meaning set forth in Section 9.15.
- "Intellectual Property" has the meaning set forth in Section 6.12.
- "Intellectual Property Rights" means all: (a) patents, patent applications, patent disclosures and inventions; (b) trademarks, service marks, trade dress, trade names, internet domain names, logos and corporate names and registrations and applications for registration thereof, together with all of the goodwill associated therewith; (c) copyrights (registered or unregistered) and copyrightable works and registrations and applications for registration thereof; (e) computer software, data, data bases and documentation thereof; (f) trade secrets and other confidential information (including, without limitation, ideas, formulas, compositions, inventions (whether patentable or unpatentable and whether or not reduced to practice), know-how, manufacturing and production processes and techniques, research and development information, drawings, specifications, designs, plans, proposals, technical data, copyrightable works, financial and marketing plans and customer and supplier lists and information); (g) other Intellectual Property Rights; and (h) copies and tangible embodiments thereof (in whatever form or medium).
- "Interest Charges" means, for any period, the sum of: (a) all interest charges and related expenses payable with respect to that fiscal period to a lender in connection with borrowed money or the deferred purchase price of assets that are treated as interest in accordance with GAAP, <u>plus</u> (b) the portion of capitalized lease obligations with respect to that fiscal period that should be treated as interest in accordance with GAAP, <u>plus</u> (c) all charges paid or payable (without duplication) during that period with respect to any hedging agreements.
- "Investment" as applied to any Person means: (a) any direct or indirect purchase or other acquisition by such Person of any notes, obligations, instruments, Membership Units and other securities of any other Person; and (b) any capital contribution by such Person to any other Person.
- "Investment Documents" means this Agreement, the agreements and instruments evidencing the Securities and any Membership Units for which Securities are exchanged or converted, the Stock and Asset Purchase Agreement, the Operating Agreement, the Employment Agreement, the Pledge Agreements, the Irrevocable Springing Proxy, the Security Agreements, the Collateral Assignment Agreement and each of the other agreements, documents and instruments expressly contemplated hereby and thereby.
 - "Irrevocable Springing Proxy" has the meaning set forth in Section 3.13(s).
 - "IRS" means the United States Internal Revenue Service.
- "Knowledge" or "Aware" regarding a Person means and includes the actual knowledge or awareness of such Person and its Subsidiaries (which shall include the actual knowledge and awareness of the officers, directors and key employees of such Person and its Subsidiaries and the general managers of each facility of such Person and its Subsidiaries).
- "Liens" means any mortgage, pledge, security interest, encumbrance, lien, charge or other restriction of any kind whatsoever (including any conditional sale or other title retention agreement or lease in the nature thereof), any sale of receivables with recourse against the Company or Affiliate of the Company, any filing or agreement to file a financing statement as debtor under the Uniform Commercial Code or any similar statute other than to reflect ownership by a third party of property leased to the Company under a lease which is not in the nature of a conditional sale or title retention agreement.
 - "LMCS" has the meaning set forth in the introductory paragraph.
 - "LMT" has the meaning set forth in Section 2.4.
- "Material Adverse Effect" means any matter or matters which would, alone or in the aggregate, have a materially adverse effect on: (a) the operating results, prospects, assets, liabilities, operations, condition (financial or or otherwise) or business of the Company and LMCS taken as a whole; or (b) the ability of the Company to perform any of its obligations under the Securities or any of the Investment Documents, but excluding (i) changes or effects arising out of or resulting from the execution or delivery of this Agreement or the public announcement thereof, the consummation of the Transactions or the identity of Purchasers, or (ii) any events, circumstances, changes or effects that result from any action (A) required to be taken pursuant to Section 5 or (B) taken at the request of Purchaser and not contemplated by this Agreement.
 - "Membership Units" means a unit of equity interest in the Company and having such rights permitted by the Operating Agreement.
- "Multiemployer Plan" shall mean a "multiemployer plan" as defined in §4001(a)(3) of ERISA, and to which the Company or any ERISA Affiliate makes, is making, or is obligated to make contributions on behalf of participants who are or were employed by any of them or to which such person has any current or potential liability.

"Notes" has the meaning set forth in Section 2.1(b).

"Officer's Certificate" means a certificate signed by the chief executive officer of the Company on behalf of the Company, stating that: (a) the officer signing such certificate has made or has caused to be made such investigations as are necessary in order to permit him to verify the accuracy of the information set forth in such certificate; and (b) such certificate does

not misstate any material fact and does not omit to state any fact necessary to make the certificate not misleading.

"PBGC" shall mean the Pension Benefit Guaranty Corporation or any successor thereto.

"Pension Plan" means a "pension plan", as such term is defined in §3(2) of ERISA, which is subject to Title IV of ERISA (other than a multiemployer plan as defined in §4001(a)(3) of ERISA), and to which the Company or any ERISA Affiliate may have liability, including any liability by reason of having been a substantial employer within the meaning of §4063 of ERISA at any time during the preceding 5 years, or by reason of being deemed to be a contributing sponsor under §4069 of ERISA.

"Permitted Indebtedness" means (a) any Indebtedness incurred pursuant to the terms of the Investment Documents; (b) Indebtedness secured by Liens permitted by clause (e) of the definition of Permitted Liens, provided, that the aggregate amount of all such Indebtedness at any time outstanding shall not exceed in the aggregate for the Company the amount provided for in the current budget approved pursuant to Section 5.1(h); and (c) Indebtedness described on the attached Indebtedness Schedule.

"Permitted Liens" means (a) Liens for Taxes or other governmental charges not yet due and payable or which are being contested in good faith by appropriate proceedings and for which adequate reserves have been established in accordance with GAAP such contest operates to suspend collections of the same; (b) Liens arising in the ordinary course of business (such as (i) Liens of carriers, warehousemen, mechanics and materialmen and other similar Liens imposed by law and (ii) Liens incurred in connection with worker's compensation, unemployment compensation and other types of social security (excluding Liens arising under ERISA) or in connection with surety bonds, bids, performance bonds and similar obligations) for sums not overdue or being contested in good faith by appropriate proceedings and not involving any deposits or advances or borrowed money or the deferred purchase price of property or services and, in each case, for which it maintains adequate reserves; (c) subject to the limitations set forth in clause (c) of the definition of Permitted Indebtedness (i) Liens arising in connection with Capitalized Lease Obligations (and attaching only to the property being leased), (ii) Liens existing on property at the time of the acquisition thereof by the Company (and not created in contemplation of such acquisition) and (iii) Liens that constitute purchase money security interests on any property securing Indebtedness incurred for the purpose of financing all or any part of the cost of acquiring such property, provided, that any such Lien attaches to such property within 60 days of the acquisition thereof and attaches solely to the property so acquired; (d) attachments, appeal bonds, judgments and other similar Liens, for sums not exceeding \$100,000 in the aggregate for the Company, arising in connection with court proceedings, provided, that the execution or other enforcement of such Liens is effectively stayed and the claims secured thereby are being actively contested in good faith and by appropriate proceedings

"Person" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization, a governmental entity or any department, agency or political subdivision thereof and any other entity.

"Plan" shall mean as required by the context at any time, an employee benefit plan, as defined in §3(3) of ERISA, which the Company or any ERISA Affiliate maintains, contributes to or has an obligation to contribute to on behalf of participants who are or were employed by any of them.

"Pledge Agreements" has the meaning set forth in Section 3.8.

"Potential Event of Default" means any event or occurrence which with the passage of time or the giving of notice or both would constitute an Event of Default.

"Pro Forma Balance Sheet" has the meaning set forth in Section 6.6(b).

"Prohibited Transaction" means any transaction set forth in §406 of ERISA or §4975 of the Code, other than a transaction that is exempt under §408 of ERISA or § 4975 of the Code.

"Public Offering" means the consummation of an underwritten public offering pursuant to an effective registration statement filed by the Company (or any successor entity to the Company) with the Securities and Exchange Commission under the Securities Act with respect to Membership Units of the Company (or any successor entity to the Company).

"Purchase Price" has the meaning set forth in Section 2.2.

"Purchased Membership Units" has the meaning set forth in Section 2.1(b).

"Purchaser Representative" has the meaning set forth in Section 5.2.

"Purchasers" has the meaning set forth in the introductory paragraph.

"Purchasers' Observation Rights" has the meaning set forth in Section 5.2.

"Put Closing" has the meaning set forth in Section 5.11(b)

"Put Equity" has the meaning set forth in Section 5.11(a).

"Put Notes" has the meaning set forth in Section 5.11(c).

"Put Notice" has the meaning set forth in Section 5.11(a).

"Put Price" has the meaning set forth in Section 5.11(a).

"Qualified Plan" means an employee pension benefit plan, as defined in §3(2) of ERISA, which is intended to be tax-qualified under §401(a) of the Code, and which the

Company or any ERISA Affiliate maintains, contributes to or has an obligation to contribute to on behalf of participants who are or were employed by any of them.

"Release" has the meaning set forth in CERCLA.

"Reportable Event" means any of the events listed in \$4043(c)(1), (2), (3), (5), (6), (8) or (9) of ERISA.

"Restricted Securities" means the Securities issued hereunder and any securities issued with respect to the Securities by way of a Dividend or split or in connection with a combination of Membership Units, recapitalization, merger, consolidation or other reorganization. As to any particular Restricted Securities, such securities shall cease to be Restricted Securities when they have (a) been effectively registered, under the Securities Act and disposed of in accordance with the registration statement covering them, (b) become eligible for sale pursuant to Rule 144(k) (or any similar provision then in force) under the Securities Act or (c) been otherwise transferred and new certificates for them not bearing any legend regarding the Securities Act. Whenever any particular securities cease to be Restricted Securities, the holder thereof shall be entitled to receive from the Company, without expense, new securities of like tenor not bearing a Securities Act legend.

"SBA" means the United States Small Business Administration.

"SBIC" means a small business investment company licensed under the SBIC Act.

"SBIC Act" means the Small Business Investment Act of 1958, as amended, or any similar federal law then in force.

"SBIC Holder" means any holder of Securities or any Membership Units for which Securities are exchanged or converted which is an SBIC.

"SBIC Regulations" means the Small Business Investment Company Act of 1958, as amended, and the regulations issued by the Small Business Administration thereunder, codified at Title 13 of the Code of Federal Regulations ("13 CFR"), 107 and 121, as amended.

"Securities" has the meaning set forth in Section 2.1(b).

"Securities Act" means the Securities Act of 1933, as amended, or any similar federal law then in force.

"Securities and Exchange Commission" includes any governmental body or agency succeeding to the functions thereof.

"Security Agreements" has the meaning set forth in Section 3.7.

"Senior Credit Facility" has the meaning set forth in Section 3.10.

"Senior Debt" means all Indebtedness owed under the Senior Credit Facility, including the line of credit or that is secured by any assets of the Company, excluding the Notes

"Senior Lender" means the Person that provides financing to the Company under a line of credit or that is secured by any assets of the Company, excluding the holders of the Notes in their capacity as such.

"Special Purpose Financial Statements" has the meaning set forth in Section 6.5(a)(ii).

"Subsidiary" means, with respect to any Person, any corporation, limited liability company, partnership, association or other business entity of which: (a) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof; or (b) if a limited liability company, partnership, association or other business entity, a majority of the partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by any Person or neore Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons shall be deemed to have a majority ownership interest in a limited liability company, partnership, association or other business entity gains or losses or shall be or control (or have the power to be or control) a managing director, manager or general partner of such limited liability company, partnership, association or other business entity.

"Tax" or "Taxes" means any federal, state, county, local, foreign or other income, gross receipts, ad valorem, franchise, profits, sales or use, transfer, registration, excise, utility, environmental, communications, real or personal property, capital stock, license, payroll, wage or other withholding, employment, social security, severance, stamp, occupation, alternative or add-on minimum, estimated and other taxes of any kind whatsoever (including deficiencies, penalties, additions to tax, and interest attributable thereto) whether disputed or not, including any obligation to pay the Taxes of others (by reason of transferee or successor liability, Treas. Reg. 1.1502-6 or similar state, local or foreign law, by contract or otherwise).

"Tax Return" means any return, information report or filing with respect to Taxes, including any schedules attached thereto and including any amendment thereof.

"Title IV Plan" means a Pension Plan which is subject to Title IV of ERISA.

"Total Assets" means, on any date, the net book value of all assets of the Company on that date, which, in accordance with GAAP, should be classified on the Company's balance sheet as assets.

"Total Debt Service" means, with respect to any period, an amount equal to the sum of (i) Interest Charges for such period plus (ii) the scheduled amortization of any Indebtedness during such period.

"Total Funded Debt" means, at any time, the aggregate amount outstanding of: (i) the Senior Debt; (ii) the Notes; plus (iii) any amounts owed to Foster Wheeler Corporation as payment of any deferred purchase price of

"Withdrawal Liability" means, at any time, the aggregate amount of the liabilities, if any, pursuant to §4201 of ERISA, and any increase in contributions pursuant to §4243 of ERISA with respect to all Multiemployer

LIQUIDMETAL TECHNOLOGIES

Exhibit 99.1

News Release

FOR IMMEDIATE RELEASE

Plans

CONTACT: Otis Buchanan

Liquidmetal Technologies 949-635-2120 otis.buchanan@liquidmetal.com

> Liquidmetal® Technologies Announces \$11.1 Million Funding Through the Organization and Capitalization Of New Coatings Business Subsidiary

> > \$10.2 Million in Convertible Notes Retired

Rancho Santa Margarita, Ca. July 26th, 2007- Liquidmetal® Technologies, Inc. (OTCBB: LQMT) today announced the completion of an \$11.1 million financing transaction in which it transferred the assets of its Liquidmetal Coatings division to a newly formed, newly capitalized subsidiary named Liquidmetal Coatings, LLC in a sale transaction. Going forward, Liquidmetal Coatings, LLC, referred to as "LMC," will own and operate the Liquidmetal Coatings business as a separate entity in which Liquidmetal Technologies, Inc. will retain a 69.25% equity interest. John Kang, Chairman of Liquidmetal Technologies, commented, "The establishment of LMC formally recognizes the strength of our Liquidmetal coating technology and the resulting market leadership of our coatings in the protective coatings market. The establishment of LMC as an independent subsidiary will provide the resources and focus to further grow this business."

LMC was funded through a \$6.5 million subordinated debt and equity investment by C₃ Capital Partners and a \$5.0 million senior credit facility with Bank Midwest, N.A., both out of Kansas City, Missouri. The remaining 30.75% of the equity of LMC will be held by C₃ Capital Partners, Larry Buffington (who will also serve as the President and CEO of LMC), and CRESO Capital Partners (the financial advisor in the transaction). The transaction resulted in \$11.1 million in gross proceeds to Liquidmetal Technologies of which \$10.2 million was used to retire convertible notes that were scheduled to become due in July and August 2007. The balance of the funds were used for transaction expenses and other debt.

This press release may contain "forward-looking statements" that involve risks and uncertainties, including statements regarding our plans, future events, objectives, expectations, forecasts, or assumptions. Any statement in this press release that is not a statement of historical fact is a forward-looking statement, and in some cases, words such as "believe," "estimate," "project," "expect," "intend," "may," "anticipate," "plans," "seeks," and similar expressions identify forward-looking statements. These statements involve risks and uncertainties that could cause actual outcomes and results to differ materially from the anticipated outcomes or result, and undue reliance should not be placed on these statements. These risks and uncertainties may include: our limited operating history in developing and manufacturing products from bulk amorphous alloys; the adoption of our alloys by customers; the commercial success of our customer's products; our ability to identify, develop, and commercialize new applications for our alloys; competition with suppliers of incumbent materials; the development of new materials that render our alloys obsolete; the ability to manage our anticipated growth; our limited direct experience in manufacturing bulk alloy products; scaling-up our manufacturing facilities; protecting our intellectual property; problems associated with manufacturing and selling our alloys outside of the United States; and other risks and uncertainties discussed in filings made with the Securities and Exchange Commission (including risks described in subsequent reports on Form 10-Q, Form 10-K, Form 8-K, and other filings). Liquidmetal Technologies disclaims any intention or obligation to update or revise any forward-looking statements, whether as a result of new information, future events, or otherwise.